WEDNESDAY, FEBRUARY 19, 1975

WASHINGTON, D.C.

Volume 40 ■ Number 34

PART II



# FREEDOM OF INFORMATION ACT

Implementation Regulations

VARIOUS EXECUTIVE AGENCIES

Note: Listed below are other documents containing Freedom of Information implementation regulations published previously in the Federal Register:

Agency	1975 Date of Issue	Vol. 40 FR, Page No.	
Cost Accounting Standards Board (correction)  Defense Department/Office of Secretary	Feb. 4 Feb. 11	4445 5135 6336	
Emergency Loan Guaranty Board Federal Maritime Commission Interstate Commerce Commission	Jan. 17 Jan. 20	6201 2983 3215	
Justice Department Cocupational Safety and Health Review Commission		<b>6494</b> 5779	

Note: Listed below are other Freedom of Information documents on file with the Office of the Federal Register which will be published at a later date:

	197	5	
Agency	Date of	Issue	
Interior Department/Office of the Secretary			
Renegotiation Board	Feb.	20	
Treasury Department/Office of the Secretary	Feb.	20	

# Title 2—Amnesty CHAPTER II—SELECTIVE SERVICE SYSTEM

# PART 200—RECONCILIATION SERVICE

Freedom of Information Act Regulations

The Director of Selective Service, pursuant to Proclamation 4313, Executive Order 11804, and the Freedom of Information Act, 50 U.S.C. 552 as amended by Pub. L. 93-502, gave public notice on January 14, 1975 (40 FR 2592), that consideration is being given to a proposed amendment to the Reconciliation Service Regulations constituting a portion of Chapter II of Title 2 of the Code of Federal Regulations.

More than thirty days have elapsed subsequent to such publication during which period comments from the public have been received and considered. The proposed amendment is being made effective without change in substance.

The amendment establishes the criteria and procedures for the release of in-

formation in a returnee's file.

Pursuant to the authority vested in me by Proclamation 4313, Executive Order 11804, and the Freedom of Information Act, 50 U.S.C. 552 as amended by Pub. L. 93–502, the Reconciliation Service Regulations constituting a portion of Chapter II of Title 2 of the Code of Federal Regulations are amended, effective February 19, 1975.

Section 200.9 is added to read as follows:

§ 200.9 Availability and use of information pertaining to a returnee.

(a) Information contained in a returnee's file and in records pertaining to an identifiable returnee may be disclosed or furnished to or examined by, the following persons:

(1) The returnee or any person having written authority dated and signed by the returnee: Provided, That whenever the time of the expiration of such authority is not specified therein, no information shall be disclosed, furnished, or examined under that authority after the expiration of a period of 1 year from its date.

(2) The legal representative of a deceased or incompetent returnee.

(3) All personnel of the Selective Service System while engaged in carrying out the functions of the Reconciliation Service Program.

(4) The Attorney General and his duly authorized representatives, including agents of the Federal Bureau of Investigation, whenever the returnee has been reported for failure to complete his reconciliation service in accord with his agreement with the Attorney General.

(5) Any other agency, official, or employee, or class or group of officials or employees of the United States upon written request in individual cases, but only when and to the extent specifically authorized in writing by the Director of Selective Service.

(b) No information shall be disclosed or furnished to, or examined by, any

person under the provisions of this section, until such person has been properly identified as entitled to obtain such information.

(c) A copy of a returnee's file and the documents contained in such file will be furnished to an individual described in paragraph (a) (1) or (2) of this section upon payment of a fee of 25¢ per page for each page in the form of a money order payable to the Treasury of the United States. A copy of other identifiable records or documents will be furnished to a person other than an individual described in paragraph (a) (3), (4), or (5) of this section entitled to receive it upon payment of a fee of 25¢ per page in the form of a money order payable to the Treasury of the United States. Search of records is made by compensated employees of the Selective Service System without charge.

(d) Whenever an employee receives a request for information or documents the disclosure of which is not clearly authorized by the provisions of this section that request will be immediately reported by telephone to the General Counsel, Selective Service System, for instructions as to its disposition.

(e) Complaints concerning possible abuse of discretion granted selective service employees under this section or failure to respond to inquiries shall be directed to the state director in the case of state headquarters or local board employees and to the Director in the case of National Headquarters employees.

(f) A requester whose request for information or documents has not been satisfied may appeal to the Director of Selective Service, 1724 F Street, NW., Washington, D.C. 20435.

(Proclamation 4313, Executive Order 11804, and the Freedom of Information Act, 50 U.S.C. 552 as amended by Pub. L. 93-502.)

BYRON V. PEPITONE, Director.

FEBRUARY 13, 1975.

[FR Doc.75-4450 Filed 2-18-75;8:45 am]

#### Title 6—Economic Stabilization

# CHAPTER VII—COUNCIL ON WAGE AND PRICE STABILITY

# PART 701—ORGANIZATION PART 702—PUBLIC ACCESS TO RECORDS

This is intended to establish a new Chapter VII, Council on Wage and Price Stability, in Title 6 of the Code of Federal Regulations and to issue, as Parts 701 and 702 of that new chapter, the regulations of the Council under the Freedom of Information Act, 5 U.S.C. Section 552.

The Council on Wage and Price Stability was established on August 24, 1974 by the enactment of the Council on Wage and Price Stability Act of 1974. The Council's principal purposes under the Act are to monitor the inflationary impact of price, wage and other activities in both the private and public sectors of the economy. In connection with these

purposes, the Council collects data related to the economy. These regulations issued today contain the criteria and procedures whereby the Council will determine to provide public access to, or alternative to preserve the confidentiality of this data, in compliance with the Freedom of Information Act.

In order that these regulations may be effective upon the February 19, 1975 effective date of the Freedom of Information Act Amendments of 1974, they are issued to become effective immediately. However, public comment on these rules is welcome, and will be considered by the Council in determining the need to amend these rules. Comments on these rules should be addressed to the General Counsel, Council on Wage and Price Stability, Room 3235, New Executive Office Building, 726 Jackson Place NW., Washington, D.C. 20506.

Issued in Washington, D.C. on February 13, 1975° and to become effective immediately.

ALBERT REES, Director.

Chapter VII is added as follows:

# CHAPTER VII—COUNCIL ON WAGE AND PRICE STABILITY

Part
701 Organization.
702 Public Access to Records.

Part 701 is added as follows:

Sec.
701.1 Purpose.
701.2 Status.
701.3 Membership.
701.4 Staff

701.3 Membership. 701.4 Staff. 701.5 Officers.

701.6 Functions of the Council.701.7 Divisions.

AUTHORITY: 5 U.S.C. 552.

# § 701.1 Purpose.

This Part is intended to provide a general description of the organization of the Council on Wage and Price Stability (the "Council"), in compliance with the Freedom of Information Act, 5 U.S.C. Section 552(a) (1). More detailed information can be obtained from the Assistant Director for Public Affairs and Congressional Relations, Council on Wage and Price Stability, Room 3225, The New Executive Office Building, 726 Jackson Place, N.W., Washington, D.C. 20506.

#### § 701.2 Status.

Pursuant to the Council on Wage and Price Stability Act of 1974, 12 U.S.C. Section 1904 note, the Council was established as an agency within the Executive Office of the President.

# § 701.3 Membership.

The Council consists of eight members and four adviser-members, all of whom are appointed by the President. The President has designated the Secretary of the Treasury to be a member and serve as Chairman of the Council, and the Assistant to the President for Economic Affairs to be a member and serve as Deputy Chairman of the Council. See Executive Order No. 11808.

## § 701.4 Staff.

The staff of the Council is headed by a Director, appointed by the President, and a Deputy Director.

# § 701.5 Offices.

The offices of the Council are at the New Executive Office Building, 726 Jackson Place, N.W., Washington, D.C. 20506. The principal office of the Council is at Room 3235 at the above address.

## § 701.6 Functions of the Council.

The principal function of the Council, pursuant to the Council on Wage and Price Stability Act, is to encourage restraint in wage, price and other activity that might have an inflationary effect upon the nation's economy. The Council serves this function by (a) monitoring wage, price and other activity within the private sector of the economy, (b) reviewing programs and policies of the various federal departments and agencies to analyze their inflationary impact upon the economy, (c) working with representatives of labor and management in various sectors of the economy to encourage restraint in their wage, price and other activity, and (d) to hold hearings and otherwise collect data related to the economy. The Council does not, however, have the authority to impose any involuntary wage, price or other restraint upon any sector of the economy.

#### \$ 701.7 Divisions.

To implement the above functions, the Council staff is organized by the following Divisions:

(a) Office of the Director. This office consists of the Director, the Deputy Director and an administrative officer. It provides general policy and guidance to

the Council staff.

(b) Office of Wage and Price Monitoring. This office is headed by the Assistant Director for Wage and Price Monitoring, and consists principally of a small staff of professional economists. The office is responsible for the Council's monitoring of wage, price and other activities in the private sector of the economy, and undertakes studies, public hearings and investigations in furtherance of that responsibility.

(c) Office of Government Operations and Research. This office is headed by the Assistant Director for Government Operations and Research and also consists principally of a small staff of professional economists. The office is responsible for the Council's review and analysis of the economic impact of rules, regulations, policies and other actions of the federal departments and agencies, and occasionally participates in formal proceedings before those departments and agencies to present the Council's views as to specific department or agency

(d) Office of General Counsel. This office consists of the General Counsel and a small legal staff. It provides the Council with legal advice and representation in connection with the various activities of the Council.

(e) Office of Public Affairs and Congressional Relations. This office is headed

by the Assistant Director for Public Affairs and Congressional Relations and has a small professional staff. It is responsible for the Council's response to Congressional inquiries about the Council's work, for inquiries from the news media, for inquiries from the public (including requests under the Freedom of Information Act, as set forth in Part 702 below), and for other activities, for example, the arrangement of public hearings.

#### Part 702 is added as follows:

#### Subpart A-General 702.1 Purpose. 702.2 Waiver. 702.3 Authority. Subpart B-Confidential Records

702.4 Request for confidential treatment. 702.5 Confidential records. Confidential determination. 702 6

# Subpart C-Request for Disclosure of Records

702.7 Written requests. 702 8 Notice of request. Initial decision. 702.9 702.10 Notice of initial decision.

702.12. Written appeal.

702.13

Preservation of requests and notices. 702.11

#### Subpart D-Appeals

Additional information. Appellate decision. Notice of appellate decision. 702.15 702.16 Access. 702.17 702.18 Prior approval or advanced deposit of 702.19 Payment of fees.

AUTHORITY: 5 U.S.C. 552.

# Subpart A-General

## § 702.1 Purpose.

The Freedom of Information Act, 5 U.S.C. Section 552, which is applicable to the Council, requires the public disclosure of records held by the Council, with the exception of specific categories of confidential records described in that Act. This Part establishes the Council's procedures for providing public access to Council records and for determining the confidential treatment of specific records submitted to it. This Part authorizes the Assistant Director for Public Affairs and Congressional Relations (hereinafter referred to in this Part as "Assistant Director"), subject to review by the Director of the Council, to receive and act upon requests for the disclosure of records.

# § 702.2 Waiver.

Whenever a waiver of any of the procedures set forth in this Part would further the purposes of the Freedom of Information Act by causing the public disclosure of nonconfidential records within the time period required by that Act, the Director or Assistant Director may, in the context of individual requests for records, waive any of the procedural requirements of this Part.

# § 702.3 Authority.

The provisions of this part are promulgated pursuant to the requirement in the Freedom of Information Act, 5 U.S.C. Section 552, that such provisions be promulgated and to 18 U.S.C. Section 1905,

both of which statutory provisions are made expressly applicable to the Council by Section 4 of the Council on Wage and Price Stability Act of 1974.

#### Subpart B-Confidential Records

#### § 702.4 Request for confidential treatment.

(a) Any person requesting the confidential treatment, pursuant to 5 U.S.C. Section 552(b) and to 18 U.S.C. Section 1905, of records submitted by that person to the Council shall make that request in writing to the Council. Such a written request shall accompany the records for which confidential treatment is quested and shall specifically identify the portions of the submitted records for which confidential treatment is requested.

 (b) To the extent possible, records for which confidential treatment is re-quested shall be separately bound or otherwise segregated from any accompanying material submitted without a

request for confidentiality.

# § 702.5 Confidential records.

The following kinds of records, exempt from the public availability requirements of 5 U.S.C. Section 552, may be provided confidential treatment by the Council, pursuant to a request made as set forth in \$ 702.1:

(a) Records specifically required by Executive order to be kept secret in the interest of the national defense or for-

eign policy;

(b) Records related solely to the internal personnel rules and practices of the Council;

(c) Records specifically exempted

from disclosure by statute;

(d) Trade secrets and commercial or financial information obtained from a person and privileged or confidential. This exemption includes, without limitation, data about the amount of sources of income, profits, losses, costs or expenditures; inventory lists; customer lists; manufacturing processes and other trade secrets; in each case to the extent that such information is not in fact pub lished or otherwise made available on a nonconfidential basis.

(e) Inter-agency or intra-agency memorandums or letters which would not be available by law to a party other than an agency in litigation with the Council. The exemption includes, without limitation, minutes and other records of the deliberations of the Council members and staff, intra- and inter-agency reports, memorandums, letters, work papers, and staff papers, except to the extent that such documents have been publicly released by the Council. To the extent that their premature release would be contrary to the public interest, such documents to be made public are included within this exemption until their public release by the Council.

(f) Personnel and medical files and similar files the disclosure of which would constitute a clearly unwarranted invasion of personal privacy. The exemption includes, without limitation, the personnel records of the Council, files containing reports, records or other material pertaining to individual cases in which disciplinary or other administrative action has been or may be taken, including records of proceedings pertaining to the conduct or performance of duties by Council personnel.

(g) Investigatory files compiled for law enforcement purposes except to the extent available by law to a party other

than an agency;

(h) Records contained in or related to examination, operating, or condition reports prepared by, on behalf of, or for the use of an agency responsible for the regulation or supervision of financial institutions; or

(i) Geological and geophysical information and data, including maps, con-

cerning wells.

#### § 702.6 Confidentiality determination.

Where a request for the confidential treatment of records submitted to the Council has been made, the General Counsel may, within a reasonable time, notify the party making that request of the General Counsel's determination whether those records are confidential records, as defined in § 702.5.

# Subpart C—Requests for Disclosure of Records

# § 702.7 Written requests.

(a) Any request for records held by the Council shall be made in writing and addressed to the Assistant Director for Public Affairs and Congressional Relations, Council on Wage and Price Stability, Room 3235 The New Executive Office Building, 726 Jackson Place, NW., Washington, D.C. 20506.

(b) Any request for records from the Council shall describe and identify, in reasonable detail, the particular documents or the type of records requested.

(c) Such a written request shall also estimate, pursuant to the fee schedule set forth in Subpart E, the maximum fee that the party making the request would be willing to pay for the search and duplication of the requested records, without further approval.

## § 702.8 Notice of request.

Whenever a request shall be made for the disclosure of records that the General Counsel has previously concluded, pursuant to § 702.6 to be confidential, notice shall be given to the party who has submitted those records that their disclosure has been requested.

#### § 702.9 Initial decision.

(a) Except as set forth in § 702.8 above, the Assistant Director shall, within ten (10) days (excepting Saturdays, Sundays and legal public holidays) of the receipt of any written request for records, determine whether to grant or deny, in whole or in part, that request. In making such a determination, the Assistant Director shall consider, among other alternatives, the possibility of separating confidential records, as defined in Subpart B, from other portions of any requested records, and disclosing the nonconfidential portions.

(b) In the event of an unusual circumstance, as defined in Paragraph (b) (1), (2), or (3) of this section, the Assistant Director may extend, by no more than ten (10) working days, the time within which he must make an initial decision in response to a written request. The Assistant Director shall immediately give written notice to the party making the request of any such extension of time, of the reasons for such an extension, and of the expected date of the initial decision. "Unusual circumstances," for this purpose, shall include:

(1) The need to search for and collect the requested records from field facilities or other establishments that are separate from the Council's office at the New Executive Office Building in Wash-

ington, D.C.

(2) The need to search for, collect, and appropriately examine a voluminous amount of separate and distinct records which are demanded in a single request.

(3) The need for consultation, which shall be conducted with all practicable speed, with another agency having a substantial interest in the determination of the request for records or among two or more components of the Council having substantial subject-matter interest therein.

(c) For purposes of this Section, a written request for records shall be deemed "received" when it has arrived in the offices of the Council in a written form that complies with the requirements of § 702.7.

## § 702.10 Notice of initial decision.

The Assistant Director shall, immediately upon making an initial decision about a request for records, give written notice of that decision to the person making the request. In the event that that initial decision is to grant a request in whole or part, the written notice shall include a statement of that decision, a brief description of the records to be made available, a statement of the times and place when such records can be available for inspection or, alternatively, of the procedure for duplication and delivery (by mail or other means) of the records to the requesting party, and an itemized statement of the total fees chargeable to the requesting person, pursuant to the fee schedule set forth in Subpart E, for the search for and duplication of the requested records. In the event, on the other hand, that the Assistant Director's initial decision is to deny a request in whole or in part, the written notice of that decision shall include a brief statement of that decision and of the reasons therefor, an identification of the name and title of the person making that decision, and also a description of the procedures, pursuant to Subpart D, for an appeal from that decision.

# § 702.11 Preservation of requests and notices.

The Assistant Director shall preserve all written requests for records and all written notices of his initial decisions in response thereto for at least one calendar year.

# Subpart D-Appeals

## § 702.12 Written appeal.

(a) Any person who has made a written request for the disclosure of records, pursuant to Subpart C, may, within 30 days of receipt of a written notice of the Assistant Director's initial decision or after the Assistant Director's failure to take any timely action upon that request, request that the Director of the Council review that decision or nonaction. Any such request for review should be made in writing, shall be addressed to the Director, Council on Wage and Price Stability at Room 3235, The New Executive Office Building, 726 Jackson Place, NW., Washington, D.C. 20506.

(b) Any written request for review

(b) Any written request for review of an initial decision shall include a copy of the party's initial request to the Assistant Director for the disclosure of records, a copy of the Assistant Director's initial decision, and a brief statement of the legal, factual or other basis for the party's objection to that initial

decision.

# § 702.13 Additional information.

Immediately upon-receipt of a written appeal, the Director may request that additional information be submitted by the party appealing or by the Assistant Director.

# § 702.14 Appellate decision.

(a) The Director shall, within twenty (20) days (excepting Saturdays, Sundays, and legal public holidays) of the receipt of any written appeal, determine whether to grant or deny that appeal.

(b) In the event of an unusual circumstance, as defined in § 702.9(b), the Director may extend, by no more than ten (10) working days, the time within which he must make an appellate decision in response to a written appeal. The Director shall immediately give written notice to the party making the request of any such extension of time, of the reasons for such an extension, and of the expected date of the appellate decision

# § 702.15 Notice of appellate decision.

The Director shall, immediately upon making an appellate decision, give written notice of that decision to the person making the request. This notice shall include a brief statement of the appellate decision and of the reasons therefor.

# Subpart E-Access to Records; Fees

## § 702.16 Access.

(a) Except as set forth in paragraph (b) of this section or as otherwise determined by the Assistant Director or Director, requested records or duplicate copies thereof, to be made available to any person shall be made available during regular business hours at the offices of the Council on Wage and Price Stability, The New Executive Office Building, Washington, D.C.

(b) Upon the Assistant Director's or the Director's approval, requested records or duplicate copies thereof can be made available by mail to the person ant to 5 U.S.C. section 552 and to § 702.9, making the request.

#### § 702.17 Fees.

(a) Fees charged by the Council for the search for and duplication of any records requested shall comply with the

following fee schedule:

(1) Search for records. \$5 per hour when the search is conducted by a clerical employee. \$8 per hour when the search is conducted by a professional employee. No charge for searches of less than one hour.

(2) Duplication of records. Records will be duplicated at a rate of \$.25 per page for all copying of four pages or more. There is no charge for duplicating

- three or less pages.
  (3) Other. When no specific fee has been established for a service, the Assistant Director is authorized to establish an appropriate fee based on "direct costs" as provided in the Freedom of Information Act and in accordance with Office of Management and Budget Circular No. A-25. Examples of services covered by this provision include searches involving computer time or special travel, transportation or communication
- (4) If records requested are stored elsewhere than at the offices of the Council, the special costs of returning those records to the Council's offices for review will be added to the search costs. Search costs are due and payable even if the record which was requested cannot be located after all reasonable efforts have been made or if it is determined that a requested record is confidential and should be withheld.
- (b) The Assistant Director or Director may determine, in connection with specific requests for records, that the public interest is best served by the provision of the requested records at no cost or at a cost below the above schedule and, in those specific instances, may waive the above schedule, in whole or in part.

## § 702.18 Prior approval or advanced deposit of fees.

- (a) Where the search, duplication or other fees anticipated to result from a request are substantially greater than the amount estimated in the written request, pursuant to § 702.7 above, or in the absence of such an estimate where the anticipated fees are greater than \$25, the person requesting the records shall be immediately notified of the estimated fees and his approval for such fees requested. Such person shall also be offered the opportunity to reformulate his or her request in order to reduce the search, duplication and other fees but yet satisfy that person's needs for records.
- (b) Where the estimated search, duplication and other fees would exceed \$25, the Assistant Director may in addition request that the person requesting records make an advance deposit of the estimated fees.
- (c) The dispatch of any such request for an estimated fee approval or advance deposit shall suspend, until a reply is re-ceived by the Council, the period, pursu-

within which the Assistant Director must respond to a written request for records.

#### § 702.19 Payment of fees.

(a) Fees actually charged a person for the search and duplication of records must be paid in full prior to issuance to him or her of those records. In the event that that person is in arrears for previous requests to the Council for records, records will not be provided for any subsequent requests until the arrears have been paid in full.

(b) Payment of fees shall be made by a personal check, postal money order or bank draft drawn on a bank in the United States, made payable to the order of the Treasurer of the United States.

[FR Doc.75-4466 Filed 2-18-75;2:04 pm]

# Title 8—Aliens and Nationality

CHAPTER I-IMMIGRATION AND NATURALIZATION SERVICE, DEPARTMENT OF JUSTICE

PART 103—POWERS AND DUTIES OF SERVICE OFFICERS; AVAILABILITY OF SERVICE RECORDS

Requests Under the Freedom of Information Act, as Amended; Policy and Pro-

Pursuant to section 552 of Title 5 of the United States Code (80 Stat. 383) as amended by Pub. L. 93-502 (88 Stat. 1561) and the authority contained in section 103 of the Immigration and Nationality Act (66 Stat. 103; 8 U.S.C. 1103), 28 CFR 0.105, and 8 CFR 2.1, amendments are being made to 8 CFR Part 103 to conform Service regulations to the requirements of the Freedom of Information Act as amended by the Act of November 21, 1974 (88 Stat. 1561).

The amendments contained in this order revise the Immigration and Naturalization Service regulations to conform with the new time limits for determining requests under the Freedom of Information Act (5 U.S.C. 552) established by the amendments to that Act contained in Pub. L. 93-502 (88 Stat. 1561). These amendments also conform with and supplement new amendments to the Freedom of Information Act regulations of the Department of Justice, 28 CFR Part 16, Subpart A, which become effective February 19, 1975. On January 16, 1975, the Service published proposed fee and proposed time limit regulations. So much of the proposed rules as concerned fees is not affected by the present amendments; the proposed rules pertaining to fees are being adopted by a separate order. The present amendments supersede so much of the proposed rules as concerned other aspects of the administration of the Freedom of Information Act.

In addition to changes necessitated by the recent amendments to the Freedom of Information Act and the amendments to 28 CFR Part 16, these amendments delegate the authority to deny requests, heretofore reserved to the Commissioner, to the Associate Commissioner, Management, and to regional commissioners and

district directors. The amendments set forth specific requirements and suggestions for the making and processing of requests. Most of these amendments are intended to insure compliance with the new time limits imposed by the amendments to the Act. There are statements regarding when time limits will commence to run. There is a provision that when time limits have been exhausted a requester will be informed of his right to treat the absence of a reply as a denial, entitling him to appeal to the Attorney General, and that he may also be asked to forego his right of appeal while the Service is continuing to process his request. A distinction is made between requests for records and requests for official verifications of facts, the latter being denoted as not being under the Freedom of Information Act.

In the light of the foregoing, the following amendments to Chapter I of Title 8 of the Code of Federal Regulations are

hereby prescribed:

# § 103.7 [Amended]

1. In § 103.7 Fees, paragraph (b) (1) is amended by adding at the end thereof the following fee:

For filing application for certification of information contained in Service records \_\_\_\_\_

2. In § 103.7(c), a new subparagraph (2) is added and the existing subparagraph (2) is redesignated as subparagraph (3). As amended, § 103.7(c) (2) and (3) reads as follows:

(2) Fees under the Freedom of Information Act, as amended, may be waived or reduced where the Service determines such action would be in the public interest because furnishing the information can be considered as primarily bene-

fiting the general public.

(3) When the prescribed fee is for services to be performed by the clerk of court under section 344(a) of the Act, the affidavit for waiver of the fee shall be filed with the district director or officer in charge of the Service having administrative jurisdiction over the place in which the court is located at least 7 days prior to the date the fee is required to be paid. If the waiver is granted, there shall be delivered to the clerk of court by a Service representative on or before the date the fee is required to be paid, a notice prepared on Service letterhead and signed by the officer granting the waiver, that the fee has been waived pursuant to this paragraph.

3. In § 103.8, the title and the introductory text are revised to read as fol-

Definitions pertaining to availability of information under the Freedom of Information Act.

Sections 103.8, 103.9, and 103.10 of this part comprise the Service regulations under the Freedom of Information Act, 5 U.S.C. 552. These regulations supplement those of the Department of Justice. 28 CFR Part 16, Subpart A. As used in this part the following definitions shall apply:

§ 103.9 [Amended]

4. In § 103.9, the title is revised to read as follows:

§ 103.9 Availability of decisions and interpretive material under the Freedom of Information Act.

5. Section 103.10 is amended in the following respects: the title is revised; paragraph (a) (1) is revised by the deletion and addition of offices to the listing therein and by the addition of two new sentences at the end of the paragraph; paragraph (a) (2) is revised in its entirety; paragraph (b) (1) is revised by the addition of a new subdivision (vi); paragraph (b) (2) is revised; paragraph (b) (4) is being revoked by a separate order and paragraphs (c), (d), and (e) are revised in their entirety; and a new paragraph (f) is added. As amended, § 103.10 is revised to read as follows:

# § 103.10 Requests for records under the Freedom of Information Act.

(a) Place and manner of requesting records.—(1) Place. Records shall be made available in the Central Office, each regional office, any district office, and the following offices: Agana, Guam; Albany, N.Y.; Cincinnati, Ohio; Dallas, Tex.; Hammond, Ind.; Harlingen, Tex.; Las Vegas, Nev.; Memphis, Tenn.; Milwaukee, Wis.; Norfolk, Va.; Pittsburgh, Pa.; Previdence, R.I.; Reno, Nev.; St. Louis, Mo.; Salt Lake City, Utah; San Diego, Calif.; Spokane, Wash.; additionally, in particular cases, a district director may designate any other Service office. Active alien records are maintained in the district or sub-office listed above where the subject of the record resides. For locations of regional and district offices, see 8 CFR 100.4.

(2) Manner of requesting records. Requests for records may be made in person or by mail. If a request is made by mail, both the envelope and its contents shall be clearly marked: "FREEDOM OF INFORMATION REQUEST" or FORMATION REQUEST". Any request for information not marked and addressed as specified in this paragraph will be so marked by Service personnel as soon as it is properly identified, and forwarded immediately to the appropriate office. A request will not be deemed to have been received for purposes of the time period set forth in 5 U.S.C. 552(a)(6) until forwarding to the appropriate office has been effected, or until such forwarding would have been effected with the exercise of due diligence by Service personnel. On receipt of a request forwarded as set forth above to the appropriate office, such office shall notify the requester of the date on which the time period commenced to run. The use of Form G-639 is recommended for rapid identification as a Freedom of Information matter and to insure expeditious handling. Each request made under this section pertaining to the availability of a record shall describe the record with sufficient specificity with respect to names, dates, subject matter and location to permit it to be identified and located. A request for all records falling within a reasonable

specific category shall be regarded as conforming to the requirement that records be reasonably described if it enables the records requested to be identifled by any process that is not unreasonably burdensome or disruptive of Service operations. If it is determined that a request does not reasonably describe the records sought, as specified above, the response rejecting the request on that ground shall specify the reasons why the request failed to meet requirements and shall extend to the requester an opportunity to confer with Service personnel in order to attempt to reformulate the request in a manner which will meet the needs of the requester and the requirements of this subsection.

(b) Authority to grant and deny requests.—(1) Grants. The Associate Commissioner, Management, may grant any type of request. The head of any office specified in paragraph (a) (1) of this section may grant the following types of requests:

(i) Requests for information and records which officers and employees of the Service prior to the enactment of 5 U.S.C. 552 customarily furnished to the public in the regular performance of their duties;

(ii) Requests for records of proceedings in deportation hearings, unless members of the public have been excluded from the hearings by direction of the special inquiry officer pursuant to § 242.16(a) of this chapter;

(iii) Requests for records of proceedings in naturalization examinations and hearings;

(iv) Requests for records of proceedings in any other proceedings before the Service which were open to the public;

(v) Requests for records of proceedings in administrative fine cases; and

(vi) Requests for any other records of the Service, including portions of records, which clearly are not within the exemptions from disclosure enumerated in 5 U.S.C. 552(b).

(2) Denials. The Associate Commissioner, Management, regional commissioners, and district directors have au-

thority to deny a request.

(3) Authority to state that a record cannot be located or does not exist. The head of any office specified in paragraph (a) (1) of this section has authority to notify a requester that a record cannot be located from the information supplied, or is known to have been destroyed or otherwise disposed of.

(c) Prompt response.—(1) Response within 10 days. Within 10 days (excluding Saturdays, Sundays and legal public holidays) of the receipt of a request by the Service (or in the case of an improperly addressed request, of its receipt by the appropriate office as specified in paragraph (a) of this section), the authorized Service official shall either comply with or deny such request unless an extension is made under 28 CFR 16.5(c). A request improperly addressed will not be deemed to have been received for purposes of the time period set forth in 5 U.S.C. 552(a) (6) until forwarding to the appropriate office has been effected, or

until such forwarding would have been effected with the exercise of due diligence by Service personnel. On receipt of a request forwarded as set forth in paragraph (a) of this section to the appropriate office, such office shall notify the requester of the date on which the time period commenced to run.

(2) Treatment of delay as a denial. If no substantive reply is forthcoming at the end of the 10-day period, or the last extension thereof under 28 CFR 16.5(c), the requester may deem his request denied, and exercise a right of appeal in accordance with 28 CFR 16.7. When no substantive reply can be made within the applicable time limit, the responsible Service office shall nevertheless continue to process the request; it shall inform the requester upon expiration of the time limit of the reason for the delay, of the date on which a determination may be expcted to be dispatched, and of his right to treat the delay as a denial and appeal to the Attorney General in accordance with 28 CFR 16.7; and it may ask the requester to forego appeal until a determination is made.

(d) Disposition of requests.—(1) Form of grant. When a requested record is available, the responsible office shall notify the requester when and where the record is available or copies will be available. The notification shall also advise the requester of any applicable fees

under 28 CFR 16.9.

(2) Form of denial. A reply denying a written request for a record in whole or in part shall be in writing, signed by the Associate Commissioner, Management, by a regional commissioner, or by a district director, and it shall include a reference to the specific exemption under the Freedom of Information Act authorizing withholding of the record. To the extent consistent with the purpose of the exemption, the notice of denial shall contain a brief explanation of how the exemption applies to the record withheld, and if the deciding official considers it appropriate, a statement of why the exempt record is being withheld. The notice of denial shall include a statement of the right of appeal to the Attorney General in accordance with 28 CFR 16.7 (see subparagraph (3) of this paragraph), and that judicial review will thereafter be available in the district in which the requester resides or has his principal place of business or the district in which the agency records are situated or the District of Columbia.

(3) Right of appeal. When a request for records has been denied in whole or in part the requester may, within 30 days of its receipt, appeal the denial to the Attorney General (Attention: Freedom of Information Appeals Unit), Department of Justice, Washington, D.C. 20530, and both the envelope and the letter shall be clearly marked: "FREEDOM OF INFORMATION APPEAL" or "INFORMA-

TION APPEAL."

(e) Copies of responses to Deputy Attorney General. A copy of each notification to a requester of a delay, notice of extension, denial or of inability to locate a requested record shall be furnished to the Deputy Attorney General.

(f) Verification of information from Service records. When a requester desires, not a copy of a record, but a formal statement by the Service verifying particular information based on Service records, his request is not being made pursuant to the Freedom of Information Act and shall be submitted on Form G-641 to the Service office having jurisdiction over his place of residence.

(Sec. 103, 66 Stat. 173; (8 U.S.C. 1103))

Compliance with the provisions of section 553 of Title 5 of the United States Code as to notice of proposed rule making and delayed effective date is unnecessary because the amendments pertain to agency policy and procedure. In addition, since many of the amendments are necessary for implementation of the amendments to the Freedom of Information Act. (5 U.S.C. 552) by the Act of November 21, 1974 (Pub. L. 93-502; 88 Stat. 1561), which become effective February 19, 1975, there is insufficient time to receive and evaluate public comment. However, in accordance with the spirit of the provisions of 5 U.S.C. 553, interested persons may submit to the Commissioner of Immigration and Naturalization, Room 7100-C, 425 Eye Street NW., Washington, D.C. 20536, written data, views, or arguments, in duplicate, with respect to these amendments. Such comments may not be presented orally in any manner. All relevant material received by March 19, 1975, will be considered in the same manner as if this document were a notice of proposed rule making.

Effective date. The amendments contained in this order shall become effective February 19, 1975.

Dated: February 14, 1975.

L. F. CHAPMAN, Jr., Commissioner of Immigration and Naturalization. IFR Doc.75-4600 Filed 2-18-75;8:45 am]

PART 103—POWERS AND DUTIES OF SERVICE OFFICERS; AVAILABILITY OF SERVICE RECORDS

# Fees and Requests for Records Under Freedom of Information Act

Reference is made to the Notice of Proposed Rule Making which was published in the Federal Register of January 16, 1975 (40 FR 2817) pursuant to the provisions of 5 U.S.C. 553 (80 Stat. 383) and 5 U.S.C. 552(a) (4), as amended by Public Law 93-502 (88 Stat. 1561), and in which there were set forth proposed amendments to 8 CFR 103.7 and 103.10 pertaining to fees and requests for record search and duplication under the Freedom of Information Act, as amended.

No representations were received in response to the proposed rules of January 16, 1975. Items 1, 2, 3, and 6 of the proposed rules are adopted, as proposed. Items 4, 5, 7, and 8 of those proposed rules, which pertain to aspects of the Freedom of Information Act other than fees, are not adopted because superseded by amendments being published

separately, to conform Service regulations to the requirements of the Act as amended by Public Law 93-502 (88 Stat.

In the light of the foregoing and pursuant to the authority contained in section 103 of the Immigration and Nationality Act, 28 CFR 0.105(b) and 8 CFR 2.1, the proposed amendments to \$\frac{1}{2}\$ 103.7 (b), 103.7(d), and 103.10(b) (4) of Part 103, Chapter I, Title 8 of the Code of Federal Regulations, as set forth below, are hereby adopted:

#### § 103.7 [Amended]

1. In § 103.7 Fees, amend paragraph (b) (1) by adding at the end thereof the following:

2. In § 103.7, amend paragraph (b) (2) by deleting in its entirety the existing material and by substituting in lieu thereof the following:

(2) Fees for production or disclosure of records under 5 U.S.C. 552 shall be charged in accordance with the regula-

tions of the Department of Justice, 28

3. Section 103.10(b) (4) is redesignated as § 103.7(d) and reads as follows:

(d) Authority to certify records. Whenever authorized under 5 U.S.C. 552 or any other law to furnish information from records to persons entitled thereto, the following officials have authority to make certifications, as follows:

(1) The Associate Commissioner, Management—copies of files, documents, and records in the custody of the Central

Office.

(2) A regional commissioner or district director—copies of files, documents, and records in the custody of his office.

(3) The Chief, Records Administration and Information Branch, Central Office—the nonexistence of an official record in the records of the Service.

(Sec. 103, 66 Stat. 173; (8 U.S.C. 1103))

The basis and purpose of the aboveprescribed rules are to conform fees for the production and disclosure of records under the Freedom of Information Act to the provisions of Public Law 93-502.

Compliance with the provisions of section 553 of Title 5 of the United States Code (80 Stat. 383) as to delayed effective date is unnecessary in this instance because the amendments to § 103.7(b) and 103.10 are in implementation of the Act of November 21, 1974 (P.L. 93-502; 88 Stat. 1561), which becomes effective on February 19, 1975.

Effective date. The amendments contained in this order shall become effective on February 19, 1975.

Dated: February 14, 1975.

L. F. CHAPMAN, Jr.,

Commissioner of

Immigration and Naturalization.

[FR Doc.75-4606 Filed 2-18-75:8:45 am]

# Title 12—Banks and Banking CHAPTER IV—EXPORT-IMPORT BANK OF THE UNITED STATES

# PART 404—DISCLOSURE OF INFORMATION

**Notice of Amended Regulations** 

Notice is hereby given that the Export-Import Bank (Eximbank) has revised Part 404 of Title 12 of the Code of Federal Regulations which sets forth Eximbank's regulations governing the disclosure of information under the Freedom of Information Act. The revision implements the provisions of Public Law 93-502 of November 21, 1974 effective February 19, 1975, which amends the Freedom of Information Act, 5 U.S.C. 552.

On January 13, 1975, Eximbank published (40 FR 2449) a proposed revision of § 404.6 which sets forth the schedule of fees for search and duplication of records under the Freedom of Information Act and invited public comment thereon.

The amendments to the regulations, including the revised fee schedule set forth in § 404.6, become effective on February 19, 1975.

Part 404 is revised as set forth below.

Issued in Washington, D.C. on February 10, 1975.

WARREN W. GLICK, Acting Executive Vice President.

404.1 Purpose and policy.

404.2 Scope.

404.3 Information and records available to the public and exempt from disclosure.

404.4 Public access to information and records.

404.5 Administrative appeal of refusal to disclose.

404.6 Fees.

404.7 Annual Report to Congress.

404.8 Appearances and testimony by Eximbank officers and employees.

AUTHORITY: 5 U.S.C. 552 and 12 U.S.C. 635.

# § 404.1 Purpose and policy.

(a) This part establishes policy and procedures governing public access to information contained in the files, documents, and records of the Export-Import Bank of the United States (Eximbank). In keeping with the spirit as well as the letter of Public Law 90-23, which codified and repealed Public Law 89-487, amending 5 U.S.C. 552, formerly section 30 of the Administrative Procedure Act, 60 Stat. 236, 5 U.S.C. 1002 (1964 Ed), and Public Law 93-502, further amending 5 U.S.C. 552, it reflects Eximbank policy that disclosure is the general rule rather than the exception. It is in addition a recognition that this policy in favor of disclosure extends in many instances to information technically exempt from disclosure under the law where such disclosure would not adversely affect some legitimate public or private interest intended to be protected by law, would not otherwise violate law or other authority, and would not impose an unreasonable burden upon Eximbank.

(b) This part is also a recognition that the soundness of many Eximbank programs, e.g. loans, guarantees and insurance, depends in large measure upon the reliability of commercial, technical, financial and business information relating to the affairs of applicants for Eximbank assistance. Since the release of such information would jeopardize the credit and competitive business position of an applicant it is essential that applicants be assured that confidential commercial or financial information which is submitted to Eximbank will not be disclosed to the public. By this assurance, applicants will be encouraged to make complete disclosure of material bearing upon an application.

# § 404.2 Scope.

This part applies to all files, documents, records, and information obtained or produced by officers and employees of Eximbank in the course of their official duties as well as all files, documents, records and other information in the custody or control of any Eximbank officer or employee. It does not purport to describe or set forth every file, document, record, or item of information which may or may not be disclosed or to incorporate every exemption from dis-closure provided by law. Material de-scribed is illustrative rather than exclusive.

- § 404.3 Information and records available to the public and exempt from disclosure.
- (a) General. All Eximbank information and records in existence which are not exempt by law are available for public inspection and copying in the manner specified in § 404.4. In addition, certain materials technically qualifying for exemption from disclosure will be made available where disclosure would not adversely affect some legitimate public or private interest, would not otherwise violate law or other authority, and would not impose an unreasonable burden on Eximbank. Reasonable requests for material not in existence may also be honored where their tabulation or compilation will not unduly interfere with Eximbank activities, programs and operations. As provided in § 404.6, a fee will be charged for Eximbank's expenses incurred in searching for, duplicating, tabulating or compiling such information and records.

(b) Information and records which

are available to the public.

The following kinds of records and information are available to the public in the manner specified in § 404.4:

- (1) Names of recipients of loans, guarantees, insurance and other assistance, (2) The kind and amount of assist-
- ance, (3) The purpose of the approved assistance in general terms,
- (4) The extent of outside participation, if any, and
- (5) Statistical data on Eximbank pro-
- (c) Information which is generally not available to the public. The following

kinds of information are generally not available to the public:

(1) Information on declined, withdrawn, or cancelled applications for assistance,

(2) Trade secrets obtained from applicants for Eximbank assistance,

(3) Privileged or confidential commercial or financial information obtained from any person, including, for example, such information contained in individual case files relating to such activities as loans, guarantees and insurance,

(4) Loan agreements, insurance policies and bank guarantee agreements relating to individual borrowers or foreign buyers receiving Eximbank assistance,

(5) Information concerning losses, delinquencies and defaults in individual cases, and

(6) Names of participating lending institutions and the terms of their participation without their consent.

(d) Minutes of the meetings of the Board of Directors. These are available for inspection and copying in Eximbank's Public Affairs Office in Room 1267, 811 Vermont Avenue NW., Washington, D.C. 20571.

(e) Personnel and similar files. Exempt from disclosure are personnel, medical and other files containing pri-vate or personal information. The vate or personal information. names, position titles, and duty stations of Eximbank employees are public information but their home addresses are not. The disclosure of private or personal information contained in other files, for example, in the files relating to members of Eximbank's Advisory Board and to applicants for Eximbank assistance, also would be exempt.

(f) Eximbank staff directives and other instructions to staff. All staff directives are considered public information except (1) those relating to audits and investigations, internal financial management and fiscal operations, and (2) portions of directives containing confidential standards and instructions, as, for example, instructions concerning processing loan, guarantee or insurance applications, negotiations or bargaining in connection with the disposition and liquidation of loans, and loan collateral held by Eximbank.

(g) Litigation materials. Copies of pleadings, motions, orders, transcripts of testimony, and documentary evidence introduced in pending or closed litigation are available once such items are a matter of public record.

(h) Internal communications. Interagency or intraagency communications routinely available to a party to litigation with Eximbank are exempt from disclosure. These would include, among other things, drafts, memoranda between officials or agencies, Exim-bank staff memoranda, opinions and interpretations prepared by Eximbank attorneys or consultants for use of Eximbank, research studies performed internally or under contract for internal management purposes, and internal management reports.

§ 404.4 Public access to information and records.

(a) Facilities. Eximbank facilities are available to the public during normal business hours for requesting, inspecting and copying information and records. Reproduction machines will also be available in or through such facilities. The Public Affairs Office is located in Room 1267, 811 Vermont Avenue NW., Washington, D.C. 20571.

(b) Materials available in Public Affairs Office. (1) For the convenience of the public, certain Eximbank materials will be maintained and readily available in the public information office. These

will include:

(i) All Eximbank directives and manuals not exempt from disclosure, (ii) Eximbank Rules and Regulations

(including Interpretations), and (iii) Index of Eximbank materials, including lists of staff directives, forms, reports, and Eximbank official actions.

(2) The public affairs office will, in addition to the above, have normally available, among other things:

(i) Pamphlets describing Eximbank Programs,

(ii) Press releases.

(iii) Names of recipients of Eximbank support and related information not exempt from disclosure,

(iv) Eximbank's Annual Report to the President and the Congress,

(v) Routine statistical reports on Eximbank activities.

(vi) Minutes of Meetings of the Board of Directors, and

(vii) Blank Eximbank forms.

(c) Requests for information and records. Requests for information, records and other materials not readily available at the Public Affairs Office are to be submitted and processed in accordance with the following procedures:

(1) Form of request. Each request shall be addressed to the Export-Import Bank of the United States, Attention: Senior Vice President—Public Affairs and Export Expansion, 811 Vermont Avenue, NW., Washington, D.C. 20571. The envelope and the letter containing the request must be clearly marked in capital letters as follows: FREEDOM OF INFORMATION ACT REQUEST. EIB Form 73-5, described in paragraph (d) below, may be used in lieu of a letter for purposes of making the request. A request submitted in an envelope which is not addressed to the Senior Vice President-Public Affairs and Export Expansion will not be deemed to have been received by Eximbank until such time as the request is forwarded to such officer. All requests must be in writing and must be marked and addressed as specified in this section.

(2) Description of material requested. Each request shall reasonably describe the document or information with respect to names; dates and subject matter to permit it to be located among the records maintained by Eximbank. A request that does not substantially comply with this subparagraph (ii) will not be deemed to have been received by Eximbank until such time as the requester has clarified his request to meet this standard. Eximbank will make every reasonable effort by telephone or by letter to assist the person making the request to be more specific in describing the document or information.

(3) Notification of Eximbank action. The person making the request normally will be notified of the availability of the material within 10 working days after the date of receipt of the request. If Eximbank determines to comply in whole or part with a request for records, the information or records shall be made available promptly provided the requirements of subparagraph (vi) regarding payment of fees are satisfied. Any denial of a request in whole or in part shall be made in writing by the Senior Vice President Public Affairs and Export Expansion. The letter shall set forth the reasons for the denial. Any person whose request for information has been denied may appeal from such determination in accordance with § 404.5.

(4) Extension of time. In certain unusual circumstances, as set forth below, the period of time within which Eximbank will respond to an initial request (10 working days) may be extended by an additional 10 working days. A determination that an extension of time to respond to a request is appropriate will be made by the Senior Vice President-Public Affairs and Export Expansion by giving written notice to the requester setting forth the reasons for the extension and the date on which a determination is expected to be made. Unusual circumstances which could necessitate the extension are the following:

(i) the need to search for and collect the requested records from field facilities or other establishments that are separate from the office processing the

request;

 (ii) the need to search for, collect, and appropriately examine a voluminous amount of separate and distinct records which are demanded in a single request;

(iii) the need for consultation, which shall be conducted with all practicable speed, with another agency having a substantial interest in the determination of the request or among two or more components of the agency having substantial subject-matter interest therein.

(5) Fees. A fee will be imposed for Eximbank expenses incurred in searching for, duplicating, tabulating, or compiling the record or information in accordance with the schedule set forth in § 404.6. A letter requesting a document or information should specifically state that all costs involved will be paid or, alternatively, thatt hey will be paid up to a specified limit. If the letter makes no reference to anticipated fees, and the request is expected to involve fees in excess of \$25, or it is estimated by Eximbank that the fee will exceed the dollar limit specified in the request, Eximbank will notify the requester of the estimated fee promptly upon receipt of the request. The request will not be deemed to have been received until Eximbank receives a reply

from the requester stating his willingness to pay the estimated fee.

(6) Method of payment. Where the anticipated fee exceeds \$25 an advance deposit of 25 percent of the anticipated fee or \$25, whichever is greater, will be required within 5 working days of Eximbank's advising the requester. Payment of the balance of the fee will be due before Eximbank furnishes the information, records, or material to the requester. All payments to Eximbank, including the deposit, shall be in the form of cash, check or money order payable to the Export-Import Bank of the United States.

(7) Waiver of fees. The Senior Vice President—Public Affairs and Export Expansion may waive or reduce the fees if such officer determines that such waiver or reduction would be in the public interest because furnishing the information can be considered as primarily benefiting the public.

(8) Deletions. If it is determined that a portion of a record is exempt from disclosure, any reasonably segregable portion of the record will be provided the requester after deletion of the exempt

portions.

(d) Forms for requesting information.
(1) EIB Form 73-5, "Request for Eximbank Forms, Documents, Records and Other Information," may be used for submitting requests to Eximbank.

Other Information, may be used for submitting requests to Eximbank.

(2) FIB Form 73-6, "Answer to Request," is a form letter or notice which may be used to advise the requester of the availability of the material, any charges involved, or the referral of the request to another office or agency.

(3) EIB Form 73-7, "Invoice," will be used to bill the requester for the appropriate fee and a copy thereof will be served as a receipt to the requester.

§ 404.5 Administrative appeal of refusal to disclose.

(a) Who may appeal. Any person whose request for information or records has been denied in whole or in part shall be entitled to submit a written appeal to Eximbank.

(b) Time for appeal. An appeal from a denial may be filed with Eximbank anytime following the date of receipt of the initial determination, in cases of denials of an entire request, or from the date of receipt of any records being made available under an initial determination,

in cases of partial denials.

(c) Form of appeal. An appeal shall be in a letter addressed to the Export-Import Bank of the United States, Attention: Executive Vice President, 811 Vermont Avenue, Washington, D.C. 20571. The envelope and the letter setting forth the appeal shall be clearly marked in capital letters: FREEDOM OF INFOR-MATION ACT APPEAL. The letter shall reasonably describe the information or records requested, the name and title of the Eximbank official or employee who denied the request, and such other pertinent facts and statements as the appellant may deem appropriate. An appeal submitted in an envelope which is not addressed to the Executive Vice President will not be deemed to have

been received until such time as the appeal is forwarded to such officer.

(d) Eximbank decision. Final Eximbank decision on appeals from denials of requests for information or records shall be made in writing by the Executive Vice President within 20 working days after the date of receipt of the request, unless an extension of up to 10 working days has been deemed necessary in accordance with the procedures set forth in § 404.4(c) (iv) above. The 10-day extension may be applied to the response to the initial request or to the appeal, or to both, but in no event shall the extension exceed a total of 10 working days. If the decision upholds the denial of the request, the appellant shall be notified in writing, which notice shall set forth the reasons for upholding the previous denial. The notification shall also refer to the provisions for judicial review of Eximbank's determination. 5 U.S.C. 552. If the Executive Vice President acts favorably on the appeal, the information or records requested shall be made available promptly provided the requirements of subparagraph (vi) regarding payment of fees are satisfied.

## §404.6 Schedule of Fees.

(a) Basis. Factors taken into account in establishing fees for the search and duplication of Eximbank's records and the tabulation or compilation of information on record in Eximbank's files are the direct costs of duplication, the cost of transportation of records, and the salary of employees based on actual time spent for performing search and duplication services and any other services necessary to tabulate or compile the information.

(b) Fees. (1) For any brochure or publication readily available in the Public Affairs Office, there is no charge.

(2) For the transportation of the records to Eximbank offices, the charge will be computed at the rate of \$5 per hour for the messenger service.

(3) For requests requiring a manual search of Eximbank records and duplication by Eximbank employees, there will be a charge of 10 cents per page for duplication and a charge based upon employee time required for the search and duplication. The charge will be computed at the rate of \$4.75 per hour for clerical time and \$9 per hour for supervisory or professional time. The rates of compensation are based on Eximbank's average hourly rate for each of the two classifications of personnel. Clerical employees will normally perform the search and duplication but the time of supervisory or professional personnel may be utilized if necessary to perform the search.

(4) For requests requiring a computer search of Eximbank records, there may be a charge based upon use of the computer and the computer operator's time. The charge will be \$100 per hour. In the event a new program must be prepared for the computer search, there may be an additional charge of \$30 per hour to cover the cost of the programming and testing. These charges will normally be in lieu of the charge for an employee

search but will be in addition to the fee

for any duplication.

(5) While there is no obligation on the part of Eximbank to tabulate or compile records or data, Eximbank will make every effort to comply with reasonable requests for these services, provided such requests do not unduly interfere with normal operations or program activities. The charge for information which has to be tabulated or compiled will include the cost of employee time and cost of computer runs or other equipment use and other overhead expense allocable thereto. The fees for tabulation or compilation services will be fair and equitable in accordance with the federal user charge statute; See 31 U.S.C. 483a.

(6) Persons may inspect and copy specifically requested documents in the Eximbank facilities which are not exempt under § 404.3 without charge except for search, duplication, tabulation, or compilation fees which may otherwise be

payable.

- (7) No fee will be charged in connection with any record which is not made available because it is found to be exempt from disclosure under § 404.3. A fee will be charged, however, for employee time spent attempting to locate any record which, after reasonable search, cannot be located.
- (8) Eximbank will waive or reduce the fees where it determines that such waiver or reduction would be in the public interest because furnishing the information can be considered as primarily benefiting the public.

## § 404.7 Annual Report to Congress.

On March 1 of each calendar year, Eximbank will report to Congress on the administration of the public requests for information and records during the prior calendar year.

§ 404.8 Appearances and testimony by Eximbank officers and employees.

Whenever an officer or employee of Eximbank is served with a subpoena demanding the disclosure of the information or the production of files, docu-ments, and records described in this part, or is requested by court, committee or other body to disclose such information, the officer or employee shall promptly inform his superior of the requirements of the subpoena or request and shall ask for instructions from the Executive Vice President with respect thereto. Such officer or employee shall appear before the court, committee or body and, if the Executive Vice President has not authorized disclosure, the employee shall respectfully decline to disclose the information or produce the files, documents, and records demanded or requested, basing such refusal upon this part.

[FR Doc.75-4213 Filed 2-18-75;8:45 am]

Title 14—Aeronautics and Space
CHAPTER II—CIVIL AERONAUTICS
BOARD

SUBCHAPTER B-PROCEDURAL REGULATIONS

[Reg. PR-142, Amdt. 6]

PART 310—INSPECTION AND COPYING OF BOARD OPINIONS, ORDERS AND RECORDS

Amendment of Part To Reflect Changes to the Board's Schedule of Fees for Document Search and Duplication (Part 389)

Adopted by the Civil Aeronautics Board at its office in Washington, D.C.,

February 12, 1975.

By Notice of Proposed Rule Making PDR-38/ODR-10, dated January 13, 1975, Docket 27378, the Board proposed, inter alia, to amend Part 389 of the Organization Regulations (14 CFR Part 389) to revise its schedule of fees for document searches and duplication, and to amend Part 310 of its Procedural Regulations to make certain editorial changes relating to the proposed revisions of Part 389. For reasons set forth in OR-94, published contemporaneously herewith, the Board has decided to adopt the proposed amendments to Parts 310 and 389.

Since the amendments herein relate only to agency procedure and practice, the Board finds that the amended rule may become effective on less than 30 days' notice, and that the rule should be made effective upon publication in

the Federal Register.

In consideration of the foregoing, the Civil Aeronautics Board hereby amends Part 310 of the Procedural Regulations (14 CFR Part 310) as set forth below, effective February 19, 1975.

1. Amend the table of contents so as to revise the caption of § 310.5, as fol-

lows:

310.5 Fees for locating and copying records.

Revise § 310.5 to read as follows:
 § 310.5 Fees for locating and copying records.

Appropriate fees for locating and copying Board records are prescribed in Part 389 of this chapter.

(Sec. 204(a), Federal Aviation Act of 1958, as amended; 72 Stat. 743, (49 U.S.C. 1324 (a)); Title V of the Act of August 31, 1951, 65 Stat. 290, (5 U.S.C. 140); (5 U.S.C. 552, as amended by Public Law 93-502, 88 Stat. 1561)).

By the Civil Aeronautics Board.

Effective: February 19, 1975.

Adopted: February 12, 1975.

[SEAL] EDWIN Z. HOLLAND,
Secretary,

[FR Doc.75-4335 Filed 2-18-75;8:45 am]

SUBCHAPTER E-ORGANIZATION REGULATIONS

[Regulation OR-93, Amdt. 45]

PART 385—DELEGATIONS AND REVIEW OF ACTION UNDER DELEGATION; NON-HEARING MATTERS

Amendment of Part To Reflect Changes to the Board's Schedule of Fees for Document Search and Duplication (Part 389)

Adopted by the Civil Aeronautics Board at its office in Washington, D.C.,

February 12, 1975.

By Notice of Proposed Rulemaking PDR-38/ODR-10, dated January 13, 1975, Docket 27378, the Board proposed inter alia, to amend Part 389 of the Organization Regulations (14 CFR Part 389) to provide for waivers or modifications of photocopying fees established therein, and to amend Part 385 of its Organization Regulations to delegate to the Managing Director the authority to process requests for such waivers or modifications. For reasons set forth in OR-94, published contemporaneously herewith, the Board has decided to adopt the proposed amendments to Parts 385 and 389.

Since the amendments herein relate only to agency procedure and practice, the Board finds that the amended rule may be made effective on less than 30 days' notice, and that the rule should be made effective upon publication in the Federal Register.

In consideration of the foregoing, the Civil Aeronautics Board hereby amends Part 385 of the Organization Regulations (14 CFR Part 385) as set forth below, effective February 19, 1975.

1. Amend § 385.12, by adding a new paragraph (d), to read as follows:

§ 385.12 Delegation to the Managing Director.

(d) Receive and determine applications for waiver or modification of prescribed fees for search and duplication services when such applications accompany requests for such services pursuant to § 389.14 of this chapter.

(Section 204(a) of the Federal Aviation Act of 1958, as amended; 72 Stat. 743, 49 U.S.C. 1324(a); Title V of the Act of August 31, 1951 (65 Stat. 290, 5 U.S.C. 140); and 5 U.S.C. 552, as amended by Public Law 93-502, 88 Stat. 1561.)

By the Civil Aeronautics Board:

Effective: February 19, 1975.

Adopted: February 12, 1975.

[SEAL] EDWIN Z. HOLLAND, Secretary.

[FR Doc.75-4333 Filed 2-18-75;8:45 am]

[Reg. OR-94, Amdt. 20]/

# PART 389—FEES AND CHARGES FOR SPECIAL SERVICES

## Amendment of Fee Schedule for **Document Search and Duplication**

Adopted by the Civil Aeronautics Board at its office in Washington, D.C.,

February 12, 1975.

By Notice of Proposed Rule Making PDR-38/ODR-10, dated January 13, 1975, Docket 27378, the Board proposed to amend Part 389 of its Organization Regulations (14 CFR Part 389) in several respects and to modify related provisions of Part 310 of the Procedural Regulations and Part 385 of the Organization Regulations (14 CFR Parts 310 and 385) to reflect the proposed changes to Part 389. Part 389 establishes a system of fees for filings, licenses, and special services, including fees for searches for documents and their duplications. In pertinent part, the proposed amend-ments relate to fees for document searches and duplications, and are designed to bring the Board's regulations into conformity with recent amendments made by Pub. L. 93-502 to the Freedom of Information Act (5 U.S.C. 552), Specifically, the proposed amendments would (1) eliminate all existing charges for services performed in connection with searches for documents requested by members of the public: (2) reduce the present fee for photocopying to 15 cents per page; (3) eliminate minimum charges for copying services; (4) provide for waivers or modifications of photocopying fees; and (5) provide for advance payment of photocopying fees in certain prescribed circumstances.

No comments have been filed in re-

sponse to the Board's notice.

The Board has decided to adopt the amendments as proposed for the reasons stated in its notice. Since the amended rule is one of agency procedure and practice, the Board finds that it may be made effective on less than 30 days' notice, and that the rule should be made effective upon publication in the FEDERAL

In consideration of the foregoing, the Civil Aeronautics Board hereby amends Part 389 of the Organization Regulations (14 CFR Part 389), effective February 19, 1975, as set forth below:

1. Amend the table of contents, so as to revise the caption of \$ 389.14 as follows.

389.14 Locating and copying records and documents.

2. Amend § 389.11, by revising paragraph (a), to read as follows:

§ 389.11 Services available. .

.

(a) Locating and copying records and documents. .

. . .

3. Revise § 389.14 to read as follows:

§ 389.14 Locating and copying records and documents.

Public records and documents on file with the Civil Aeronautics Board will be located, and such copies thereof as may be practicable to furnish will be provided, upon request therefor and payment of fees as set forth below:

(a) There shall be no charge in connection with searches for records or doc-

uments under this chapter.

(b) Copies of records or documents are made by Board facilities, or by non-Government contractors.

(1) The fee for photocopying will be at the rate of 15 cents per page.

(2) The fee for copying by Government contractors will be that established in the contracts with the Board and will be billed directly by such contractors.

(c) Where the fee for service requested hereunder will exceed \$100.00, the service will not be performed until payment has been received. In such cases, the requester will be notified promptly of the amount of the fee, and the requested service will be performed as expeditiously as practicable following receipt of such payment.

(d) Applications may be filed requesting waivers or modifications of any fees required to be paid under this section in

accordance with the following:

(1) Each applicant shall set forth briefly and succinctly the relief that it seeks and the reasons why such relief should be granted. Waivers or modifications of stated fees shall be granted only where it is demonstrated that such action is in the public interest because furnishing of the information requested can be considered as primarily benefiting the general public.

(2) Applications requesting waivers or modifications of fees under this section shall be addressed to the Managing Director, who has been delegated authority by the Board to decide such applications in § 385.12 of this chapter, and shall accompany the request for service under

this section.

(3) The Managing Director shall either rule on the application or, at his discretion, pass the matter on to the Board for its determination. In acting upon such applications the Managing Director and the Board, where applica-ble, shall be guided by the procedures and requirements of § 310.9(d) of this

(4) A decision by either the Managing Director or the Board pursuant to paragraph (d) (3) of this section is final and will not be subject to petitions for reconsideration.

(Section 204(a), Federal Aviation Act of 1958, as amended; 72 Stat. 743 (49 U.S.C. 1324(a)); Title V of the Act of August 31, 1951, 65 Stat. ·290 (5 U.S.C. 140); (5 U.S.C. 552, as amended by Public Law 93-502, 88 Stat. 1561)).

By the Civil Aeronautics Board. Effective: FEBRUARY 19, 1975.

Adopted: FEBRUARY 12, 1975.

EDWIN Z. HOLLAND. Secretary.

[FR Doc.75-4334 Filed 2-18-75;8:45 am]

Title 32—National Defense CHAPTER I-OFFICE OF THE SECRETARY OF DEFENSE SUBCHAPTER P-RECORDS

PART 286a—AVAILABILITY OF INFORMATION TO THE PUBLIC

-OFFICE OF THE SECRETARY OF DEFENSE/ORGANIZATION OF THE JOINT CHIEFS OF STAFF IMPLEMENTA-TION OF THE DEPARTMENT OF FENSE FREEDOM OF INFORMATION **PROGRAM** 

The Deputy Secretary of Defense approved Part 297. This Part 297 implements the provisions of 32 CFR Part 286 pursuant to Title 5, United States Code, section 552, as amended by Pub. L. 93-502, assigns responsibilities, and sets forth procedures for the effective administration of the Freedom of Information Program within the Office of the Secretary of Defense including the Organization of the Joint Chiefs of Staff.
Part 286a is superseded and cancelled as noted in § 297.2.

Part 297 is added to read as follows:

Sec.

297.1 Purpose. Cancellation.

Applicability and scope. 297.3

Policy.

Responsibilities and functions. 297.5

297.6 Procedures 297.7 Effective date.

AUTHORITY: The provisions of this Part 297 issued under 5 U.S.C. 552, as amended by Pub. L. 93-502.

# § 297.1 Purpose.

This Part implements the provision of 32 CFR 286 pursuant to Title 5, United States Code, section 552, as amended by Pub. L. 93-502. assigns responsibilities. and sets forth procedures for the effective administration of the Freedom of Information Program within the Office of the Secretary of Defense, including the Organization of the Joint Chiefs of Staff.

# § 297.2 Cancellation.

Part 286a of this subchapter is hereby superseded and cancelled.

# § 297.3 Applicability and scope.

(a) The provisions of this Part apply to the Office of the Secretary of Defense (Director, Defense Research and Engineering, Assistant Secretaries of Defense. Assistants to the Secretary of Defense, or equivalent), the Organization of the Joint Chiefs of Staff, and other activities assigned to OSD for administrative support (hereinafter referred to collectively as "OSD Components").

(b) The provisions of this Part apply to the release of records to the general public and are not applicable to requests from Members of Congress, the General Accounting Office, and other Federal. State and local governmental organizations.

# § 297.4 Policy.

(a) It is the policy of the Department of Defense, consistent with the letter and spirit of Title 5, U.S.C., section 552, the Freedom of Information Act, to make available to the public the maximum amount of information concerning its operations and activities.

(b) Unclassified information, documents and forms which have heretofore been provided to the public as part of normal service will continue to be made available in accordance with previously

established criteria.

(c) Authority to deny an initial request for a record resides in the heads of

OSD Components.

(d) Materials required by the Freedom of Information Act for public inspection will be maintained, published, and made available in a public reading room located in room 2C757, the Penta-

(e) Fees will not be charged for access by the public to the public reading room or to the records provided for inspection therein, but fees in accordance with a schedule published in the FEDERAL REG-ISTER may be charged for furnishing copies.

# § 297.5 Responsibilities and functions.

(a) The Assistant Secretary of Defense (Public Affairs) shall:

(1) Direct and administer the Department of Defense Freedom of Information Program for OSD Components.

(2) Establish standards and procedures to assure compliance with and implementation of title 5, United States Code, section 552, as amended by Pub. L. 93-502 and 32 CFR 286.

(3) Develop and submit the consolidated OSD Component Annual Report as required by title 5, United States Code, section 552, as amended by Pub. L. 93-502 and 32 CFR 286.

(4) Serve as the central contact office for the receipt of all requests for records

from OSD Components.

(5) Forward requests for records from the public to the OSD Component having proprietary responsibility for the record, and maintain records control procedures that insure action within time limits established in title 5, United States Code, section 552, as amended by Pub. L. 93-502

(6) Respond to requesters and collect for deposit with the Assistant Secretary of Defense (Comptroller) fees entailed in the search and reproduction of

records.

(7) Establish a facility where the public may inspect and copy records as provided for in 32 CFR Part 286.

(8) Provide for educational programs on the requirements and implementation of the Freedom of Information Act.

(9) Evaluate and decide, in coordination with the General Counsel, appeals from denial of records by OSD Components-

(10) Maintain data on Freedom of Information activity in a format that meets the requirements of 32 CFR 286.15 and in a manner that affords rapid retrieval.

(b) The Assistant Secretary of De-

fense (Comptroller) shall:

(1) Maintain, publish, and make available for public inspection current OSD Component indices of materials as required by title 5, United States Code, section 552, as amended by Pub. L. 93-502.

(2) Coordinate and arrange for information to be published in the FEDERAL REGISTER pursuant to title 5, United States Code, section 552, as amended by Pub. L. 93-502, 32 CFR 286 and 296.

(c) The General Counsel shall:

(1) Coordinate, as appropriate, with the Department of Justice on all final denials of appeals for requests for records when litigation is likely.

(2) Provide advice and assistance to the ASD(PA) as may be required in the discharge of his appellate responsibilities.

(d) Heads of OSD Components shall: (1) Designate the official authorized to deny initial requests for records and designate an office as the point-of-contact for Freedom of Information matters.

(2) Provide to the ASD(PA), when the request for a record is granted, a copy of the record and a statement of search and reproduction charges.

(3) Provide to the ASD(PA), when a record is denied in whole or in part, the justification for denial.

(4) Provide to the ASD(C) quarterly updates of indices as required by 32 CFR 286.11.

(5) Coordinate with the Office of the General Counsel on all proposed denials of records.

(6) Consult with other OSD and DoD Components which have a significant interest in the requested record prior to a final determination. When consultation with an agency outside DoD is appropriate because of their interest in the requested record, a procedure for effecting that consultation shall be developed in cooperation with the ASD(PA).

(7) Refer cases to the ASD(PA) for review and evaluation when the issues raised are of unusual significance, precedent setting, or otherwise require special

attention or guidance.

(8) Provide the record to the ASD (PA), when the initial denial of such record has been appealed by the requester, or at the time of initial denial when appeal seems likely.

(9) Establish procedures to mark record copies and notify holders of classifled records that have been declassified as a result of review under this program.

(10) Instruct employees, who act in Freedom of Information matters, in the provisions of this part and 32 CFR Part

## § 297.6 Procedures.

(a) General. A request for a record under the Freedom of Information Act may be denied only upon determining

(1) The record is subject to one or more of the exemptions set forth in 32

CFR 286.5, and a significant and legitimate governmental purpose is served by withholding.

(2) The record cannot be found because it has not been described with sufficient particularity to enable a responsible authority to locate it with a reasonable amount of effort.

(3) The requester has unreasonably failed to comply with the procedural requirements imposed by this Part.

(b) Instructions to a requester. A request for a record from an OSD Component can be made in person or in writing.

(1) Indices of current records will be made available for public inspection in Room 2C757, the Pentagon, between 9 a.m. and 4 p.m., except weekends and government holidays. Persons not authorized access to the Pentagon will be escorted to the public reading room and will not be permitted to visit other areas of the building.

(2) Copies of records which are listed in these indices will be provided upon payment of any search and copying fees. A copy of the standard schedule of fees previously published in the FEDERAL REGISTER will be posted in the public reading room. Checks are payable to the

Treasury of the United States.

(3) A written request for a record should be addressed to: Directorate for Freedom of Information, Office of the Assistant Secretary of Defense (Public Affairs), Room 2C757, Pentagon, Washington, D.C. 20301. The record should be described as completely as possible to facilitate its retrieval from the gross file of Defense Department records and to reduce search fees, which may be borne by requester.

(4) A requester will state his intention to pay fees that may be charged for search and reproduction of a record. Time spent to assure willingness to pay fees will be in addition to the time allot-

ted to provide a record.

(5) A record will not be released prior to receipt of fees. Failure to pay fees may be cause not to accept a subsequent request for a record from the same person.

(6) Not more than ten working days after notification that a request has been accepted, normally the requester will be notified that his request has been granted or denied. In unusual circumstances, however, such notification may convey that additional time, not to exceed ten working days, is required to make a determination.

(7) Notification of denial will cite the exemption under which the request was denied, will detail the supporting reasons, will identify the denial authority, and will set forth the course of action to be followed for an administrative appeal.

(8) Administrative appeals of initial denials should be addressed to: Assistant Secretary of Defense (Public Affairs), ATTN: Freedom of Information Appeal Board, Pentagon, Washington, D.C. 20301. Such an appeal should offer justification for a reversal of the initial denial.

(9) Not more than twenty working days after notification that an administrative appeal has been accepted, normally the requester will be notified that his appeal has been sustained or denied. In unusual circumstances, however, such notification may convey that additional time, not used in an extension of time to repsond to the initial request and not to exceed ten working days, is required to make a determination.

(10) Notification of a final denial will cite supporting reasons, will set forth a course of action that may be followed to seek judicial relief, and will be signed by the official exercising final denial au-

thority.

(c) Instructions to OSD components. The Directorate for Freedom of Information, Office of the Assistant Secretary of Defense (Public Affairs, hereinafter referred to as the Directorate, will be the office of record and point of contact for OSD Components in matters pertaining to the Freedom of Information Act.

(1) The Directorate will deliver a request for a record to the OSD Component having proprietary responsibility. Redirect without delay any request for a record that originates from a source other

than the Directorate.

(2) A request for a record will be delivered to the office of contact, together with a letter of instruction and a cover sheet. The cover sheet, of distinctive design and overprinted with instructions vital to Freedom of Information requests. will not be separated from the request packet to which it is attached.

(3) A request for a record from the Directorate may not be declined, except, when proprietary responsibility is challenged, it may be returned by a memorandum signed by the denial authority which fixes the proprietary responsibility with a specific OSD Component or other agency of the government. A lack of specificity by the requester that suggests records of other components or agencies may also fit the broad category of the request, will not be sufficient cause to chal-

lenge proprietary responsibility. (4) A suspense will be assigned to each request that must be met by either a decision to grant the request in whole or in part, a decision to deny, or by a petition from the denial authority stating the specific number of additional days required and the reason, provided such petition is made to the Directorate three working days prior to the suspense. Extension of time will be granted on a case by case basis depending upon the ability to justify "unusual circumstances," and will normally not exceed

five working days.

(5) When a request is granted a copy of the record and a completed statement of fees will be delivered to the Directorate. When a copy of the record can-not be made available in the time allotted, a statement explaining the delay and establishing the date of its availability will accompany the statement of fees. In every case where a request for a record is granted, a statement of fees will be provided in the time allotted.

(6) When a request is denied in part, an excised copy of the record, a statement of fees, and a letter explaining the reasons for denial and summarizing the coordination/consultation history, will

be delivered to the Directorate. When denial is based on classification in the interest of national defense or foreign policy, the explanation shall convey that a current review of the record supports continued classification according to the criteria and rationale of DoD Regulation 5200.1-R 1 for specifically stated reasons that logically support those specific criteria and rationale in the context of the record. Letters of denial will be signed by the denial authority.

(7) When a record is denied in its entirety, all procedures set out in subparagraph (6) of this paragraph apply except, a statement of fees and a copy of

the record are not provided.

(8) Cases referred to the Directorate in accordance with § 297.5(d) (7), will be transmitted by memorandum establishing the unusual or precedent setting conditions. Unless notified to the contrary, Public Affairs review and evaluation action will not relieve the OSD Component of any responsibility to reach an initial decision in the time originally allotted. Public Affairs action will, however, be accomplished so as to provide timely recommendation and guidance.

(9) Components are urged to use the direct lines of communication that have been established by the formation of OSD Component offices to focus on Freedom of Information matters and the Directorate for Freedom of Information. Early in the time allotted, any problems associated with a request that contact with the requester or agencies outside the DoD might resolve, should be presented to the Staff Director, Directorate for Freedom of Information. Direct contact between OSD Components and requesters, or between OSD Components and agencies outside the DoD, is not au-

§ 297.7 Effective date.

This part is effective February 19, 1975.

MAURICE W. ROCHE, Director Correspondence and Directives OASD (Comptroller).

FEBRUARY 12, 1975.

[FR Doc.75-4299 Filed 2-18-75;8:45 am]

Title 14—Aeronautics and Space

CHAPTER V-NATIONAL AERONAUTICS AND SPACE ADMINISTRATION

PART 1206-AVAILABILITY OF AGENCY RECORDS TO MEMBERS OF THE PUBLIC

Freedom of Information Regulations

The purpose of this revision of Part 1206 of the National Aeronautics and Space Administration Regulations is to implement the Freedom of Information Act, 5 U.S.C. 552, as amended on November 21, 1974, by Pub. L. 93-502.

Subpart 7 of this Part 1206 is the same, except for minor editorial changes, as the proposed Subpart 7 which was published as a Notice of Proposed Rulemaking in the FEDERAL REGISTER, Vol. 40, No. 10, January 15, 1975, and on which writ-

ten comments were solicited. Comments received on the Proposed Rulemaking have been considered.

Part 1206 of Title 14 of the Code of Federal Regulations is revised in its entirety as follows:

Subpart 1—Basic Policy

1206.100 Scope of part. 1206,101 Definitions. 1206.102 General Policy. Subpart 2-Records Available

1206.200 Types of records to be made avail-1206.201 Records which have been pub-

lished. Deletion of segregable portions of 1206.202 a record.

1206 203 Creation of records.

1206.204 Records of interest to other agencies.

1206 205 Incorporation by reference. 1206.206 Availability for copying. 1206.207 Copies.

1206,208 Release of exempt records.

Subpart 3—Exemptions

1206.300 Exemptions. 1206.301 Limitation of exemptions.

Subpart 4 Location for Inspection and Request of Agency Records

1206.400 Information centers. 1206:401 Location of NASA information centers.

Documents available for inspec-1206.402 tion at NASA information centers

1206.403 Duty hours.

Subpart 5-Responsibilities 1206.500 Assistant Administrator for Public

Affairs. 1206.501 General Counsel. 1206.502 Field and component installations. 1206 503

NASA Headquarters 1206.504 Delegation of authority.

Subpart 6-Procedures

1206.600 Requests for records. 1206.601 Mail requests. 1206,602 Requests in person.

Procedures and time limits for 1206.603 initial determinations.

1206.604 Request for records which exist elsewhere.

1206.605 Appeals. 1206.606 Request for additional records. 1206.607

Action on appeals.

Time extensions in unusual cir-1206.608 cumstances.

1206.609 Litigation.

Subpart 7—Search and Duplication Fees

1206,700 Schedule of fees. 1206.701 Advance estimate or deposit before incurring costs.

1206.702 Waiver of fees. Certification. 1206.703 1206.704 Form of payment.

Subpart 8—Failure To Release Records to the

1206,800 Failure to release records to the public.

Subpart 9-Annual Report

1206.900 Requirements for annual report.

AUTHORITY: Sec. 203, National Aeronautics and Space Act of 1955, as amended, 72 Stat. 429, 42 U.S.C. 2473 and 5 U.S.C. 552 as amended by Pub. L. 93-502, 88 Stat. 1561.

# Subpart 1—Basic Policy

# § 1206.100 Scope of part.

This Part 1206 establishes the policies, responsibilities, and procedures for the release to members of the public of

<sup>1</sup> Filed as part of original.

agency records which are under the jurisdiction of the National Aeronautics and Space Administration. This part applies to information and agency records located at NASA Headquarters, at NASA Field Installations, and at NASA component installations, as defined in Part 1201 of this chapter.

# § 1206.101 Definitions.

For the purposes of this part, the fol-

lowing definitions shall apply:
(a) The term "agency records" or "records" includes all books, papers, maps, photographs, or other documentary materials, regardless of physical form or characteristics, made or received by NASA in pursuance of Federal law or in connection with the transaction of public business and preserved by NASA as evidence of the organization, functions, policies, decisions, procedures, operations, or other activities or because of the informational value of data contained therein. It does not include tangible objects or articles, such as structures, furniture, paintings, sculptures, exhibits, models, vehicles or equipment; library or museum material made or acquired and preserved solely for reference or exhibition purposes; or records of another agency, a copy of which may be in NASA's possession.
(b) The term "initial determination"

means a decision by a NASA official, in response to a request by a member of the public for an agency record, on whether the record described in the request can be identified and located after a reasonable search and, if so, whether the record (or portions thereof) will be made available under this part or will be withheld from disclosure under Subpart 3

of this part.

(c) The term "appeal" means a request by a member of the public to the Administrator or his designee for reversal of any adverse initial determination the requester has received in response to a request for an agency record.

(d) The term "final determination"

means a decision by the Administrator

or his designee on an appeal.

(e) The term "working days" means all days except Saturdays, Sundays and

legal public holidays.

(f) As used in § 1206.608, the term "unusual circumstances" means, but only to the extent reasonably necessary to the proper processing of a particular request for agency records-

(1) The need to search for and collect the requested records from field facilities or other establishments that are separate from the NASA Information Center processing the request (see Sub-part 6 of this part for procedures for processing a request for agency records);

(2) The need to search for, collect, and appropriately examine a voluminous amount of separate and distinct records which are demanded in a single request;

(3) The need for consultation, which shall be conducted with all practicable speed, with another agency having a substantial interest in the determination of the request or among two or more

components of NASA having substantial subject-matter interest therein.

# § 1206.102 General policy.

(a) In accordance with section 203 (a) (3) of the National Aeronautics and Space Act of 1958 (42 U.S.C. 2473(a) (3)), it has been and continues to be NASA policy to provide for the "widest practicable and appropriate dissemination of information concerning its activities and the results thereof.'

(b) In compliance with the "Freedom of Information" amendments to the Administrative Procedure Act (5 U.S.C. 552, as amended by Public Laws 90-23 and 93-502), a positive and continuing obligation exists for NASA to make available upon request by members of the public to the fullest extent practicable all agency records under its jurisdiction, as described in Subpart 2 of this part, except to the extent that they may be exempt from disclosure under Subpart 3 of this

## Subpart 2-Records Available

§ 1206.200 Types of records to be made available.

(a) Records required to be published in the FEDERAL REGISTER. The following records are required to be published in the FEDERAL REGISTER, for codification in Title 14, Chapter V, of the Code of Federal Regulations.

(1) Description of NASA Headquarters and field organization and the established places at which, the employees from whom, and the methods whereby, the public may secure information, make submittals or requests, or obtain decisions:

(2) Statements of the general course and method by which NASA's functions are channeled and determined, including the nature and requirements of all formal and informal procedures avail-

(3) Rules of procedure, descriptions of forms available or the places at which forms may be obtained, and instructions as to the scope and contents of all papers, reports, or examinations;

(4) Substantive rules of general applicability adopted as authorized by law, and statements of general policy or interpretations of general applicability formulated and adopted by NASA;

(5) Each amendment, revision, or repeal of the foregoing.

(b) Agency opinions, orders, state-ments, and manuals. (1) Unless they are exempt from disclosure under Subpart 3 of this part, or unless they are promptly published and copies offered for sale, NASA shall make available the following records for public inspection and copying or purchase:

(i) All final opinions (including concurring and dissenting opinions) and all orders made in the adjudication of cases, such as opinions of the NASA Board of Contract Appeals;

(ii) Those statements of NASA policy and interpretations which have been adopted by NASA and are not published in the FEDERAL REGISTER:

(iii) Administrative staff manuals (or similar issuances) and instructions to staff that affect a member of the public.

(2) In connection with all records required to be made available or published under this paragraph (b), identifying details shall be deleted to the extent required to prevent a clearly unwarranted invasion of personal privacy. However, in each case the justification for the deletion shall be explained fully in writing. A copy of such justification shall be attached to the front of the portion of the record made available to the requester.

(3) In connection with all records required to be made available or published under this paragraph (b) which are issued, adopted, or promulgated after July 4, 1967, except to the extent they are exempt from disclosure under Subpart 3 of this part, current indexes providing identifying information will be maintained and made available for public inspection and copying or purchase

(see § 1206.402).

(c) Other agency records. In addition to the records made available or published under paragraphs (a) and (b) of this § 1206.200, NASA shall, upon request for other records made in accordance with this part, make such records promptly available to any person, unless they are exempt from disclosure under subpart 3 of this part, or unless they may be purchased from other readily available sources, as provided in § 1206.201.

§ 1206.201 Records which have been published.

Publication in the FEDERAL REGISTER is a means of making certain agency records available to the public. Also, the Commerce Business Daily, Synopsis of U.S. Government Proposed Procurement. Sales and Contract Awards (Department of Commerce) is a source of information concerning agency records or actions. Various other NASA publications and documents, and indexes thereto, are available from other sources, such as the U.S. Superintendent of Documents, the National Technical Information Service (Department of Commerce), and the Earth Resources Observation Systems (EROS) Data Center (Department of the Interior). Such publications and documents are not required to be made available or reproduced in response to a request therefor unless they cannot be purchased readily from available sources. If a publication or document is readily available from a source other than NASA. the requester shall be informed of the procedures to follow to obtain the publication or document.

§ 1206.202 Deletion of segregable portions of a record.

If a record requested by a member of the public contains both information required to be made available and that which is exempt from disclosure under Subpart 3 of this part, and the portion of the record that is required to be made available is reasonably segregable from the portion that is exempt, the portion that is exempt from disclosure shall be deleted and the balance of the record shall be made available to the requester. If the non-exempt portion of the record appears to be unintelligible or uninformative, the requester shall be informed of that fact, and such non-exempt portion shall not be sent to the requester unless he thereafter specifically requests it.

# § 1206.203 Creation of records.

Records will not be created by compiling selected items from the files at the request of a member of the public, nor will records be created to provide the requester with such data as ratios, proportions, percentages, frequency distributions, trends, correlations, or comparisons.

## § 1206.204 Records of interest to other agencies.

If a NASA record is requested and another agency has a substantial interest in the record, such an agency shall be consulted on whether the record shall be made available under this part (see § 1206.101(f)(3)). If a record is requested that is a record of another agency, the request shall be returned to the requester, as provided in § 1206.-604(c).

# §-1206.205 Incorporation by reference.

Matter which is reasonably available to the members of the public affected thereby shall be deemed published in the FEDERAL REGISTER when incorporated by reference in material published in the FEDERAL REGISTER (pursuant to the FED-ERAL REGISTER regulation on incorporation by reference, 1 CFR Part 51).

#### § 1206.206 Availability for copying.

Except as provided in § 1206.201, the availability of a record for inspection shall include the opportunity to extract information therefrom or to purchase copies.

# § 1206.207 Copies.

The furnishing of a single copy of the requested record will constitute compliance with this part.

# § 1206.208 Release of exempt records.

If a record which has been requested is exempt from disclosure under Subpart 3 of this part, the record may nevertheless be made available under the procedures of Subpart 6 of this part if it is determined by an official authorized to make either an initial determination or a final determination that such action would not be inconsistent with a purpose of the exemptions set forth in Subpart 3 of this

# Subpart 3—Exemptions

# § 1206.300 Exemptions.

(a) Under 5 U.S.C. 552(b) agency records falling within the exemptions of paragraph (b) of this section are not required to be made available under this part. Such records may nevertheless be made available if it is determined that such actions would not be inconsistent with a purpose of the exemption (see § 1206.208).

(b) The requirements of this part to make agency records available do not apply to matters that are-

(1) (i) Specifically authorized under criteria established by an Executive order to be kept secret in the interest of national defense or foreign policy and (ii) are in fact properly classified pursuant to such Executive order:

(2) Related solely to the internal personnel rules and practices of NASA;
(3) Specifically exempted from dis-

closure by statute;

(4) Trade secrets and commercial or financial information obtained from a person and privileged or confidential;

(5) Interagency or intraagency memorandums or letters which would not be available by law to a party other than an agency in litigation with NASA;

(6) Personnel and medical files and similar files the disclosure of which would constitute a clearly unwarranted inva-

sion of personal privacy;

(7) Investigatory records compiled for law enforcement purposes, but only to the extent that the production of such records would (i) interfere with enforcement proceedings, (ii) deprive a person of a right to a fair trial or an impartial adjudication, (iii) constitute an unwarranted invasion of personal privacy, (iv) disclose the identity of a confidential source and, in the case of a record compiled by a criminal law enforcement authority in the course of a criminal investigation, or by an agency conducting a lawful national security intelligence investigation, confidential information furnished only by the confidential source, (v) disclose investigative techniques and procedures, or (vi) endanger the life or physical safety of law enforcement personnel.

(8) Contained in or related to examination, operating, or condition reports prepared by, on behalf of, or for the use of an agency responsible for the regulation or supervision of financial institu-

tions; or

(9) Geological and geophysical information and data, including maps, concerning wells.

## § 1206.301 Limitation of exemptions.

(a) This Part 1206 does not authorize the withholding of information or the availability of records to the public, except as specifically stated in this part.

(b) Nothing in this Part 1206 shall be construed as authority to withhold information from Congress.

## Subpart 4—Location for Inspection and **Request of Agency Records**

# § 1206.400 Information centers.

NASA will maintain Information Centers as set forth in this subpart.

## § 1206.401 Location of NASA information centers.

NASA will maintain the following Information Centers, at which agency records may be inspected, from which copies of agency records may be requested and at which copies of agency forms may be obtained:

(a) NASA Headquarters Information Center, National Aeronautics and Space Administration, Washington, DC 20546.

(b) NASA Information Center, Ames

Research Center, Moffett Field, CA 94035.

(c) NASA Information Center, Flight Research Center, Post Office Box 273, Edwards, CA 93523.

NASA Information Center. Goddard Space Flight Center, Greenbelt,

MD 20771.

(e) NASA Information Center, John F. Kennedy Space Center, Kennedy Space Center, FL 32899.

(f) NASA Information Center, Langley Research Center, Hampton, VA 23365. Langley Station,

(g) NASA Information Center, Lewis Research Center, 21000 Brookpart Road, Cleveland, OH 44135.

(h) NASA Information Center, Lyndon B. Johnson Space Center, Houston, TX 77058.

(i) NASA Information Center, George C. Marshall Space Flight Center, Huntsville, AL 35812.

(j) NASA Information Center, Michoud Assembly Facility, Post Office Box 29300, New Orleans, LA 70126.

(k) NASA Information Center, National Space Technology Laboratories, Bay St. Louis, MS 39520.

(1) NASA Information Center, Pasadena Office (JPL), 4800 Oak Grove Drive, Pasadena, CA 91103.

(m) NASA Information Center, Wallops Flight Center, Wallops Island, VA

# § 1206.402 Documents available for inspection at NASA information cen-

(a) Each NASA Information Center will have available for inspection, as a minimum, a current version of the following documents:

(1) The Freedom of Information Act (5 U.S.C. 552, as amended by Pub. L. 90-

23 and Pub. L. 93–502);
(2) Title 14, Chapter V, and Title 41. Chapter 18, Code of Federal Regulations, and material published in the FEDERAL REGISTER for codification but not yet included in the Code of Federal Regulations:

(3) A master list and index of NASA Issuances, and a copy of all such Is-

suances;

(4) A list and index of the management issuances of the NASA installation at which the Information Center is located, and a copy of such issuances;

(5) NASA's Scientific and Technical Aerospace Reports (STAR) and current

indexes thereto;

(6) Cumulative Index to Selected Speeches and News Releases issued by NASA Headquarters:

(7) Index/Digest of Decisions, NASA

Board of Contract Appeals;
(8) Decisions of the NASA Contract Adjustment Board and a current index thereto:

(9) NASA Handbook NHB 5500.1A containing an index to decisions of the NASA Inventions and Contributions Board on Petitions for Patent Waivers;

(10) Copies of Environmental Impact Statements filed by NASA under the National Environmental Policy Act of 1969;

(11) Collection of all issues of "NASA Activities"; and

(12) List of licenses granted under

NASA-owned patents.

(b) Because the indexes listed in paragraph (a) of this section are voluminous and because current versions thereof will be available for inspection at NASA Information Centers, from which copies of the indexes may be requested under § 1206.603, it is determined and so ordered that publication of the indexes quarterly in the Federal Register would be unnecessary and impractical.

# § 1206.403 Duty hours.

The NASA Information Centers listed in § 1206.401 shall be open to the public during all regular workdays, from 9:00 a.m. to 4:00 p.m.

# Subpart 5—Responsibilities

§ 1206.500 Assistant Administrator for Public Affairs.

The Assistant Administrator for Public Affairs is responsible for the following:

(a) Providing overall supervision and coordination of the implementation of the policies and procedures set forth

in this Part 1206;
(b) After consultation with the General Counsel, making final determinations under § 1206.607, within the time

limits specified in Subpart 6 of this

(c) Providing consultation on whether unusual circumstances exist under § 1206.608 as would justify the extension of the time limit for an initial determination:

(d) Determining whether unusual circumstances exist under § 1206.608 as would justify the extension of the time limit for a final determination; and

(e) Preparing the annual reports required by § 1206.900, including establishing reporting procedures throughout NASA to facilitate the preparation of such report.

# § 1206.501 General Counsel.

The General Counsel is responsible for the interpretation of 5 U.S.C. 552 and this part, and for the handling of litigation in connection with a request for an agency record under this part.

# § 1206.502 Field and component in-

(a) The Director of each NASA Field Installation or the Official-in-Charge of each Component Installation is respon-

sible for the following:

- (1) After consultation with the Chief Counsel or the Counsel charged with providing legal advice to a Field or a Component Installation, making initial determinations under §§ 1206.603 and 1206.604:
- (2) After consultation with the Assistant Administrator for Public Affairs, determining whether unusual circumstances exist under § 1206.608 as would justify the extension of the time limit for an initial determination; and

(3) Insuring that requests for agency records under the cognizance of his respective installation are processed and initial determinations made within the time limits specified in Subpart 6 of this part.

(b) If so designated by the Director or Official-in-Charge of the respective installation, the principal Public Affairs Officer at the installation may perform the functions set forth in paragraphs (a) (1) and (2) of this section.

# § 1206.503 NASA Headquarters.

(a) The Director of Headquarters Administration is responsible for the following:

(1) After consultation with the Office of General Counsel, making initial determinations under \$\ \frac{1}{2} \ 1206.603 \ \text{and}

1206.604:

(2) After consultation with the Assistant Administrator for Public Affairs, determining whether unusual circumstances exist under § 1206.608 as would justify the extension of the time limit for an initial determination; and

(3) Insuring that requests for agency records under the cognizance of Head-quarters are processed and initial determinations made within the time limits specified in Subpart 6 of this part.

(b) The functions set forth in paragraphs (a) (1) and (2) of this section may be delegated by the Director of Headquarters Administration to a Public Affairs Officer designated by the Assistant Administrator for Public Affairs.

# § 1206.504 Delegation of authority.

Authority necessary to carry out the responsibilities specified in this subpart is delegated to the officials named in this subpart.

# Subpart 6-Procedures

# § 1206.600 Requests for records.

A member of the public may request an agency record by mail or in person from the Information Center having cognizance over the record requested or from the NASA Headquarters Information Center.

# § 1206.601 Mail requests.

In view of the time limits under 5 U.S.C. 552(a) (6) for an initial determination on a request for an agency record (see § 1206.603), a request by mail must meet the following requirements:

(a) The request must be addressed to an appropriate NASA Information Center (see § 1206.401) or otherwise be clearly identified on the envelope and in the letter as a request for an agency record under the "Freedom of Information Act".

(b) The request must identify the record requested or reasonably describe it in such a manner as to enable a professional NASA employee who is familiar with the subject area of the request to identify and locate the record with a reasonable amount of effort. NASA need not comply with a blanket or categorical request (such as "all matters relating to"

a general subject) where it is not feasible reasonably to determine what is sought. NASA will in good faith endeavor to identify and locate the record sought and will consult with the requester when necessary and appropriate for that purpose. However, as provided in § 1206.203, NASA will undertake no obligation to compile or create information or records not already in existence at the time of the request.

(c) If a fee is chargeable under Subpart 7 of this part for search or duplication costs incurred in connection with a request for an agency record, and the requester knows the amount of the fee at the time of the request, the request should be accompanied by a check or money order payable in that amount to "Treasury of the United States." Cash or stamps should not be sent by mail. If the amount of the fee chargeable is not known at the time of the request, the requester will be notified in the initial determination (or in a final determination) of the amount of the fee chargeable (see § 1206.603(c)). For circumstances in which an advance estimate of fees shall be provided to the requester, see § 1206.701.

#### § 1206.602 Requests in person.

(a) A member of the public may request an agency record in person at a NASA. Information Center (see § 1206.401) during the duty hours of the center.

(b) A request at an Information Center must identify the record requested or reasonably describe it as provided in

1206.601(b).

(c) If the record requested is located at the Information Center or otherwise readily obtainable, it shall be made available to the requester upon the payment of any fees that are chargeable (see Subpart 7 of this part), which fees may be paid in cash or by a check or money order payable to the "Treasury of the United States." If the record requested is not located at the Information Center or otherwise readily obtainable, the request will be docketed at the Information Center and processed in accordance with the procedures of \$\frac{8}{1206.603}\$ and 1206.604, with any fee chargeable being handled in accordance with \frac{1}{206.601}\$

# § 1206.603 Procedures and time limits for initial determinations.

(a) Except as provided in § 1206.608, an initial determination on a request for an agency record, addressed in accordance with § 1206.601(a) or made in person at a NASA Information Center, shall be made, and the requester shall be sent notification thereof, within ten working days after receipt of the request, as required by 5 U.S.C. 552(a) (6).

(b) An initial determination on a request for an agency record by mail not addressed in accordance with § 1206.601 (a) shall be made, and the requester shall be sent notification thereof, within ten working days after the correspondence is recognized as a request for an agency record under the "Freedom of In-

formation Act" and received by the appropriate NASA Information Center. With respect to such a request, unless an initial determination can reasonably be made within ten working days of the original receipt, the request will be promptly acknowledged and the requester notified of the date the request was received at that Information Center and that an initial determination on the request will be made within ten working days of that date.

(c) If it is determined that the requested record (or portions thereof) will be made available and the fee chargeable is \$25.00 or less, a copy of the record requested (or portions thereof) shall be sent to the requester with the initial determination or, if that is not feasible, promptly after the initial determination is made. If the fee chargeable exceeds \$25.00 (unless the requester has indicated a willingness to pay a higher fee), the requester will be informed of the fee in the initial determination, and the requested record will be sent to the requester promptly upon receipt of the required fee.

(d) Any notification of an initial determination that does not comply fully with the request for an agency record shall include a statement of the reasons for the adverse determination, include the name and title of the person making the initial determination, and notify the requester of his right to appeal to the Administrator under § 1206.605.

# § 1206.604 Request for records which exist elsewhere.

(a) If a request for an agency record is received by an Information Center not having cognizance of the record (for example, where a request is submitted to one NASA installation or Headquarters and the requested record exists only at another NASA installation), the Information Center receiving the request shall promptly forward it to the NASA Information Center having cognizance of the record requested. That center shall acknowledge the request and inform the requester that an initial determination on the request will be sent within ten working days from the date of receipt by such center.

(b) If a request is received for agency records which exist at two or more centers, the center receiving the request shall undertake to comply with the request, if feasible, or to forward the request (or portions thereof) promptly to a more appropriate center for processing. The requester shall be kept informed of the actions taken to respond to the request.

(c) If a request is received by a NASA Information Center for a record of another agency, the requester shall promptly be informed of that fact, and the request shall be returned to the requester, with advice as to where the request should be directed.

# § 1206.605 Appeals.

(a) A member of the public who has requested an agency record in accordance with §§ 1206.601 or 1206.602, and who has received an initial determination which does not comply fully with

the request, may appeal such an adverse initial determination to the Administrator under the procedures of this section.

(b) The appeal must:

(1) Be addressed to the Administrator, NASA, Washington, D.C. 20546:

(2) Be identified clearly on the envelope and in the letter as an "Appeal under the Freedom of Information Act";

(3) Include a copy of the request for the agency record and a copy of the ad-

verse initial determination;

(4) To the extent possible, state the reasons why the requester believes the adverse initial determination should be reversed; and

(5) Be sent to the Administrator within 30 calendar days of the date of receipt of the initial determination.

(c) An official authorized to make a final determination may waive any of the requirements of paragraph (b) of this section, in which case the time limit for the final determination (see § 1206.607 (a)) shall run from the date of such waiver.

# § 1206.606 Request for additional records.

If, upon receipt of a record (or portions thereof) following an initial determination to comply with a request, the requester believes that the materials received do not comply with the request, he may elect either to request additional records under the procedures of §§ 1206.601 or 1206.602, or to file an appeal under the procedures of § 1206.605, in which case the appeal must be sent to the Administrator within 30 days of receipt of the record (or portions thereof), unless good cause is shown for any additional delay.

# § 1206.607 Actions on appeals.

(a) Except as provided in § 1206.608, the Administrator or his designee shall make a final determination on an appeal, and notify the requester thereof, within twenty working days after receipt of the appeal.

(b) If the final determination reverses in whole or in part the initial determination, the record requested (or portions thereof) shall be made available promptly to the requester, as provided

in the final determination.

(c) If the final determination sustains in whole or in part an adverse initial determination, the notification of the final determination shall:

(1) Explain the basis on which the record (or portions thereof) will not be made available;

(2) Include the name and title of the person making the final determination;

(3) Include a statement that the final determination is subject to judicial review under 5 U.S.C. 552(a) (4); and

(4) Enclose a copy of 5 U.S.C. 552(a)

# § 1206.608 Time extensions in unusual circumstances.

(a) In "unusual circumstances" as that term is defined in § 1206.101(f), the time limits for an initial determination (see §§ 1206.603 and 1206.604) and for a final determination (see § 1206.607) may

be extended, but not to exceed a total of ten working days in the aggregate in the processing of any specific request for an agency record.

(b) If an extension of time under this section would be required, the requester shall be promptly notified of the reasons therefor and the date when a determina-

tion will be sent.

(c) If a record described in a request cannot be located within the ten-working-day time limit for an initial determination, after consultation with a professional NASA employee who is familiar with the subject area of the request, that fact normally will justify an initial determination that the record requested cannot be identified or located, rather than a decision that an extension of time under this section would be appropriate.

(d) In exceptional circumstances, if it would be impossible to complete a search for or review of agency records within the ten-working-day period for an initial determination, an official authorized to make an initial determination or his designee may seek an extension of time from the requester. If such an extension of time can be agreed upon, that fact should be clearly documented and the initial determination made within the extended time period; if not, an initial determination that the record cannot be identified or located, or reviewed, within the ten-working-day time limit shall be made under § 1206.603.

#### § 1206.609 Litigation.

In any instance in which NASA is sued in connection with a request for an agency record under this part, the matter shall promptly be referred to the General Counsel together with a report on the details and status of the request. In such a case, if a final determination with respect to the request has not been made, such a determination shall be made as soon as possible, under procedures prescribed by the General Counsel in each case.

# Subpart 7—Search and Duplication Fees § 1206.700 Schedule of fees.

The fees specified in this section shall be charged for searching for or duplicating an agency record made available in response to a request under this part.

(a) Copies. For copies of documents, such as letters, memoranda, statements, reports, contracts, etc., \$0.10 per copy of each page. For copies of oversize documents, such as maps, charts, blueprints, etc., \$0.15 for each reproduced copy per square foot. These charges for copies include the time spent in duplicating the documents.

(b) Clerical searches. For each one quarter hour spent by clerical personnel in searching for an agency record in response to a request under this part. \$1.00.

(c) Nonroutine, nonclerical searches. Where a search cannot be performed by clerical personnel, for example, where the task of determining which records fall within a request and collecting them requires the time of professional or managerial personnel, and where the amount of time that must be expended in the search and collection of the requested

records by such higher level personnel is substantial, charges for the search may be made at a rate in excess of the clerical rate, namely for each one quarter hour spent by such higher level personnel in searching for a requested record,

(d) Computerized records. Because of the diversity in the types and configurations of computers which may be required in responding to requests for agency records maintained in whole or in part in computerized form, it is not feasible to establish a uniform schedule of fees for search and printout of such records. In most instances, records maintained in computer data banks are available also in printed form and the standard fees specified in paragraph (a) of this section shall apply. If the request for an agency record required to be made available under this part requires a computerized search or printout, the charge for the time of personnel involved shall be at the rates specified in paragraphs (b) and (c) of this section. The charge for the computer time involved and for any special supplies or materials used, such as magnetic tape or Hollerith cards, shall not exceed the direct cost to NASA. Before any computer search or printout is undertaken in response to a request for an agency record, the requester shall be notified of the applicable unit costs involved and the total estimated cost of the search or printout.

(e) Other search and duplication costs. Reasonable standard fees, other than as specified in paragraphs (a) through of this section, may be charged for additional direct costs incurred in searching for or duplicating an agency record in response to a request under this part. Charges which may be made under this paragraph include, but are not limited to, the transportation of NASA personnel to places of record storage for search purposes or freight charges for transporting records to the personnel searching for or duplicating a requested

record.

Unsuccessful or unproductive (f) searches. Search charges, as set forth in paragraph (b) and (c) of this section, may be made even when an agency record which has been requested cannot be identified or located after a diligent search and consultation with a professional NASA employee familiar with the subject area of the request, or if located, cannot be made available under Subpart 3 of this part. Ordinarily, however, fees will not be charged in such instances unless they are substantial and the requester has consented to the search after having been advised that it cannot be determined in advance whether any records exist which can be made available (see § 1206.701) and that search fees will be charged even if no record can be located and made available.

(g) Examination and related tasks in screening records. No charge shall be made for the time spent in resolving legal or policy issues in connection with a request for an agency record under this part, in examining records to determine

whether they will be made available to the requester, or in monitoring the inspection of requested records by a member of the public.

#### § 1206.701 Advance estimate or deposit before incurring costs.

(a) In the circumstance specified in subparagraphs (1) through (4) of this paragraph, before NASA undertakes to search for an agency record and thereby subjects the requester to a fee under this subpart, NASA will promptly notify the requester of the estimated fees chargeable for searching for and duplicating the requested record. Such an advance estimate shall be provided in the following circumstances:

(1) If the requester specifically requests an estimate;

(2) If the search involves computerized records (see § 1206.700(d));

(3) If the search and duplication would involve transportation costs (see § 1206.700(e)); or

(4) If the estimated cost of the search for an agency record exceeds \$25.00 and it cannot be determined in advance if the record requested can be identified or located and made available to the re-

(b) In appropriate cases, an advance deposit may be required before NASA will undertake to search for and duplicate an agency record.

(c) A notice of estimated fees or request for advance deposit shall also extend an offer to the requester to confer with knowledgeable NASA personnel in an attempt to reformulate the request in a manner which will reduce the fees and meet the needs of the requester.

(d) If an estimate is provided under this section, the request for the record shall not be deemed to have been received until the requester agrees to bear the estimated cost or, if an advance deposit is requested, provides such a deposit.

# § 1206.702 Waiver of fees.

(a) The NASA official making an initial or final determination on a request for an agency record may waive all or part of a fee which would be chargeable under this subpart if the official determines that such action is in the public interest because making the record available can be considered as primarily benefiting the general public.

(b) Ordinarily, fees will not be charged where they would amount in the aggregate, for a request or series of related requests, to less than \$3.00.

# 1206.703 Certification.

In accordance with the provisions of 31 U.S.C. 483a, a charge of \$1.00 shall be made for each certification of true copies of agency records.

# § 1206.704 Form of payment.

Payment by mail shall be made by check or money order payable to the "Treasury of the United States" and sent to the NASA Information Center which processed the request.

Subpart 8-Failure To Release Records to the Public

## \$ 1206.800 Failure to release records to the public.

(a) Except to the extent that a person has actual and timely notice of the terms thereof, a person may not in any manner be required to resort to, or be adversely affected by, a matter required to be published in the FEDERAL REGISTER under § 1206.200(a) and not so published.

(b) A final order, opinion, statement of policy, interpretation, or staff manual or instruction that affects a member of the public may be relied upon, used, or cited as precedent by NASA against any member of the public only if it has been indexed and either made available or published as provided by § 1206.200(b) or if the member of the public has actual and timely notice of the terms thereof.

(c) Failure to make available an agency record required to be made available under this part could provide the jurisdictional basis for a suit against NASA under 5 U.S.C. 552(a) (4) (B) through (G), which provides as follows:

(B) On complaint, the district court of the United States in the district in which the complainant resides, or has his principal place of business, or in which the agency records are situated, or in the District of Columbia, has jurisdiction to enjoin the agency from withholding agency records and to order the production of any agency records im-properly withheld from the complainant. In such a case the court shall determine the matter de novo, and may examine the contents of such agency records in camera to determine whether such records or any part thereof shall be withheld under any of the exemptions set forth in subsection (b) of this section, and the burden is on the agency to sustain its action.

(C) Notwithstanding any other provision of law, the defendant shall serve an answer or otherwise plead to any complaint made under this subsection within thirty days after service upon the defendant of the pleading in which such complaint is made unless the court otherwise directs for good cause shown.

(D) Except as to cases the court considers of greater importance, proceedings before the district court, as authorized by this subsection, and appeals therefrom, take precedence on the docket over all cases and shall be assigned for hearing and trial or for argument at the earliest practicable date and expedited in every way.

(E) The court may assess against the United States reasonable attorney fees and other litigation costs reasonably incurred in e under this section in which the

complainant has substantially prevailed.

(F) Whenever the court orders the production of any agency records improperly withheld from the complainant and assesses against the United States reasonable at-torney fees and other litigation costs, and the court additionally issues a written finding that the circumstances surrounding the withholding raise questions whether agency personnel acted arbitrarily or capriciously with respect to the withholding, the Civil Service Commission shall promptly initiate a proceeding to determine whether discipli-mary action is warranted against the officer aployee who was primarily responsible for the withholding. The Commission, after investigation and consideration of the evidence submitted, shall submit its findings and recommendations to the administrative authority of the agency concerned and shall send copies of the findings and recommendations to the officer or employee or his repre-sentative. The administrative authority shall take the corrective action that the Commission recommends.

(G) In the event of noncompliance with the order of the court, the district court may punish for contempt the responsible em-ployee, and in the case of a uniformed service, the responsible member.

## Subpart 9—Annual Report

## § 1206.900 Requirements for annual report.

On or before March 1 of each calendar year, NASA shall submit a report covering the preceding calendar year to the Speaker of the House of Representatives and President of the Senate for referral to the appropriate committees of the Congress. The report shall include the following information:

(a) The number of determinations made by NASA not to comply with requests for records made to the Agency under Subpart 6 of this part and the reasons for each such determination;

(b) The number of appeals made by persons under Subpart 6 of this part, the result of such appeals, and the reason for the action upon each appeal that results in a denial of information;

(c) The names and titles or positions of each person responsible for the denial of records requested under this part, and the number of instances of participation for each;

(d) The results of each proceeding conducted pursuant to 5 U.S.C. 552(a) (4) (B) through (G) (see § 1206.800(c)), including a report of the disciplinary action taken against the officer or employee who was primarily responsible for improperly withholding records or an explanation of why disciplinary action was not taken:

(e) A copy of the current version of this Part 1206 and any other rules or regulations made by NASA regarding 5 U.S.C. 552;

(f) A copy of the current fee schedule and the total amount of fees collected by NASA for making records available under this Part 1206; and

(g) Such other information as indicates efforts by NASA to administer fully this part.

Effective date: This Part 1206 shall become effective on February 19, 1975.

> JAMES C. FLETCHER, Administrator.

[FR Doc.75-4449 Filed 2-18-75;8:45 am]

# CHAPTER III—NATIONAL TRANSPORTA-TION SAFETY BOARD

[NTSB Reg. OR-1 Amdt. 4]

# PART 401—PUBLIC AVAILABILITY OF INFORMATION

Fee Schedule

On January 13, 1975, a notice was published in the FEDERAL REGISTER (40 FR

2446) stating that the National Transportation Safety Board was proposing an amendment to 14 CFR Part 401, which would add Appendix A-Fee Schedule, applicable to copying and other special services relating to requests for records of

Interested persons were given 30 days to submit written comments. No comments or objections have been received, and the following amendment is hereby adopted without change:

## APPENDIX A-FEE SCHEDULE

1. Special Services Fees (pursuant to 31 U.S.C. 483a). Upon request, services relating to public documents are available at the following fees:

(a) Subscriptions (calendar year)
(1) Initial decisions of the Administrative
Law Judges—\$40.00

(2) Board safety enforcement opinions and orders-\$20.00

the Board.

(3) Board aircraft accident (probable cause) reports, brief format—\$40.00 (U.S.) and \$50.00 (foreign)

(4) Aircraft accident reports, narra--\$27.50 (U.S.) and \$34.50 (foreign). tive-

(5) Surface accident reports, narrative-\$27.50 (U.S.) and \$34.50 (foreign).

Note.—Subscripton orders for (a) (3), (4) and (5) above, should be forwarded to the National Technical Information Service, P.O. Box 1551, Springfield, Virginia 22151.

(b) Document certification under the Board's seal-\$4.00.

(c) Computer tapes and services for aviation accidents. Duplication of computer tapes (or a fraction thereof)-\$40.00.

Note.—Computer tape requests should be addressed to the Chief, Information Systems Branch, Bureau of Aviation Safety, National Transportation Safety Board, Washington,

(d) The basic fees set forth provide for ordinary first-class postage prepaid. If registered, certified, air, or special delivery mail is used, postal fees therefor will be added to the basic fee. Also, if special handling or packaging is required, such costs will be added to the basic fee.

(e) Subscription fees for (a) above are waived for qualifying foreign countries, international organizations, nonprofit public safety entities, State and Federal transportation agencies, and colleges and universities, after approval by the Board's General Manager.

2. Commercial Reproduction Fees and Search Fees. Pursuant to 5 U.S.C. 552, as amended, reproduction of the documents listed below is accomplished by commercial contract, and the reproductions are mailed from the contractor together with a billing for the costs. Requests must be forwarded to the Accident Inquiry Unit, National Transportation Safety Board, Washington, D.C. 20591, together with an agreement to pay the reproduction cost. The search fee (if required) must be enclosed. Fees are subject to change depending upon the Board's annual contract award.

Current fees are:

# (a) Photocopy:

Size:	Pe	r page
81/2"x11"		80, 12
81/2"x14"		. 24
10"x14"		. 30
14"x18"		. 60
18"x24"		. 90
(h) Photog	renhic neinte:	

Size: Un	it price
8"x10" black/white	\$1.05
5"x7" color	3. 25
8"x10" color	6.50
2"x2" color slide	1.50
2"x2" black/white slide	1.25
4"x10" X-ray	6.50

(c) Minimum charge for reproduction--\$4.00.

(d) Special expedite service-80 percent surcharge for 24-hour service and 50 percent surcharge for 48-hour service.

(e) Document search fee-The Board has determined that it is in the public interest to limit search fees to documents which require commercial reproduction. Further, the fee has been reduced to the labor cost for a minimum time-search. Therefore, a \$4.00 search fee is required only for:

(1) Factual accident investigation reports, statements, photographs, and other material contained in the Board's accident investigation files.

(2) Accident investigation information not in the Board's public files,

Note.—The \$4.00 search fee will be included in the commercial reproduction firm's

(f) Documents requiring commercial reproduction for copies:

Transcripts of public hearings.
 Factual accident investigation reports

and information (see (e) (1) and (2) above). 3. Reproduction fees for the public reading room. All documents in the Board's public files may be examined, without charge, in the Board's public reading room, located in the Accident Inquiry Unit. There is a self-service duplicator in the reading room, which is available to the public for reproduction at a nominal cost.

4. Documents available without commercial reproduction cost until limited supplies are exhausted:

(1) Press releases

(2) Aviation accident reports, narrative, and brief format probable cause reports (on request for specific accidents)

(3) Surface accident reports (4) Special studies

(5) Safety Board regulations (14 CFR 400 through 440)

(6) Indices to initial decisions, Board opinions and orders, and staff manuals and instructions

(7) Statistical data published by the Board

(8) Safety recommendations
5. Documents for sale by the Government Printing Office:

(1) Board's annual report (2) Volume I, National Transportation Safety Board Decisions (1967-1972)

Effective date: This fee schedule is effective February 19, 1975, as Appendix A to 14 CFR 401, and supersedes any

conflict therewith.

JOHN H. REED. Chairman.

FEBRUARY 14, 1975.

[FR Doc.75-4499 Filed 2-18-75;8:45 am]

# Title 16—Commercial Practices CHAPTER I—FEDERAL TRADE COMMISSION

SUBCHAPTER A—PROCEDURES AND RULES
OF PRACTICE

PART 4-MISCELLANEOUS RULES Schedule of Fees for Document Search, **Duplication and Certification** 

On page 2450 of the Federal Register of January 13, 1975, there was published a notice of initiation of a proceeding for the amendment of § 4.8(c) of its miscellaneous rules to establish a uniform schedule of fees applicable to all constituent units of the Federal Trade Commission. The proceeding was initiated as the result of recent amendments to the Freedom of Information Act (Administrative Procedure, 5 U.S.C. 552) which made significant changes to the law pertaining to the fees an agency may charge for search and duplication services performed for those requesting records. The fees proposed for certification services were in accordance with 31 U.S.C. 483(a). The proposal set forth fees for single copies of paper reproduction, microfilm services, duplication of microfilm, information retrieval by computer service, and clerical, paraprofessional, and professional search fees. The proposal also set forth instances under which charges or a portion thereof would not be made. Interested persons were given 30 days in which to submit written comments concerning the proposed amendment.

No written comments were received, and the proposed amendment of § 4.8(c) is hereby adopted without change and is set forth below. Minor changes were made in paragraph (b) to provide for a written request and to delete the ref-erences to identifiable records. Paragraph (b) as amended is also set forth below:

In § 4.8, paragraphs (b) and (c) are revised as follows:

# § 4.8 Availability of public information.

(b) Reasonable facilities for copying and for producing copies are provided at each of the offices of the Commission. Subject to appropriate limitations and the availability of facilities, any person may copy without charge any of the public records available for inspection at each of those offices, or reproductions of any such records will be provided by the Commission to any person upon request in person or upon written request and the payment of the prescribed duplication fees. All written requests for reproductions of public records should be addressed to the Secretary of the Commission, and should specify as clearly and

provision of subpart C which may be in accurately as reasonably possible the records desired. With respect to records which cannot be specified with complete clarity and particularity, identification requirements will be met by descriptions sufficient to enable qualified Commission personnel to locate the requested records. In any instance the Commission, the Secretary of the Commission, or the official in charge of each office may restrict the use of the Commission's facilities to providing only one reproduction of any public record, or may refuse to permit the use of those facilities for copying and producing copies of records which are published or are publicly available at places other than the offices of the Commission, or may supply to any person only one reproduction of any record when copies thereof may be obtained from the Commission without charge.

(c) (1) User fees pursuant to 31 U.S.C. 483(a) and 5 U.S.C. 552(a), as amended by Sec. (b) (1) of Public Law 93-502, shall be charged according to the schedule contained in paragraph (2) of this section for services rendered in responding to requests for Commission records under this subpart unless the Secretary initially or the Commission on appeal determines, in conformity with the provisions of 5 U.S.C. 552(a) amended by Public Law 93-502 and 31 U.S.C. 483(a) that such charges or a portion thereof are not in the public interest. Such a determination will ordinarily not be made unless the service to be performed will benefit primarily to the public as opposed to the requester, or unless the requester is a government agency or indigent. Fees shall not be charged for a request where they would amount, in the aggregate, to less than \$10.00. Ordinarily, fees will not be charged if the requested records are not found, or if all the records found are withheld as exempt. However, if the time expended in processing such a request is substantial, or if the requester has been notified of the estimated cost, pursuant to \$ 4.11 of the Rules, or has been advised that it cannot be determined in advance whether any records will be made available, fees may be charged. The Secretary, with the approval of the Commission, shall establish such fees.

(2) The following uniform schedule of fees applies to all constitutent units of the Commission, each requesting party being limited to one copy:

## REPRODUCTION

Paper copy \_\_\_\_\_ 10 cents per page.

> MICROFILM SERVICES—PRODUCTION OF MICROPILM

16 mm .... 4 cents per frame. Microfiche 4" x 6" \_\_\_\_\_ 5 cents per frame

#### DUPLICATION OF MICROFILM

\$3.75 per 100-ft 16 mm roll. Microfiche 4" x 6" \_\_\_\_\_ 15 cents each. \$1.15 each. 

COMPUTER SERVICES-INFORMATION RETRIEVAL

Use of terminal (each 25 cents.

request).
Hard copy (paper) of 30 cents.
each request. Programer \_\_\_\_\_ \$6.50 per hour.

#### SEARCH FEES

Clerical 1st hour, free. 2d and subsequent hours\_ \$4.20 per hour. Peraprofessional \_\_\_\_\_\_2d and subsequent hours\_ 1st hour, free. \$5.20 per hour. Professional .... 1st hour, free. 2d and subsequent hours\_ \$11 per hour. Certification \_\_\_\_\_ 82.75 each.

(3) Payment should be made by check or money order payable to the Treasury of the United States.

(Authority: 15 U.S.C. 41, et seq. (5 U.S.C. 552, 31 U.S.C. 483(a)))

Effective date, February 19, 1975.

By direction of the Commission dated February 14, 1975.

[SEAL] CHARLES A. TOBIN, Secretary.

[FR Doc.75-4507 Filed 2-18-75;8:45 am]

-Conservation of Power and **Water Resources** 

CHAPTER I-FEDERAL POWER COMMISSION

SUBCHAPTER A-GENERAL RULES

[Docket No. RM75-15; Order 524]

PART 1—RULES OF PRACTICE AND PROCEDURE

RT 3—ORGANIZATION; OPERATION; INFORMATION AND REQUESTS; MIS-CELLANEOUS CHARGES; ETHICAL **STANDARDS** 

Freedom of Information Act Regulations

FEBRUARY 10, 1975.

On January 2, 1975, the Commission issued a notice proposing certain changes in its general rules governing requests for public information. The notice was published in the FEDERAL REGISTER on January 6, 1975 (40 FR 1077). The changes involve, among other things, specific timetables and procedures to be followed in the event the Commission withholds requested information, the procedure to be followed for the search and duplication of requested documents, and rules governing the maintenance and dissemination of a current index of public information. The purpose of the amendments is to conform the Commission's procedures to the requirements of the Freedom of Information Act, 5 U.S.C. 552, as amended by the passage of H.R. 12471, Pub. L. No. 93-502.

The Commission received one response to the notice on January 20, 1975, from the Salt River Project, Phoenix, Arizona. Ted. B. Hardy, Manager, Power Resource Development, writing on behalf of the Project, questioned whether the \$2 fee for a certified copy means \$2 per document or \$2 per page. The language of Section 3.102(b) will be revised accordingly to clarify the Commission's intent to charge \$2 per document for certification. A conference was not requested.

The Commission finds:

(1) The notice and opportunity to par-ticipate in this rulemaking proceeding with respect to the matters presently before this Commission through the submission, in writing, of data, views, comments, and suggestions are in accordance with the procedural requirements prescribed by 5 U.S.C. 553.

(2) In view of the intent of Pub. L. No. 93-502 that the amendments adopted herein become effective on February 19. 1975, good cause exists for making them

effective on that date.

(3) The amendments of the Commission's General Rules as prescribed herein are in the public interest and are necessary and appropriate for the administration of the Federal Power Act and the

Natural Gas Act.

The Commission, acting pursuant to the provisions of the Federal Power Act, as amended, particularly Section 309 (49 Stat. 858-859; 16 U.S.C. 825h), the Natural Gas Act, as amended, particularly Section 16 (52 Stat. 830; 15 U.S.C. 7170), and Pub. L. No. 93-502 (88 Stat. 1561, amending 5 U.S.C. 552) orders:

#### § 1.36 [Amended]

A. Section 1.36 in Part 1, Subchapter A of Chapter I, Title 18 of the Code of Federal Regulations is amended as follows:

1. Paragraph "(c) Public records" is amended as follows:

a. The introductory portion is revised to read:

- (c) Public records. The public records of the Commission, available for inspection and copying upon a request reasonably describing the document, during regular business hours in the public reference room maintained by the Office of Public Information, include:
- b. Paragraph (c) (14) (i) is revised to read:
- (i) Specifically authorized under criteria established by an Executive order to be kept secret in the interest of national defense or foreign policy and are in fact properly classified pursuant to such Executive order:

c. Paragraph (c) (14) (vii) is revised to read:

(vii) Investigatory records compiled for law enforcement purposes, but only to the extent that the production of such records would interfere with enforcement proceedings, deprive a person of a right to a fair trial or an impartial adjudication, constitute an unwarranted invasion of personal privacy, disclose the identity of a confidential source and, in the case of a record compiled by a criminal law enforcement authority in the course of a criminal investigation, or by an agency conducting a lawful national security intelligence investigation, confidential information furnished only by the confidential source, disclose investigative techniques and procedures, or endanger the life or physical safety of law enforcement personnel; or

d. Paragraph (c) (15), and all references thereto, is redesignated as paragraph (c) (16) and immediately after paragraph (c) (14) (viii) the following new paragraph (c) (15) is inserted:

(15) Any reasonably segregable portion of a record after deletion of the portions which are exempt under this section.

2. Paragraph (f), and all references thereto, is redesignated as paragraph (g), and two new paragraphs (e) and (f) inserted immediately after paragraph (d). As so revised, the amended portions read as follows:

(e) Index of Commission actions. The Office of Public Information will maintain and make available for public ininspection and copying current indexes providing identifying information for the public as to any matter issued, adopted, or promulgated after July 4, 1967, and required by this section to be made available or published. The index will be published quarterly and copies or supplements thereto will be distributed by sale

or otherwise.

(f) Timetables and procedures in event of withholding of public records. (1) The Director of Public Information will determine within ten days (except Saturdays, Sunday, and legal public holidays) after receipt of a request for public records whether to comply with such request and will immediately notify the person making such request of such determination and the reasons therefor, and of the right of such person to appeal to the Chairman any adverse determination by petition filed pursuant to § 1.7.

(2) The Chairman, in his capacity as administrative head of the agency pursuant to Section 1 of Reorganization Plan No. 9 of 1950, will make a determination with respect to any appeal within twenty days (except Saturdays, Sundays, and legal public holidays after the receipt of such appeal. If on appeal the denial of the request for records is in whole or in part upheld, the Chairman will notify the person making such request of the provisions for judicial review of that determination.

(3) In unusual circumstances, the time limits prescribed in this section may be extended by the Secretary by written notice to the person making such request setting forth the reasons for such extension and the date on which a determination is expected to be dispatched. No such notice will specify a date that would result in an extension for more than ten working days. "Unusual circumstances' as used in this paragraph means, but only to the extent reasonably necessary to the proper processing of the particular request:

(i) the need to search for and collect the requested records from field facilities or other establishments that are separate from the office processing the request:

(ii) the need to search for, collect, and appropriately examine a voluminous amount of separate and distinct records which are demanded in a single request;

(iii) the need for consultation, which will be conducted with all practicable speed, with another agency having a substantial interest in the determination of the request or among two or more components of the agency having substantial subject-matter interest therein.

(4) If the Commission fails to comply with the applicable time limit provisions of this section, the Commission can request the court for additional time to complete its review of the records by showing that exceptional circumstances exist and that it is exercising due diligence in responding to the request.

(5) Any notification of denial of any request for records under this section will set forth the names and titles or positions of each person responsible for

the denial of such request.

(6) Upon any determination to comply with a request for records, the records will be made promptly available to such person making such request.

(g) Procedure in event of subnoena

B. Section 3.102 in Part 3, Subchapter A of Chapter I, Title 18 of the Code of Federal Regulations is amended as fol-

1. Paragraph (b) is revised to read:

§ 3.102 Public information requests, and assistance; miscellaneous charges.

(b) During the Commission's regular business hours, the public may examine in the Office of Public Information in Washington, D.C., copies of public information filed with the Commission. Pursuant to the competitive advertised procurement procedure of the Federal Power Commission, responsibility for the search and duplication of public documents is contracted out each year, and a new schedule of fees is prescribed. Any person may obtain a copy of the schedule of fees by coming in person to the Office of Public Information, by telephone, or by mail. Where practicable, self-service duplication of requested documents may also be made in the Office on duplicating machines by the person requesting the documents. Where data has been extracted from the Commission's public records on magnetic tape computer files, copies of the tape files may be secured on a reimbursable basis, upon a written request to the Office of Public Information. The fee will vary for each requirement, depending on size and complexity. Documents will be furnished without charge or at a reduced charge where the Secretary determines that waiver or reduction of the fee is in the public interest. Except for requests made by Government agencies certification of copies of any official Commission record shall be accom-panied by a fee of \$2.00 per document Inquiries and orders may be made to that office personally, by telephone, or by mail.

C. The amendments ordered herein are effective as of February 19, 1975.

D. The Secretary shall cause prompt publication of this order to be made in the FEDERAL REGISTER.

By the Commission.

KENNETH F. PLUMB, Secretary.

[FR Doc.75-4503 Filed 2-18-75;8:45 am]

#### CHAPTER VI--WATER RESOURCES COUNCIL

# PART 701-COUNCIL ORGANIZATION Freedom of Information Act Regulations

Part 701, Subpart D of Chapter VI, Title 18 of the Code of Federal Regulations is hereby amended by deleting the current Subpart D entirely and substituting therefor a new Subpart D as shown below. This rulemaking is done by authority of 5 USC 552, as amended by the Freedom of Information Act Amendments of 1974 (the "1974 Amendments") (Pub. L. 93-502, 88 Stat. 1561 (5 USC 552)) and 42 USC 1962d-1.

The Water Resources Council (WRC) regulations implementing the Freedom of Information Act (the "Act") (5 USC 552) were published on pages 40761 and 40762 of the Federal Register of November 20, 1974 (39 FR 40761). On November 21, 1974, the Congress passed over the veto of President Ford a series of Amendments to the Act which go into effect on February 19, 1975. The 1974 Amendments render the current WRC Freedom of Information regulations anachronistic and necessitate extensive revision, which is expedited by the deletion of the current Subpart D and substitution of a new Subpart D that reflects the changes in the Act by the 1974 Amendments. These changes include three provisions whose nature is "substantive" in the sense that they affect what records are subject to compulsory disclosure under the Act, as well as 'procedural" provisions affecting how requests for records shall be processed or litigated.

The substantive changes are a revision of the 1st exemption (pertaining to documents classified under an Executive order for reasons of defense or foreign policy), a revision of the 7th exemption (investigating law enforcement records), and a provision on the availability of "reasonably segregable" portions of records from which exempt matter has been deleted. These substantive changes are reflected in § 701.201 of the proposed regulations.

The procedural changes include provisions for: how and to whom requests for information are made (§ 701.202 (a), (b), (c), (d), and (e)); a uniform schedule of fees and when they will or will not be charged (§ 701.203); time limits for initial determinations (§ 701.204), appeals (§ 701.205), and final determina-tions (§ 701.206), regarding requests for information; administrative extensions of time limits in unusual circumstances (§ 701.207); and judicial extension of time limits in exceptional circumstances (§ 701.208).

in the previous regulations (§ 701.204), the new § 701.209 encourages WRC river basin commissions and Field Committees to establish procedures for public availability of information consistent with 5 USC 552, as amended, and the new subpart D.

The policy of the WRC with regard to availability of information (previous and new § 701.200) has not changed. WRC records and informational materials are available to the fullest extent possible consistent with the Act and will be promptly furnished to any member of the public.

This subpart is intended to be consistent with, but separate from, the public participation provisions of the Council's principles and standards (38 FR 174) and subsequent implementing procedures.

Notice and public procedure on this amendment are deemed unnecessary and impracticable since they merely conform to the requirements of the 1974 Amendments.

Dated: February 13, 1975.

Effective date: February 19, 1975.

## WARREN D. FAIRCHILD, Director U.S. Water Resources Council.

Part 701, Subpart D, of Chapter VI, Title 18 of the Code of Federal Regulations is hereby amended by deleting the current Subpart D entirely and substituting therefor a new Subpart D as set forth below:

## Subpart D-Availability of Information

701.200 Statement of policy.

Availability of records and informational materials. 701.201

701.202 Procedure for requests for information.

701.203 Schedule of fees.

701.204 Time limits for WRC initial determinations regarding requests for information.

701.205 Time limit for requester to appeal

an initial adverse determination.
701.206 Time limits for WRC final determinations regarding requests for information appealed by the requester from an initial adverse determination.

701.207 Extension of time limits for WRC initial and final determinations. 701.208

WRC petition for judicial extension of time. 701.209 River Basin Commissions and Field Committees.

AUTHORITY: 5 U.S.C. 552 as amended by Pub. L. 93-502, 88 Stat. 1561; 42 U.S.C. 1962d-1.

# § 701.200 Statement of policy.

Water Resources Council records and informational materials are available to the fullest extent possible consistent with 5 U.S.C. section 552, as amended, and will be promptly furnished to any member of the public.

#### § 701.201 Availability of records and informational materials.

(a) Except for records and materials exempted from disclosure pursuant to paragraph (b) of this section, any person

may inspect and copy any document in the possession and custody of the Water Resources Council in accordance with the procedure provided in § 701.202.

(b) The provisions of 5 U.S.C. section 552 which require that agencies make their records available for public inspection and copying do not apply to matters which are:

(1) (a) Specifically authorized under criteria established by an Executive order to be kept secret in the interest of national defense or foreign policy and (b) are in fact properly classified pursuant to such Executive order;

(2) Related solely to the internal personnel rules and practices of an agency;

(3) Specifically exempted from disclosure by statute;

(4) Trade secrets and commercial or financial information obtained from a person and privileged or confidential;

(5) Inter-agency or intra-agency memorandums or letters which would not be available by law to a party other than an agency in litigation with the

agency;
(6) Personnel and medical files and similar files the disclosure of which would constitute a clearly unwarranted in-

vasion of personal privacy;

(7) Investigatory records compiled for law enforcement purposes but only to the extent that the production of such records would (A) interfere with enforcement proceedings, (B) deprive a person of a right to a fair trial or an impartial adjudication, (C) constitute an unwarranted invasion of personal privacy, (D) disclose the identity of a confidential source and, in the case of a record compiled by a criminal law en-forcement authority in the course of a criminal investigation, or by an agency conducting a lawful national security intelligence investigation, confidential information furnished only by the confidential source, (E) disclose investigative techniques and procedures, or (F) endanger the life or physical safety of law enforcement personnel;

(8) Contained in or related to examination, operating, or condition reports prepared by, on behalf of, or for the use of an agency responsible for the regulation or supervision of financial institu-

tions: or

(9) Geological and geophysical information and data, including maps, concerning wells.

Any reasonably segregable portion of a record shall be provided to any person requesting such record after deletion of the portions which are exempt under this subsection.

# §701.202 Procedure for requests for information.

(a) A member of the public who requests records or materials from the Water Resources Council must provide a reasonable description of the records or materials sought so that such records or materials may be located without undue search or inquiry.

(b) Requests which reasonably describe the records or materials sought should be directed to the Public Inforton, D.C. 20037.

- (c) To insure that requests for information are processed as expeditiously as possible, all Freedom of Information Act (FOIA) requests should be clearly identifled by the requester as such on the envelope and in the letter.
- (d) Records or materials will be available for inspection and copying in person during normal business hours or by mail.
- (e) Requests for records which originate in or concern matters which originate in another department or agency may be forwarded to the department or agency primarily concerned and the requester so notified.

# § 701.203 Schedule of fees.

- (a) The Public Information Officer will to the extent practicable, encourage the widest possible distribution of information by permitting requests for inspection or copies of records or materials to be met without cost to the person making the request.
- (b) Fees will be charged in the case of requests which are determined by the Public Information Officer to involve a burden on staff or facilities significantly in excess of that normally accepted by the Council in handling routine requests for information.
- (c) In all instances where the Public Information Officer determines that a request for information can be considered as primarily benefiting the general public (despite a § 701.203 determination of burden), such request shall be met either without cost wherever practicable or at a reduced cost to the requester. Any such reduction shall be determined by the Public Information Officer on the basis of the balance between the benefit to the general public and the cost to the Water Resources Council.
- (d) Fees shall be limited to recovery of only direct costs of search and duplication but in no event shall the fee for search and duplication exceed \$2.50 per half hour, nor shall the fee for copying exceed \$0.25 per page (maximum per page dimension of 8x14 inches).
- (e) Unless a request for information specifically states that whatever cost is involved will be acceptable, or acceptable up to a specified limit that covers anticipated costs, a request that is expected to involve an assessed fee in excess of \$50.00 will not be deemed to have been received until the requester is advised promptly upon physical receipt of the request of the anticipated cost and agrees to bear it.
- (f) When anticipated fees exceed \$50.00, a deposit for 25% of the amount must be made within 10 days of the notice to the requester of the initial determination.
- (g) The Council reserves the right to limit the number of copies of any document that will be provided to any one person.

# mation Officer, Water Resources Council, § 701.204 Time limits for WRC initial Suite 800, 2120 L Street NW., Washingfor information.

(a) An initial determination to grant or deny each request for information will be made within ten (10) working days of receipt of such request.

(b) The requester shall be notified immediately of the initial determination

and the reasons therefor.

(c) The Public Information Officer will make initial determinations to grant requests for information.

(1) In those instances where the initial determination by the Public Information Officer is to grant the request and the information is immediately supplied such action will serve as both notice of determination and compliance with the request.

(2) In those instances where the initial determination by the Public Information Officer is to grant the request, but the information is not immediately available, the Public Information Officer will send immediate notice of the determination to comply, and the approximate date the information will be forwarded.

(d) The Public Information Officer will make initial determination to deny the requests only with the concurrence of the General Counsel. The requester shall be notified immediately of the initial adverse determination, the reasons therefor, and the right to appeal the initial adverse determination to the Director.

#### § 701.205 Time limit for requester to appeal an initial adverse determination.

(a) The requester shall have thirty (30) calendar days to file with the Director an appeal from an initial adverse determination. The appeal must be in

(b) The thirty (30) day period of appeal shall run from receipt of the initial adverse determination (in cases of denials of an entire request) and from receipt of any records being made available pursuant to the initial adverse determination (in cases of partial denials).

§ 701.206 Time limit for WRC final determinations regarding requests for information appealed by the requester from an initial adverse determination.

The Director shall make a final determination with respect to any appeal within twenty (20) working days after receipt of such appeal. If the initial adverse determination is in whole or in part upheld by the Director, the requester shall be notified of the final adverse determination and the provisions for judicial review of that determination as stated in the Freedom of Information Act, as amended (see 5 U.S.C. 552(a) (4) et seq.; as amended by Pub. L. 93-502).

# § 701.207 Extension of time limits for WRC initial and final determina-

(a) In unusual circumstances, as specified in this section, the time limits pre-

scribed in either § 701.203 or § 701.204 may be extended by written notice from the responsible WRC official (i.e., the Public Information Officer in instances of initial requests and the Director in instances of appeals) to the requester setting forth the reasons for such extension and the date on which a determination is expected to be dispatched. No such notice shall specify a date that would result in an extension for more than ten (10) working days, and in no event shall the total extended time exceed ten (10) working days with respect to a particular request.

(b) As used in this section, "unusual circumstances" means, but only to the extent reasonably necessary to the proper processing of the particular

request:

(1) the need to search for and collect the requested records from field facilities or other establishments that are separate from the office processing the request:

(2) the need to search for, collect, and appropriately examine a voluminous amount of separate and distinct records which are demanded in a single request;

(3) the need for consultation, which shall be conducted with all practicable speed, with another agency having a substantial interest in the determination of the request or among two or more components of the agency having substantial subject-matter interest therein.

#### § 701.208 WRC petition for judicial extension of time.

The provisions of \$ 701,206 notwithstanding, the Director may petition for judicial extension of time when exceptional circumstances warrant action.

## § 701.209 River basin commissions and field committees.

(a) River basin commissions established pursuant to Title II of the Water Resources Planning Act are encouraged to establish, pursuant to section 205(c) of that Act, procedures for public availability of information that are consistent with 5 U.S.C. 552, as amended, and this subpart.

(b) Field committees will be governed by the procedures adopted by the lead Federal agency to implement 5 U.S.C. 552, as amended; except that if the lead agency of a field committee is a non-Federal entity, the standards of this sub-

part shall apply.

(c) Requests for documents and informational materials may be made to the chairmen of the field committees and river basin commissions at the following addresses.

(1) River Basin Commissions:

Great Lakes Basin Commission, P.O. Box 999. Ann Arbor, Michigan 48106;

New England River Basins Commission, 55 Court Street, Boston, Massachusetts 02108; Ohio River Basin Commission, 36 East 4th Street, Suite 208-220, Cincinnati, Ohio 45202:

Pacific Northwest River Basins Commission, Box 908, Vancouver, Washington 98660:

Upper Mississippi River Basin Commission, Federal Office Building, Room 510, Fort Snelling, Twin Cities, Minnesota 55111; Missouri River Basin Commission, 10050 Regency Circle, Suite 403 Omaha, Nebraska 68114.

(2) Field Committees:

Arkansas-White-Red Inter-Agency Commit-tee, Room 4030, Federal Building, Albuquerque, New Mexico 87101;

Pacific Southwest Inter-Agency Committee, 630 Sansome Street, Room 1216, San Francisco, California 94111;

Southeast Basins Inter-Agency Committee, 402 Walton Building, Columbus, Georgia

[FR Doc.75-4576 Filed 2-18-75;8:45 am]

# Title 20—Employees' Benefits CHAPTER II-RAILROAD RETIREMENT BOARD

#### PART 200-PROCEDURES AND **FORMS**

Freedom of Information Act Regulations

The Railroad Retirement Board has determined that it would be impractical and unnecessary for it to publish an index of all records and documents in its custody. Such an index is maintained by the Railroad Retirement Board at its headquarters and copies of the index may be obtained for a fee equivalent to the costs of reproduction. Requests for such indexes should comply with the form for requests for information as described in § 200.3(h).

Pursuant to the general authority provided by section 7 of the act of October 16, 1974 (88 Stat. 1338; 45 U.S.C. 231f) and by section 12 of the act of June 25, 1938 (52 Stat. 1107, as amended; 45 U.S.C. 362), paragraphs (c), (e) (1), (f), (g), and (h) of § 200.3 of 20 CFR Part 200 of the regulations under such acts are revised and paragraphs (i), (j), (k), (l), and (m) are added to such section by Board Order 75-19, dated Feb-

ruary 7, 1975, to read:

§ 200.3 Availability of information to public.

(c) There shall be maintained in the Board's library a current index of the materials referred to in paragraph (a) of this section which will have been issued, adopted, or promulgated subsequent to July 4, 1967. This index shall be available for public inspection and copying at the Board's headquarters offices located at 844 Rush Street, Chicago, Illinois, during the normal business hours of the Board. Copies of the index or any portion thereof may be obtained for a fee equivalent to the costs of reproduction by submitting a written request therefor. Such request should comply with the form for requests as described in paragraph (h) of this section.

(e) The copies of manuals and instructions made available for public copying and inspection shall not include:

(1) Confidential statements, standards, and instructions which do not affect the public, and

(f) With the exception of records specifically excluded from disclosure by section 552(b) of Title 5, United States Code, or other applicable statute, any records of or in the custody of this agency, other than those made available under paragraphs (a), (c), and (d) of

this section, shall, upon receipt of a written request reasonably describing them, promptly be made available to the person requesting them.

(g) The Chief Executive Officer may charge the person or persons making a request for records under paragraph (f) of this section a fee in an amount not to exceed the costs actually incurred in complying with the request. After February 21, 1975, any such fees charged by

the Chief Executive Officer shall be in accordance with the following schedule. (1) The fee for copies shall be \$.10 per

copy per page. (2) The charge for making a manual search for records shall be \$6.00 per man hour.

(3) The maximum computer search charge shall be \$268.00 per hour (\$4.50 per minute).

(4) Any or all fees may be waived or reduced by the Chief Executive Officer whenever he or she determines that it is in the public interest to do so.

(5) All fees shall be paid to the Board's Director of Budget and Fiscal Operations.

(h) Any person or organization requesting records pursuant to this section shall submit such request in writing to the Chief Executive Officer, Railroad Retirement Board, Room 536, 844 Rush Street, Chicago, Illinois 60611. All such requests should be clearly and prominently identified as requests for information under the Freedom of Information Act. If submitted by mail or otherwise submitted in an envelope or other cover, requests should be clearly and prominently identified as such on the envelope or cover.

(i) The Chief Executive Officer, or any other individual specifically authorized to act on behalf of the Chief Executive Officer, shall have the authority to grant or deny a request for information submitted under this section. The Chief Executive Officer or such authorized representative shall, within 10 working days following the receipt of a request, except as provided in subparagraph (1) of paragraph (j) of this section, make a determination granting or denying the request and notify the requester of his or her decision and if a denial, the reasons therefor. The requester shall be further advised that a total or partial denial may be appealed to the Board as provided in

paragraph (j) of this section. (j) In cases where a request for information is denied, in whole or in part, by the Chief Executive Officer or his or her authorized representative, the party who originally made the request may appeal

such determination to the Board by filing a written appeal with the Secretary of the Board within 20 working days following receipt of the notice of denial. The Board shall render a decision on an appeal within 20 working days following receipt of the appeal except as provided in subparagraph (1) of this paragraph. The requester shall promptly be notified of the Board's decision and, in cases where the denial is upheld, of the provisions for judicial review of such final administrative decisions.

(1) Ir. unusual circumstances, as enumerated in § 552(a) (6) (B) of Title 5, United States Code, the time restrictions of paragraphs (i) and (j) of this section may be extended in the aggregate by no more than 10 days by notice to the requester of such extension, the reasons therefor, and the date on which a determination is expected to be dis-

patched.

(2) For purposes of paragraphs (i) and (j) of this section, a request shall be received by the Chief Executive Officer of the Board when it arrives at the Board's headquarters. Provided, however, That when the estimated fee to be assessed for a given request exceeds \$30.00, such request shall be deemed not to have been received by the Chief Executive Officer until the requester is advised of the estimated cost and agrees to bear it. Provided further,. That a request which does not fully comply with all the provisions of paragraph (h) of this section shall be deemed to have been received by the Chief Executive Officer on the day it actually reaches his or her office.

(k) Any person in the employ of the Railroad Retirement Board who receives a request for any information, document or record of this agency, or in the custody thereof, shall advise the requester to address such request to the Chief Executive Officer. If the request received is in writing, it shall be immediately referred for action to the Chief Executive Officer.

(1) The Chief Executive Officer shall

maintain records of

(1) The total amount of fees collected by this agency pursuant to this section;

(2) The number of initial denials of requests for records made pursuant to this section and the reason for each;

(3) The number of appeals from such denials and the result of each appeal, together with the reason(s) for the action upon each appeal that results in a denial of information;

(4) The name(s) and title(s) or position(s) of each person responsible for each initial denial of records requested and the number of instances of action on a request for information for each such

person:

(5) The results of each proceeding conducted pursuant to section 552(a) (4) (F) of Title 5, United States Code, including a report of any disciplinary action against an official or employee who was determined to be primarily responsible for improperly withholding records, or an explanation of why disciplinary action was not taken:

(6) Every rule made by this agency affecting or in implementation of section 552 of Title 5, United States Code;

(7) The fee schedule for copies of resords and documents requested pursuant

to this regulation; and

(8) All other information which indicates efforts to administer fully the letter and spirit of section 552 of Title 5. United States Code.

(m) The Board shall, prior to March 1 of each year, prepare and submit a report to the Speaker of the House of Representatives and the President of the Senate covering each of the categories of records maintained in accordance with the foregoing for the preceding calendar

Dated: February 12, 1975. By authority of the Board,

R. F. BUTLER. Secretary of the Board.

IFR Doc.75-4571 Filed 2-18-75:8:45 am1

# Title 22—Foreign Relations

# CHAPTER I-DEPARTMENT OF STATE SUBCHAPTER A-GENERAL

[Departmental Reg. 108.711]

# PART 6—FREEDOM OF INFORMATION POLICY AND PROCEDURES

#### **Miscellaneous Amendments**

In accordance with the 1974 amendments to the Freedom of Information Act 5 U.S.C. 552 (Pub. L. 93-502), the following amendments and revisions are made to Part 6 of Title 22 of the Code of Federal Regulations.

Sec.

6.1 Definitions.

6.2 Availability of records. Executive Order 11652.

63 Records which may be exempt from 6.4 disclosure.

6.5 Classified records and information from other agencies.

6.6 Authority to release and certify; au-thority to withhold records.

6.7 Time limits.

Appeals. Public reading room. 6.8 6.9

6.10 Manner of requesting records.

Requests addressed to Foreign Service posts.

6.12 Closing requests,

6.13 Subsequent requests for same records.

Schedule of fees and method of pay-6.14 ment for services rendered.

Opening of records for nonofficial research.

6.16 Activities of advisory committees.

AUTHORITY: Sec. 4 of the Act of May 26, 1949, as amended (63 Stat. 111) (22 U.S.C. 2658); E.O. 11652, 37 FR 5209 (5 U.S.C. 552) (Pub. L. 93-502).

As used in this Part, the following definitions shall apply:

#### § 6.1 Definitions.

(a) The term "identifiable" means, in the context of a request for a record, a description which reasonably identifies a particular record sought. Such a description, if possible, should include date, format, subject matter, country concerned, office or mission originating or receiving the record, and the name of

any person to whom the record is known to relate.

(b) The term "record" includes all books, papers, maps, photographs, or other documentary material, or copies thereof, regardless of physical form or characteristics, made in or received by the Department of State (including Foreign Service posts abroad) and preserved as evidence of the organization, functions, policies, decisions, procedures, operations, or other activities of the Department or the Foreign Service. It does not include copies of the records of other Government agencies (except those which have been expressly placed under the control of the Department of State upon termination of another agency), foreign governments, international organizations, or non-governmental enti-ties unless they evidence organization, functions, policies, decisions, procedures, operations, or activities of the Department of State. It does not include records in the Berlin Document Center.

#### § 6.2 Availability of records.

(a) All identifiable records of the Department of State shall be made available to the public upon compliance with the procedures established in this Part. except to the extent that a determination is made to withhold a record exemptable under 5 U.S.C. 552(b). That determination shall be made in accordance with § 6.6(b).

(b) Unclassified information, documents, and forms which have previously been provided to the public as part of the normal services of the Department of State will continue to be made available on the same basis as before. Any Departmental officer who receives a request for records through normal channels of contact with the public, media, or the Congress which would not normally be made available shall advise the requester that, if the requester so wishes, the request will be referred to the Director, Freedom of Information Staff, Bureau of Public Affairs (hereinafter the FOI Director). If the Depart mental officer has reason to believe that the request is intended to be one under the Freedom of Information Act (see § 6.10), the officer shall refer it immediately to the FOI Director and so advise the requester.

# § 6.3 Executive Order 11652.

A request for classification review under Executive Order 11652 (37 FR 5209) which is identified as specifically and exclusively so intended will be processed in accordance with that Order and the implementing National Security Council Directive (37 FR 10053).

#### Records which may be exempt from disclosure.

(a) The following categories of records maintained by the Department of State may be exempted from disclosure:

(1) Records specifically authorized under criteria established by an executive order to be kept secret in the interest of national defense or foreign policy and in fact properly classified pursuant to such executive order.

(2) Records related solely to the internal personnel rules and practices of an

agency.

(3) Records specifically exempted from disclosure by statute, included in this category are records relating to the officers and employees of the Foreign Service, including efficiency records (§ 612 of the Foreign Service Act of 1946, as amended, 22 U.S.C. 986), the records of the Department of State or of diplomatic and consular officers of the United States pertaining to the issuance or refusal of visas or permits to enter the United States (§ 222(f) of the Immigration and Nationality Act of 1952, as amended, 8 U.S.C. 1202(f)), "Restricted Data" under section 224 of the Atomic Energy Act (42 U.S.C. 2274), and records subject to section 102(d) of the National Security Act of 1947 (61 Stat. 498).

(4) Records of trade secrets and commercial or financial information obtained from a person and privileged or con-

fidential.

(5) Records which are inter-agency or intra-agency memorandums, letters, telegrams, or airgrams which would not be available by law to a party other than an agency in litigation with the agency.

(6) Records such as personnel and medical files and similar files the public disclosure of which would constitute a clearly unwarranted invasion of personal

privacy.

(7) Investigatory records compiled for law enforcement purposes, but only to the extent that the production of such records would: (i) Interfere with enforcement proceedings; (ii) deprive a person of a right to a fair trial or an impartial adjudication; (iii) constitute an unwarranted invasion of personal privacy; (iv) disclose the identity of a confidential source and, in the case of a record compiled by a criminal law enforcement authority in the course of a criminal investigation, or by an agency conducting a lawful national security intelligence investigation, confidential information furnished only by the confidential source; (v) disclose investigative techniques and procedures; or (vi) endanger the life or physical safety of law enforcement personnel.

(8) Records contained in or related to examination, operating, or condition reports prepared by, on behalf of, or for the use of an agency responsible for the regulation or supervision of financial

(9) Geological or geophysical information and data, including maps, con-

cerning wells.

institutions.

(b) Any reasonably segregable portion of a record shall be provided to any person requesting such record after deletion of the portions which are exempt under paragraph (a) of this section. Normally a portion of a record shall be considered reasonably segregable when segregation can produce an intelligible record which is not distorted out of context and does not contradict the record being withheld.

## § 6.5 Classified records and information from other agencies.

(a) The applicability of the exemption for classified information (§ 6.4(a) (1))

requires a determination that the record in question is specifically authorized under the criteria established by Executive Order 11652 to be kept classified and is in fact properly classified pursuant to that order. This determination shall be made whenever possible before the initial denial under § 6.6(b). It must in any case be made prior to the decision of an appeal under § 6.8. No denial should be based solely on the existence of a classification marking on the record, and there shall be a substantive review of the validity of the classification to the maximum extent feasible within the time limits for

a denial under § 6.7.

(b) When a request for a Depart-mental record encompasses classified information originated by another department or agency, the request for that information shall be referred to the originator. The requester should be advised of the date and the addressee of the referral. Classified information involving intelligence sources and methods is subject to the control of the Director, Central Intelligence Agency; FBI information (whether or not classified) is subject to the control of the FBI; "Restricted Data" is subject to the control of the Energy Research and Development Administration or the Nuclear Regulatory Commission, as the case may be; and National Security Council documents are subject to the control of the Council. The Bureau of Intelligence and Research shall be consulted on records which appear to be in the first category; the Office of Security shall be consulted on records which appear to be in the second category; and the Executive Secretariat of the Department shall be consulted on records which appear to be in the last

(c) Requests for documents of other departments or agencies in the files of the Department of State which are not records of the Department (such as FBI reports) shall be referred to the appropriate department or agency whether or not classified. The requester shall be advised of the date and the addressee of

the referral.

# § 6.6 Authority to release and certify; authority to withhold records.

(a) Except as provided in § 6.15, the FOI Director is authorized to furnish copies of any record requested under these regulations which is not the subject of a denial letter under paragraph (b) of this section, and this authority shall include the authority to declassify any such record. The FOI Director is also authorized upon request to provide certified copies of any such record in accordance with Part 131 of this Chapter. In order to determine whether a record requested under this Part may be furnished, the FOI Director shall refer the request to the bureau, office, or other unit of the Department of State which he believes has primary responsibility, normally the originating office, or, in the case of documents originated elsewhere, the office with action responsibility for that document. If the office to which the request is referred does not accept re-

sponsibility, it shall immediately inform the FOI Director which Departmental office should have responsibility. The office which recommends referral to another office is required to send the request by hand, if possible, to that office. Whichever office accepts responsibility shall examine the requested record and, in consultation with the Assistant Legal Adviser designated to provide legal advice to the responsible office, determine which, if any, of the exemptions in § 6.4(a) are applicable and determine whether to claim an applicable exemption. If there is no applicable exemption, or it is determined not to exercise an exemption, the responsible office shall advise the FOI Director, who will, in turn, so advise the requester of the record and be responsible for collection of any

fees and release of the record.

(b) If the responsible office believes that one or more exemptions is applicable and should be exercised, it shall so advise the FOI Director immediately. The responsible office shall also complete whatever coordination within the Department or with other departments or agencies is required under normal Departmental practice. After informing the FOI Director of its intention to recommend denial of a requested record, the responsible office shall send its recommendation in writing to the designated Freedom of Information officers of the bureau with supervisory responsibility over the responsible office. (The designated Freedom of Information Officer shall be at the Deputy Assistant Secretary level or equivalent rank.) The designated Freedom of Information Officer shall consult with the FOI Director before making his decision on the recommendation to deny a record. If the Freedom of Information Officer decides to deny a record requested under this Part, he shall directly advise the requester in writing. This shall be considered the initial denial by the Department of State, and the initial denial letter must be sent within 10 working days from the date actually received by the FOI Director, unless the FOI Director grants an extension of time under § 6.7 (b). The denial letter shall state that the denial may be appealed to the Council on Classification Policy, the exemption or exemptions claimed, and the reason for doing so; it shall also enclose a copy of the regulation on appeals (§ 6.8). If a request is denied in whole or part but the responsible office is prepared to give further consideration to release of the requested record, the letter shall also state this fact and indicate the date by which this further consideration is expected to be completed. Whenever it is not possible to complete substantive review of the classification of classified records within the time allocated by the FOI Director, the letter should so state and indicate that this review is continuing and when it will be completed. If the designated Freedom of Information Officer decides that the requested record will not be denied, he shall so inform the FOI Director: this decision must be made also within the time allocated by the FOI Director.

§ 6.7 Time limits.

(a) An initial denial under § 6.6(b) shall be made within 10 days (excepting Saturdays, Sundays, and legal public holidays) after the receipt of a request for a record under this Part by the FOI Director. An appeal under § 6.8 shall be decided within 20 days (excepting Saturdays, Sundays and legal public holidays) after the receipt of such an appeal by the Assistant Secretary of State for

Public Affairs.

(b) In unusual circumstances as specified in this paragraph, the FOI Director may extend the time limits in paragraph (a) by written notice to the person requesting a record under this Part, which notice shall set forth the reasons for such extension and the date on which a determination or appeal decision is expected to be dispatched. No such notice shall specify a date which would result in an extension of either the initial determination period, or the appeal period, or both, for a total of more than 10 working days. As used in this paragraph "unusual circumstances" means. but only to the extent reasonably necessary to the proper processing of the particular request-

(1) The need to search for and collect the requested records from overseas posts or other establishments that are separate from the office processing the

request:

(2) The need to search for, collect, and appropriately examine a voluminous amount of separate and distinct records which are demanded in a single request;

- (3) The need for consultation, which shall be conducted with all practicable speed, with another agency having a substantial interest in the determination of the request or among two or more components of the Department of State having substantial subject matter interest therein.
- (c) Where the responsible office believes that further consideration of a request may result in the release of additional records or portions thereof, the expiration of the time period allocated by the FOI Director does not require the termination of that consideration, and it should be continued with a view to maximum disclosure of requested records within a reasonable period of time.

# § 6.8 Appeals.

- (a) Review of an initial denial under § 6.6(b) may be requested by the person who submitted the original request for a record. The review (hereinafter the appeal) must be requested in writing within 60 days of the date that the requester is informed either: (1) that the request is denied completely, or (2) that all records which are being furnished in response to his request have been released and he has been so informed.
- (b) The appeal must be sent by certified mail to the Assistant Secretary of State for Public Affairs, Chairman, Council on Classification Policy, Department of State, Washington, D.C. 20520. The time for decision on appeal begins on the date the appeal is actually received in

the Office of the Assistant Secretary. The request shall include a statement of the circumstances, reasons, or arguments advanced for insistence upon disclosure of the originally requested record and a copy of the initial denial letter.

(c) The Council on Classification Policy shall decide the appeal; there shall be a written record of the reasons for the decision and the views of each member. The decision shall be communicated in writing to the requester by a letter from the Assistant Secretary for Public Affairs. If the decision is to withhold. the letter shall identify the exemption or exemptions exercised under § 6.4, the reason, and, if the Council is prepared to give further consideration notwithstanding expiration of the time allocated for decision, the letter shall so state and indicate when that consideration is expected to be completed.

# § 6.9 Public reading room.

A public reading room or area where records may be made available is located in the Department of State, 2201 C Street, N.W., Washington, D.C. 20520. The receptionist will refer the applicant to the proper room. Fees will not be charged for access by the public to this room or the records contained therein, but fees, in accordance with § 6.14, will be charged for furnishing copies thereof. The Assistant Secretary of State for Public Affairs, in his or her discretion, may authorize persons to utilize their own portable copying equipment. Any arrangements for the use of such equipment must be consistent with security regulations of the Department of State and are subject to the availability of personnel to monitor such copying.

#### § 6.10 Manner of requesting records.

(a) Identifiable records may be requested by the public in person from 10 a.m. to 4 p.m., Department of State, 2201 C Street, N.W., Washington, D.C., where the receptionist will refer the applicant to the proper office for service and the necessary forms for making a request. Requests by mail and referrals from other agencies should be addressed to the Director, Freedom of Information Staff, Bureau of Public Affairs, Department of State, Washington, D.C. 20520.

(b) A request will not be considered as a request under this Part until actually received by the FOI Director either under paragraph (a) of this section or by referral under § 6.2(b) or § 6.11.

(c) Original or record copies of records will not be permitted to leave the custody of the lawful custodian thereof. Copies, duly certified upon request, will be furnished in lieu thereof in accordance with established fees.

(d) The burden of adequately identifying the record so requested lies with the requesting person. Such person may seek appropriate assistance from the FOI Director or a member of his or her staff, in identifying the record sought.

# § 6.11 Requests addressed to Foreign Service posts.

As indicated in § 6.10(b), requests for records will not be considered under

this Part until actually received by the FOI Director. This is true of requests addressed directly to overseas posts of the Department and Foreign Service. Accordingly, any such post receiving a request for records which it has reason to believe is intended to be a request under this Part shall send that request by the fastest means available to the FOI Director and shall so inform the requester. In the case of a request for a record which is available only at that post, the post shall forward a copy of the requested record, together with its recommendation and a copy of the request, to the FOI Director. The post may submit its recommendations in any case and, in that event, shall do so as soon as possible. A post receiving such a request, however, may review the requested record and, if it is unclassified and the post believes it should be released, the post may forward the record to the requester on its own authority and so advise the FOI Director.

## § 6.12 Closing requests.

(a) The FOI Director is authorized to consider a request closed if, within 60 working days after he has sent a letter to a requester advising the latter either of a complete denial or that some or all of the requested records have been released and are available upon payment of fees, the requester fails to respond.

(b) The Assistant Secretary of State for Public Affairs is authorized to refuse to initiate a search for records requested by any person who has previously made a request under this Part and who, upon being informed by letter that some or all of the requested records are available and the fees for services performed, refused to pay or failed to respond within 60 working days from the date of the letter.

# § 6.13 Subsequent requests for same records.

When, pursuant to request under this Part, records are released to any person, any other person subsequently requesting the same records will also receive the records unless there would be an unwarranted invasion of privacy as to one but not to the other. In such cases, search costs for subsequent requests would normally be minimal.

# § 6.14 Schedule of fees and method of payment for services rendered.

(a) The following specific fees shall be applicable with respect to services rendered to members of the public under this part:

- (3) Copies made by photostat or otherwise (per page) maximum of 5 copies will be provided...... 0.10

-(7) For each signed statement of non-availability of record—no fee

availability of record—no fee....
(8) Duplication of architectural photographs and drawings......

(b) If records requested under this Part are stored elsewhere than the headquarters of the Department of State at 2201 C Street NW., Washington, D.C., the special costs of returning such records to the headquarters shall be included in the search costs. These costs will be computed at the actual cost of transportation of either a person or the requested record between the place where the record is stored and Departmental headquarters when, for time or other reasons, it is not feasible to rely on Government mail service or diplomatic pouch. The most economical means of transportation available and consistent with the purpose of incurring special costs shall be utilized. Search costs are due and payable even if the record which was requested cannot be located after all reasonable efforts have been made, or if the Department determines that a record which has been requested, but which is exempt from disclosure under this Part, is to be withheld.

(c) Where it is anticipated that the fees chargeable under this section will amount to more than \$25 and the requester has not indicated in advance his willingness to pay fees as high as are anticipated, the requester shall be promptly notified of the amount of the anticipated fee or such portion thereof as can readily be estimated. In appropriate cases, an advance deposit may be required. The notice or request for an advance deposit shall extend an offer to the requester to confer with knowledgeable Departmental personnel in an attempt to reformulate the request in a manner which will reduce the fees and meet the needs of the requester. Dispatch of such a notice or request shall suspend the running of the period for response by the Department until a reply is received from the requester.

(d) The Assistant Secretary of State for Public Affairs, or an officer designated by the Assistant Secretary, may waive all or part of any fee provided for in this section when the Assistant Secretary or the designated officer considers it appropriate to do so.

(e) Fees for services performed other than under this Part, as described in § 6.2(b), will be charged in accordance with Parts 21 and 22 of this chapter.

(f) When no specific fee has been established for a service, or the request for a service does not fall under one of the above categories due to the amount or size or type thereof, the Assistant Secretary for Public Affairs is authorized to establish a appropriate fee, pursuant to the criteria established in Bureau of Budget Circular No. A-25, entitled "User Charges."

(g) Fees must be paid in full prior to issuance of requested copies.

(h) Remittances shall be in the form either of a personal check or bank draft drawn on a bank in the United States, a postal money order, or cash. Remittances shall be made payable to the order of the Treasury of the United States and mailed to the Director, Freedom of Information Staff, Bureau of Public Affairs, Department of State, Washington, D.C. 20520. The Department will assume no responsibility for cash which is lost in the mail.

(i) A receipt for fees paid will be given only upon request. Refund of fees paid for services actually rendered will not

be made.

(j) Notwithstanding the foregoing, materials may be furnished without charge to foreign governments, other governmental agencies or units, non-profit educational organizations, or any other applicant whenever the Assistant Secretary for Public Affairs determines that such action would further the performance of the functions of the Department of State.

# § 6.15 Opening of records for nonofficial

(a) General policy. The Department will open its records on an equitable basis to all individuals engaged in nonofficial research as soon as such action may be taken without adversely affecting the national security, the maintenance of friendly relations with other nations, and the efficient operation of the Department. The opening of departmental records to nonofficial researchers will ordinarily not take place until after the records have been transferred to the custody of the National Archives and Records Service. The opening of records for nonofficial research will generally take place on the basis of large blocks of records defined by years and/or major subject categories.

(b) Open records. (1) The records of the Department, with the exceptions stated in subparagraph (2) of this paragraph, shall be automatically declassified and opened to research when such records are 30 years old. The records for an entire year shall be opened normally on the first of January each year.

(2) Certain categories of records falling within the exemptions and specified in § 6.4 and requiring continuing protection under Executive Order 11652 or otherwise, as defined in restriction statements issued by the National Archives and Records Service, are not open for unofficial research purposes until they

are 50 or 75 years old.

(3) When, in the opinion of the Director of the Historical Office, it is administratively feasible and consistent with the national security to open blocks of records less than 30 years old, he shall so recommend to the Council on Classification Policy. On approval by the Council, such records shall be declassified and treated administratively as if they were open records as defined in subparagraph (1) of this paragraph, except that classified papers which have been identified as requiring special handling shall not be opened to researchers until they are declassified either by specific action of the Department in each case or by the automatic declassification after 30 years, as

provided for in subparagraph (1) of this paragraph.

(c) Restricted records. (1) Records of the Department other than those declassified and opened in accordance with procedures set forth above are not available to access by nonofficial researchers except to the limited extent that may be permitted under §§ 9.22 and 9.25 of this chapter.

(2) Nothing in this subsection affects the rights of persons to request and to receive copies of identifiable foreign policy documents of any age, unclassified or declassified, in accordance with the procedures set forth in §§ 6.1 through 6.14.

(d) Information on administration of this section. Correspondence with regard to open records of the Department should be addressed to the Chief, Diplomatic Branch, Civil Records Division, National Archives and Record Service, Washington, D.C. 20400. Correspondence with regard to the availability of other records of the Department under this section should be addressed to the Director, Historical Office, Bureau of Public Affairs, Department of State, Washington, D.C. 20520.

## § 6.16 Activities of advisory committees.

Any determination under section 10 (d) of the Federal Advisory Committee Act (Public Law 92-463, 86 Stat. 770) that activities of an advisory committee are concerned with matters listed in section 552(b) of Title 5 of the United States Code and that the Public interest requires such activities to be withheld from disclosure shall be made by the Deputy Under Secretary for Management. This determination shall be in writing and is final.

Public notice. In view of the imminence of the present compliance date, and since these amendments relate to Departmental organization, procedure and practices, notice and public procedure thereon (except with respect to § 6.14) are found impractical and unnecessary, and they may be made effective in less than 30 days after publication in the Feb-ERAL REGISTER. On January 13, 1975 (40 FR 2443), the Department published in the FEDERAL REGISTER a "notice of proposed rulemaking with respect to Freedom of Information Fees". Public comments have been received and carefully considered, and the regulation (then § 6.8) described in that notice, subject only to a change in paragraph (b) to more specifically describe fees for special costs, is set forth above as § 6.14. Public comment is invited for consideration in connection with possible further amendments. Such comments may be addressed to the Assistant Legal Adviser for Management, Department of State, Washington, D.C. 20520. Any comments received on or before March 21, 1975, will be considered; comments received after that time may be considered.

Effective date. These amendments are effective February 19, 1975.

Amendments to 22 CFR 6 (Departmental Regulation 108.698) published in

the FEDERAL REGISTER on April 22, 1974 (39 FR 14195) are cancelled.

Dated: February 13, 1975.

For the Secretary of State,

LAWRENCE S. EAGLEBURGER,

Acting Deputy Under

[FR Doc.75-4587 Filed 2-18-75;8:45 am]

Secretary for Management.

# Title 24—Housing and Urban Development

SUBTITLE A—OFFICE OF THE SECRETARY

[Docket No. R-75-194]

# PART 15—PRODUCTION OR DISCLOSURE OF MATERIAL OR INFORMATION

#### Interim Rule

The Department of Housing and Urban Development is amending Part 15 of Subtitle A of Title 24 of the Code of Federal Regulations in order to effectuate the amendments to section 552 of Title 5, United States Code, known as the Freedom of Information Act, enacted November 21, 1974, as Public Law 93-502.

Section 15.21(a) (7) is revised and expanded to specifically recite those classes of investigatory records compiled for law enforcement purposes production of which is exempted. This revision follows the express wording of section

2(b) of the statute.

In accordance with section 2(c) of the statute, § 15.21(b) expressly states the Department's presently implicit policy of providing reasonably segregable portions of a record after deletion of exempt portions.

In addition, certain time limits have been decreased and other proposed technical changes are set forth principally to comply with the new statutory require-

ments.

The public interest dictates that these amendments be made effective as soon as possible and, therefore, are being issued as an interim rule which will become effective on February 19, 1975, the effective date of Public Law 93-502. However, interested persons are invited to submit such written relevant comments or suggestions as they may desire. Communications should be filed, using the above docket number and title, with the Rules Docket Clerk, Office of General Counsel, Room 10245, Department of Housing and Urban Development, 451 Seventh Street SW., Washington, D.C. 20410. All relevant material received on or before April 4, 1975, will be considered by the Secretary before adoption of a final rule. Copies of comments submitted will be available during business hours, both before and after the specified closing date, at the above address, for examination by interested persons.

It should be noted that the Department has presently in effect a uniform schedule of fees for document search and duplication (24 CFR 15.14). It is intended that this fee schedule remain in

effect unchanged.

Part 15 is amended as follows:

1. In § 15.1 the first sentences of paragraphs (a) and (d) are revised to read as follows:

## § 15.1 Definitions.

As used in this Part-

(a) "Act" means section 552 of Title 5, United States Code, as amended by Public Law 90–23, 81 Stat. 54, June 5, 1967, and Public Law 93–502, 88 Stat. 1561, November 21, 1974. \* \*

(d) "Organizational unit" means any one of the several offices, staffs, divisions, or administrations of the Department including the Government National Mortgage Association (GNMA), the Federal Insurance Administration (FIA), the Office of Interstate Land Sales Registration, the Community Development Corporation, and the Federal Disaster Assistance Administration.

# § 15.3 [Amended]

- 2. The second sentence of § 15.3 is
- 3. Section 15.12(c) is revised to read as follows:
- § 15.12 Materials not published in Federal Register.
- (c) The Department shall also maintain and make available for public inspection and copying current indexes providing identifying information for the public as to any matter issued, adopted, or promulgated after July 4, 1967, and required by this section to be made available or published. The Department shall promptly publish, quarterly and distribute (by sale or otherwise) copies of each index or supplements thereto unless it determines by order published in the FEDERAL REGISTER that the publication would be unneces-sary and impracticable, in which case the Department shall nonetheless provide copies of such index on request at a cost not to exceed the direct cost of duplication.
- 4. Section 15.13 is amended by revising the heading and by revising paragraph (b) as follows:
- § 15.13 Records produced upon request when reasonably described.
- (b) When a request is made which reasonably describes a record of the Department which has been stored in the National Archives or other record centers of the General Services Administration, such record will be requested by the Department if it otherwise would be available under this part.
- 5. Section 15.14 is amended by adding an undesignated paragraph at the end to read as follows:

# § 15.14 Schedule of fees.

Records shall be furnished without charge or at a reduced charge where the

Department determines that waiver or reduction of the fee is in the public interest because furnishing the information can be considered as primarily benefiting the general public.

6. Section 15.21 is amended by revising paragraphs (a) (1) and (a) (7) and by adding a new paragraph (b) as follows:

§ 15.21 Exemptions authorized by 5 U.S.C. 552.

. (1) Specifically authorized under criteria established by an Executive order to be kept secret in the interest of national defense or foreign policy and are in fact properly classified pursuant to such Executive order.

(7) Investigatory records compiled for law enforcement purposes, but only to the extent that the production of such records would (A) interfere with enforcement proceedings, (B) deprive a person of a right to a fair trial or an impartial adjudication, (C) constitute an unwarranted invasion of personal privacy, (D) disclose the identity of a confidential source and, in the case of a record compiled by a criminal law enforcement authority in the course of a criminal investigation, or by an agency conducting a lawful national security intelligence investigation, confidential in-formation furnished only by the confidential source, (E) disclose investigative techniques and procedures, or (F) endanger the life or physical safety of law enforcement personnel.

(b) Any reasonably segregable portion of a record shall be provided to any person requesting such record after deletion of the portions which are exempt under this section.

7. In § 15.31 paragraph (b) is revised to read as follows:

# § 15.31 Information centers.

- (b) The Department also maintains an information center-
- (1) In each of its Regional Offices as

Region I (Boston)-Room 800, John F. Ken-

nedy Building, Boston, Mass. 02203.
Region II (New York)—26 Federal Plaza,
Room 3541, New York, N.Y. 10007.

Region III (Philadelphia)—Curtis Building, 6th and Walnut Streets, Philadelphia, Pa. 19106.

Region IV (Atlanta)-Room 211, Pershing Point Plaza, 1371 P eachtree Street NE., At-

lanta, Georgia 30309. egion V (Chicago—300 South Wacker Region

Region V (Chicago—300 South Wacker Drive, Chicago, Illinois 60606. Region VI (Dallas)—Room 1402, Earle Cabell Federal Building. U.S. Courthouse, 1100 Commerce Street, Dallas, Texas 75202. Region VII (Kansas City)—Federal Office Building, Room 300, 911 Walnut Street, Venesc City Viscousi 64106.

Kansas City, Missouri 64106.

Region VIII (Denver)—Federal Building, 1961 Stout Street, Denver, Colorado 80202. Region IX (San Francisco)—450 Golden Gate Avenue Post Office Box 36003, San Fran-

cisco, California 94102.

Region X (Seattle)—Arcade Plaza Building, 1321 Second Avenue, Seattle, Wash. 98101.

- (2) With respect to the mortgage loan activities of GNMA, in each FNMA Regional Office as follows:
- 5 Penn Center Plaza, Philadelphia, PA 19103. 100 Peachtree Street NW., Atlanta, GA 30303.
- 150 South Wacker Drive, Chicago, IL 60606. 2001 Bryan Tower, Suite 1200, Dallas, TX
- 3435 Wilshire Boulevard, Los Angeles, CA 90010.
- 8. Section 15.41 is revised to read as follows:

# § 15.41 Requests for records.

(a) Requests for copies of records may be made in person during normal business hours at information centers listed in § 15.31 or by mail addressed to such centers. Although oral requests may be honored, a requester may be asked to submit his request in writing.

(b) Each request must reasonably describe the desired record including the name, subject matter, and number or date, where possible, so that the record may be identified and located. In order to enable the Department to comply with the time limitations set forth in § 15.42, the envelope containing a written request and the letter itself should both clearly indicate that the subject is a Freedom of Information Act request.

(c) The request shall be accompanied by the fee or an offer to pay the fee as determined pursuant to § 15.14. At its discretion, the Department may refuse to furnish records prior to receipt of the required fee.

(d) Copies of available records shall be made as promptly as possible. Copying service shall be limited to not more than 10 copies of any single page. Records which are published or available for sale need not be reproduced.

9. A new § 15.42 is added to Subpart E as follows:

# § 15.42 Time limitations.

(a) Upon receipt of a request for records, the head of the appropriate organizational unit shall determine within ten days (excepting Saturdays, Sundays, and legal public holidays) whether to comply with such request and shall immediately notify in writing the requester of such determination and the reasons therefor and of the right of such person to request a review by the Secretary of any adverse determination.

(b) When a request for records is misdirected by the requester, the Department official receiving same shall promptly refer it to the head of the appropriate organizational unit and advise the requester that time of receipt for processing purposes will be the time when it is received by the appropriate official.

(c) A determination with respect to a request for review by the Secretary pursuant to § 15.61 shall be made within twenty days after receipt (excepting Saturdays, Sundays, and legal public holidays) and shall be immediately communicated to the person requesting review.

(d) Upon any determination to comply with a request for records, the records shall be made promptly available to the requester.

(e) In unusual circumstances as specifled in this paragraph, and subject to the concurrence of the General Counsel, the time limits prescribed in either paragraph (a) or (c) may be extended by written notice to the requester setting forth the reasons for the extension and the date on which a determination is expected to be dispatched. No such notice shall specify a date that would result in an extension for more than ten working days. As used in this paragraph, "unusual circumstances" means (but only to the extent necessary to the proper processing of the particular request) that there is a need:

 to search for and collect the requested records from field facilities or other establishments that are separate from the office processing the request;

(2) to search for, collect, and appropriately examine a voluminous amount of separate and distinct records which are demanded in a single request; or

(3) for consultation, which shall be conducted with all practicable speed, with another agency having a substantial interest in the determination of the request or among two or more organizational units of the Department having a substantial interest in the subject matter of the request.

10. The second sentence in § 15.52 is revised to read as follows:

§ 15.52 Authority to deny requests for records.

\* \* \* Any denial shall be made in writing, contain a simple statement of reasons for the denial, state that a review of the denial may be requested, set forth the steps for obtaining that review in accordance with § 15.61, and shall be signed by the official responsible for such denial. \* \* \*

11. In § 15.61 paragraph (b) is revised to read as follows:

.

§ 15.61 Administrative review.

.

(b) The decision after review will be in writing, will constitute final action of the Department on the request, and, if the denial of the request for records is in whole or in part upheld, the decision shall notify the person making the request of his right to seek judicial review under 5 U.S.C. 552(a) (4).

(5 U.S.C. 552 and sec. 7(d), 79 Stat. 670; 42 U.S.C. 3535(d); and Public Law 93-502, 88 Stat. 1561)

Effective date. These amendments shall be effective February 19, 1975.

JAMES L. MITCHELL, Acting Secretary of Housing and Urban Development. [FR Doc.75-4594 Filed 2-18-75;8:45 am] Title 28—Judicial Administration
CHAPTER I—DEPARTMENT OF
JUSTICE

[Order No. 596-75]

PART O—ORGANIZATION OF THE DEPARTMENT OF JUSTICE PART 16—PRODUCTION OR DISCLOSURE

OF MATERIAL OR INFORMATION
Freedom of Information Act Requests
Procedure and Policy

Note: This document is reprinted from the Federal Register of Wednesday, February 12, 1975 (40 FR 6494).

These amendments revise Department of Justice regulations in the light of past experience under the Freedom of Information Act and conform the regulations to the requirements of the Act as amended by Pub. L. 93-502, 88 Stat. 1561. On 13 January 1975, the Department published in the FEDERAL REGISTER (40 FR 2443) proposed fee regulations and that subject is therefore not covered by these amendments.

Present system. The present system of processing Freedom of Information requests furnishes the framework on which the amendments build. Under existing regulations requests for records are sent to the Deputy Attorney General (except that requests for records of the Immigration and Naturalization Service, the Bureau of Prisons, and the Board of Immigration Appeals are sent directly to those divisions). The Deputy Attorney General then sends the requests to the division which has primary concern with the records requested. (The term "division" includes all divisions, bureaus, offices, services, administrations, and boards of the Department, the Pardon Attorney, and Federal Prisons Industries except as otherwise provided.) That division then responds by either granting the request, denying it, or granting and denying it in part. There is a ten-day period for this initial response, with provision for extension of time. The Deputy Attorney General is specifically charged with insuring timely response and may be petitioned when a request is not answered within the applicable time limit.

Denials of initial requests may be appealed to the Attorney General within thirty days of receipt by the requester. The Attorney General then has twenty days to act on the appeal unless the time is extended. The Deputy Attorney General maintains files on all requests for information (except for requests directed to the Immigration and Naturalization Service, the Bureau of Prisons, and the Board of Immigration Appeals), and on all appeals. There is a provision for classification review of national security information more than ten years old.

Amended regulations. Discussed below are the major changes made by these amendments. There are additional minor changes, not discussed, which are either

necessary under the amended Act, or are desirable in light of the experience of the Department with the present regulations.

The amended regulations establish in the Office of the Deputy Attorney General a new unit, designated the Freedom of Information Appeals Unit, which will assist the Attorney General in processing administrative appeals of initial denials of requests under the Freedom of Information Act. This assistance was formerly provided by the Office of Legal Counsel. The organizational location of this new unit is consistent with the overall responsibility of the Deputy Attorney General to supervise the processing of Freedom of Information requests. The Office of Legal Counsel will continue to process appeals in cases where the Deputy Attorney General has participated in the initial denial, and will continue its function of providing requested guidance to all divisions of the Department on difficult issues of law associated with Freedom of Information requests.

The amended regulations, while maintaining the same general system as presently exists, establish more specific requirements for the making and processing of requests. Most of these requirements are intended to insure compliance with the new time limits imposed by the amendments to the Act. Thus, the amended regulations require that requests under the Act be clearly marked, so as to enable prompt processing by the Department. The burden of forwarding misdirected requests to the appropriate office ordinarily the Office of the Deputy Attorney General) is placed on Department personnel; but time limits do not commence to run on such requests until they have been so delivered or, with the exercise of due diligence, should have

The amended regulations make clear that authority to deny a request lies only in the division head unless otherwise specified by regulation. Allowing delegation of this authority by regulation permits divisions with widely scattered components to decentralize the initial decision authority in order to meet the time limits of the Act.

Authority to extend (in increments of five working days) is limited to division heads. It was felt that limiting this authority still further-to the Attorney General and the Deputy Attorney General-would entail more delay than the time limits of the Act permit, while placing the authority in officials lower than division heads would not be consistent with the strict control that is necessary. This procedure may result in the use of the entire ten-day extension time at the division level, but in view of the shorter standard period available at that level this appears to be the most appropriate policy. Where the extension authority is abused, the Deputy Attorney General may remove it.

The Office of the Deputy Attorney General is required to keep records of all time extensions taken. This serves the dual purpose of insuring that time requirements are met and of monitoring extensions so that the ten-day extension limit in the Act is not exceeded.

The amended regulations provide that when the time for initial reply, including any extensions, has expired and no determination has been made, the requester must be informed that he may deem this a denial and appeal to the Attorney General. This provision assures the preservation of the twenty-day period alloted to the Attorney General for appeals. In cases of unavoidable delay such as that occasioned by requests for voluminous records, a requester may be asked to defer appeal so long as the division is making diligent efforts to process the request. When a requester who accepts such a disposition believes that a division is not making diligent efforts, immediate appeal to the Attorney General provides the opportunity for relief. For this reason, provisions in the present regulations allowing petition to the Deputy Attorney General to complain of delay are eliminated as unnecessary.

The section on responses by the divisions is amended to require the letter denying a request to include the name and title or position of the person responsible for the denial. This person would be the division head or other person authorized by regulation to issue final denials. It is possible, however, that a person from another agency or division may share that status—under the regulations and the Act—if the denial is made at the instance of that person, out of regard for the primary interest or expertise of his agency or division, and with notice to him that his judgment will be relied on.

Two new paragraphs are added to the section dealing with appeals to the Attorney General, covering extensions of time and the handling of appeals where the time limits have been exhausted. In the latter case the requester will be informed of his right to treat the failure to complete administrative review as an exhaustion of administrative remedies, enabling immediate suit for judicial review. He will also be advised, however, of the reason for the delay and of the date by which a response may be expected. When the delay is unavoidable, it is anticipated the requester will avoid unnecessary litigation expenses by foregoing judicial review until the Department completes processing of the request, since a court may and presumably would grant an extension in such circumstances. No appeal to the Attorney General is provided with respect to denials by the Special Prosecutor, who is authorized, if he wishes, to establish an appeals procedure within his Office.

Extensive changes are made in Department procedures regarding the processing of requests for classified records. The present regulations provide for classification review of records only when they are over ten years old. The

amended regulations require that in all cases of requests for records classified pursuant to Executive Order 11652 or its predecessors, the Department must review the information to determine if it continues to warrant classification. This requirement applies whenever classified matter is requested, even if other exemptions are also to be asserted. Because under Executive Order 11652 national security material can ordinarily be declassified only by the originating agency, documents classified by another agency are to be treated as records of that agency for purposes of FOIA decisionmaking authority; requests pertaining to such documents are to be referred to such agency for disposition (not only as to the classified information exemption but as to all other exemptions which might be asserted); the requester is to be informed that this referral has occurred and that he may expect a reply from that other agency.

The Appeals Unit will refer classified

The Appeals Unit will refer classified material included in an appeal to the Attorney General to the Departmental Review Committee, established under part 17 of chapter I. The Committee will complete its review in ten working days unless that time is extended by the Deputy Attorney General.

Because the amendments pertain to matters of procedure and policy, the relevant provisions of the Administrative Procedure Act (5 U.S.C. 553) requiring notice of proposed rulemaking, opportunity for public participation, and delay in effective date are inapplicable. In addition, since many of these changes are necessary to achieve compliance with the amendments to the Freedom of Information Act (5 U.S.C. 552) which become effective on February 19, 1975, there is not sufficient time to receive and evaluate public comment. However, in accordance with the spirit of the public policy set forth in 5 U.S.C. 553, interested persons may submit written comments on these amendments to the Office of Legal Counsel, Department of Justice, Washington, D.C. 20530, no later than March 19, 1975. Comments should identify this order (596–75). Arrangements to inspect copies of written comments may be made by calling the Office of Robert Saloschin, Chairman, Freedom of Information Committee, Department of Justice at 202-739-2069. All comments received in this manner will be evaluated and acted upon in the same manner as if this document were a proposal.

In consideration of the above, Title 28, chapter I, parts 0 and 16 are amended as set forth below.

Effective date. These amendments become effective on February 19, 1975.

Edward H. Levi, Attorney General.

1. Section 0.18 is added to read as follows:

§ 0.18 Freedom of Information Appeals Unit.

The Freedom of Information Appeals Unit is established in the Office of the Deputy Attorney General, under the supervision of the Deputy Attorney Gen-

eral, to assist the Attorney General in acting on Freedom of Information appeals under § 16.7 of this chapter, except that in the case of appeals from initial decisions in which the Deputy Attorney participated this assistance shall be provided by the Office of Legal Counsel.

2. The last sentence of § 16.2 is revised to read as follows:

. .

§ 16.2 Public reference facilities.

Each of these public reference facilities will maintain, make available for public inspection and copying, and publish quarterly (unless the applicable division determines by order published in the Federal Register that the publication would be unnecessary or impracticable), a current index of the materials available at that facility which are required to be indexed by 5 U.S.C. 552(a) (2).

3. Section 16.3 is amended by revoking paragraph (e) and by revising paragraphs (a), (b) and (d) as follows:

§ 16.3 Requests for identifiable records and copies.

(a) How made and addressed. A request for a record of the Department which is not customarily made available and which is not available in a public reference facility as described in § 16.2, shall be made in writing, with the envelope and the letter clearly marked "FREEDOM OF INFORMATION REQUEST." Or "INFORMATION REQUEST." All such requests shall be addressed to the Deputy Attorney General, Department of Justice, Washington, D.C. 20530, except that requests for records of the following divisions shall be addressed as follows:

Bureau of Prisons (including Federal Prison Industries)—Director, Bureau of Prisons, 320 First Street NW., Washington, D.C. 20534.

Board of Immigration Appeals—Chairman, Board of Immigration Appeals, Department of Justice, Washington, D.C. 20530.

Law Enforcement Assistance Administration—Administrator, Law Enforcement Assistance Administration, 633 Indiana Ave. NW., Washington, D.C. 20531. Immigration and Naturalization Service—As

set forth in 8 CFR Part 103.

Any request for information not marked and addressed as specified in this not paragraph will be so marked by Department personnel as soon as it is properly identified, and forwarded immediately to the appropriate office as specified above. A request improperly addressed will not be deemed to have been received for purposes of the time period set forth in 5 U.S.C. 552(a) (6) (A) (i) until forwarding to the appropriate office has been effected, or until such forwarding would have been effected with the exercise of due diligence by Department personnel. On receipt of an improperly addressed request forwarded as set forth above to the appropriate office, such office shall notify the requester of the date on which the time period commenced to

(b) Request should reasonably describe the records sought. A request for access to records should sufficiently

identify the records requested to enable or removal of the authority of a division Department personnel to locate them with a reasonable amount of effort. Where possible, specific information regarding dates, titles, file designations, and other information which may help identify the records should be supplied by the requester. If the request relates to a matter in pending litigation, the court and its location should be identifled.

(d) Categorical Requests—(1) Records must be reasonably described. A request for all records falling within a reasonably specific category shall be regarded as conforming to the requirement that records be reasonably described if it enables the records requested to be identified by any process that is not unreasonably burdensome or disruptive of

Department operations.

- (2) Assistance in reformulating nonconforming requests. If it is determined that a request does not reasonably describe the records sought, as specified in paragraph (d) (1) of this section, the response denying the request on that ground shall specify the reasons why the request failed to meet the requirements of paragraph (d) (1) of this section and shall extend to the requester an opportunity to confer with Department personnel in order to attempt to reformulate the request in a manner which will meet the needs of the requester and the requirements of paragraph (d) (1) of this section.
  - (e) [Revoked]
- 4. Section 16.4 is revised to read as follows:
- § 16.4 Requests referred to division primarily concerned.
- (a) Referral to responsible division. The Deputy Attorney General shall, promptly upon receipt of a request for Department records, forward the request to the division of the Department which has primary concern with the records requested. As used in this subpart, the term "division" includes all divisions, bureaus, offices, services, administrations, and boards of the Department, the Pardon Attorney and Federal Prison Industries, except as otherwise expressly provided. As used in this subpart, the term 'responsible division' means, with respect to a particular request, the division to which the Deputy Attorney General forwards the request pursuant to this paragraph or, if the request is not one which is to be addressed to the Deputy Attorney General under § 16.3(a), the division to which the request is properly addressed thereunder.

(b) Deputy Attorney General shall assure timely response. The Deputy Attorney General shall periodically review the practices of the divisions in meeting the time requirements set out in \$ 16.5. including the granting of extensions of time, and shall take such action to promote timely responses as he deems appropriate. Such action may include, but is not limited to, removal from a division of a request or class of requests

to grant extensions, as specified in § 16.5(f).

- (c) Records to be kept by Deputy Attorney General. The Deputy Attorney General shall retain or be furnished with a file copy of each request which is required to be addressed to him pursuant to \$16.3(a). With respect to such requests he shall maintain records to show the date of receipt by the Department (and, in the case of improperly addressed requests, the date of receipt by the appropriate office after forwarding pursuant to \$16.3(a)), the responsible division to which it was forwarded under this section, and the date of such forwarding. The Board of Immigration Appeals, the Bureau of Prisons, the Immigration and Naturalization Service and the Law Enforcement Assistance Administration, respectively, shall retain or be furnished with file copies of requests required to be addressed to them pursuant to § 16.3(a), and shall maintain records to show the date of receipt by the Department (and, in the case of improperly addressed requests, the date of receipt by the appropriate office after forwarding pursuant to § 16.3(a)).
- 5. Section 16.5 is revised to read as follows:
- § 16.5 Prompt response by responsible division.
- (a) Response within ten days. Within ten days (excluding Saturdays, Sundays and legal public holidays) of the receipt of a request by the Department (or, in the case of an improperly addressed request, of its receipt by the appropriate office after forwarding pursuant to § 16.3(a)) the responsible division shall determine whether to comply with or to deny such request and dispatch such determination to the requester unless an extension is made under paragraph (c) of this section.
- (b) Authority to deny request. Unless otherwise specified by division regulation, only the head of a division may deny a request, and is the "person responsible for the denial" within the meaning of 5 U.S.C. 552(a). When a denial is made at the request of another agency or division, and out of regard for its primary interest or expertise, the person in that agency or division responsible for the request to deny may also be a "person responsible for the denial" if, before his final recommendation is accepted, he is advised that he will be so designated under § 16.6(b)(2).
- (c) Extension of time. In unusual circumstances as specified in this paragraph, the head of a division may extend the time for initial determination on requests up to a total of ten days (excluding Saturdays, Sundays, and legal public holidays). Extensions shall be granted in increments of five days or less and shall be made by written notice to the requester which sets forth the reason for the extension and the date on which a determination is expected to be dispatched. As used in this paragraph "unusual circumstances" means, but only

to the extent necessary to the proper processing of the request-

(1) The need to search for and collect the requested records from field facilities or other establishments that are separate from the office processing the request;

(2) The need to search for, collect, and appropriately examine a voluminous amount of separate and distinct records which are demanded in a single request;

(3) The need for consultation, which shall be conducted with all practicable speed, with another agency or another division having substantial interest in the determination of the request, or the need for consultation among two or more components of the responsible division having substantial subject matter interest

- (d) Treatment of delay as a denial. If no determination has been dispatched at the end of the ten-day period, or the last extension thereof, the requester may deem his request denied, and exercise a right of appeal in accordance with § 16.7. When no determination can be dispatched within the applicable time limit, the responsible division shall nevertheless continue to process the request; on expiration of the time limit it shall inform the requester of the reason for the delay, of the date on which a determination may be expected to be dispatched, and of his right to treat the delay as a denial and to appeal to the Attorney General in accordance with § 16.7; and it may ask the requester to forego appeal until a determination is made.
- (e) Copies of extension notices and delay advisories maintained by Deputy Attorney General. Copies of all extension notices issued under paragraph (c) of this section and delay advisories issued under paragraph (d) of this section shall be supplied to and maintained by the Deputy Attorney General.
- (f) Removal by Deputy Attorney General. The Deputy Attorney General may remove any request or class of requests from the division to which it is referable under this part. The Deputy Attorney General may remove from a division the authority to grant extensions of time under this section. In event of such action the Deputy Attorney General shall perform the functions of the head of that division with respect to the removed requests or authority.
- 6. Section 16.6 is amended by revising paragraphs (b) and (d) as follows:
- § 16.6 Responses by divisions: Form and content.
- (b) Form of denial. A reply denying a written request for a record shall be in writing, signed by the head of the responsible division (or other person authorized by regulation to deny requests) and shall include:
- (1) Exemption category. A reference to the specific exemption or exemptions under the Freedom of Information Act authorizing the withholding of the record, a brief explanation of how the

exemption applies to the record withheld, and, where relevant, a brief statement of why a discretionary release is

not appropriate; and

(2) Person responsible. The name and title or position of the person or persons responsible for the denial under § 16.5 (b), provided, that no person not an employee of the responsible division shall be so designated unless he has been advised that he will be so designated before his final recommendation is accepted; and

- (3) Administrative appeal and judicial review. A statement that the denial may be appealed under § 16.7(a) within thirty days by writing to the Attorney General (Attention: Freedom of Information Appeals Unit), Department of Justice, Washington, D.C. 20530, that the envelope and letter should be clearly marked: "FREEDOM OF INFORMA-TION APPEAL" or "INFORMATION APPEAL," and that judicial review will thereafter be available in the district in which the requester resides or has his principal place of business or the district in which the agency records are situated or the District of Columbia. Provided, however, that a denial by the Office of the Watergate Special Prosecution Force shall instead of the foregoing describe any internal appeals procedure which it may establish or, in absence of such procedure, advise the requester that judicial review is available in the districts set forth above.
- (d) Copy of response to Deputy Attorney General. The Deputy Attorney General shall be provided and shall maintain a copy of each denial letter; each notification under paragraph (c) of this section; and each letter advising the requester of the determination to grant the request, except such grant letters issued by the Board of Immigration Appeals, the Bureau of Prisons, the Immigration and Naturalization Service, and the Law Enforcement Assistance Administration.
  - 7. Section 16.7 is revised as follows:

# § 16.7 Appeals to the Attorney General from initial denials.

(a) Appeals to the Attorney General. When a request for records has been denied in whole or in part by a head of a division or other person authorized to deny requests, the requester may, within thirty days of its receipt, appeal the denial to the Attorney General; except that no appeal to the Attorney General shall lie from a denial of a request for records of the Office of the Watergate Special Prosecution Force, which is hereby authorized to establish an internal appeals procedure. Appeal to the Attorney General shall be in writing, addressed to the Attorney General (Attention: Freedom of Information Appeals Unit), Department of Justice, Washington, D.C. 20530, and both the envelope and the letter shall be clearly marked: "FREEDOM OF INFORMATION AP-PEAL" or "INFORMATION APPEAL." An appeal not so addressed and marked will be so marked by Department per-

sonnel as soon as it is properly identified, and forwarded immediately to the Freedom of Information Appeals Unit. An appeal improperly addressed will not be deemed to have been received for purposes of the time period set forth in 5 U.S.C. 552(a) (6) (A) (ii) and for purposes of paragraph (b) of this section until the Appeals Unit receives the request or would have done so with the exercise of due diligence by Department personnel.

(b) Action within twenty working days. The Attorney General will act upon the appeal within twenty days (excluding Saturdays, Sundays and legal public holidays) of its receipt, unless an extension is made under paragraph (c)

of this section.

(c) Extension of time. In unusual circumstances as specified in this paragraph, the time for action on an appeal may be extended up to ten days (excluding Saturdays, Sundays, and legal public holidays) minus any extension granted at the initial request level pursuant to § 16.5(c). Such extension shall be made by written notice to the requester which sets forth the reason for the extension and the date on which a determination is expected to be dispatched. As used in this paragraph "unusual circumstances" means, but only to the extent necessary to the proper processing of the appeal-

(1) The need to search for and collect the requested records from field facilities or other establishments that are separate from the office processing

the request;

(2) The need to search for, collect, and appropriately examine a voluminous amount of separate and distinct records which are demanded in a single request;

(3) The need for consultation, which shall be conducted with all practicablespeed, with another agency or another division having substantial interest in the determination of the request or the need for consultatio namong two or more components of the responsible division having substantial subject matter inter-

est therein.

(d) Treatment of delay as a denial. If no determination on the appeal has been dispatched at the end of the twenty-day period or the last extension thereof, the requester is deemed to have exhausted his administrative remedies, giving rise to a right of review in a district court of the United States as specified in 5 U.S.C. 552(a) (4). When no determination can be dispatched within the applicable time limit, the appeal will nevertheless continue to be processed; on expiration of the time limit the requester shall be informed of the reason for the delay, of the date on which a determination may be expected to be dispatched, and of his right to seek judicial review in the United States district court in the district in which he resides or has his principal place of business, the district in which the Department records are situated or the District of Columbia. The requester may be asked to forego judicial review until determination of the appeal.

(e) Form of action on appeal. The Attorney General's determination on appeal shall be in writing. An affirmance in whole or in part of a denial on appeal shall include (1) a reference to the specific exemption or exemptions under the Freedom of Information Act authorizing the withholding of the record, a brief explanation of how the exemption applies to the record withheld, and, where relevant, a brief statement of why a discretionary release is not appropriate; and (2) a statement that judicial review of the denial is available in the district in which the requester resides or has his principal place of business, the district in which the agency records are situated or the District of Columbia.

(f) Copies to Deputy Attorney General. Copies of all appeals, all actions on appeal, all extension notices issued under paragraph (c) of this section, and all delay advisories issued under paragraph (d) of this section shall be supplied to and maintained by the Deputy Attorney

General

8. Section 16.8(c) is revised to read as follows:

# § 16.8 Maintenance of files.

- (c) Protection of privacy. Where the release of the identity of a requester, or other identifying details related to a request, would constitute a clearly unwarranted invasion of personal privacy, the Deputy Attorney General shall delete identifying details from the copies of documents maintained in the public file established under paragraph (b) of this section.
- 9. Section 16.10(b) is revised to read as follows:

# § 16.10 Exemptions.

(b) (1) In processing requests for information classified pursuant to Executive Order 11652, the responsible division shall review the information to determine whether it continues to warrant classification under the criteria of sections 1 and 5 (B), (C), (D), and (E) of the Executive Order. Information which no longer warrants classification under these criteria shall be declassified and shall not be withheld on the basis of 5 U.S.C. 552(b) (1). No record remaining classified after such review shall be withheld by a division on the basis of any exemption other than 5 U.S.C. 552(b)(1) unless in addition to such other exemption it is also asserted that the record is exempt under 5 U.S.C. 552(b) (1).

(2) The Freedom of Information Appeals Unit shall, upon receipt of any appeal from an initial denial based in whole or in part upon 5 U.S.C. 552(b) (1), refer to the Departmental Review Committee, established in Part 17 of this chapter, any portion of the request as to which that exemption was asserted at the initial level. Within ten days (excluding Saturdays, Sundays and legal public holidays) of receipt of such referral (unless such period is extended by the Deputy Attorney General), the Committee shall advise

the Appeals Unit whether all or any portion of the material referred warrants continued classification under the criteria of Executive Order 11652.

(3) When a request for Department records encompasses information classified by another agency, or by a division of the Department other than the responsible division, the responsible division shall refer that portion of the request to the originating agency or division for determination as to all issues in accordance with the Freedom of Information Act. In the case of a referral to another agency under this paragraph, the requester shall be notified that such portion of his request has been so referred and that he may expect a determination from that agency. In the case of a referral to another division under this paragraph, the requester need not be notified, the original date of receipt of the request as established under this section shall continue to govern for purposes of all time limits, and the originating division shall advise the division receiving the request of its determination. (5 U.S.C. 301; 5 U.S.C. 552 as amended by Public Law 93-502, 88 Stat. 1561)

[FR Doc.75-4043 Filed 2-10-75;12:38 pm]

Order No. 599-751

# PART 16-PRODUCTION OR DISCLOSURE OF INFORMATION .

## Fees for the Production or Disclosure of Information

On January 13, 1975 there was published in the FEDERAL REGISTER (40 FR 2443) a notice of proposed rulemaking with proposed amendments to § 16.9 of title 28 of the Code of Federal Regulations. The proposed amendments revised Justice Department fee schedules to bring them into conformity with the 1974 amendments to the Freedom of Information Act, Public Law 93-502, 88 Stat. 1561. Three comments were received. One suggested a further change to § 16.-9(a) which would incorporate the language of the 1974 amendments to the Freedom of Information Act regarding reduction or elimination of fees in the public interest. To accommodate this suggestion the phrase "because furnishing the information primarily benefits the general public" is added to the pro-posed language for the last sentence of § 19.9(a) directing waiver of fees when it is in the public interest.

Another comment questioned the propriety of that part of § 16.9(c) which states that where costs in excess of \$25 are anticipated, and the requester has not indicated his willingness to bear such costs, the request will not be deemed re-

ceived until the requester is notified of the anticipated costs and agrees to bear them. (Such notification must be given promptly and in any event within five working days.) That portion of the regulation was written to protect a requester from unexpectedly burdensome fees without placing the agency in violation of the strict time limits for initial response which the 1974 amendments provide. The actual delay involved is slight, because anticipated costs must be promptly determined. Moreover, a requester may avoid application of the provision entirely by stating in his request that he is willing to bear all costs involved, or all costs up to a specified limit which in fact is adequate to cover the anticipated fee. For these reasons, no change is made on the basis of this comment.

In response to the third comment, the provision of the present regulations requiring checks to be made payable to the Treasury of the United States, which is now correct, is left unchanged.

With the revisions discussed above, § 16.9 is hereby amended as set forth below.

Effective date. These amendments are effective February 19, 1975.

Dated: February 13, 1975.

EDWARD H. LEVI, Attorney General.

Section 16.9 of 28 CFR Part 16 is amended as set forth below.

§ 16.9 [Amended]

1. By revising the first sentence in paragraph (a) to read as follows:

(a) When charged. Fees pursuant to 31 U.S.C. 483a and 5 U.S.C. 552 shall be charged according to the schedules contained in paragraph (b) of this section for services rendered in responding to requests for Justice Department records under this subpart unless the official of the Department making the initial or appeal decision determines that such charges, or a portion thereof, are not in the public interest because furnishing the information primarily benefits the general public.

2. By substituting "\$1.00" for "\$1.25"

in paragraph (b) (2).

3. By revoking and reserving paragraph (b) (3).

4. By substituting "\$2.00" for "\$3.75" in paragraph (b) (6).

5. By deleting the word "ordinarily" in the second sentence of paragraph

6. By deleting the third and fourth sentences of paragraph (b) (7).

7. By amending and revising paragraph (b) (8) to read as follows:

(8) Computerized Records.

(i) Computer time charges (includes personnel cost).

por dotate code).
<ol> <li>Central processor charge per hour. \$188.00</li> <li>Main storage charge per 1,000</li> </ol>
bytes per hour
3. Channel charges per hour
4. Card reading per 1,000 cards 20
5. Printing per 1,000 lines
6. Card punching per 1,000 cards 10.76
7. Tape mount50
8. Specific device charges:
a. IBM 2260 Cathode ray tube or
equivalent per hour 4.20
b. IBM 3330 Disk storage or
equivalent per hour 39.72
c. IBM 2314 Disk storage or
equivalent per hour 39.72
d. IBM 3420 Tape Drive or equiva-
lent per hour 44.59
(ii) Material charges.
1. One-part paper per 1,000 \$11.00
2. Two-part paper per 1,000 17.63
3. Three-part paper per 1,000 28.95
4. Four-part paper per 1,000: 37.52
5. Five-part paper per 1,000 50.83
6. Stock Hollerith cards per 1,000 1.78
7. Magnetic tape per reel 9.50
8. Disk pack, each 775.00
8. By inserting after paragraph (b)

(8), two new paragraphs (b) (9) and (b) (10) as follows: (9) Tape recordings and other audio

records. (i) Personnel charges. Personnel charges in connection with the duplication of audio records shall be charged in accordance with paragraph (b) (2) or (b) (6) of this section, whichever is

appropriate. (ii) Material charges.

1.	45	minute	cassette	\$0.56
2.	60	minute	cassette	. 60
3.	90	minute	cassette	. 77

(10) Other charges. When a response to a request requires services or materials other than the common ones described in paragraphs (b) (1) through (b) (9) of this section, the direct cost of such services or materials to the government may be charged, but only if the requester has been notified of such cost before it is incurred.

9. By amending and revising para-

graph (c) to read as follows:

(c) Notice of anticipated fees in excess of \$25. Where it is anticipated that the fees chargeable under this section will amount to more than \$25, and the requester has not indicated in advance his willingness to pay fees as high as are anticipated, the requester shall be notifled of the amount of the anticipated fee or such portion thereof as can readily be estimated. In such cases, a request will not be deemed to have been received until the requester is notified of the anticipated cost and agrees to bear it.

Such a notification shall be transmitted as soon as possible, but in any event within five working days, giving the best estimate then available. The notification shall offer the requester the opportunity to confer with Department personnel with the object of reformulating the request so as to meet his needs at lower cost.

10. By inserting after paragraph (d) a new paragraph (e) as follows:

(e) Advance deposit.

(i) Where the anticipated fee chargeable under this section exceeds \$25, an advance deposit of 25% of the anticipated fee or \$25, whichever is greater, may be required.

(ii) Where a requester has previously failed to pay a fee under this section, an advance deposit of the full amount of the anticipated fee may be required.

11. By adding a new paragraph (f) as follows:

(f) Other services. Nothing in this section shall be construed to entitle any person, as of right, to any services or materials to which such person is not entitled under 5 U.S.C. 552.

(5 U.S.C. 301, 552, as amended by Pub. L. 93-502, 88 Stat. 1561)

[FR Doc.75-4463 Filed 2-18-75:8:45 am]

## Title 29—Labor

# SUBTITLE A-OFFICE OF THE SECRETARY OF LABOR

# PART 70—EXAMINATION AND COPYING OF DEPARTMENT OF LABOR RECORDS

The Freedom of Information Act was amended by Pub. L. 93-502, 88 Stat. 1561, effective February 19, 1975, to clarify and liberalize the requirements for, and to expedite, the disclosure of information.

On January 15, 1975, a proposed revision of Subpart B of 29 CFR Part 70 was published at 40 FR 2705 with a request for comments on or before February 5, 1975.

One comment has been received objecting to the charge of \$2.50 for each quarter hour for a search requiring the services of professional or supervisory personnel as provided in § 70.62(a) (ii) on the ground that this charge will prevent legitimate researchers from obtaining access to Department of Labor records. The fee is based upon the average hourly rate of those professional and supervisory personnel who would be conducting the search. In addition, § 70.67 of these regulations provides for the waiver or reduction of fees by disclosure officers under proper circumstances.

The amended Freedom of Information Act also necessitates a revision of Sub-parts A and C of 29 CFR Part 70 to bring them into conformity with the amendments made by Pub. L. 93-502. As the changes in the rules in these subparts as set forth below consist only of textual adjustments to conform to the amended language of the statute, and of interpretative rules, general statements of policy, and rules of procedure and practice to implement the amended Act, it is unnecessary that the revised Sub-

parts A and C of Part 70 be published for notice and comment.

The more important changes made by the revision of the rules in this part other than those made in the adoption of the revised Subpart B relating to charges for searching and copying services, give effect to the new requirements of the statute concerning publication of indexes of specified materials, to new time limitations for responding to requests for access to records and to appeals from denials of access (including procedures for expediting, noting, and acknowledging receipt of requests and appeals), to new provisions requiring identification of the persons responsible for denials of requests, and to the amendments made to the Act's exemptions protecting materials classified pursuant to executive order and investigatory records compiled for law enforcement purposes.

The general effect of these regulations is to implement the liberalized provisions of the amendments. The benefit of the reduced fees should be given immediately to the public, in accordance with the provisions of the law. Other provisions of benefit to the public are effective under the statute on February 19, 1975, and should be implemented immediately. I find the above reasons constitute good cause for making these regulations effective immediately. I further find that it is contrary to the public interest to delay these benefits for 30 days. These amendments shall be effective February 19, 1975.

29 CFR Part 70 is Accordingly, 29 amended as follows:

1. The part heading is amended to read as set forth above.

2. The section table and text of Part 70 are revised to read as follows:

# Subpart A-General

#### INTRODUCTORY

Purpose and scope of this part. 70.1 Definitions.

#### AVAILABILITY OF PUBLISHED INFORMATION

Information published in the FEDERAL 70.5 70.6 Information in Department of Labor

publications.

Published indexes.

#### INFORMATION AVAILABLE ON REQUEST

Policy of disclosure.

Records of administrative proceedings.

70.13 Evaluation reports on external pro-

Policy statements and interpretations. Staff manuals and instructions, 70.15

Indexes to certain records.

#### RESTRICTIONS ON DISCLOSURE

Records not disclosable. 70.21

Records disclosure of which may be refused.

Internal rules and practices. 70 23

Trade secrets and privileged or confidential information. 70 24

Inter-agency and intra-agency mem-70.25 oranda and letters.

Personnel, medical, and similar files. 70.26 Investigatory records compiled for law 70.27 enforcement purposes.

70.28 Well information. 70.29 Partial disclosure.

70.30

Withdrawal of originals.
Record of concern to more than one agency.

#### FACILITIES FOR DISCLOSURE

Where information may be obtained. Titles and addresses of the responsible officials of various agencies. 70.36.

## PROCEDURE FOR DISCLOSURE

Applicability of procedures. 70 41

Submittal of requests for access to 70.42 records.

To whom to direct requests. 70.43

70.44 Description of information requested. Deficient descriptions. 70.45

Requests for categories of records. 70.46

Receipt by agency of request; ac-knowledgment. 70.47

Action on request. 70.49 Form of denials.

70.50 Appeals from denial of requests.

Receipt by Solicitor of appeal; acknowledgment.

70.52 Action on appeals.

# Time Limitations for Action on Requests and Appeals

70.53 Period within which action on request shall be taken.

Period within which action on appeal 70.54 shall be taken.

Extension of period for taking action on request or appeal.

# Subpart B—Copies of Records and Fees for Services

#### SPECIAL SEARCHING AND COPYING SERVICES

70.61 Charges for services, generally.

70.62 Search and copying charges.

70.63 Computerized records.

70.64 Payment of fees.

70.65 Standard fees not charged in certain circumstances.

70.66 Services performed without charge. 70.67 Waiver or reduction of fees by disclosure officer.

AUTHENTICATION; SPECIAL STUDIES AND COMPILATIONS

Authentication of copies.

70.69 Special studies and compilations.

# Subpart C—Special Rules Applicable to Certain Specific Records

70.71 Authority of agency officials in Department of Labor.

70.72 Supplementary regulations currently in force.

70.73 Bureau of Labor Statistics.

Employment Standards Administration.

70.75 Employees' Compensation Appeals 70.76 Labor-Management Services Admin-

istration. Manpower Administration 70.77

70.78

Occupational Safety and Health Administration.

Bureau of International Labor Affairs.

AUTHORITY: (5 U.S.C. 301, 552, as amended by Pub. L. 93-502), 88 Stat. 1561; (29 U.S.C. 9(b)); Reorganization Plan No. 6 of 1950, 64 Stat. 1263 (5 U.S.C. Appendix).

# Subpart A-General

# INTRODUCTORY

# § 70.1 Purpose and scope of this part.

This part contains the general rules of the Department of Labor providing for public access to information from records of the Department. These regula-tions implement 5 U.S.C. 552, the Free-dom of Information Act, as amended, and the policy of the Department of the Secretary may invest any officer or Labor to disseminate information on employee of the Department with the aumatters of interest to the public and to thority to disclose information from disclose to members of the public on request all information contained in records in its custody insofar as is compatible with the discharge of its responsibilities and consistent with law. This part sets forth generally the categories of records accessible to the public, the types of records subject to prohibitions or restrictions on disclosure, and the places at which and procedure whereby members of the public may obtain access to and inspect and copy information from records in the custody of the Department of Labor. Rules providing for the copying of records made available for inspection and for special services furnished in locating and copying records and fees therefor are set forth in Subpart B. Rules applying the general provisions of this part to requests for disclosure of particular types of records concerned with certain specific programs of the Department are set forth in Subbart C or in the titles and parts of the Code of Federal Regulations referred to in that subpart.

### § 70.2 Definitions.

As used in this part—

(a) The terms "agency," "person," "party," "rule," "rulemaking," "order," and "adjudication" have the meaning attributed to these terms by the definitions in 5 U.S.C. 551, except where the context demonstrates that a different meaning is intended, and except that for purposes of the Freedom of Information Act the term "agency" as defined in 5 U.S.C. 551 includes any executive department. military department, Government corporation, Government controlled corporation, or other establishment in the executive branch of the Government (including the Executive Office of the President) or any independent regulatory agency.

(b) Having "custody" of a Government record means having such charge of it as to be responsible for keeping it available for governmental use as required and protecting, preserving, and exercising such control over it as may be necessary for that purpose. Custody of a record is not synonymous with, and does not require, actual physical possession of the record and a record may be in the custody of an office or officer at a place other than that where the record

is located.

(c) The terms "disclosure officer," "responsible officials," and "officer authorized to disclose information from Department records" refer generally to those officials in the Department of Labor to whom requests to inspect or copy record information in their custody may be addressed as provided in \$ 70.43 and who must pass on such requests as provided in §§ 70.46-70.49. The Solicitor of Labor, in the case of appeals as provided in §§ 70.50-70.52, and with respect to withdrawal of originals as provided in § 70.30, is the disclosure officer for such purposes, as is the Secretary of Labor when he takes action under § 70.30. As the head of the Department,

particular records.

#### AVAILABILITY OF PUBLISHED INFORMATION

§ 70.5 Information published in the Federal Register.

(a) General. In accordance with the provisions of 5 U.S.C. 552(a) (1), basic information concerning the organization, operations, functions, substantive and procedural rules and regulations, officials, office locations, and allocation of responsibilities for functions and programs of the Department of Labor is published in the FEDERAL REGISTER for the guidance of the public. Among other things, such published information describes the central and field organization of the Department and its component units, identifies the persons in charge of its various programs and offices to whom submittals or requests may be made and from whom decisions may be obtained and the places at which they are stationed, outlines generally the course and method by which the various functions of the Department are channeled and determined, and makes plain the nature and requirements of available formal and informal procedures for official action. Indexes to the FEDERAL REGISTER are published in each daily issue and compiled currently on a monthly, quarterly, and annual basis. Copies of the FEDERAL REG-ISTER and its indexes are available in many libraries, may be examined in reference facilities and offices of the Department of Labor in which reference copies are maintained, and may be purchased from the Superintendent of Documents, Government Printing Office, Washington, D.C. 20402. No formal request to examine documents published in the Federal Register is necessary to inspect them at the place where they are kept. Much of the information published in the FEDERAL REGISTER is available in compiled form as stated in paragraphs (b) and (c) of this section.

(b) Government Organization Manual. The Government Organization Manual, published annually as a special edition of the FEDERAL REGISTER provides a compact source of information concerning the general and field organization of the Department of Labor, location of Department offices, division of functions among various component units of the Department, and the names and addresses of major officers and those in charge of particular programs or offices. Copies of the Manual may be examined without formal request in any office of the Department of Labor and are available for purchase from the Superintendent of Documents.

(c) Code of Federal Regulations. Titles 20, 29, and 41 of the Code of Federal Regulations, which are revised annually, contain a compilation of documents published by the Department of Labor in the FEDERAL REGISTER which set forth sub-

stantive and procedural rules and regulations of the Department and statements of general policy or interpreta-

tions of general applicability formulated and adopted by the Department. The published rules include descriptions of forms available or the places at which forms and instructions for their use may be obtained, and provide guidance as to the scope and contents of any required papers, reports or examinations. Copies of the pertinent titles of the Code of Federal Regulations are available in many libraries and may be purchased from the Superintendent of Documents. Reference copies maintained in offices of the Department of Labor are available for examination without formal request.

#### § 70.6 Information in Department of Labor publications.

Copies of informational material, such as press releases, pamphlets, and other material ordinarily made available to the public without cost as part of a public information program, including reprints of individual parts of the Code of Federal Regulations relating to programs affecting substantial segments of the general public, shall be available upon oral or written request so long as an adequate supply exists. Copies of informational publications of the Department which may be purchased from the Superintendent of Documents may be inspected in those offices of the Department in which reference copies are available. Compliance with the formal procedures provided in this part for obtaining access to Department records is not necessary for access to the foregoing materials.

## § 70.7 Published indexes.

The informational publications available from the Department may include indexes to materials published or contained in its records. Pursuant to the provisions of 5 U.S.C. 552(a)(2) as amended, they will in any event include the current indexes required by the Freedom of Information Act to be maintained and made available for inspecting and copying, except as otherwise provided by published order, as noted below. These indexes provide identifying information for the public as to any matter in the following categories issued. adopted, or promulgated after July 4, 1967: Final opinions and orders made in the adjudication of cases (including concurring and dissenting opinions if any); statements of policy and interpretations adopted but not published in the Federal Register; and administrative staff manuals and instructions to staff that affect a member of the public. As promptly as possible after adoption of this part, such indexes will be made available in published form and offered for sale or otherwise distributed to make them readily available to members of the public. Thereafter, updated indexes or supplements shall be published not less frequently than quarterly to keep the indexes current. In the event that publication of any such index is determined to be unnecessary and impracticable, the order of the agency head so determining shall be published in the FEDERAL REGISTER. Where current indexes of the required matter in the foregoing categories are published commercially and made readily available for public use by the agency no separate publication by the Department will be necessary.

## Information Available on Request

## § 70.11 Policy of disclosure.

(a) Records made available pursuant to Freedom of Information Act. Upon the request of any person submitted in accordance with provisions of this part for access to records reasonably described by the requester which are records of the Department of Labor or any of its constituent agencies in the custody of any official of the Department, such records shall be made available as provided in this part for inspection and copying unless specifically exempt from disclosure under the provisions of 5 U.S.C. 552, subsection (b) thereof, and

§§ 70.21 through 70.28. (b) Records made available notwithstanding exemption from requirements of Freedom of Information Act. Except in the case of records of which disclosure is prohibited, as described in § 70.21. each officer of the Department authorized to disclose information from Department records under the provisions of this part shall make available for inspection and copying pursuant to the applicable procedures any document from the records in his custody if he determines that, notwithstanding the applicability or possible applicability of an exemption from disclosure, the requested inspection or copying furthers the public interest and does not impede the discharge of any of the functions of

the Labor Department. (See § 70.22(b).)
(c) Implementation of disclosure policy. To provide for a maximum of availability of Department records under the
foregoing policy, the provisions of § 70.29
of this part will be applied where appropriate. Rules to effectuate the application
of this policy of disclosure to specific
types of information from Department
records are set forth in §§ 70.12 through
70.15.

## § 70.12 Records of administrative proceedings.

(a) Rulemaking proceedings. All papers and documents made a part of the official record in administrative proceedings conducted by the Department of Labor in connection with the issuance, amendment, or revocation of rules and regulations or determinations having general applicability or legal effect with respect to members of the public or a class thereof shall be made available for public inspection and copying at reasonable times during business hours by the officer responsible for their custody at the place where such records are kept. Arrangements for such inspection may be made in response to oral requests, but a register shall be kept identifying the persons who inspect the records and the times at which they do so. Official records of other rulemaking proceedings shall be made available in accordance with the policies set forth in § 70.11 and the special regulations contained or cited in Subpart C.

(b) Adjudication proceedings. All final opinions, including concurring and dissenting opinions, as well sa orders, made in the administrative adjudication of cases, shall be made available for inspection and copying upon request pursuant to the provisions in this part. Except where an exemption provided by 5 U.S.C. 552 must be asserted in the public interest to prevent a clearly unwarranted invasion of personal privacy or violation of law or to ensure the proper discharge of functions of the Department of Labor, all papers and documents made a part of the official record in adjudication proceedings conducted by the Department shall be made available for public inspection and copying on request as provided in this part or in specific regulations pertinent to such adjudications as referred to in Subpart C.

## § 70.13 Evaluation reports on external programs.

Except in the case of records of which disclosure is prohibited, as described in § 70.21, each officer of the Department authorized to disclose information from Department records under the provisions of this part shall make available for inspection and copying promptly after receipt thereof, any document or part thereof containing a final evaluation by an independent contractor of any external program or activity carried out by the Department of Labor, upon written request submitted as provided in this part.

## § 70.14 Policy statements and interpre-

Statements of policy and interpreta-tions affecting a member of the public which have been adopted by the Department of Labor or a constituent agency thereof in connection with the administration of any program of the Department and which have not been published in the Federal Register shall be made available for public inspection and copying in accordance with the policy set forth in § 70.11 and the provisions and procedures in this part. The policies and interpretations made available pursuant to this section shall include any policy or interpretation concerning a particular fact situation, if that policy or interpretation can reasonably be expected to have precedential value in any case involving a member of the public in a similar situation. If the statement of policy or interpretation sought is contained in an available publication of the Department of Labor or has been placed in the public domain by release to private publishers of Government information services, it will be made available for examination in the form in which it was published or released without any formal request. at any office of the Department in which a copy or copies are kept.

## § 70.15 Staff manuals and instructions.

Any administrative staff manual or instruction to staff issued by the Department of Labor or a constituent agency thereof in connection with the administration of any program of the Department, containing matter (such as statutory interpretations) affecting a member

of the public shall be made available for public inspection and copying in accordance with the policy set forth in § 70.11 and the provisions and procedures in this part. These are the manuals and instructions to staff or portions thereof which delineate the basis for administrative action affecting members of the public, and are to be distinguished from manuals, portions thereof, or instructions provided for the guidance of Government personnel engaged in such law enforcement activities as investigating, inspecting, or auditing to determine whether violations of law exist, or prosecuting law violations in courts or administrative tribunals. The criteria or guidelines for staff engaged in such procedures, with respect to such matters as the selection, handling, defense, prosecution, or settlement of cases, operational tactics, allowable tolerances, investigation or litigation techniques and strategy, are traditionally and necessarily of a confidential nature, as is recognized in the legislative history of the Freedom of Information Act, and are subject to protection under the provisions of 5 U.S.C. 552(b), which restrict the disclosure requirements by the exemptions explained in the following sections of this part. Any special provisions for access to administrative staff manuals and instructions applicable to specific programs may be ascertained by reference to Subpart C.

#### § 70.16 Indexes to certain records.

(a) Current indexes identifying final opinions and orders in adjudicated cases (see § 70.12(b)), statements of policy and interpretations adopted by the agency but not published in the FEDERAL REGISTER (see § 70.14), and administrative staff manuals and instructions (see § 70.15), which have been issued, adopted, or promulgated after July 4, 1967, are normally available to the public in published form as provided in § 70.7. Such indexes, whether or not published, are made available for inspection and copying on request. If published copies of a particular index are at any time not available or if publication of such index has been determined to be unnecessary and impracticable by order published in the FEDERAL REGISTER, copies thereof will be furnished on request upon payment of the direct cost of duplication as provided in § 70.62(d).

(b) To effectuate the provisions of 5 U.S.C. 552(a) (2), § 70.7 of this part and paragraph (a) of this section, each officer identified in § 70.35 and § 70.36 shall maintain, make available for public inspection and copying, and provide for the required publication of current indexes of all materials in his custody which are subject to the indexing and publication requirements of 5 U.S.C. 552(a) (2). Each such officer shall forward a copy of each page of his index, including information as to pages which have become obsolete, to the Office of Information, Publications, and Reports in the Main Labor Building, 200 Constitution Avenue NW., Washington, D.C. 20210, which shall provide a central repository for, and make available for public inspection, all such indexes and supplements thereto compiled by

such officers, and shall provide for distribution of the published indexes and supplements.

#### RESTRICTIONS ON DISCLOSURE

### § 70.21 Records not disclosable.

(a) Pursuant to the provisions of 18 U.S.C. 1905, every officer and employee of the Department of Labor is prohibited from publishing, divulging, disclosing, or making known in any manner or to any extent not authorized by law any information coming to him in the course of his employment or official duties or by reason of any examination or investigation made by, or return, report or record made to or filed with the Department or any agency or officer or employee thereof, which information concerns or relates to the trade secrets, processes, operations, style of work, or apparatus, or to the identity, confidential statistical data, amount or source of any income, profits, losses, or expenditures of any person, firm, partnership, corporation, or association. No officer or employee of the Department of Labor shall disclose records in violation of this provision of law.

(b) No records of the Department of Labor with respect to matters specifically required by statute to be kept secret shall be made available for inspection or copying under the provisions of this part. By virtue of the exclusionary language in 5 U.S.C. 552(b) (3) the disclosure requirements of the Freedom of Information Act do not apply to or authorize the disclosure of records with respect to any matters specifically exempted from dis-

closure by statute.

(c) No records of the Department of Labor with respect to matters specifically authorized under criteria established by Executive order to be kept secret in the interest of the national defense or foreign policy and properly classified pursuant to such order shall be made avallable for inspection or copying under the provisions of this part. Records concerning such matters are expressly excluded from the application of the disclosure requirements of the Freedom of Information Act by the provisions of 5 U.S.C. 552(b) (1).

## § 70.22 Records disclosure of which may be refused.

(a) Records exempt from statutory disclosure requirements. The Freedom of Information Act, as codified in 5 U.S.C. 552, lists nine categories of records (in 5 U.S.C. 552(b)) to which the disclosure requirements of the statute do not apply. The first and third of these relate to the records described in § 70.21 which are not disclosable because protected from disclosure by the express provisions of a statute or a secret classification authorized by Executive order in the interest of national defense or for-eign policy. The other seven categories of records excluded from the statutory disclosure requirements are set forth in §§ 70.23 through 70.28, inclusive. Information from records in these seven categories may, however, be made available for inspection and copying as provided in paragraph (b) of this section.

(b) Disclosure of protected records; conditions precedent. Although the Department of Labor is not required by the Freedom of Information Act to make available for inspection or copying any materials or documents included in its records which are within the categories described in 5 U.S.C. 552(b) (2), (4), (5), (6), (7), (8), or (9) (see §§ 70.23-70.28) under the Department's disclosure policy set forth in § 70.11 particular records requested which come within these categories, or portions thereof, shall nevertheless be made available to the extent, but only to the extent, that the appropriate officer authorized to disclose information from Department records determines that the disclosure will further the public interest and will not impede the discharge of any of the functions of the Department of Labor, Such a determination shall be made with due regard not only to the public interest in accessibility to the people of information regarding operations of their Government but also to the public interest in protecting citizens from impairment of their rights to privacy or from harassment, injury, or the dissemination of information concerning them which is privileged or has been submitted by them to the Government on a confidential basis. In determining whether access to such records will be permitted, due consideration shall also be given to the public interest in preventing disclosure of information which would handicap, obstruct, or jeopardize effective performance of the Department's functions under statutes or Executive orders, including its duties with respect to law enforcement.

## § 70.23 Internal rules and practices.

(a) Pursuant to exemption (2) set forth in 5 U.S.C. 552(b), and as provided in § 70.22, the disclosure from Department of Labor records of matters that are related solely to the internal personnel rules and internal practices of the Department may be refused. The records protected by this exemption include memoranda pertaining to personnel matters such as staffing policies and policies and procedures for the hiring, training, promotion, demotion, and discharge of employees. Also included are records concerning operating rules, practices, guidelines, and procedures for departmental investigators, inspectors, compliance officers, examiners, and attorneys, the release of which would substantially impair the effective performance of their duties.

(b) The purposes of exemption (2) include the protection from public disclosure of any record that is designed only for the guidance of Department personnel, including internal rules and practices that cannot be disclosed to the public without substantial prejudice to the effective performance of a significant Department function. A negotiator cannot bargain effectively if his instructions and limitations are known to the person with whom he is negotiating. Similarly, the effectiveness of an authorized but unannounced inspection or audit would be destroyed if the circumstances under which that inspection or

audit is to be held become public knowledge.

(c) Although access to particular records concerning matters within the purview of exemption (2) may be permitted as provided in § 70.22 if the officer authorized to disclose records determines that the disclosure would serve the public interest and not impede the discharge of any function of the Department, such a determination ordinarily cannot be made in the case of internal rules and instructions relating to investigations and enforcement activities concerned with questions of compliance with or violations of provisions of law.

# § 70.24 Trade secrets and privileged or confidential information.

(a) Pursuant to exemption (4) set forth in 5 U.S.C. 552(b), and as provided in § 70.22, the disclosure from Department of Labor records of matters that are trade secrets, and of commercial and financial information obtained from a person and privileged or confidential, may be refused. Legal requirements of secrecy and prohibitions of disclosure may apply to such records as set forth in § 70.21. Disclosure shall be refused where these mandatory restrictions apply to the records sought. Even where denial of access is not required by these restrictions, access to records exempted from the disclosure requirements by exemption (4) cannot be granted under the policy expressed in §§ 70.11 and 70.22 unless the disclosure officer, in balancing the right of the public to know how the Government operates against the need of the Government to keep information in confidence and the right of the person from whom it was obtained to have privileges and confidences respected, is able to determine that disclosure will serve the public interest and not impede the discharge of any function of the Department of Labor.

(b) The intent of the exemption set forth in paragraph (a) of this section to protect privileged or confidential information is, according to the committee reports in both Houses of Congress, not restricted to the trade secrets and commercial or financial information specifically mentioned in the statue. Information the disclosur: of which may be refused pursuant to this exemption is, according to the legislative history, intended to include information customarily subject to a doctor-patient, lawyer-client, or other such privilege.

(c) Information "obtained from any person" would include information obtained from a person inside as well as outside the Government. The applicability of this exemption does not depend on whether the record contains information obtained from the public at large, from a particular person, from within the Department, or from another agency. While information which is confidential in the hands of one agency retains its protected character in the hands of agencies to which it is subsequently furnished, the exemption does not sanction the rendering of documents confidential by the expedient of transferring them among agencies.

(d) Except as otherwise provided in this part (e.g., § 70.13), disclosure in certain circumstances may be refused of material such as formulae, designs, drawings, research data, and the like, which are significant not as records but as items of valuable property. These may have been developed by or for the Government for its use and at its expense. Nothing in the legislative history suggests that the Freedom of Information Act was intended to give away such valuable property to any person willing to pay the price of making a copy. Where similar property in private hands would be held in confidence, the public interest would appear to require that such property in the hands of an agency should

be protected under exemption (4).

(e) This exemption is further intended to extend protection to other information in Government records which has been furnished and accepted in confidence and which would not customarily be released to the public by the person from whom the Government obtained it. See, for example, the House Report (H. Rept. 1497, 89th Cong., 2d Sess.) and the President's signing statement. Accordingly, the exemption assures the confidentiality of information thus obtained by the Department of Labor through questionnaires and required reports to the extent that the information would not customarily be made public by the person from whom it was obtained. Nothing in the Freedom of Information Act necessitates a disregard of the right of individuals or groups to rely in good faith on an understanding of confidentiality for which a Government agency has reasonably afforded a basis. Maintenance of citizens' respect for governmental fairness requires that such understandings be given due consideration. At the same time, Department representatives should be alert to discourage the development of such understandings where not clearly warranted by departmental responsibilities.

(f) Pursuant to exemption (8) set forth in 5 U.S.C. 552(b), and as provided in § 70.22, the disclosure from Department of Labor records of any matter contained in, or related to, examination, operating, or condition reports prepared by, on behalf of, or for the use of an agency responsible for the regulation or supervision of financial institutions, may be refused. Exemption (8) emphasizes the application to financial institutions of the protection from disclosure afforded by the exemption set forth in paragraph (a) of this section, and makes plain the intent to protect information relating to such institutions which may be prepared for or used by any agency responsible for the regulation or supervision of such institutions. Access to any such informa-tion in records of the Labor Department will be refused where disclosure is not legally permitted or where a determination to disclose is inappropriate for reasons discussed in paragraph (a) of this section. In cases where another agency is concerned with information protected by exemption (8) in 5 U.S.C. 552(b) which is sought from records of the De-

partment, the provisions of § 70.31 are also applicable.

## § 70.25 Inter-agency and intra-agency ment of Labor.

(a) Pursuant to exemption (5) set forth in 5 U.S.C. 552(b), and as provided in § 70.22, the disclosure from Department of Labor records of matters in inter-agency or intra-agency memoranda or letters which would not be available by law to a party other than the agency in litigation with the agency may be refused. The exemption is intended essentially to protect the full and frank exchange in writing of ideas, views, and opinions necessary for the effective functioning of the Government and the making of informed decisions by its officers.

(b) The protection from disclosure afforded by exemption (5) to the internal records of the Government described in paragraph (a) of this section is limited to those communications which, in litigation with a Government agency, would not be routinely available by law to another party to the proceeding. The legislative history and decisions of the courts make it clear that this provision is intended to insure that memoranda or letters not protected from disclosure by some other exemption would be available to the general public for inspection and copying if they "would routinely be disclosed" to such a party "through the discovery process" in such litigation. (See H. Rept. 1497, 89th Cong., 2d Sess.) The internal memoranda and letters protected from disclosure by exemption (5) are accordingly those which would not be released as a matter of course in litigation where discovery is sought by a party other than the agency under the Federal Rules of Civil Procedure. Since the granting of discovery of internal documents is typically a very extraordinary step, not normally a "routine" one, it is only in a limited category of situations that such documents would be routinely available by law to another party in litigation with the agency.

(c) Examples of the type of record information protected from disclosure by the exemption set forth in paragraph (a) of this section include opinions, advice, deliberations, or recommendations made in the course of developing official action by the Department of Labor or any of its component units, and other internal communications which would not be routinely available through the discovery process to a party in litigation with the Government.

(d) In the case of inter-agency memoranda and letters protected by this exemption, the officer authorized to disclose records shall not make a determination to allow access to such matters under the policy set forth in §§ 70.11 and 70.22 if to do so would conflict with the provisions of § 70.31.

§ 70.26 Personnel, medical, and similar files.

(a) Pursuant to exemption (6) set forth in 5 U.S.C. 552(b), and as provided in § 70.22, the disclosure from De-

partment of Labor records of matters in personnel and medical files and similar files the disclosure of which would constitute a clearly unwarranted invasion of personal privacy" may be refused. In view of the congressional concern expressed in the legislative history regarding the protection of individuals' privacy, each disclosure officer of the Department shall apply the disclosure policy set forth in §§ 70.11 and 70.22 with due regard to the apparent intent of the statutory language to characterize the invasion of personal privacy typically involved in disclosure of personnel and medical files as "clearly unwarranted." "Similar files" for which protection is provided under this exemption appear to refer to any files the disclosure of which would invade personal privacy to such a degree that the disclosure would be as "clearly unwarranted" as the disclosure of personnel or medical files.

(b) Among the records in Department files protected from disclosure by the exemption set forth in paragraph (a) of this section are (1) personnel and background records personal to any officer or employee of the Department, including his home address and telephone number; (2) medical histories and medical records concerning individuals; (3) any other detailed record containing personal information identifiable with a particular individual where it appears that such person's right to have such information protected from public dissemination is clear; and (4) private or personal information in other files which, if disclosed to the public, would amount to a clearly unwarranted invasion of the privacy of any person, including members of the family of the person to whom the information pertains. The disclosure of information about a person to that same person is not, of course, an invasion of such person's privacy.

(c) Related regulations implementing this part and the application of this exemption to personal and medical information in the files of the Employment Standards Administration relating to claims of injured employees for workmen's compensation benefits are referred to in Subpart C.

§ 70.27 Investigatory records compiled for law enforcement purposes.

(a) Restrictions on public access authorized. Pursuant to the provisions of exemption (7) set forth in 5 U.S.C. 552 (b), as amended by Pub. L. 93-502, 88 Stat. 1563, effective February 19, 1975, the disclosure from Department of Labor records of matters that are "investigatory records compiled for law en-forcement purposes" and to which access by the public would be detrimental to such purposes or to rights of privacy as specified in the statute, may be refused. Access to such records may be refused for any one or more of several specific reasons. Thus, exemption (7) protects from the public access requirements of the Freedom of Information Act investigatory records compiled for law enforcement purposes whenever their disclosure to any person requesting them would-

(1) interfere with enforcement proceedings; and/or

\$ 8. A

(2) deprive a person of a right to a fair trial or an impartial adjudication; and/or

(3) constitute an unwarranted invasion of personal privacy; and/or

(4) disclose (i) the identity of a confidential source and (ii), in the case of a record compiled by a criminal law enforcement authority in the course of a criminal investigation, or by an agency conducting a lawful national security intelligence investigation, confidential information furnished only by the confi-

dential source; and/or
(5) disclose investigative techniques

and procedures; and/or

(6) endanger the life or physical safety

of law enforcement personnel.

Where one or more of the foregoing consequences would ensue from the disclosure of particular records to any person requesting access to investigatory records compiled for law enforcement purposes, no determination to grant access to such records may be made by the disclosure officer under the disclosure policy set forth in §§ 70.11 and 70.22 in the ordinary case because in such circumstances it would not ordinarily be possible to determine that disclosure would serve the public interest and would not impede any of the functions of the Labor Department.

(b) "Law enforcement purposes." The exemption set forth in paragraph (a) of this section expresses the public interest in preventing disclosure detrimental to the enforcement, whether civil or criminal in nature, of applicable laws and in assuring, to this end, that such enforcement will not be jeopardized by disclosing information which would be harmful to citizens who aid the Government in law enforcement investigations and prosecutions, would endanger the persons or handicap the essential enforcement activities of investigators and prosecutors, would violate the rights of persons who may be charged with violations, or would otherwise interfere with investigations and other proceedings to enforce the laws. "Law enforcement" as used in the statute, according to the legislative history, is used in the broadest sense to include the enforcement not only of criminal statutes, but of all laws establishing rules of conduct, whether by statute or by Executive order or by a duly promulgated regulation having the force and effect of law. Moreover, "enforcement" is not limited to enforcement by adversary proceedings, and includes other types of Government law enforcement activities as well; the work of a policeman or a compliance officer is law enforcement even if he does not participate in adversary proceedings. On the other hand, "enforcement" does not include all activities conducted in order to carry out the laws, but only those intended to counteract past, present, or future violations.

(c) "Investigatory" records. The protection afforded investigatory records under the exemption set forth in paragraph (a) of this section also extends, according to the legislative history, to

those files related to the investigation which are prepared in connection with related Government litigation and adjudicative proceedings. One of the purposes of the exemption is to preserve the position of the Government in litigation or potential litigation, in accordance with the rules governing discovery in cases before courts and administrative agencies. It ensures that the litigant is not given any earlier or greater access to investigatory material than he would have directly in litigation or other enforcement proceedings.

(d) Records "compiled" for law enforcement purposes. Whenever the disclosure of investigatory records requested by any person would have any of the consequences specified in the exemption set forth in paragraph (a) of this section, access to such records may be refused if the records have been "compiled" for law enforcement purposes, irrespective of the nature of the action, if any, to carry out such purposes which may have been contemplated at the time of their compilation or may have been taken thereafter as a result of the compilation. Investigatory records shown to have been compiled for law enforcement purposes retain their status as such and continue to be protected from disclosure in the specified circumstances as provided in this exemption whether or not enforcement proceedings are contemplated at the time of investigation or are instituted thereafter or, if brought, have been completed before the request is made for the records. None of these factors negate the need, in the public interest and in the interest of persons who have been the subject of or who have furnished information to help the law enforcement activities of the Government, for continued protection against a prejudicial disclosure from such records of information which may still constitute an unwarranted invasion of personal privacy or may reveal the identity of a confidential source. The subject of an investigation which has been closed because charges of law violation could not be substantiated is entitled to protection from the opprobrium which might follow public disclosure of the record of the charges without opportunity to demonstrate their falsity in an adjudicatory proceeding. An informant whose identity as the source of confidential information furnished in aid of law enforcement activities is made public by disclosure of investigative records may be made subject to retaliatory action by others and in any event may not inform without assurance that private persons will not, through access to the agency's records, be able to identity the informant as the source of the information furnished. Even years after enforcement proceedings are concluded. the public interest may require protection of investigatory records which would identify a confidential source or would reasonably lead to such a disclosure. Any disclosure of such records would soon become a matter of common knowledge and few individuals, if any, would come forth to embroit themselves in controversy or possible recrimination by notifying

the Government of something which might justify investigation. Even more important in view of the increasing concern today over the conflict between a citizen's right of privacy and the need of the Government to investigate, it is unthinkable that rights of privacy, whether of subjects or confidential sources, should be jeopardized. In addition, the disclosure of investigatory records may reveal investigative techniques and procedures or endanger the life or physical safety of law enforcement personnel. In either case, the ability of the Government to carry out its law enforcement functions effectively would be seriously impaired.

### 70.28 Well information.

Pursuant to exemption (9) set forth in U.S.C. 552(b), and as provided in § 70.22, the disclosure from Department of Labor records of matters consisting of geological and geophysical information and data, including maps, concerning wells, may be refused. This exemption supplements the exemption set forth in § 70.23 by removing any doubt that disclosure of this specific type of information is protected under the Act.

### § 70.29 Partial disclosure.

(a) Deletions to protect personal privacy. To the extent required to prevent a clearly unwarranted invasion of personal privacy, the officer authorized to disclose information from a record may delete identifying details when he makes available or publishes an opinion, statement of policy, interpretation, or staff manual or instruction, provided that in every case the justification for the deletion is fully explained in writing.

(b) Records containing both disclosable and nondisclosable information. If a requested record or document contains some materials which are protected from disclosure and other materials which are not so protected, identifying details or protected matters shall be deleted whenever analysis indicates that such deletions are feasible. Any reason-ably segregable portion of a record or document shall be provided after deletion of protected matter.

(c) Deletions to protect identity of requester. Where the identity of an applicant, or other identifying details related to a request, would constitute an unwarranted invasion of personal privacy if made generally available, as in the case of a request to examine one's own medical files, identifying details shall be deleted from copies of the request and written responses to it that are made available to requesting members of the public.

## § 70.30 Withdrawal of originals.

No document or record in the custody of the Department of Labor, or of any agency or officer thereof, shall on any occasion be taken or withdrawn by any agent, attorney, or any other person not officially connected with the Department; no exception will be made without the written consent of the Secretary or the Solicitor of Labor.

§ 70.31 Record of concern to more than

If the release of a record in custody of the Department of Labor would be of concern not only to the Department but also to another Federal agency, the record will be made available by the Department only if its interest in the record is the primary interest and only after coordination with the other interested agency. If the interest of the Department in the record is not primary, the request will be transferred promptly to the agency having the primary interest, and the applicant will be so notified. The release of information received from another agency and the release to another agency of information collected from persons outside the Government shall be subject to the conditions and restrictions imposed by 44 U.S.C. 3508.

### FACILITIES FOR DISCLOSURE

§ 70.35 Where information may be obtained.

(a) Any person desiring to examine or copy records of the Department of Labor known to be situated in any office of the Department and providing information adequate to permit identification of the records sought may obtain from the head of such office information as to the availability of the derived records. Such person will be advised whether such records are available for examination under the provisions of this part and any applicable supplemental regulations (see Subpart C) without the submission of a formal request or whether such a request must be submitted for consideration by an officer responsible for custody of the records under the procedures set forth in §§ 70.42-70.52. If such a request must be submitted, information as to the name, title, and address of the officer to whom it should be addressed will also be furnished. The officers to whom such requests should be addressed are identified in general terms in § 70.43. Such officers may provide access to records at the place where they are kept or, where appropriate, at another location.

(b) If the person desiring access to particular records is in doubt as to the place where they are kept or the agency of the Department listed in paragraph (e) of this section which is responsible for their custody, the necessary information may be obtained by addressing an inquiry, identifying as specifically as possible the records sought, to the Assistant Secretary for Administration and Management, U.S. Department of Labor, 200 Constitution Avenue NW., Washing-

ton, D.C. 20210.

(c) Indexes, normally in published form, are available to members of the public to assist them in locating records of certain types, as provided in §§ 70.7 and 70.16 of this part.

(d) Public reference facilities for examination and copying of certain Department records are maintained in some offices. Their locations and the types of records available for examination therein

may be ascertained by addressing an inquiry to the chief officer of the agency of the Department listed in paragraph (e) of this section which has custody of records of the types sought to be examined, or by communicating with the nearest officer of such agency who is authorized to act on requests for its records. If the agency of the Department responsible for custody of the records is not known, the Assistant Secretary for Administration and Management will provide the necessary information on request.

(e) Offices in Washington, D.C., are maintained by the following component agencies of the Department of Labor. Field offices are maintained by some of these, as listed in the Government Organization Manual (see § 70.5(b)).

(1) Office of the Secretary of Labor. (2) Office of the Solicitor of Labor.

(3) Office of Information, Publications, and Reports. (4) Bureau of International Labor

(5) Bureau of Labor Statistics.

(6) Employment Standards Administration.

(7) Labor-Management Services Administration.

(8) Manpower Administration.

(9) Occupational Safety and Health Administration.

(10) Employees' Compensation Appeals Board.

(11) Wage Appeals Board. (12) Benefits Review Board. (13) Board of Contract Appeals.

(14) Office of Administrative Law

Judges.

The heads of the foregoing agencies shall make available for inspection and copying in accordance with the provisions of this part, records in their custody or in the custody of component units within their organizations, either directly or through their authorized representatives in particular offices and locations.

§ 70.36 Titles and addresses of the responsible officials of various agencies.

The titles of the responsible officials of the various independent agencies in the Department of Labor are listed below. Unless otherwise specified, the mailing address for these officials shall be:

U.S. Department of Labor, Washington, D.C. 20210.

Chief Administrative Law Judge.

Chairman, Employees Compensation Appeals Board.

Deputy Under Secretary for International Labor Affairs.

Associate Deputy Under Secretary for International Labor Affairs. Associate Deputy Under Secretary for Trade

and Adjustment Policy. Director, Office of Information, Publication and Reports.

Executive Assistant for Assistant Secretary for Policy, Evaluation and Research.

Deputy Assistant Secretary for Policy Development, Office of Policy Development. Deputy Solicitor, Office of the Solicitor.

Assistant Secretary for Administration and Management.

Office of Assistant Secretary for Administration and Management.

Associate Manpower Administrator for the Office of Administration and Management, Patrick Henry Building, Room 4000, 601 D Street NW., Washington, D.C. 20213.
Associate Manpower Administrator for the

Office of Policy, Evaluation and Research, Room 9000.

Associate Manpower Administrator for the

Bureau of Apprenticeship and Training, Room 5000. Associate Manpower Administrator for the Office of Field Direction and Management,

Room 10100. Associate Manpower Administrator for the United States Employment Service, Room

Associate Manpower Administrator for the Unemployment Insurance Service, Room

Associate Manpower Administrator for the Office of Manpower Development Programs, Room 600.

Assistant Regional Director for Manpower: Region I.—Room 1703, J. F. Kennedy Federal Building, Center, Boston, Mass. 02203. (Connecticut, Maine, Massachusetts, New Hampshire, Rhode Island, and Vermont.)

Region II.—1515 Broadway, N.Y. 10036. (New York, New Jersey, Puerto Rico, and U.S.

Virgin Islands.)

Region III.—Post Office Box 8796, Philadel-phia, Pa. 19101. (Gateway Building, 3535 Market St., Philadelphia, Pa. 19101. Do not use street address for mailing purposes.) (Delaware, District of Columbia, Maryland, Pennsylvania, Virginia, and West Virginia.)

Region IV.—Room 405, 1371 Peachtree Street NE., Atlanta, Ga. 30309. (Alabama, Florida, Georgia, Kentucky, Mississippi, North Carolina, South Carolina, and Tennessee.)
Region V.—230 South Dearborn, Chicago, Ill.

60604. (Illinois, Indiana, Michigan, Minnesota, Ohio, and Wisconsin.)

Region VI.—Federal Building, 7th Floor, 1100 Commerce Street, Dallas, Tex. 75202. (Arkansas, Louisiana, New Mexico, Oklahoma, and Texas.)

Region VII.-Room 3000, Federal Building. 911 Walnut Street, Kansas City, Mo. 64106. (Iowa, Kansas, Missouri, and Nebraska.)

Region VIII.—Room 16015, Federal Office Building, 1961 Stout Street, Denver, Colo. 80202. (Colorado, Montana, North Dakota,

South Dakota, Utah, and Wyoming.)
Region IX.—Federal Building, 450 Golden
Gate Avenue, Box 36084, San Francisco, Calif. 94102. (Arizona, California, Guam, Hawaii, and Nevada.)
Region X.—Arcade Plaza, 1321 Second Ave-

nue, Seattle, Wash. 98101. (Alaska, Idaho, Oregon, and Washington.)

Assistant Secretary for Manpower, U.S. Department of Labor, Manpower Administration, Room 1000, 601 D Street NW., Washington, D.C. 20213.

Assistant Secretary for Occupational Safety Health, Occupational Safety and and Health Administration.

Assistant Secretary for Employment Standards, Employment Standards Administra-

Administrator, Wage-Hour Division. Director, Office of Federal Contract Com-

pliance. Associate Directors, OFCC, Teams I-IV. Associate Director, OFCC, Policies, Plans, and

Programs. Associate Director, OFCC, Construction Compliance Division.

Associate Director, OFCC, Training Administrative Support Staff.

Director, Women's Bureau.

Associate Assistant Regional Director for Wage-Hour Division, Associate Assistant Regional Director for Office of Federal Contract Compliance and Associate Assistant Regional Director for Women's Bureau.

## RULES AND REGULATIONS

Addresses: JFK Building, Government Center, Boston, Massachusetts 02203. Everett M. Dirksen Building, 219 South Dearborn Street, Chicago, Illinois 60604. 1515 Broadway, New York, New York 10036. Federal Building, U.S. Court House, 1100 Commerce Street, Dallas, Texas 75202. Gateway Building, 3535 Market Street, Philadlephia, Pennsylvania 19104. Federal Office Building, 911 Walnut Street, Kansas City, Missouri 64106. 1371 Peachtree Street NE., Atlanta, Georgia 30309. Federal Office Building, 1961 Stout Street, Denver, Colorado 80202. Federal Building, 450 Golden Gate Avenue, San Francisco, California 94102. Smith-Tower, 506 Second Avenue, Seattle, Washington 98104. Area Director, OFCC, Federal Building, 1240 East Ninth Street, Cleveland, Ohio 44199. Area Director, OFCC,

Director, Office of Rehabilitation Employment Compliance and Service. Director, Office of Program Development and Accountability.

Federal Building, 300 North Los Angeles Street,

\*Los Angeles, California 90012.

The responsible field office for Office of Rehabilitation Employment Compliance and Service and Office of Program Development and Accountability is Assistant Regional Director for Employment Standards Administration. Requests may be made to the above addresses

Commissioner, Bureau of Labor Statistics.
Assistant Secretary for Labor-Management
Services.

Labor Management Services Administration.

Director, Office of Labor Management Standards Enforcement.

Director, Office of Federal Labor Management Petitions.

Director, Office of Labor-Management Relations Services.

Assistant Regional Directors for Labor-Management Services:

Director, 8757 Georgia Avenue, Room 1449 Silver Spring Plaza, Silver Spring, Maryland 20910. Assistant Regional Director, 1371 Peachtree Street NE., Room 300, Atlanta, Georgia 30309. Assistant Regional Director,

Assistant Regional Director, 230 S. Dearborn Street, 10th Floor, Federal Office Building, Chicago, Illinois 60604.

Assistant Regional Director, 911 Walnut Street, Room 2200 Federal Office Building,

Kansas City, Missouri 64106. Assistant Regional Director,

1515 Broadway, Room 3515, New York, New York 10036. Assistant Regional Director, 3535 Market Street,

Room 14120 Gateway Building, Philadelphia, Pennsylvania 19104. Assistant Regional Director, 450 Golden Gate Avenue, Room 9061 Federal Office Building, San Francisco, California 94102.

Area Directors for Labor-Management Services:

Area Director, 1371 Peachtree NE., Room 303, Atlanta, Georgia 30309.

Area Director, 100 Tremont Street, Room 211 New Studio Building,

Boston, Massachusetts 02108. Area Director, 111 W. Huron Street, Room 616 Federal Building, Buffalo, New York 14202.

Buffalo, New York 14202.

Area Director,
230 S. Dearborn Street,

Room 700 Federal Office Building, Chicago, Illinois 60604. Area Director,

1240 E. 9th Street,
Room 821 Federal Office Building,
Cleveland, Ohio 44199.
Area Director,

P.O. Box 239, Bryan and Ervay Streets, Room 301 Post Office Building, Dallas, Texas 75221.

Area Director, 1961 Stout Street, 2320 Federal Office Building,

Denver, Colorado 80202. Area Director,

234 State Street, Room 1906, Washington Boulevard Building,

Detroit, Michigan 48226.

Area Director, 1833 Kalakaua Avenue, Room 601, Honolulu, Hawaii 96815.

Area Director, 911 Walnut Street, Room 2200, Federal Office Building, Kansas City, Missouri 64106.

Area Director, 300 N. Los Angeles Street, Room 7731 Federal Building, Los Angeles, California 90012.

Area Director, 18350 Northwest Second Avenue, P.O. Box 3750, Norland Branch, Miami, Florida 33169.

Area Director, 110 S. Fourth Street, Room 110, Federal Courts Building, Minneapolis, Minnesota 55401.

Area Director, 1808 West End Building, Room 825, Nashville, Tennessee 37203.

Area Director, 9 Clinton Street, Room 305,

Newark, New Jersey 07102. Area Director,

Room 940, Federal Office Building, New Orleans, Louisiana 70130.

Area Director, 26 Federal Plaza, Room 1751, New York, New York 10007.

New York, New York 10007. Area Director,

600 Arch Street, Room 4256, Federal Office Building, Philadelphia, Pennsylvania 19106. Area Director, 1000 Liberty Avenue,

1000 Liberty Avenue, Room 1436, Federal Office Building, Pittsburgh, Pennsylvania 15222. Area Director, 210 N. Twelfth Boulevard, Room 570. St. Louis, Missouri 63101.

Area Director, 100 McAllister Street, Room 1604, San Francisco, California 94102.

Area Director, 605 Condado Avenue,

Room 704, Condominio San Alberto, Santurce, Puerto Rico 00907. Area Director, 506 Second Avenue, Room 3301, Smith Tower Building,

Seattle, Washington 98104. Area Director, 1111 20th Street NW., P.O. Box 19257, Boom 509, Vanguard Building, Washington, D.C. 20036.

Deputy Director, Office of Employee Benefits

The following is a list of the offices in which the Disclosure Officers for Occupational Safety and Health Administration are located in Washington, D.C.:

Office of Executive Assistant to the Assistant Secretary.

Office of Special Assistant for Public Affairs.
Office of Federal Agency Safety Programs.
Office of Training and Education.

Office of Training and Education.

Office of Standards Development.

Office of Compliance Programming.

Office of State Plan Review and Operation.

Office of State Plan Review and Operation Office of Field Performance Evaluation.

Office of Management Data Systems.
Office of Public Hearing Transcripts.

Office of Public Hearing Transcripts.

Office of Planning, Evaluation, and Research.

Office of Equal Employment Opportunity.

Office of Personnel Management.
Office of Administratve Management.
Office of Financial Management.
Office of Publications and Visual Aids.

Office of Committee Management.
Director, Office of Information, Publications

and Reports.
Assistant Regional Director for Information,
John F. Kennedy Federal Building, Government Center, Boston, Massachusetts

John F. Kennedy Federal Building, Government Center, Boston, Massachusetts 02203. Assistant Regional Director for Information, 1515 Broadway, New York, New York 10036.

1515 Broadway, New York, New York 10036. Assistant Regional Director for Information, Gateway Building, 3535 Market Street, Philadelphia, Pennsylvania 19104. Assistant Regional Director for Information,

1371 Peachtree St. NE., Atlanta, Georgia 30309.

Assistant Regional Director for Information, 300 S. Wacker Drive, 16th Floor, Chicago, Illinois 60606. Assistant Regional Director for Information,

Federal Bullding, U.S. Court House, 1100 Commerce Street, Dallas, Texas 75202. Assistant Regional Director for Information,

Assistant Regional Director for Information, Federal Office Building, 911 Walnut Street, Kansas City, Missouri 64106.

Assistant Regional Director for Information, Federal Office Building, 1961 Stout Street, Denver, Colorado 80202.

Assistant Regional Director for Information, Federal Building, 450 Golden Gate Avenue, San Francisco, California 94102.

Assistant Regional Director for Information, Room 2078 Arcade Plaza Building, 1321 Second Avenue, Seattle, Washington 98101.

Information Officer, 300 N. Los Angeles Street, Room 7641, Los Angeles, California 90012.

The following is a list of the responsible field officials in the Occupational Safety and Health Administration:
Assistant Regional Directors:

18 Oliver Street, Boston, Massachusetts 02110.

## RULES AND REGULATIONS

1515 Broadway (1 Astor Plaza), Room 3445, New York, New York 10036. 3535 Market Street, Philadelphia, Pennsylvania 19104. 1375 Peachtree St. NE., Atianta, Georgia 30309. 32nd Fioor, Room 3259, 230 S. Dearborn Street, Chicago, Illinois 60604. Texaco Building, 7th Floor, 1512 Commerce Street, Dallas, Texas 75201. 911 Walnut Street, Rm. 3000, Kansas City, Missouri 64106. Federai Building, Room 15010, 1961 Stout Street. Denver, Colorado 80202. 9470 Federal Building, 450 Golden Gate Avenue, P.O. Box 36017, San Francisco, Calif. 94102. 1808 Smith Tower Building, 506 Second Avenue, Seattle, Washington 98104.

Area Directors:

Custom House Building,
Room 703,
State Street,
Boston, Massachusetts 02109.
Federal Building, Rm. 426,
55 Pieasant Street,
Concord, New Hampshire 03301.
Federal Building, Rm. 617B,
450 Main Street,
Hartford, Connecticut 06103.
U.S. Post Office and Courthouse Building,
Rm. 501,

U.S. Post Office and Courthouse Rm. 501, 436 Dwight Street, Springfield, Mass. 01103. 370 Old Country Road, Garden City, L.I., N.Y. 11530. 90 Church Street, Room 1405, New York, New York 10007. 970 Broad Street, Room 1435C, Newark, New Jersey 07102. 605 Condado Avenue, Room 328, Santurce, Puerto Rico 00907.

Midtown Plaza, Room 203, 700 E. Water Street, Syracuse, New York 13210.

Federal Building, Room 1110A, Charles Center, 31 Hopkins Piaza, Baltimore, Maryland 21201.

Charieston National Plaza, Room 1726, 700 Virginia Street, Charleston, West Virginia 25301.

Jonnet Building, Room 802, 4099 William Penn Highway, Monroeville, Pennsylvania 15146.

Wiiiiam J. Green, Jr., Federal Building, Room 4456, 600 Arch Street, Philadelphia, Pa. 19106.

Room 8015, Federal Building (P.O. Box 10186), 400 N. 8th Street, Richmond, Virginia 23240.

Todd Mall, 2047 Canyon Road, Birmingham, Alabama 35216.

1710 Gervais Street, Rm. 205, Columbia, South Carolina 29201.

Bridge Building, Room 204, 3200 E. Oakland Park Blvd., Fort Lauderdale, Fla. 33308.

57601–55 North Frontage Rd. East, Jackson, Mississippi 39211, Art Museum Plaza, Suite 4, 2809 Art Museum Drive. Jacksonville, Florida 32207. 600 Federal Place. Suite 554-E, Louisville, Kentucky 40202. Riverside Piaza Shopping Ctr., 2720 Riverside Drive, Macon, Georgia 31204. Commerce Building, Room 600, 118 North Royal Street, Mobile, Alabama 36602. 1600 Hayes Street, Suite 302, Nashvilie, Tennessee 37203. Federal Office Building, Rm. 406, 310 New Bern Avenue, Raieigh, North Carolina 27601. Enterprise Building, Suite 204, 6605 Abercon Street Savannah, Georgia 31405. 650 Cieveland Street, Rm. 44,

Clearwater, Florida 33515.
Building 10, Suite 33,

La Vista Perimeter Office Park, Tucker, Georgia 30084. 230 South Dearborn, 10th Floor.

Chicago, Illinois 60604.
Federal Office Building, Rm. 4028,

550 Main Street, Cincinnati, Ohio 45202. Federal Office Building, Room 847,

1240 East Ninth Street, Cleveland, Ohio 44199. 360 S. Third Street, Rm. 109,

Columbus, Ohio 43215. Michigan Theatre Building,

Room 626, 220 Bagley Avenue, Detroit, Michigan 48226.

U.S. Post Office and Courthouse, Room 423, 46 East Ohio Street, Indianapoiis, Indiana 46202.

Clark Building, Room 400, 633 West Wisconsin Avenue, Milwaukee, Wisconsin 53203.

110 South Fourth Street, Rm. 437, Minneapoiis, Minnesota 55401. Federal Office Building, Rm. 734,

Federal Office Building, Rm. 734, 234 North Summit Street, Toledo, Ohio 43604.

Federal Building, Room 302, 421 Goid Avenue SW., P.O. Box 1428, Albuquerque, N. Mexico 87103. Adolphus Tower, Suite 1820, 1412 Main Street, Daiias, Texas 75202. 2320 La Branch Street, Rm. 2118, Houston, Texas 77004. Federal Building, Room 421, 1205 Texas Avenue, Lubbock, Texas 79401.

Donaghey Building, Rm. 526, 103 East 7th Street, Little Rock, Arkansas 72201. 546 Carondelet Street, Rm. 202, New Orleans, La. 70130.

Petroieum Building, Rm. 512, 420 South Boulder, Tulsa, Oklahoma 74103.

1015 Jackson Keller Road, Room 122, San Antonio, Texas 78213.

210 Walnut Street, Room 643, Des Moines, Iowa 50309. 1627 Main Street, Rm. 1100, Kansas City, Missouri 64108. 113 West 6th Street, North Platte, Nebraska 69101. Harney & 16th Street, Rm. 803, City National Bank Building, Omaha, Nebraska 68102. 210 North 12th Boulevard, Room 554, St. Louis, Missouri 63101. Petroleum Building, Suite 312, 221 South Broadway Street, Wichita, Kansas 67202. Petroleum Building, Suite 525, 2812 1st Avenue North, Biiiings, Montana 59101. Squire Plaza Building, 8527 West Coifax Avenue Lakewood, Coiorado 80215. U.S. Post Office Building, Room 452, 350 South Main Street, Salt Lake City, Utah 84101. Court House Plaza Building, Room 408, 300 North Dakota Avenue, Sioux Falls, South Dakota 57102. 1100 East William Street, Carson City, Nevada 89701. 333 Queen Street, Suite 505, Honolulu, Hawaii 96813. Hartwell Building, Room 401, 19 Pine Avenue. Long Beach, California 90802. Amerco Towers, Suite 318, 2721 North Central Avenue, Phoenix, Arizona 85004. 100 McAliister Street, Rm. 1706, San Francisco, California 94102. Federal Building, Room 227, 605 West 4th Avenue, Anchorage, Alaska 99501. 121 107th Street NE., Believue, Washington 98004. 228 Idaho Building, 216 N. 8th Street, Boise, Idaho 83702. Pittock Block, Room 526, 921S W. Washington Street, Portland, Oregon 97205.

District Supervisors:

Federal Building, Room 503A, U.S. Courthouse, Providence, Rhode Island 02903. Stanwick Building, Room 111, 3661 Virginia Beach Blvd., Norfolk, Virginia 23502. 600 Leopard Street, Suite 1322, Corpus Christi, Texas 78401.

## PROCEDURE FOR DISCLOSURE

## § 70.41 Applicability of procedures.

Requests for inspection or copying of information from records in the custody of the Department of Labor which are available under the provisions of this part shall be made and acted upon as provided in the following sections of this subpart. The prescribed procedure shall be followed in all cases where access is sought to official records pursuant to the provisions of the Freedom of Information Act, except with respect to records for which a less formal disclosure procedure is provided specifically in this part or in any supplemental regulation referred to in Subpart C. Officers and employees of the Department may be authorized by the Secretary, or by the head of the agency in the Department

having custody of the record, to continue to furnish to the public, informally and without compliance with these procedures, information and copies from its records which prior to enactment of the Freedom of Information Act (5 U.S.C. 552) were customarily furnished in the regular performance of their duties.

## § 70.42 Submittal of requests for access to records.

(a) Any person who desires to inspect or copy any record covered by this part, including any writing, drawing, map, recording, tape, film, photograph, or other documentary material by which information is preserved, shall submit a written request to that effect to the agency of the Department of Labor (see the listings in §§ 70.35(e) and 70.36) which has custody of the record.

(b) Standard forms for making a request are not required. However, to expedite the processing of requests submitted in accordance with paragraph (a) of this section each such request should clearly indicate on the envelope and on the request the following: F.O.I.A. re-

quest.

(c) To avoid delay in receipt of a sufficiently complete request at the agency in cases where the search and copying charges are not known or estimated and agreed upon in advance but may be substantial, the request in such a case should, as provided in \$ 70.64(b), state specifically that whatever costs will be involved pursuant to \$\$ 70.62-70.63 will be acceptable, or will be acceptable up to an amount not exceeding a named figure.

## § 70.43 To whom to direct requests.

(a) A request for records in the custody of the Bureau of Labor Statistics shall be directed to the Commissioner of Labor Statistics, Department of Labor, 200 Constitution Avenue NW., Washington, D.C. 20210. In the case of records in the custody of other agencies of the Department, the request shall be directed to the chief of such agency, unless the record asked for is kept in one of its field offices. If the desired record is kept in a field office, the request shall be directed to the field official having the principal responsibility to the agency for the custody of the records in that office. who may, for brevity, be referred to as the "responsible field official." The responsible field official may be identified as follows:

(1) In the case of records kept in a regional office, the chief of such office (for aid in identification, see § 70.36).

(2) In the case of records relating to compensation claims under laws providing compensation for work injuries to employees and under the War Hazards Act and Defense Base Act, the deputy commissioner of the district office in which the records are kept.

(3) In the case of records kept in any other field office, the chief of the regional office or of the highest field office which, pursuant to authority from the agency of the Department administering the program to which the record relates, is responsible for the direct supervision of operations under such pro-

gram in the field office where the record is kept. If the person making the request does not know where the record is located, he may direct his request to the Assistant Secretary of Labor for Administration and Management, Department of Labor, 200 Constitution Avenue NW., Washington, D.C. 20210, for appropriate handling.

(b) The agencies of the Department of Labor, referred to in paragraph (a) of this section, are, or are constituent units of, the agencies listed in § 70.35 (e). Heads and responsible field officials of these agencies are further identified in § 70.36. For purposes of submittals of requests for records under this section, the chief of the Office of the Secretary shall be considered to be the Assistant Secretary for Administration and Management, and the chief of the Office of the Solicitor shall be considered to be the Deputy Solicitor. The general organization of the Department of Labor, its principal officers, and addresses of its regional and other field offices may be found in the "U.S. Government Organization Manual" referred to in

## § 70.44 Description of information requested.

Each request should reasonably describe the record or records sought; i.e., in sufficient detail to permit identification and location thereof with a reasonable amount of effort. So far as practicable, the request should specify the subject matter of the record, the date or approximate date when made, the place where made, the person or office that made it, and any other pertinent identifying details.

## § 70.45 Deficient descriptions.

If the description is insufficient so that a professional employee who is familiar with the subject area of the request cannot locate the record with a reasonable amount of effort, the officer processing the request will notify the applicant and, to the extent possible, indicate the additional information required. Every reasonable effort shall be made to assist an applicant in the identification and location of the record or records sought. Records will not be withheld merely because it is difficult to find them.

## § 70.46 Requests for categories of records.

Requests calling for all records falling within a reasonably specific category will be regarded as reasonably described within the meaning of \$70.44 and the statutory requirement if the Department is reasonably able to determine which records come within the request and to search for and collect them, at least where such search can be accomplished without unduly interfering with Department operations because of staff time consumed or the resulting disruption of files. If undue disruption of agency operations would result from fulfilling the request, the officer to whom the request was submitted shall promptly give the applicant notice thereof and the

opportunity to confer with him in an attempt to reduce the request to manageable proportions by reformulation and by outlining an orderly procedure for the production of the records.

## § 70.47 Receipt by agency of request; acknowledgment.

(a) Since the officer responsible for acting upon a request for access to records of the Department or a constituent agency thereof must take the necessary action promptly after receipt of the request and a determination regarding disclosure must in any event be dispatched to the requester within a prescribed period after such receipt as set forth in § 70.53, such officer shall, upon receipt of such request, have the date and time of such receipt immediately inscribed thereon and give contemporaneous notice to the requester that the request was received on such date. Such notice shall also advise the requester of the time within which it is anticipated that a response will be made. If it appears from the request that the unusual circumstances set forth in § 70.55(b) may require extension of the normal time limit for responding to the request, and if such extension is authorized pursuant to \$ 70.55(c), the acknowledgment of receipt shall contain the information required by § 70.53(b).

(b) Each agency of the Department of Labor shall make provision for transmittal procedures which will expedite on a priority basis the receipt by the proper disclosure officer of incoming requests for access to records on which such officer must pass but which initially come to some other office or agency of the Department. Requesters can assist the Department in this regard by proper marking of their requests as indicated in

§ 70.42(b).

(c) In accordance with the rules set forth in Subpart B of this part, the request by a person for access to records requiring substantial search and/or copying services must be accompanied by payment or assurance of payment of any fees therefor applicable under the statute and this part, and action thereon may not be required in the absence of such payment or assurance. As provided in § 70.64(b), and to protect requesters from an unexpected accrual of liability greater than they may wish to assume for access to requested records when they do not know or have any estimate of the applicable fees and the agency anticipates that search and/or copying charges may exceed \$25, no receipt of a complete request will be deemed to have occurred (and, therefore, the provisions of paragraph (a) of this section will not be applicable) unless or until (1) the requester specifically states that whatever costs will be involved pursuant to \$\$ 70.62-70.63 will be acceptable or will be acceptable up to an amount not exceeding a named figure, or (2) prompt notification has been given to the requester of the agency's estimate of the costs and the requester has perfected the request by payment or assurance of payment thereof. Upon completion of the request by such payment or assurance of payment, the re-

quest will be deemed to have been made in accordance with the published rules as to fees and receipt will be acknowledged as provided in paragraph (a) of this section.

## § 70.48 Action on request.

(a) The officer responsible for determining whether a request for agency records will be complied with in whole or in part shall cause any necessary search to be made promptly upon receipt of a request made in accordance with the rules published in this part and, after location of the requested records and such examination as may be necessary, shall proceed, as expeditiously as possible and within the time limitations set forth in § 70.53, to determine whether and to what extent the request may be granted pursuant to the policy of disclosure set forth in § 70.11 and the provisions of §§ 70.21, 70.22, and 70.29, and to advise the requester immediately of such determination.

(b) When a determination to grant the request with respect to all or any portion of the records requested is made, such records shall be made available to the requester at the time he is advised of the determination or as promptly thereafter as possible.

## § 70.49 Form of denials.

A reply denying a written request for a record or portion thereof shall be in writing and shall contain a brief statement of the reasons for the denial, including (except in case of a denial for the reasons set forth in § 70.53(c)) a reference to the specific exemption or exemptions under the Freedom of Information Act authorizing the withholding of the record and an explanation of how the exemption applies to the matter withheld. The denial shall also include the name and title or position of the person(s) responsible for the denial and an outline of the appeal procedure available.

### § 70.50 Appeals from denial of requests.

An applicant whose request for a record or portion thereof has been denied pursuant to \$ 70.49 may file an appeal within 90 days from the date of the denial or in the case of a partial denial, 90 days from the date the material is received by the requester. The appeal shall state, in writing, the grounds for appeal, including any supporting statements or arguments. The appeal shall be addressed to the Solicitor of Labor, Department of Labor, 200 Constitution Avenue NW., Washington, D.C. 20210. To expedite the processing of the appeal, each such appeal should clearly indicate on the envelope and on the appeal the following: F.O.I.A. Appeal.

#### § 70.51 Receipt by Solicitor of appeal; acknowledgment.

Since action on an appeal from a denial of a request for records must be taken promptly after the receipt of the appeal by the Solicitor and in any event within the prescribed time limits following such receipt as set forth in § 70.54,

each such appeal, when received by the Solicitor, shall have the date and time of such receipt inscribed thereon and the appellant shall immediately be notified of the time of receipt as so inscribed. To expedite the handling of appeals as much as possible, the Department will maintain internal procedures designed to assure transmittal to and receipt by the Solicitor of incoming appeals on a priority basis. Appellants can assist the Department in this regard by proper marking of their appeals as indicated in \$ 70.50.

## § 70.52 Action on appeals.

(a) As promptly as possible after receipt of an appeal and within the time limits specified in § 70.54, the Solicitor shall review the appellant's supporting papers and pass on the appeal. In his review of the matter on appeal, the Solicitor is authorized to determine de novo, in the light of the disclosure policy set forth in § 70.11, whether the denial of appellant's request for access to records was proper and in accord with the applicable provisions of the statute and the pertinent regulations. In the event that the denial appealed from is one made pursuant to the provisions of paragraph (c) of § 70.53 by reason of inability of the disclosure officer to make an informed determination within the specified time limits, the Solicitor's decision shall take into consideration any supplementary determination by disclosure officer made pursuant to the provisions of such paragraph.

(b) The Solicitor may proceed as provided in paragraph (a) to take the necessary action on the appeal notwithstanding the pendency of any action for judicial relief against withholding of records requested pursuant to this part, unless otherwise directed by the court. The Solicitor may take such action whether the judicial relief was sought prior to any determination by the disclosure officer to grant or deny the request in whole or in part, or was applied for before or after the filing of the appeal. In the event that a requester seeks review by a court of the denial by a disclosure officer of a request for agency records without first filing an appeal to the Solicitor as provided in this part, the Solicitor may, unless otherwise or-dered by the court, consider such action as the filing of an appeal and issue a decision thereon within the period specified in § 70.54 and in accordance with the provisions of paragraph (a) of this section. In any of the foregoing circumstances a final decision by the Solicitor to grant access to records would make unnecessary a ruling by the court with respect thereto.

(c) The Solicitor shall issue a decision in writing granting or denying the appeal in whole or in part and, if the appeal is granted with respect to any or all of the requested records, the Solicitor shall order such records to be made available promptly to the appellant. If the appeal is denied wholly or in part, the decision shall set forth each exemption provided in 5 U.S.C. 552 which is

relied on, how it applies to the record or portion thereof which had been withheld, and the reasons for asserting it. The decision of the Solicitor shall be the final action of the Department of Labor with respect to the request. The notification to the requester of any decision upholding a denial in whole or in part of the request shall include notice of the provisions for judicial review set forth in 5 U.S.C. 552(a) (4).

(d) Copies of both grants and denials on appeal shall be collected in one file open to the public (subject to provisions of § 70.29(c)) and indexed in accordance

with §§ 70.7 and 70.16.

TIME LIMITATIONS FOR ACTION ON RE-QUESTS AND APPEALS

§ 70.53 Period within which action on request shall be taken.

(a) As soon as possible and within a period not to exceed 10 working days after receipt (see § 70.47) of a request for agency records reasonably described and requested in accordance with the regulations published in this Part 70 and such supplementary regulations referred to in Subpart C hereof as may be pertinent, the officer responsible for acting on the request shall determine that such request will be granted or denied in whole or in part as provided in this Part 70, and shall thereupon immediately notify the requester of such determination and the reasons therefor. The provisions of paragraphs (b), (c), and (d) of this section are applicable where a final determination to grant or deny the request has not been made within the 10-day period with respect to all the records requested.

(b) As provided in § 70.55, an extension of the 10-day period for action on the request may be determined to be necessary because of unusual circumstances. If such an extension has been authorized as provided in § 70.55(c), the officer responsible for acting on the request shall notify the requester in writing within such 10-day period of the extension and the reasons therefor, specifying the date on which the determination to grant or deny the request is expected to be dispatched.

(c) If the disclosure officer is unable to grant the request as to all or any part of the records sought within the 10-day period specified in paragraph (a) or such extended period as may be specified pursuant to paragraph (b), because some or all of the records have not been located or made available for examination and consideration in time to make an informed determination, the officer may, within such period, respond to the request with respect to such records by denving access thereto at such time, with notification to the requester of such reasons and of the right to appeal such denial pursuant to the provisions of the Freedom of Information Act and this part within the 90-day period provided in § 70.50. In such event the officer shall further advise the requester that the search or examination will be continued and that the denial will be subject to withdrawal, modification, or confirmation by a supplementary determination to be made as soon as processing of the request can be completed. If an appeal is filed from the initial denial, the Solicitor shall act thereon as provided in § 70.52.

(d) If the officer who must pass upon the request fails to make a determination to grant or deny access to requested records as provided in paragraph (a) or (c) within the 10-day period specified in paragraph (a) or an extension thereof as specified pursuant to paragraph (b), the requester shall be deemed to have exhausted his administrative remedies and may apply for judicial relief against the withholding of such records as provided in the Freedom of Information Act. In such event, however, the court may allow additional time as provided in the Act, upon a showing of exceptional circumstances and the exercise of due diligence in responding to the request. In view of this, where despite due diligence exceptional circumstances have prevented a timely determination, the disclosure officer shall so advise the requester, explaining fully the facts and circumstances and that processing of the request will continue with the expectation that a determination can be made by a stated date, and shall solicit agreement by the requester to delay any application for judicial relief until such date. Processing of the request shall continue until such determination is made. however, irrespective of whether the requester agrees to defer court action or applies for judicial relief. If it is finally determined to grant access to any such records, no judicial relief would be necessary with respect thereto.

# § 70.54 Period within which action on appeal shall be taken.

(a) After receipt (see § 70.51) of an appeal from a denial of a request for agency records, the Solicitor shall as soon as possible within a period not to exceed 20 working days or such extended period as may be authorized pursuant to § 70.55 (d) issue the decision on the appeal as

provided in § 70.52.

(b) If a decision on the requester's appeal from a denial of access to agency records is not made by the Solicitor within the normal or extended period applicable under paragraph (a), the requester shall be deemed to have exhausted his administrative remedies and may apply for judicial relief against the withholding of the requested records as provided in the Freedom of Information Act. In such event, however, the court may allow additional time as provided in the Act, upon a showing of exceptional circumstances and the exercise of due diligence in responding to the request. In view of this, where despite due diligence exceptional circumstances have prevented a timely decision on the appeal, the Solicitor shall so advise the requester. explaining fully the facts and circumstances and that processing of the appeal will continue with the expectation that a decision will be made by a stated date, and shall solicit agreement by the requester to delay any application for judicial relief until such date. Processing of the appeal shall continue until a decision thereon is made, however, irrespective of whether the requester agrees to

defer court action or applies for judicial relief. If it is finally determined to grant the appeal as to any of the requested records, no judicial relief will be necessary with respect thereto.

## § 70.55 Extension of period for taking action on request or appeal.

(a) Extension in unusual circumstances, generally. Pursuant to paragraph 6(B) of 5 U.S.C. 552(a) as amended, the maximum period of 10 working days normally allowed (see § 70.53) for making the determination in response to a request for agency records submitted as provided in this part, or the maximum period of 20 working days normally allowed (see § 70.54) for deciding an appeal from a denial of the request for any such records in whole or in part, may be extended in the unusual circumstances described in paragraph (b) of this section for an additional period by written notice from the disclosure officer or the Solicitor, as the case may be, to the requester setting forth the reasons for such extension and the date on which a determination is expected to be dispatched. Such additional time may be provided only to the extent reasonably necessary to the processing of the particular request and no extension for more than an additional 10 working days is permitted. Such extensions shall be allowed only as provided in paragraphs (c) and (d) of this section, so that the total additional time made available thereby for determining the action to be taken on a particular request, whether by the disclosure officer or by the Solicitor in the event of appeal, or by both together, will not exceed 10 working days.

(b) Unusual circumstances which may justify extended time. The unusual circumstances which may justify an extension, as provided in paragraph (a), of the time for taking action on a request for records or an appeal from a denial of the request in whole or in part are:

(1) The need to search for and collect the requested records from field facilities or other establishments that are separate from the office processing the request:

(2) The need to search for, collect, and appropriately examine a voluminous amount of separate and distinct records which are demanded in a single request; or

(3) The need for consultation, which shall be conducted with all practicable speed, with another agency having a substantial interest in the determination of the request or among two or more components of the agency having substantial subject-matter interest therein.

Consultations between an agency unit and its legal counsel, the public information staff, or the Department of Justice are not unusual circumstances within the meaning of this paragraph and should not be considered the basis for an extension as provided in paragraph (a) and paragraph (c) or paragraph (d).

(c) Extension of period for disclosure officer to determine action to be taken on request. The disclosure officer passing upon the request for records may, to the extent reasonably necessary to the

proper processing of such request in the unusual circumstances set forth in paragraph (b), extend the period for determining the action to be taken thereon beyond the normal 10-day limit as provided in paragraph (a), but only with the prior approval of the Solicitor and for such period up to but not in excess of 10 additional working days as the Solicitor shall approve. Notification shall be given to the requester as provided in § 70.53(b).

(d) Extension of period for Solicitor to decide appeal from denial of access to requested records. The Solicitor may, to the extent necessary to the proper processing of an appeal from a denial of a request for agency records, extend the period for deciding such appeal beyond the normal 20-day limit by written notice to the appellant as provided in paragraph (a), but only in the unusual circumstances set forth in paragraph (b) and in no event for more than 10 additional working days or such portion thereof as remains after any extension previously made pursuant to paragraph (c).

(e) Further provisions for extension of time for taking action. As noted in § 70.53(d) and § 70.54(b), notwithstanding the fact that a requester is deemed to have exhausted his administrative remedies and may apply for judicial relief under the Freedom of Information Act when the disclosure officer or the Solicitor fails to take the necessary action with respect to a request for records or an appeal from a denial thereof within the normal or extended time limits prescribed in the Act, the court in any such action may, upon a showing of exceptional circumstances and the exercise of due diligence in responding to the requester, allow additional time to complete the necessary review. While any deferral of an application for judicial relief under such circumstances is within the discretion of the requester, the responsible officer will, pursuant to the rules in this part, continue to proceed to a determination of the matter before him and, if the determination is to grant access to the requested records, the necessity for judicial relief will be obviated.

## Subpart B—Copies of Records and Fees for Services

## SPECIAL SEARCHING AND COPYING SERVICES

§ 70.61 Charges for services, generally.

(a) Pursuant to the provisions of the Freedom of Information Act, as amended, the payment of standard charges as set forth in the fee schedule in § 70.62 will, except as otherwise provided in this subpart, be required of the requester to cover the direct costs of searching for and duplicating records requested under the Act from the Department or any of its constituent agencies. Where the direct cost to the agency of making the search is substantial, fees for searching as provided in the schedule will be charged even if the record searched for is not found or if, after it is found, it is determined that the request to inspect

it must be denied under the provisions of 5 U.S.C. 552(b) and the regulations in this part.

(h) Circumstances under searching and copying facilities or services may be made available to the requester without charge or at a reduced charge are delineated in §§ 70.65-70.67. Recoupment of agency costs for determining whether a requested record is disclosable under the statute and this part and for making deletions of portions exempted from disclosure by the Act has been excluded from consideration in arriving at the standard charges contained in the fee schedule and no charge is made to a requester for the cost of any such services. However, where a requester desires the agency to provide such services as certification, authentication, or other special services not required under the Freedom of Information Act with respect to requested records, fees in addition to the standard charges for search and copying will be charged to cover the agency costs as indicated in §§ 70.68-70.69 and as authorized by the special provisions in Subchapter II of Title 29 of the United States Code and the general user-charge statute, section 483a of Title 31, United States Code.

## § 70.62 Search and copying charges.

(a) Fee schedule for searching records. Where agency records must be searched to locate a requested record, charges applicable under this subpart to the search will be made according to the following fee schedule:

(1) Search time: (1) Ordinary search by custodial or clerical personnel, \$1.25 for each one-quarter hour or fraction thereof of employee worktime in excess of the first quarter-hour required to reach or obtain the records to be searched and to make the necessary search; and

(ii) Search requiring services of professional or supervisory personnel to locate requested record, \$2.50 for each such quarter-hour of such services required.

(2) Additional search costs. If the search for a requested record requires transportation of the searcher to the location of the records or transportation of the records to the searcher, at a cost in excess of \$5, actual transportation costs will be added to the search time cost.

(3) Search in computerized records. Special fees to cover direct personnel and machine time costs of such searches will be charged as set forth in the above fee schedule and § 70.63.

(b) Fee schedule for copying of records. The fees payable pursuant to this subpart for obtaining requested copies of records which have been made available for inspection under the Freedom of Information Act will be computed on the following basis and subject to the following conditions:

(1) Standard copying fee: \$0.10 per page of record copies furnished. This has been determined to be a reasonable standard charge to cover direct costs to the Department of duplicating records where the volume of page copies furnished is not extraordinary. The stand-

ard fee is applicable both where the copies are reproduced by the person desiring them, using Government-furnished reproduction equipment, such as coin-operated machines, which may be made available for use by any number of the public, and where, in the absence of availability of such facilities, the copies are reproduced by agency personnel. This standard fee is also applicable to the furnishing of copies of available computer printouts as stated in § 70.63(a) (1).

(2) Voluminous material. If the volume of page copy desired by the requester is such that the reproduction charge at the standard page rate would be in excess of \$200, the person desiring reproduction may request special rate quotation from the Office of the Assistant Secretary for Administration and Management.

(3) Limit of service. Not more than 10 copies of any document will be furnished.

(4) Manual copying by requester. No charge will be made for manual copying by the requesting party of any document made available for inspection under the provisions of this part. The Department shall provide facilities for such copying without charge at reasonable times during normal working hours.

'(c) Transcripts of Where pursuant to the provisions of this part there is made available for inspection and copying the transcript of a hearing or other proceeding pertaining to matters within the purview of the Department of Labor or any of its programs, constituent agencies, or advisory committees, no search charge will be made. Copies of such transcripts will normally be furnished upon payment of the actual cost of duplication as computed pursuant to the fee schedule in paragraph (b). If, however, the transcript was furnished to this Department under the provisions of a contract executed prior to January 5, 1973, the effective date of the Federal Advisory Committee Act, under contractual provisions which prohibit the Department from reproducing the transcript, copies of such transcripts prepared by persons or firms reporting the proceedings may be obtained upon application to the reporter and payment of the reporter's fees at the rate provided in the contract. The foregoing fees are not applicable to the furnishing of any copies of such transcripts to any person entitled to receive such copies without charge pursuant to a regulation or order issued by the Secretary or his authorized representative.

(d) Indexes. Pursuant to 5 U.S.C. 552(a) (2), copies of indexes or supplements thereto which are maintained as therein provided but which have not been published will be provided on request at a cost not to exceed the direct cost of duplication as computed pursuant to the fee schedule in paragraph (b) of this section.

## § 70.63 Computerized records.

(a) Information available in whole or in part in computerized form which is disclosable under the Freedom of Information Act is available to the public as follows:

(1) When there is an existing printout from the computer which permits copying the printout, the material will be made available at the per page rate stated in § 70.62(b) (1) for each 8½- by 11-inch page.

(2) When there is not an existing printout of information disclosable under the Freedom of Information Act, a printout shall be made provided the applicant pays the cost to the Department as here-

inafter stated.

(3) Obtaining information from computerized records frequently involves a minimum computer time cost of approximately \$100 per request. Multiple requests involving the same subject may cost less per request. Services of personnel in the nature of a search shall be charged for at rates prescribed in § 70.62 (a). A charge shall be made for the computer time involved based upon the prevailing level of costs to Government organizations and upon the particular types of computer and associated equipment and the amounts of time on such equipment that are utilized. A charge shall also be made for any substantial amounts of special supplies or materials used to contain, present or make available the output of computers based upon the prevailing levels of costs to Government organizations and upon the type and amount of such supplies and materials that are used.

(b) Information in the Department's computerized records which could be produced only by additional programming of the computer, thus producing information not previously in being, is not required to be furnished under the Freedom of Information Act. In view of the usually heavy workloads of the computers used by the Department, such a service cannot ordinarily be offered to

the public.

## § 70.64 Payment of fees.

(a) Medium of payment. Payment of the applicable fees as set forth in §§ 70.62 and 70.63 shall be made in cash, by U.S. postal money order, or by check payable to the Secretary of Labor. Postage stamps will not be accepted in lieu of cash, checks, or money orders as payment for fees specified in the schedule. Cash should not be sent by mail.

(b) Advance payment or assurance. Payment of the known and officially estimated searching and copying fees shall be made or assured to the satisfaction of the disclosure officer prior to the performance of substantial searching or copying services. Where the requester does not know and has no official estimate of the search and copying costs at the time the request is made, the request should specifically state that whatever costs will be involved pursuant to §§ 70.62-70.63 will be acceptable, or will be acceptable up to an amount not exceeding a named figure. A request made without such specific assurance of payment of fees, in an amount at least equal to the charges under §§ 70.62-70.63 which the Department anticipates will be necessary, will, if such estimated charges are in excess of \$25, not be deemed to have been received by the Department until the requester has been notified (promptly upon physical tender of the request) of the Department's cost estimate and has perfected the request by paying the estimated charges or giving satisfactory assurance that payment will be made. In the event that a request is made only for inspection of a record, advance payment or assurance of payment as set forth in this paragraph is applicable only with respect to searching fees; charges for copying will not be made unless or until

copies are requested.

(c) Adjustment of fees. Where an estimated fee paid by the requester in advance exceeds the fee chargeable under the applicable schedule for the search or copying services actually performed, the balance will be refunded by the Department. Where the actual fees due for the services are in excess of the estimate, the requester will be required to remit the difference. In cases where the estimated costs required under the fee schedule for responding to a request are such that an advance deposit is deemed necessary, and it appears that the information sought by the requester might be made available at less cost by revision of the request, the Department's advice to the requester of the estimated costs and the need for an advance deposit will be accompanied by extension of an offer to the requester to confer with knowledgeable Department personnel with a view to reformulation of the request in a manner which will reduce the fees and meet the needs of the requester.

(d) Post-copying costs. The scheduled fees for furnishing copies of records made available pursuant to the Act for inspection and copying cover the costs of furnishing the copies at the place of duplication. Where requests for such copies are made by mail, no postage charge will be made for transmitting by regular mail a single copy of the requested record to the requester, or for mailing additional copies where the total postage cost does not exceed \$1. However, where the volume of page copy or method of transmittal requested is such that transmittal charges to the Department are in excess of \$1, the transmittal costs will be added to the copying fee specified in accordance with the schedule, unless appropriate stamps or stamped envelopes are furnished with the request, or authorization is given for collection of shipping charges on delivery.

§ 70.65 Standard fees not charged in certain circumstances.

The searching and copying fees set forth in this subpart are intended to cover agency costs of furnishing services that are special to the requester and above and beyond those which are generally provided to the public at large. Where furnishing the information requested can be considered as primarily benefiting the general public, the Freedom of Information Act provides that the agency may determine that waiver or reduction of the standard charges for searching and copying is in the public

interest, in which event requested records shall be made available without charge or at a reduced charge. On this basis as well as for other reasons which have satisfied the Department that it is in the public interest to do so, the standard fees for making a requested record available will, in certain circumstances, not be charged or will be reduced as stated in the sections following. In addition, where the disclosure officer granting the request for a record determines that the waiver or reduction of the standard charge would be in the public interest because of the primary benefit to the general public of disclosure of the requested information, the disclosure officer shall advise the requester that the requested record will be made available without charge or at a reduced charge which he shall specify.

§ 70.66 Services performed without charge.

(a) No searching charge under § 70.62 (a) shall be made for routine procurement for inspection from Department records, not requiring more than one-quarter hour of personnel time, of any record required by the Freedom of Information Act to be made available for

public inspection on request.

(b) The searching fees set forth in § 70.62(a) shall not be charged for the production of any requested record material maintained in a public reference facility of the Department or any of its component units, or for searching for any records available for public inspection without formal request pursuant to the rules in this part and supplemental regulations issued as provided in Subpart C, or for producing any record required by statute to be made available for inspection without charge.

(c) The fees provided for copies in § 70.62(b) do not apply to copies of materials reproduced for the purpose of distribution to the public without charge or to copies of published materials available for purchase from the Superintendent of Documents which the Department makes available in single copies or in limited quantities without charge to persons whose access to such materials may serve the public interest and assist the Department in administering statutes or carrying out programs for which it is responsible.

§ 70.67 Waiver or reduction of fees by disclosure officer.

In appropriate circumstances the disclosure officer may waive or reduce fees otherwise applicable under § 70.62 for services in searching for and copying record information. Thus, where no major expenditure of staff time or burden on reproduction facilities is involved, single copies of disclosable documents readily accessible in Departmental files may be furnished without charge to persons properly and directly concerned with the matters therein (e.g., where an individual seeks a copy of a record pertaining to Departmental action or reference material directly concerning that individual), and to persons in special circumstances where inability to pay is

demonstrated and it is clear that a significant public interest would be served by providing the service free of charge. Similarly, searching and copying services may be performed at a reduced charge determined by the disclosure officer to be justified by the public interest in disclosure of a requested record (see § 70.65) or by the benefit to the public that may accrue from dissemination of information likely to assist the Department in carrying out a statutory program.

AUTHENTICATION; SPECIAL STUDIES AND COMPILATIONS

§ 70.68 Authentication of copies.

(a) Fees. The Freedom of Information Act does not require certification or attestation under seal of copies of records furnished in accordance with its provisions. Pursuant to provisions of the general user-charger statute, 31 U.S.C. 483a and Subchapter II of Title 29 of the United States Code, the following charges will be made where such services are requested:

(1) For certification of true copies,

each \$1.

(2) For attestation under the seal of the Department, each \$3.

(b) Authority and form for attestation under seal. Authority is hereby given to any officer or officers of the Department of Labor designated as authentication officer or officers of the Department to sign and issue certificates of authentication under the seal of the Department of Labor. The form of authentication shall be as follows:

I hereby certify, that \_\_\_\_\_\_\_ is now and was at the time of signing (title) \_\_\_\_\_\_ and has legal custody of the official documents of the U.S. Department of Labor therein attested and that full faith and credit should be given to his act as such.

> (Authentication Officer, Department of Labor)

§ 70.69 Special studies and compila-

Pursuant to the provisions of Subchapter II of Title 29 of the United States Code (29 U.S.C. 9-9b), the Department may, upon the written request of any person and within the discretion of the Secretary of Labor, undertake special statistical studies relating to employment, hours of work, wages and other conditions of employment, may prepare from its records statistical compilations, and may furnish transcripts of its studies, tables, and other records, upon the payment of the actual cost of such work by the person requesting it. Transcripts of existing records requested and made available under the Freedom of Information Act will be furnished subject to payment of the applicable searching and copying fees as set forth earlier in this subpart. However, services as above described which are not within the contemplation of the Freedom of Information Act and are performed in the discretion of the Secretary of Labor will be charged for on the basis of a computation of actual total cost. Nothing in the Freedom of Information Act is construed to require the compiling of records other than the indexes prescribed by 5 U.S.C. 552(a) (2) or the preparation of documents other than those required by 5 U.S.C. 552(a) (1) to be published in the FEDERAL REGISTER.

### Subpart C—Special Rules Applicable to Certain Specific Records

§ 70.71 Authority of agency officials in Department of Labor.

Each agency of the Department of Labor for which an officer or officers have authority to issue rules and regulations may through such officers promulgate supplementary regulations, not inconsistent with this part, governing the disclosure of particular or specific records which are in the custody of that departmental unit. Agencies of the Department which do not or have not promulgated special supplementary regulations governing disclosure of particular records shall disclose such records pursuant only to the provision of Subparts A and B of this Part 70.

## § 70.72 Supplementary regulations currently in force.

Duly promulgated regulations curently in force governing the disclosure of records in the custody of a subdivision of the Department shall continue to be operative insofar as such regulations are consistent with the provisions of this part. Agencies of the Department which have promulgated supplementary regulations or to which special regulations apply are listed in this subpart. Disclosure by such agencies of records not subject to such regulations shall be governed by the provisions of this Part 70.

## § 70.73 Bureau of Labor Statistics.

(a) The records described in paragraph (b) of this section are in the custody of the Bureau of Labor Statistics at the address indicated. The right of inspection and copying provided in this Part 70 may be exercised at that office. See § 70.77(b) concerning rules specially applicable to the disclosure of statistical information compiled by the Bureau pursuant to the Williams-Steiger Occupational Safety and Health Act of 1970.

(b) Collective bargaining agreements and other available agreements and actions thereunder settling or adjusting labor disputes, maintained by the Department of Labor pursuant to section 211(a) of the Labor-Management Relations Act of 1947 (61 Stat. 156; 29 U.S.C. 181); other available statistical information, tables, studies, and reports, collected, collated, and published or reported by the Bureau of Labor Statistics pursuant to the provisions of title 29, chapter I of the United States Code. As required by the governing statute and as provided in § 70.21, requests to inspect or copy any

such document shall be denied with respect to any specific information submitted to the Department in confidence.

Address: U.S. Department of Labor, Bureau of Labor Statistics, General Accounting Office, 441 G Street NW., Washington, D.C. 20210.

## § 70.74 Employment Standards Administration.

(a) Records relating to compensation claims under statutes administered by the Employment Standards Administration for work injuries to employees and for detention benefits under the War Hazards Act may be made available for inspection as provided in 20 CFR 1.22, 31.22, 41.21, 51.1, 61.19, 81.1, and 91.1, and for copying as provided in Subpart B and the supplemental rules in 20 CFR 1.22.

(b) Special supplementary regulations governing the disclosure of records respecting the certification of eligibility to apply for worker assistance under the Trade Expansion Act of 1962 (76 Stat. 872; 19 U.S.C. 1801) and Executive Order 11075 (28 FR 473) may be found in §§ 90.18-19 of this title. The proceedings and reports of advisory committees established under the Trade Act (88 Stat. 1996; 19 U.S.C. 2155) may, as provided in Section 135(f) (2) of that Act, be nondisclosable whenever and to the extent it is determined in accordance with law that such meetings will be concerned with matters the disclosure of which would seriously compromise the Government's negotiating objectives or bargaining positions on the negotiation of any trade agreement.

(c) The disclosure of records in the custody of the Office of Federal Contract Compliance or its Compliance Agencies shall be governed by the supplementary regulations published as new Part 60–40 of Title 41 of the Code of Federal Regulations. These will be rescinded or modified in the near future so as to conform to the Freedom of Information Act Amendments of 1974.

(d) The regulations of the Office of Workers' Compensation Programs relating to release of information, which will appear in the near future in the Federal Register under Title 20, Chapter I, Part 10, with reference to the Federal Employees' Compensation Act, as amended; at 20 CFR 702.112-702.114 with reference to the Longshoremen's and Harbor Workers' Compensation Act, as amended and extended; and at 20 CFR 715.301 with reference to the Black Lung Benefits Program.

# § 70.75 Labor-Management Services Administration.

(a) The following documents are in the custody of the Office of Labor-Management Standards Enforcement at the address indicated below, and the right of inspection and copying provided in this Part 70 may be exercised at such offices:

(1) Data and information contained in any report or other document filed pursuant to Sections 201, 202, 203, 211, and 301 of the Labor-Management Reporting and Disclosure Act of 1959 (73)

Stat. 524-528, 530, 79 Stat. 888; 29 U.S.C. 431-433, 441, 461).

(2) Data and information contained in any report or other document filed pursuant to the reporting requirements of Part 204 of this title, which are the regulations implementing Executive Order 11491. The reporting requirements are found in Part 204, of this title, § 204.3.

Address: U.S. Department of Labor, Office of Labor-Management Standards Enforcement, Public Documents Room 8757, Georgia Avenue, Silver Spring, MD 20216.

(b) Upon request of the Governor of a State for copies of any reports or documents filed pursuant to Sections 201, 202, 203, 211 of the Labor-Management Reporting and Disclosure Act of 1959 (73 Stat. 524-528, 79 Stat. 888; 29 U.S.C. 431-433, 441), or for information contained therein, which have been filed by any person whose principal place of business or headquarters is in such State, the Office of Labor-Management Standards Enforcement shall:

(1) Make available without payment of a charge to the State agency designated by law or by such Governor, such requested copies of information and data, or;

(2) Require the person who filed such reports and documents, to furnish such copies or information and data directly to the State agency thus designated.

## § 70.76 Employees' Compensation Appeals Board.

In matters before the Employees' Compensation Appeals Board the Board shall make available, or deny access to, records in the proceedings to employees or their attorneys, or in case of death, to their beneficiaries or authorized representatives, in compliance with standards set out in 20 CFR 501.8.

## § 70.77 Occupational Safety and Health Administration.

(a) Supplementary regulations governing the disclosure of records in the custody of the Occupational Safety and Health Administration pursuant to the Williams-Steiger Occupational Safety and Health Act of 1970 (84 Stat. 1590; 29 U.S.C. 651) have been published in Part 1913 of this title. These will be rescinded or modified in the near future so as to conform to the Freedom of Information Act Amendments of 1974.

(b) Disclosure of statistical information compiled by the Bureau of Labor Statistics pursuant to the Williams-Steiger Occupational Safety and Health Act of 1970 shall be governed by the supplementary regulations referred to in paragraph (a) of this section. These will be rescinded or modified in the near future so as to conform to the Freedom of Information Act Amendments of 1974.

Signed at Washington, D.C., this 11th day of February 1975.

PETER J. BRENNAN, Secretary of Labor.

[FR Doc.75-4328 Filed 2-18-75;8:45 am]

## CHAPTER X—NATIONAL MEDIATION BOARD

# PART 1208—AVAILABILITY OF INFORMATION

## Freedom of Information Regulations

Pursuant to the authority contained in 44 Stat. 577, as amended (45 U.S.C. 151-163), the National Mediation Board herein amends Part 1208 of its rules by revising § 1208.2.

These amendments revise National Mediation Board regulations to conform to the requirements of the Freedom of Information Act as amended by Pub. L. 93-502, 88 Stat. 1561.

The amendments require that requests for information be mailed to the Executive Secretary of the Board (except that requests for records of the National Railroad Adjustment Board be sent directly to its offices in Chicago). The Executive Secretary responds to the request, in writing, by either granting the request, in writing, by either granting the request, denying it, or granting and denying it in part. There is a ten-day period for this response, with provision for extension of time. The Executive Secretary is specifically charged with insuring timely response.

In cases of a complete or partial denial by the Executive Secretary, the amendments provide for an appeal to the Chairman of the Board within thirty days of receipt of the denial. The Chairman of the Board has twenty days to act on the appeal. Provision is made for extension of the 20-day period and handling of appeals if and when time limits are exhausted.

The Executive Secretary is charged with maintaining records and files on the processing of all requests and appeals. He must also keep records of all extensions of time taken. This insures that the ten-day extension limit in the Act is not exceeded. Authority for granting extensions of time rests with the official authorized to make a decision on the request.

The amended regulations provide that when the time for initial reply, including any extensions, has expired and no determination has been made, the requester must be informed in writing that he may deem this a denial and appeal to the Chairman of the Board. This provision assures the preservation of the twenty-day period allotted to the Chairman for deciding appeals. In cases of unavoidable delay, such as that occasioned by requests for voluminous records the requester may be asked to defer appeal so long as the Board is making diligent efforts to process the request.

The amendments require that at either stage a letter denying a request must include the name and title or position of the person responsible for the denial, as well as a statement of reasons for it.

In a case where time limits have expired at the appeal level the requester will be informed of his right to treat the failure to complete administrative review as an exhaustion of administrative remedies, permitting immediate suit for ju-

dicial review. He will also be advised, however, of the reason for the delay and of the date by which a response may be expected. When the delay is unavoidable, it is anticipated the requester will avoid unnecessary litigation expenses until processing is completed, since a court may grant an extension in such circumstances.

Because the amendments pertain to matters of procedure and policy, the relevant provisions of the Administrative Procedure Act (5 U.S.C. 553) requiring notice of proposed rulemaking, opportunity for public participation, and delay in effective date are inapplicable. In addition, since these changes are necessary to achieve compliance with the amendments to the Freedom of Information Act (5 U.S.C. 552), which become effective on February 19, 1975, there is not sufficient time to receive and evaluate public comment. However, in accordance with the spirit of the public policy set forth in 5 U.S.C. 553, interested persons may submit written comments on these amendments to the Executive Secretary, National Mediation Board, Washington, D.C. 20572 no later than March 19, 1975. All comments received by that date will be evaluated and acted upon as if this document were a proposal.

On January 13, 1975, the Board published in the FEDERAL REGISTER (40 FR 2451) proposed fee regulations (§ 1208.6). That subject is therefore not covered by these amendments.

In consideration of the above, § 1208.2 is amended in the manner set forth below.

These amendments, together with the proposed fee regulations referred to above, become effective on February 19, 1975.

#### ROWLAND K. QUINN, Jr., Executive Secretary.

Part 1208 of Title 29 of the Code of Federal Regulations is amended by revising § 1208.2 to read as follows:

## § 1208.2 Production or disclosure of material or information.

(a) Requests for identifiable records and copies. (1) All requests for National Mediation Board records shall be filed in writing by mailing the request or delivering it to the Executive Secretary, National Mediation Board, 1230 16th Street, NW., Washington, D.C. 20572, except that requests for records of the National Railroad Adjustment Board shall be in writing and addressed to the Administrative Officer, National Railroad Adjustment Board, 220 South State Street, Chicago, Illinois 60604.

(2) The request shall reasonably describe the records being sought in a manner which permits identification and location of the records.

(i) If the description is insufficient to locate the records, the National Mediation Board will so notify the person making the request and indicate the additional information needed to identify the records requested.

(ii) Every reasonable effort shall be made by the Board to assist in the iden-

tification and location of the records sought.

(3) Upon receipt of a request for records the Executive Secretary shall maintain records in reference thereto which shall include the date and time received, the name and address of the requester, the nature of the records requested, the action taken, the date the determination letter is sent to the requester, appeals and action thereon, the date any records are subsequently furnished, the number of staff hours and grade levels of persons who spent time responding to the request and the payment requested and received.

(4) All time limitations established pursuant to this section with respect to processing initial requests and appeals shall commence at the time a written request for records is received at the Board's offices in Washington, D.C., except for requests directed to the National Railroad Adjustment Board pursuant to § 1208.2(a) (1) in which case the time limit shall commence when the request is received at the NRAB's office in Chicago.

(i) An oral request for records shall not begin any, time requirement.

(b) Processing the Initial Request. (1) Time limitations. Within 10 working days (excepting Saturdays, Sundays and working holidays) after a request for records is received, the Executive Secretary shall determine and inform the requester by letter whether or the extent to which the request will be complied with, unless an extension is taken under paragraph (3) of this subsection.

(2) Such reply letter shall include:

 A reference to the specific exemption or exemptions under the Freedom of Information Act authorizing the withholding of the record, a brief explanation of how the exemption applies to the

record withheld.

(ii) The name or names and positions of the person or persons, other than the Executive Secretary, responsible for the denial.

(iii) A statement that the denial may be appealed within thirty days by writing to the Chairman of the Board, National Mediation Board, Washington, D.C. 20572, and that judicial review will thereafter be available in the district in which the requester resides, or has his principal place of business, or the district in which the agency records are situated, or the District of Columbia.

(3) Extension of time. In unusual circumstances as specified in this paragraph, the Executive Secretary may extend the time for initial determination on requests up to a total of ten days (excluding Saturdays, Sundays, and legal public holidays). Extensions shall be granted in increments of five days or less and shall be made by written notice to the requester which sets forth the reason for the extension and the date on which a determination is expected to be dispatched. As used in this paragraph "unusual circumstances" means, but only to the extent necessary to the proper processing of the request-

(i) The need to search for and collect the requested records from field facilities or other establishments that are separate from the office processing the re-

(ii) The need to search for, collect, and appropriately examine a voluminous amount of separate and distinct records which are demanded in a single request;

(iii) The need for consultation, which shall be conducted with all practicable speed, with another agency or another division having substantial interest in the determination of the request, or the need for consultation among components of the agency having substantial subject

matter interest therein.

(4) Treatment of delay as a denial. If no determination has been dispatched at the end of the ten-day period, or the last extension thereof, the requester may deem his request denied, and exercise a right of appeal, in accordance with § 1208.2c. When no determination can be dispatched within the applicable time limit, the responsible official shall nevertheless continue to process the request; on expiration of the time limit he shall inform the requester of the reason for the delay, of the date on which a determination may be expected to be dispatched, and of his right to treat the delay as a denial and to appeal to the Chairman of the Board in accordance with § 1208.2(c) and he may ask the requester to forego appeal until a determination is made.

(c) Appeals to Chairman of the Board. (1) When a request for records has been denied in whole or in part by the Executive Secretary or other person authorized to deny requests, the requester may, within thirty days of its receipt, appeal the denial to the Chairman of the Board. Appeals to the Chairman of the Board shall be in writing, addressed to the Chairman, National Mediation Board, Washington, D.C.

20572.

(2) The Chairman of the Board will act upon the appeal within twenty working days (excluding Saturdays, Sundays and legal public holidays) of its receipt unless an extension is made under par-

agraph (c) (3) of this section.

(3) In unusual circumstances as specified in this paragraph, the time for action on an appeal may be extended up to ten days (excluding Saturdays, Sundays and legal public holidays) minus any extension granted at the initial request level pursuant to § 1208.2(b) (3). Such extension shall be made by written notice to the requester which sets forth the reason for the extension and the date on which a determination is expected to be dispatched. As used in this paragraph "unusual circumstances" means; but only to the extent necessary to the proper processing of the appeal-

(i) The need to search for and collect the requested records from field facilities or other establishments that are separate from the office processing the request:

appropriately examine a voluminous

amount of separate and distinct records which are demanded in a single request;

(iii) The need for consultation, which shall be conducted with all practicable speed, with another agency or another division having substantial interest in the determination of the request or the need for consultation among components of the agency having substantial subject

matter interest therein.

(4) Treatment of delay as a denial. If no determination on the appeal has been dispatched at the end of the twenty-day period or the last extension thereof, the requester is deemed to have exhausted his administrative remedies, giving rise to a right of review in a district court of the United States as specified in 5 U.S.C. 552(a) (4). When no determination can be dispatched within the applicable time limit, the appeal will nevertheless continue to be processed; on expiration of the time limit the requester shall be informed of the reason for the delay, of the date on which a determination may be expected to be dispatched, and of his right to seek judicial review in the United States district court in the district in which he resides or has his principal place of business, the district in which the Board records are situated or the District of Columbia. The requester may be asked to forego judicial review until determination of the appeal.

(d) Indexes of Certain Records. (1) The National Mediation Board at its office in Washington, D.C. will maintain, make available for public inspection and copying, and publish quarterly (unless the Board determines by order published in the Federal Register that such publication would be unnecessary or impracticable) a current index of the materials available at the Board offices which are required to be indexed by 5 U.S.C. 552

(a) (2).

(i) A copy of such index shall be available at cost from the National Mediation Board, Washington, D.C. 20572.

(2) The National Railroad Adjustment Board at its offices in Chicago, Illinois will maintain, make available for public inspection and copying, and publish quarterly (unless the Board determines by order published in the FEDERAL REGISTER that such publication would be unnecessary or impracticable) a current index of the materials available at the Board offices which are required to be indexed by 5 U.S.C. 552(a)(2).

[FR Doc.75-4467 Filed 2-18-75:8:45 am]

Title 32—National Defense CHAPTER XII-DEFENSE SUPPLY AGENCY

SUBCHAPTER B-MISCELLANEOUS [DSAR 5400.14; RCS DD-PA(A) 1365]

PART 1285—AVAILABILITY TO THE P LIC OF OFFICIAL INFORMATION -AVAILABILITY TO THE PUB-

Freedom of Information Regulations; **Revision of Part** 

FEBRUARY 13, 1975.

Pursuant to the provisions of Pub. L. (ii) The need to search for, collect, and 93-502, enacted November 21, 1974, notice is hereby given of the publication

of revised DSA Regulation 5400.14 implementing 5 U.S.C. 552, as amended by Pub. L. 93-502, Freedom of Information Act. This regulation will become effective February 19, 1975. However, public comments and recommendations submitted to Defense Supply Agency, ATTN: DSAH-XA, Cameron Station, Alexandria, Virginia 22314 by April 18, 1975. will be considered in determining the need for modification of the regulation.

Effective date: February 19, 1975.

By order of the Director Defense Sunply Agency.

> GEORGE W. JOHNSON, Jr. Colonel, USAF, Staff Director, Administration.

Part 1285, Subchapter B, Chapter XII of Title 32 of the Code of Federal Regulations is revised to read as follows:

References. 1285.2 Purpose and Scope.

Policy. 1285.3

1285.4 Definitions.

Background. 1285.6 Responsibilities.

1285.7 Procedures.

Forms and Reports.

Appendix A

AUTHORITY: 5 U.S.C. 552, as amended by Pub. L. 93-502.

## § 1285.1 References.12

(a) 5 U.S.C. 552 as amended by Pub. L. 93-502.

(b) DoD 5200.1-R, Information Security Program Regulation.

(c) DSAR 5400.12, Administration of Congressional Matters.

(d) DSAR 4185.13, Management of Technical Data.

(e) DSAR 4185.5, DSA Technical Data Management Program.

(f) 5 U.S.C. 551 (Section 2 of the Administrative Procedure Act).

(g) DSAM 5205.1, Safeguarding Classified Information.

(h) 18 U.S.C. 1905 Disclosure of Confidential Information Generally.

(i) Pub. L. 86-36 (50 U.S.C. 401 note) National Security Information Exemption.

(j) 35 U.S.C. 181-188 Secrecy of Certain Inventions and Filing Applications in Foreign Countries.

(k) Pub. L. 90-620, 44 U.S.C. 3301 Definition of Records.

(1) Pub. L. 93-579, U.S.C. 552a, The Privacy Act of 1974.

(m) 42 U.S.C. 2162 Classification and Declassification of Restricted Data Periodic Determinations.

(n) DoD Directive 5400.9, Publication of Proposed and Adopted Regulations Affecting the Public.

### § 1285.2 Purpose and scope.

To announce policy and prescribe procedures governing the release to the public of DSA documentary material or records, or information therefrom,

<sup>1</sup>Referenced material may be obtained from Department of Defense, Defense Supply Agency, and Superintendent of Documents. <sup>2</sup>This publication supersedes Part 1285 published in Volume 38, No. 172, dated

Thursday, September 6, 1973.

pursuant to the provisions of 5 U.S.C. 552 as amended by Pub. L. 93-502, and to implement DoD Directive 5400.7, Availability to the Public of Department of Defense Information. This DSAR is applicable to HQ DSA and DSA field activities.

## § 1285.3 Policy.

(a) General. (1) It is the policy of DSA to make available to the public the maximum amount of information concerning its operations and activities. Exceptions to the requirements for disclosure will be made in accordance with the provisions of paragraph (d) of this section and the release procedures in § 1285.7.

(2) Information exempt from public disclosure, under provisions of paragraph (d) of this section, will be made available to the public when its disclosure is not inconsistent with statutory requirements or with DSAM 5205.1 and when officials determine that no significant purpose would be served by withholding the information. The determination of whether a significant purpose is served by withholding information under provisions of paragraph (d) of this section is within the sole discretion of the DSA official authorized by this Part 1285 to deny release.

(3) Records and other documents or related material may be withheld from the public only as authorized by this Part 1285. In no event will the determination that requested information comes within any of the specific exemptions in paragraph (d) of this section, or that information has not been properly requested, be influenced by the possibility that its release might suggest administrative error or inefficiency, or might embarrass DSA or an official of DSA.

(b) Availability of records. (1) Subject to the exemptions in paragraph (d) of this section, and the procedural requirements of § 1285.7, any record in the possession of DSA will be made available upon the request of any person. A request for information reasonably interpretable as being made pursuant to 5 U.S.C. 552 as amended by Pub. L. 93-502 will be processed in accordance with this Part 1285. Records which are published in accordance with DoD Directive 5400.9 or made available for inspection and copying in accordance with DoD Directive 5400.9 will, when practicable, also be copied and forwarded by the custodian of the records to those who request copies.

(2) A request for a record will be honored if:

(i) The requester describes the record sought with sufficient particularity to enable DSA to locate the record with a reasonable amount of effort, and

(ii) The requester is willing and able to pay the cost associated with locating and providing a copy of the record sought, as determined by enclosure 1.

(3) In order for a record to be considered "identifiable", it must exist at the time of the request. There is no obligation to "create" a record for the purpose of satisfying a request for informa-

tion. When the information requested exists in the form of several records at several locations, the activity receiving the request will normally compile the records. However, the applicant may be referred to the custodian of the records if gathering the information would be unduly burdensome on the DSA activity.

(4) When the record requested was originated by another agency, the request will be referred promptly and directly to that agency for disposition. The originating agency of a record should consult with other agencies having a significant interest in the content of a requested document before determining whether to make it available. A member of the public requesting material primarily originated by a Member of Congressional Committee will be advised to direct his request to the Member or Committee concerned.

(5) DSA activities will not create procedural obstacles when internal DSA or Department of Defense organizational questions arise, particularly where reorganization or transfer of functions contributes to an improperly directed request. DSA personnel will make all reasonable efforts to assist private persons in directing requests for information to the appropriate authorities.

(c) Inspection and copying of opinions, orders, and manuals. (1) Subject to the exemptions set forth in paragraph (d) of this section, DSA will make available for public inspection and copying in an appropriate facility or facilities in accordance with the procedures in § 1285.7, the following materials, unless such materials are published and copies offered for sale:

(i) All final opinions (including concurring and dissenting options) and orders in adjudication, as defined in 5 U.S.C. 551 (Section 2 of the Administrative Procedure Act) that may be cited, used, or relied upon as precedents in future adjudication.

(ii) Statements of policy and interpretations of less than general applicability affecting the public but not published in the FEDERAL REGISTER.

(iii) Administrative staff manuals and instructions, or portions of such, which establish DSA policy or interpretation of policy that is determinative of the rights of members of the public. This provision does not apply to instructions for employees on the tactics and techniques to be used in performing their duties, or to instructions relating only to the internal management of DSA. Examples of manuals and instructions not normally made available are:

(a) Those issued for audit and inspection purposes.

(b) Those which prescribe operational tactics, standards of performance, or criteria for defense, prosecution or settlement of cases.

(c) Operations and maintenance manuals and technical information concerning munitions, equipment, and systems.

(2) The cost to the DSA activity of copying any such opinion, order, or statement of policy or interpretation

shall be imposed on the person requesting the copy in accordance with enclosure 1.

(3) When feasible, all material published in the Federal Register will be made available for inspection and copying, along with any available index of that published material, in the same facility or facilities provided for inspection and copying of opinions, orders and manuals.

(4) Identifying details which if revealed would create a clearly unwarranted invasion of personal privacy may be deleted from any final opinion, order, statement of policy, interpretation, staff manual, or instruction made available for inspection and copying. In every such case, the justification for the deletion must be fully explained in writing.

(5) No order, opinion, statement of policy, interpretation, staff manual, or instruction which is issued, promulgated, or adopted after 4 July 1967 which is not indexed and either made available or published, may be relied upon, used, or cited as a precedent against any member of the public unless he has actual and timely notice of terms. If the order, opinion, statement of policy, interpretation, staff manual, or instruction was issued, promulgated, or adopted before 4 July 1967, it need not be indexed but must be made available in accordance with paragraph (c) (1) of this section. In determining whether an order, opinion, statement of policy, interpretation, staff manual, or instruction is likely to be used or relied upon as precedent, the primary test will be whether it is intended to provide binding guidance for decisions or evaluations by subordinates or for future decisions by the same authority in adjudications of cases affecting the public, where similar facts or issues are presented.

(d) Exemptions. (1) Documentary material which would otherwise have to be made available under paragraphs (b) and (c) of this section may be withheld from public disclosure if it comes within a specific exemption. However, even exempted material will be made available if, in the judgment of the releasing authority, no significant purpose would be served by withholding it from the requester under an applicable exemption and its release is not inconsistent with statutory requirements or DSAM 5205.1 and DoD Directive 5200.1-R.

(2) The following types of records will be withheld from public disclosure:

(i) Those properly classified in the interest of national defense or foreign policy, as specifically authorized under the criteria established by Executive Order and implemented by regulations, such as DSAM 5205.1 and DoD Directive 5200.1-R.

(ii) Those containing rules, regulations, orders, manuals, directives, and instructions relating to the internal personnel rules or to the internal practices of the Department of Defense if their release to the public would substantially hinder the effective performance of a significant function of the Department of Defense. Examples include:

(a) Operating rules, guidelines, and manuals for Department of Defense investigators, inspectors, auditors, or examiners, and schedules or methods which cannot be disclosed to the public without substantial prejudice to the effective performance of a significant function of the Department of Defense. Some of those materials would reveal:

(1) Negotiating and bargaining tech-

niques.

(2) Bargaining limitations and positions.

(3) Inspection schedules and methods. Audit schedule and methods. (4)

(b) Personnel and other administrative matters such as examination questions and answers used in training courses or in the determination of the qualifications of candidates for employment, entrance to duty, advancement or promotion.

(iii) Those containing information which statutes authorize or require to be withheld from the public. The authorization or requirement may be found in the terms of the statute itself or in Executive Orders or in regulations authorized by, or in implementation of, the statute.

Examples include: (a) National Security Agency information, governed by Pub. L. 86-36 (50 U.S.C. 401 note) (National Security

Agency Reporting Exemption).

(b) Any records containing information relating to inventions which are the subject of patent applications on which Patent Secrecy Orders have been issued, governed by 35 U.S.C. 181-188 (Secrecy of Certain Invention and Filing Applications in Foreign Countries).

(c) Information to be held in confidence under the Privacy Act of 1974.

(d) Information protected as restricted data, 42 U.S.C. 2162.

(iv) Those containing information which DSA receives from anyone including an individual, a foreign nation, an international organization, a State or local government, a corporation, or any other organization with the understanding that it will be retained on a privileged or confidential basis, or similar commercial or financial records which DSA develops internally, if they are in fact the kinds of records which are normally considered privileged or confidential, and when release of the record would adversely affect the competitive position of the source of the information. Such records include documents containing:

(a) Commercial or financial information received in confidence in connection with loans, bids, or proposals, as well as other information received in confidence or privileged, such as trade secrets and inventions and discoveries.

(b) Statistical data and commercial financial information concerning contract performance, income, profits, losses, and expenditures, if received in confidence from a contractor or potential contractor.

(c) Information customarily considered privileged or confidential under the rules of evidence in the Federal Courts.

(d) Personal statements given in the course of inspections, investigations, or audits, where such statements are received in confidence from the individual and retained in confidence because they cover trade secrets or commercial or financial information normally considered confidential or privileged or because they are essential to an effective inspection, investigation, or audit and could not otherwise be obtained.

(v) Except as provided in subdivisions (a) through (d) of this subparagraph, internal communications within and among elements of Federal agencies and of the Department of Defense.

(a) Examples include:

(1) Staff papers containing staff ad-

vice, opinions, or suggestions.

(2) Information received or generated by DSA preliminary to a decision or action, including draft versions of documents, where premature disclosure would interfere with the authorized appropriate purpose for which the records were created.

(3) Advice, suggestions, or reports prepared on behalf of DSA by boards, committees, councils, groups, panels, conferences, commissions, task forces, or other similar groups that are formed by DSA to obtain advice and recommendations or by individual consultants.

(4) Those portions of DSA activities evaluations of contractors and their products which contain recommendations or advice by Government employees about the contractor or product.

(5) Advance information on such matters as proposed plans to procure, lease or otherwise acquire and dispose of materials, real estate, facilities, or functions, when such information would provide undue or unfair competitive advantage to private personal interests.

(6) Records which are exchanged among agency personnel or within and among elements of Federal agencies and the Department of Defense preparing for anticipated legal proceedings before any Federal, State, or military court or before any regulatory body.

(7) Reports of inspections, audits, investigations, or surveys which pertain to safety, security or the internal management, administration, or operation of the

Department of Defense.

(b) If any such intra- or inter-agency record, or reasonably segregable portion of such record would routinely be made available through the discovery process in the course or litigation with the agency (i.e., the process by which litigants obtain information from each other that is relevant to the issues in a trial or hearing) then it will not be withheld from the general public. If, however, the record or parts of the record would only be made available through the discovery process by special order of the court based on the particular needs of a litigant balanced against the interests of the agency in maintaining its confidentiality, then the record or part of the record will not be made available to a member of the general public.

(c) A direction or order from a superior to a subordinate, though contained

in internal communication, is generally not withholdable from a requester if it constitutes policy guidance or a decision, as distinguished from a discussion of preliminary matters that would compromise the decision-making process.

(d) An internal communication concerning an event or decision which has subsequently been made a matter of public record, or was between officers or employees (or members of the armed forces) who no longer are actively employed by the DSA activity will normally be made available to a requester unless it is determined that, because of special circumstances, release would prejudice the current decision-making process.

(vi) Information in personnel and medical files, as well as information in similar files that, if disclosed to a member of the public, would result in a clearly unwarranted invasion of personal

privacy.

(a) Examples of files similar to per-

sonnel and medical files include:

(1) Those compiled to evaluate or adjudicate the suitability of candidates for civilian employment and the eligibility of individuals, civilian, military or industrial, for security clearances.

(2) Files containing reports, records, and other material pertaining to personnel matters in which administrative action, including disciplinary action, may

be taken.

- (b) In determining whether the release of information would result in a "clearly unwarranted invasion of personal privacy," consideration will be given to the stated or assumed purpose of the request. When determining whether a release is "clearly warranted," the public interest in satisfying this purpose must be balanced against the sensitivity of the privacy interest being threatened.
- (c) When the only basis for withholding information is protection of the personal privacy of an individual who is the subject of the record, the informa-tion will not be withheld from him or from his designated legal representative. A clearly unwarranted invasion of the privacy of others discussed in that record may, however, constitute a basis for deleting reasonably segregable portions of the record even when providing it to the subject of the record.

(d) An individual's personnel, medical, or similar files may be withheld from him or from his designated legal representative only to the extent consistent with Pub. L. 93-579, 5 U.S.C. 552a, The Pri-

vacy Act of 1974.

(vii) Those investigative records compiled for the purpose of enforcing civil, criminal, or military law, including the implementation of Executive Orders, or regulations validly adopted pursuant to law.

(a) But only to the extent that their release would:

(1) Interfere with enforcement proceedings.

(2) Deprive a person of the right to a fair trial or impartial adjudication.

(3) Constitute an unwarranted invasion of personal privacy.

(4) Disclose the identity of a confidential source.

(5) Disclose confidential information furnished only from a confidential source obtained by a criminal law enforcement authority in a criminal investigation or by an agency conducting a lawful national security intelligence investigation.

(6) Disclose investigative techniques and procedures not already in the public domain and requiring protection against public disclosure to ensure their

effectiveness.

(7) Endanger the life or physical safety of law enforcement personnel.

(b) Examples include:

(1) Statements of witnesses and other material based on the information developed during the course of the investigation and all materials prepared in connection with related Government litigation or adjudicative proceedings.

(2) The identity of firms or individuals suspended from contracting with the Department of Defense or being investigated for alleged irregularities when no indictment has been obtained nor any civil action filed against them by the

United States.

- (3) Information obtained in confidence in the course of a criminal investigation by a criminal law enforcement agency or office within a DSA activity, or a lawful national security intelligence investigation conducted by an authorized agency or office within a DSA activity for the purpose of obtaining affirmative or counterintelligence information, or background investigation information needed to determine suitability for employment or eligibility for access to classified information.
- (c) The right of individual litigants to investigative records currently available by law is not diminished.
- (d) When the subject of an investigative record is the requester of that record it may be withheld only in accordance with P.L. 93-579, 5 U.S.C. 552a, (The Privacy Act of 1974), and implementing regulations.

(viii) Those contained in or related to examination, operating or condition reports prepared by, on behalf of, or for the use of any agency responsible for the regulation or supervision of financial institutions.

(ix) Those containing geological and geophysical information and data (including maps) concerning wells.

### § 1285.4 Definitions.

- (a) Records. (1) In determining whether documentary material qualifies as a "record", consideration should be given to Pub. L. 90–620 44 U.S.C. 3301 Definitions of Records which defines the word "record" for records disposal purposes as follows:
- [It] includes all books, papers, maps, photographs or other documentary materials, regardless of physical form or characteristics, made or received by any agency of the United States Government in pursuance of federal law or in connection with the transaction of public business and preserved or appropriate for preservation by that agency or its legitimate successor as evidence of the orga-

nization, functions, policies, decisions, procedures, operations, or other activities of the Government or because of the informational value of data contained therein.

(2) The term "records" does not include objects or articles such as structures, furniture, paintings, sculpture, three-dimensional models, vehicles, equipment, etc., whatever their historical value or value as "evidence". Records are not limited to permanent or historical documents but include contemporaneous documents as well.

(3) Records include data stored in

omputers.

(4) Formulae, designs, drawings, research data, computer programs, technical data, and so forth are not considered "records" within the Congressional intent of 5 U.S.C. 552 as amended by Pub. L. 93-502. Because of development costs, utilization, or value, these items are considered property, not preserved for informational value nor as evidence of agency functions, but as exploitable resources to be utilized in the best interest of all the public. Requests for copies of such material will be evaluated in accordance with policies expressly directed to the appropriate dissemination or use of such property.

(b) Record custodian. The Heads of DSA primary level field activities and Heads of HQ DSA principal staff elements who have custody of the records.

## § 1285.5 Background.

Section 552 of Title 5 of the United States Code, as amended by Pub. L. 93-502, describes the kinds of official records and information that must be made available to the public and the kinds of information that need not be made available to the public, requires facilities be made available to the public for the examination and copying of records, and provides procedures for the review of denials to release records and information to the public. Requests from Members of Congress are governed by DSAR 5400.12.

## § 1285.6 Responsibilities.

(a) (HQ DSA. (1) The Chief, Publications Division, Office of Administration (DSAH-XM) will review DSA publications to assure that those which meet the criteria for publication in the Federal Register are prepared in proper form and transmitted promptly for publication in the Federal Register as required by DSAR 5025.19, Publication of Material in the Federal Register.

(2) The Chief, Administrative Management Division, Office of Administra-

tion (DSAH-XA) will:

(i) Serve as the point of contact for referring members of the public, except the news media, desiring to examine and copy records to the appropriate staff element within HQ DSA, or within the DSA Administrative Support Center having custody of the records.

(ii) Maintain for examination and copying by the public a copy of the material and documents referred to in subparagraph (b) (3) of this section.

(iii) Process requests for records received under provisions of § 1285.7(a) (i).

(iv) Maintain this DSAR in a current status and review it annually.

(3) The Special Assistant for Public Affairs (DSAH-B) will serve as the central point of contact for the release of information to the news media.

(4) The Counsel (DSAH-G) will:(i) Provide advice and assistance in

determining releasability of records.

(ii) Process appeals to the Director,
DSA, of denials to provide records.

(iii) Coordinate denials of releases with Office of the General Counsel, Department of Defense, and the Department of Justice, as appropriate.

(5) The Heads of HQ DSA Principal

Staff Elements will:

(i) Review all instructions for which they are the proponent to ensure that such instructions are not inconsistent with the provisions of this Part 1285.

(ii) Ensure that the provisions of this Part 1285 are followed in processing requests for records from members of the

public.

(iii) Coordinate with other HQ DSA staff elements, to the extent considered necessary, requests from the public for

information.

- (iv) Arrange for the collection of fees prescribed in enclosure 1 associated with locating and providing copies of documentary material requested. Screening records and transporting records for screening purposes may constitute a search cost payable by the requester, in advance of release, along with costs of duplication in accordance with enclosure 1.
- (v) Furnish DSAH-XA a copy of all denials to provide a record made under the provisions of § 1285.7(b) (13). Copies will be identified using the office symbol, calendar year, Freedom of Information Act (FOIA) and serially numbered, e.g., DSAH-X-75-FOIA-1.

(vi) Prepare and submit reports in ac-

cordance with § 1285.8(b).

(vii) Make initial determination to deny release after coordination with DSAH-G.

(b) The Heads of DSA Primary Level

Field Activities will:

(1) Ensure that the provisions of this Part 1285 are followed in processing requests for records from members of the public.

(2) Provide convenient facilities where members of the public may examine and copy documents to which they are en-

titled.

(3) Maintain for examination and copying by the public a copy of the following:

(i) All publications listed in § 1285.1.
 (ii) DSAH 5025.2, DSA Field Establishment Directory, which reflects the mailing addresses of all activities of DSA.

(iii) DSAH 5025.1, Defense Supply Agency Index of Publications (HSIs, Handbooks, Manuals and Regulations) which contains, among other things, a topical index of the publications issued by HQ DSA.

(iv) DSAM 5015.1, Files Maintenance and Disposition, which contains a functional description of all records and files generated by DSA. This manual also

contains a topical index.

(v) A copy of such local indexes of opinions, orders, statement of policy, and publications that may exist, or that may

be prepared in the future.

(4) Arrange for the collection of fees prescribed in enclosure 1 associated with locating and providing copies of documentary material requested. Screening records and transporting records for screening purposes may constitute a search cost payable by the requester, in advance of release, along with costs of duplication in accordance with enclosure 1.

(5) Furnish DSAH-XA a copy of all denials to provide a record made under provisions of § 1285.7(b) (13). In the case of the Defense Contract Administration Services Regions, the copy will be forwarded through the Deputy Director for Contract Administration Services. Copies will be identified using activity office symbol, calendar year, FOIA and serially numbered, e.g., DCSC-75-FOIA-1.

(6) Prepare and submit reports in ac-

cordance with § 1285.8(b).

(7) Refer cases of significance to DSAH-G for review and evaluation when the issues raised are unusual, precedent setting, matters of disagreement among components, or otherwise requiring special attention or guidance.

(8) Establish safeguards to ensure that the official records of the activity are properly safeguarded during the time they are made available for examination

by a member of the public.

(9) Ensure that internal operating procedures provide for prompt response to all requests for records.

(10) Establish a training program for those personnel that may be involved in passing on requests from the public for records.

### § 1285.7 Procedures.

(a) Requests for records or for permission to examine records. (1) Members of the public may make requests in writing for copies of records or for permission to examine or copy records directly to the Head of the DSA activity having custody of the records, if the location of the document is known. If the location is not known, and it is reasonably certain that the document is in the custody of DSA, the requester should submit the request to HQ DSA, ATTN: DSAH-XA, Cameron Station, Alexandria, Virginia 22314.

(2) Requests must identify each record with sufficient particularity to enable the custodian to locate the record with a reasonable amount of effort. Information as to where the record originated, its subject, date, number, or other identification that would enable the custodian to locate the document should be provided by the requester when possible. The DSA activity may require the requester to complete a form to facilitate efforts to locate a record not otherwise reasonably described.

(3) Because certain information and documents are exempt from the imposition of fees under provisions of enclosure 1. the requester need not submit payment for services with the initial written re-

quest. When it is anticipated that the cost of the record search and reproduction of the documents may exceed \$10.00, the requester will be so advised and requested to submit payment prior to furnishing the records. In other instances, the requester will be advised at the time the record is found and prior to the release as to the charges involved.

(4) Refusal to make a record available may be made only by the Heads of DSA primary level field activities or the Heads of HQ DSA principal staff elements. The refusal may be appealed to the Director,

DSA.

(b) Processing requests for records or for permission to examine records. (1) Upon receipt of a request for records or for permission to examine records, the DSA activity having custody of the records will collect the documents, determine whether they are releasable under provisions of this DSAR, determine the fees to be charged, and advise the requester accordingly.

(2) If the request is for permission to examine releasable records, the requester will be advised as to where and when during normal working hours he may appear for this purpose. Every reasonable effort will be made to accommodate individuals granted permission to examine records; however, overtime is not authorized for this purpose.

(3) The DSA official having custody and control of any DSA record, requested by a member of the public, is authorized to make such record available unless the record falls within one of the exemptions listed in § 1285.3(d). In such case the request will be referred promptly to the Head of the primary level field activity or the Head of a HQ DSA principal staff element, as appropriate. The marking or absence of the marking "FOR OFFICIAL USE ONLY" does not relieve the official who is authorized to release the record from the responsibility of reviewing the requested record for the purpose of determining whether an exemption under § 1285.3(d) is applicable.

(4) The official designated by DSA to make initial determinations, if not a public affairs officer, should consult with the appropriate public affairs officer to determine if the subject matter is considered to be newsworthy. The Public Affairs Officer will be advised of all requests from news media representatives. In addition, public affairs officers will be informed in advance whenever a record containing potentially newsworthy material is to be released or to be withheld when it is likely that the withholding action will be publicly challenged.

(5) Initial determinations will normally be made within 10 working days of the date a request is received by the official designated to respond for the type of record sought, providing the requester indicates a willingness to reimburse the DSA activity for any search and duplication costs incurred in providing the record. If the willingness of the requester to reimburse the DSA activity for any required search and duplication costs is not expressed in the request, this issue must be resolved before

the time for responding begins to run. The requester will be contacted as expeditiously as possible, usually by telephone, for assurance of his intent to reimburse the Government for these costs.

(6) When the request is received by someone other than the records custodian, it will be forwarded promptly to the records custodian with the period for response commencing upon his receipt.

(7) When a request is received for records which were obtained by DSA from a non-U.S. government source; or contain information · obtained by DSA from non-U.S. government source and because of the source and the nature of the records or information, there is reason to believe that the source of the information or records may object to release and may have an enforceable right to prevent release, prompt notification of intended release shall be given to the source. Release will normally be withheld until the source has a reasonable time to comment on the proposed release. Comments received will be considered in determining the releasability of the document. When the source advises that it is seeking a restraining order or other court action to prevent release, release will normally not be made pending the outcome of the court action.

(8) When a decision is made to release a record, it will be forwarded promptly to the requester, upon receipt of any required payment for search and dupli-

cation.

(9) In all cases where the time for response may become an issue, the official responsible for replying will acknowledge to the requester the date of the receipt of the request for purposes of determining time limits.

(10) If additional time is needed in unusual circumstances to respond to requests for records, the DSA activity will acknowledge the request in writing within the 10 day period, briefly cite one of the unusual circumstances requiring delay, and indicate the anticipated date for substantive response which may not exceed 10 additional working days. Unusual circumstances that may justify delav are:

(i) The requested records are located in whole or part at places other than the office processing the request.

(ii) The request requires the collection and coordination of a substantial number of records.

(iii) Consultation is required with other DoD components or agencies having substantial interest in the subject matter of the requested records to determine whether the records requested in whole or part are exempt from disclosure under section VI of DoD Directive 5400.7 or should be released as a matter of discretion.

(11) The extension of time for responding to an initial request must be approved on a case by case basis by DSAH-G.

(12) Requests for records will be denied only by the Head of a DSA primary level field activity or the Head of a HQ DSA principal staff element upon a determination after consultation with the appropriate Office of Counsel that:

(i) The record is subject to one or more of the exemptions set forth in § 1285.3(d) of this Part 1285 and a significant and legitimate Governmental purpose is served by withholding it, or

(ii) The record cannot be found because it has not been described with sufficient particularity to enable the DSA activity to locate it with a reasonable

amount of effort.

(13) When a request for a record or records is denied in whole or part, the designated official who has made the determination will explain to the requester in writing (with at least one additional copy) the basis for the determination, that he may appeal the denial to the Director, DSA and that such appeals should contain the basis for disagreement with the denials.

(i) Inability to process any part of the request within the specified time will be explained to the requester, with notification that he may treat this delay as an initial denial with a right to appeal, or that he may agree to await a substantive response by an anticipated date. It will be made clear that any such agreement does not prejudice the right of the requester to appeal the initial

decision after it is made.

(ii) The explanation of the substantive basis for a denial will include both specific citation of the exemption applied under the authority of § 1285.3(d), and a short discussion of the significant and legitimate Governmental purpose served by invoking an exemption. Reference to the marking "FOR OFFICIAL USE ONLY" on the requested record does not constitute a proper citation or explanation of the basis for invoking an exemption.

(iii) The name and title or position of the official responsible for the denial will be included in the written response

to the requester.

(iv) When the initial denial is based in whole or part on a security classification pursuant to § 1285.3(d) (2) (i), the explanation will include a summary of the paragraph or paragraphs contained in DSAM 5205.1 which set forth the criteria or rationale for the current classification of the requested record. In addition, it will advise the requester of his optional right under DSAM 5205.1 and DoD Directive 5200.1–R to seek declassification review of a record more than 10 years old.

(v) Copies of all initial denials will be maintained by each DSA activity in a form suitable for rapid retrieval, periodic statistical compilation, and man-

agement evaluation.

(vi) When there is good reason to believe that the requester will appeal the denial, the record, the letter of denial, and such other information as will assist the Director, DSA, will be promptly forwarded to DSAH-G in order to permit timely response of the anticipated appeal.

(14) Final determination on appeals will normally be made within 20 working days of the receipt of the appeal by

DSAH-G. Misdirected appeals will be forwarded promptly to the Director, DSA, with the period for response commencing upon his receipt, which should be acknowledged. If additional time is needed to decide the appeal because of unusual circumstances, as described in subparagraph 10 of this paragraph, the final determination may be delayed for the number of working days, not to exceed 10, which were not utilized as additional time for responding to the initial request.

(15) A final denial to provide a requested record will be made in writing by the Director, DSA. Such a denial will be made in accordance with appeal procedures prescribed in this DSAR and will include, as a minimum, the follow-

ing elements:

(i) The basis for the denial will be explained to the requester, in writing, both with regard to the applicable exemption under § 1285.3(d), and the significant and legitimate governmental purpose served by its withholding. More particularly:

(a) When the final denial is based in whole or part on a security classification, pursuant to § 1285.3(d) (2) (i), the explanation will include a statement that the record meets the cited criteria and rationale of DSAM 5205.1 and that this determination is based on a declassification review. A brief explanation of why that review confirmed the continuing validity of the security classification will also be included.

(b) The requester will be advised of his optional right to seek declassification of the record by the appropriate component Classification Review Committee established pursuant to DoD 5200.1–R.

(c) Appeals from a denial by the appropriate component Classification Review Committee may be made to the Interdepartmental Classification Review Committee, establish pursuant to Executive Order 11652, March 8, 1972, in lieu of immediate judicial review.

(d) The written final denial will include the name and title or position of the official responsible for the denial and of the provision for judicial review of the

denial set forth in § 1285.8.

(ii) No final denial will be made without prior consultation with the Office of the General Counsel of the Department of Defense when there is reason to believe that the requester will file a complaint in a U.S. District Court to force release of the denied record.

(iii) Copies of all final denial letters will be maintained by DSAH-XA in a central repository. Whenever a complaint is filed in a U.S. District Court to force release of the record, a copy of the final denial letter from DSA will be forwarded to the General Counsel of the Department of Defense. DSAH-G will also furnish to the General Counsel a copy of the requester's complaint and the DSA litigation report.

(iv) When the denial to provide the record is based in whole or part on a security classification, pursuant to \$ 1285.3(d) (2) and \$ 1285.3(d) (2) (iii), the litigation report will include an affidavit from the Director, DSA, for this

purpose explaining in as much detail as national security interests permit the basis under applicable statute, executive order, and regulations for the current security classification of the requested record.

(16) The costs of searching for and duplicating a requested record must be paid or waived in accordance with enclosure 1. The time limits for responding to requests will be computed from the date of receipt from the requester of clear evidence of willingness to pay any anticipated search and duplication costs under the schedule of fees set forth in enclosure 1 for providing the requested record. The record need not be forwarded until actual receipt of payment.

(c) Release and authentication of copies of official records. (1) Records available to a person, requesting them under § 1285.3(b), will be authenticated with an appropriate seal whenever necessary to fulfill an official governmental

or other legal function.

(2) Records exempt from release to a person requesting them may, nevertheless, be authenticated on request, and released in accordance with this DSAR to local, State, or other Federal Governmental bodies, whether legislative, executive, administrative, or judicial, as follows:

(i) To the courts: whenever ordered as appropriate to the proper administra-

tion of justice.

(ii) To the Congress: in accordance with DSAR 5400.12.(iii) To local and State legislative

bodies: in accordance with the determi-

nation of the Director, DSA.

(iv) To other Federal agencies, both executive and administrative as determined by the Director, DSA, as consistent with efficient administration and in accordance with law, including Pub. L. 93–579, 5 U.S.C. 552a, The Privacy Act of 1974.

(v) To local and State executive and administrative agencies as determined

by the Director, DSA.

## § 1285.8 Forms and reports.

(a) An annual report will be prepared by DSAH-XA for the preceding calendar year on DSA's implementation of 5 U.S.C. 552 as amended by Pub. L. 93-502 as follows:

(1) The number and reasons for initial denial of records requested from

DSA.

- (2) The number of appeals from initial denials and the disposition of each such appeal; with the reasons for any final denial of records requested from DSA.
- (3) The names and titles or positions of each person primarily responsible for an initial denial or for a final denial on appeal of a request for a record under 5 U.S.C. 552 as amended by Pub. L. 93–502 and the number of such denials by each such person.
- (4) The results of any disciplinary proceeding, including an explanation of a decision not to discipline that was initiated against an officer or employee because a court determined that a record

requested from DSA was arbitrarily or capriciously withheld.

(5) A copy of the current DSAR implementing 5 U.S.C. 552 as amended by Pub. L. 93-502.

(6) Total amount collected for search and duplication under this DSAR as accurately and as reasonably as can be determined.

(7) A brief description of every court case brought against DSA to force the release, or withhold a record requested under 5 U.S.C. 552 as amended by Pub. Li. 93-502, including the status of the case, whether attorney fees have been awarded to the private party, and whether the court found that the circumstances of withholding raised questions of arbitrary and capriclous personnel conduct.

(8) A description of all efforts undertaken by DSA or by personnel of DSA to instruct and educate employees or the public on the requirements of 5 U.S.C. 552 as amended by Pub. L. 93-502.

(9) Such data on the costs of processing requests under 5 U.S.C. 552 as amended by Pub. L. 93-502 as can reasonably be ascertained or estimated.

(10) Any other information that demonstrates efforts by DSA to implement 5 U.S.C. 552 as amended by Pub. L. 93–502, including problems with the implementation and any proposed solutions for those problems.

(11) Six copies of the annual report will be furnished to the Assistant Secretary of Defense (Public Affairs), on or before 1 February of each year for transmittal to the Speaker of the House of Representatives, and to the President of the Senate.

(b) An annual feeder report on the actions of the preceding year will be prepared by Heads of DSA primary level field activities and Heads of HQ DSA principal staff elements for submission, in duplicate to DSAH—XA on or before 15 January each year. The report will contain the following:

(1) The number of initial denial of records requested.

(2) The total amount collected for search and duplication under this DSAR as accurately and as reasonably as can be determined.

(3) Such data on other costs of processing requests under 5 U.S.C. 552 as amended by Pub. L. 93-502 as can reasonably be ascertained or estimated.

(4) A description of all efforts undertaken by the DSA activity personnel to instruct and educate employees or the public on the requirements of 5 U.S.C. 552 as amended by Pub. L. 93-502 and this Part 1285.

(5) Any other information that demonstrates efforts by the DSA activity to implement 5 U.S.C. 552 as amended by Pub. L. 93-502, including problems with the implementation and any proposed solutions for those problems.

(6) The Reports Control Symbol for these annual reports will be DD-PA(A) 1365.

#### APPENDIX A

UNIFORM AGENCY FEES FOR SEARCH AND DUPLI-CATION UNDER THE FREEDOM OF INFORMATION ACT AMENDMENTS PUB. L. 93-502

#### SCHEDULE OF FEE

Duplication, Publications, Forms and Reports

Shelf stock of printed or microfiche medium (requesters may be furnished more than one copy of a publication or form if it does not deplete stock levels below projected planned usage).

Minimum fee, per requestPlus:	\$2.00
Forms, per copy Publications, per printed page	
Microfiche, per fiche Reports, per printed page	. 06
rechorms, her brimeen bagarring	

(Examples: Cost of 20 forms, \$3; cost of a printed publication with 100 pages, \$3; cost of a microfiche publication consisting of 10 fiche, \$2.60.)

13.00

Minimum charge

Computer service charges will be
based on actual computer configuration used and be based on
direct costs only if the Central
Processing Unit plus Input/Output Devices plus Memory Capacity

quest) per hour\_\_\_\_

Exceptions.—1. In general, charges may be waived in the public interest when:

a. The recipient of the benefits is engaged in a nonprofit activity designed for public safety, health or welfare.

b. Payment of the full costs or fee by a state, local government or nonprofit group would not be in the interest of the program.

c. The incremental cost of collecting the fees would be an unduly large part of the receipts from the activity.

2. A refusal to waive charges by the official responsible for the initial decision on the request for the record may be appealed to the Director, DSA, for purposes of final approval.

Collections.—1 Normally, collection of charges and fees will be made in advance of rendering the service. In some instances, it may be more practical to collect charges and fees at the time of conveying the service or property to the recipient, but only in those instances where the request specifically states that whatever cost involved will be acceptable or acceptable up to a specified limit that covers anticipated costs. The time for responding to a request begins to run upon agreement to pay for anticipated costs of search and duplication or upon actual receipt of payment.

Collection of scheduled fees and charges will normally be deposited to Miscellaneous Receipts of the Treasury.

3. Search fees are assessable even when no records responsive to the request, or no records not exempt from disclosure are found, provided the requester is advised of the requirement at the time the estimated charges are presented to the requester for approval and he agrees to incur the costs of search.

IDENTIFICATION AND MARKING "FOR OFFICIAL USE ONLY"

1. At the time of their origination, records which are not classified under DSAM 5205.1 but which are authorized by DoD Directive 5400.7 to be withheld from the general public disclosure under paragraph III D of this DSAR, and which for a significant reason should not be given general circulation, will be considered as being "FOR OFFICIAL USE ONLY" (FOUO). No other record may be considered as being FOUO.

2. A record that is considered FOUO will be marked "FOR OFFICIAL USE ONLY" when such marking is deemed necessary or desirable to ensure that all persons having access to the record are aware that it should not be publicly released and should not be handled indiscriminately. Individual folders, records, and files covering specific kinds of subject matter, normally falling within the exemptions of paragraph III D of this DSAR, such as personnel and medical files, bids, proposals, and the like, which are covered by rules and regulations specifying what may be released publicly, do not require the FOUO marking unless handled under circumstances where marking is desirable to ensure protection of the information involved.

a. The marking may not be used on records or portions of records which are classified under DSAM 5205.1 and DoD Directive 5200.1-R, but, if otherwise proper under this DSAR, may be applied to information or material which has been declassified.

b. The marking may not be employed as a less stringent security designation under conditions where classification under DSAM 5205.1 is not warranted.

c. Information contained in a technical document for which a determination has been made that a distribution statement under DSAR 4185.5 is appropriate will not be marked FOUO.

3. Material which is considered FOUO must be safeguarded from general disclosure irrespective of whether the material is physically marked with the term "FOR OFFICIAL USE ONLY."

4. Whenever necessary to assure proper understanding, or appropriate as a means of facilitating segregation of exempt information in a lengthy record, individual paragraphs which contain FOUO information will be marked with the symbol FOUO. In classified documents, this marking will be applied only to paragraphs which contain FOUO information and do not contain classified information. Instructions regarding marking, safeguarding, and transmitting FOUO materials are set forth in 42 U.S.C. 2162 (Classification and Declassification of Restricted Data Periodic Determination).

### TITRICITAL ACTION

1. A requester has exhausted his administrative remedy after he has been refused a record by the Director, DSA, or when the DSA activity fails to respond to his request within any of the time limits set forth in paragraph VII of this DSAR. The requester may then seek an order from a United States

District Court, in the district in which he resides or has his principal place of business; in the district in which the record is situated to produce the record; or in U.S. District Court for the District of Columbia.

2. The burden is on DSA to justify its denial to produce the record, and its justi-fication will be evaluated de novo by the District Court, which may examine any requested record in camera to determine whether the denial of the record, in whole or part, is justified under 5 U.S.C. 552 as amended by P.L. 93-502.

3. The United States must answer or otherwise plead to the complaint within 30 days. Consideration of such cases by both trial courts and courts of appeal will be expedited in every way, except as to cases the court considers of greater importance.

4. A court may retain jurisdiction and allow DSA additional time to complete its review of records to determine their availability to the requester when DSA demonstrates due diligence in exceptional circum-

5. If the court orders production of a record and the responsible official designated in accordance with paragraph VII of this DSAR refuses to produce it, the court has statutory authority to punish the official whose decision and order govern the determination of whether to release the record. In addition, the court may assess against the United States reasonable attorney fees and other litigation costs when the requester has substantially prevailed. In such situations the court may also issue a written finding that the circumstances of withholding, raise a question of whether DSA personnel acted arbitrarily or capriciously, and the Civil Service Commission must determine whether disciplinary action against responsible DSA officials, or employees is warranted.

[FR Doc.75-4501 Filed 2-18-75;8:45 am]

### SUBCHAPTER P-RECORDS

### PART 298-FREEDOM OF INFORMATION **PRACTICES**

The Director. Defense Investigative Service (DIS) approved Part 298. This Part 298 has been added based upon a requirement contained in Congressional enacted amendments to Section 552 of Title 5. United States Code. The act requires that agencies publish in the Fen-ERAL REGISTER their rules of procedure for compliance with the Freedom of Information Act. In consideration of the foregoing the following are procedures in effect within the Defense Investigative Service, a component of the Department of Defense.

298.1 Purpose.

DIS mission. 298.2

298.3 DIS organization.

Records maintained by DIS. 298.4

Procedure for release of DIS records.

298.6 Effective date.

AUTHORITY: The provisions of Part 298 issued under 5 U.S.C. 552 as amended by Public Law 93-502.

## § 298.1 Purpose.

This part sets forth the mission of the Defense Investigative Service (DIS), its organization, and policy and procedures for the public to obtain information from DIS records.

## § 298.2 DIS mission.

(a) The Defense Investigative Service was established on April 18, 1972, by Department of Defense (DoD) Directive 5105.42 (38 FR 7131, March 16, 1973) and became operational on October 1, 1972. DIS is a separate operating agency within the DoD, under control of the Secretary of Defense with staff supervision provided by the Assistant Secretary of Defense (Comptroller).

(b) The mission of DIS is to provide the Department of Defense (and, when authorized by the Secretary of Defense, other U.S. Government activities), with a single centrally directed personnel security investigative service. DIS also conducts criminal investigations and crime prevention surveys, and participates in joint investigative task force operations, in support of the Defense Supply Agency (DSA). Finally, it conducts such special investigations as the Secretary of Defense may direct. DIS operates the Defense Central Index of Investigations (DCII), a consolidated index of investigations conducted by all components of the DoD, and the DoD National Agency Check Center (NACC).

#### § 298.3 DIS organization.

DIS organization includes a headquarters located in Washington, DC, as well as 20 district offices and over 200 subordinate operating locations throughout Continental United States (CONUS), Alaska, Hawaii, and Puerto Rico. A copy of the DIS Regulation 01-13,1 "DIS Mailing List," showing addresses of all offices is available to the public upon request and may be obtained from the DIS Public Information Officer, Defense Investigative Service (DO102), Washington, DC 20314.

## § 298.4 Records maintained by DIS.

It is the expressed policy of DIS to make available to the public all information which may be released within the spirit of the Freedom of Information Act with due regard to protecting the privacy of any individual whom it investigates. In this regard DIS maintains the following records which may be of interest to the public:

(a) The DCII, which contains references to investigative records created and held by DoD components. The records indexed are primarily those prepared by the investigative agencies of the Military departments and DIS, covering criminal. fraud, counterintelligence, and personnel security information. Information from the DCII is not available to the general public since release would violate the privacy of individuals whose names are indexed therein.

(b) Records created as required by DoD Directive 5105.42, "Charter for the Defense Investigative Service" (38 FR 7131, March 16, 1973).

(c) Investigative records created in the furnishing of criminal investigative support to DSA.

(d) Publications listed in DIS Regulation 00-1,1 "Numerical Index of Standard DIS Publications". A copy of DISR 00-1 may be obtained upon request from

the DIS Public Information Officer, Defense Investigative Service (D0102), Washington, DC 20314. DISR 00-1 will be provided for the convenience of possible users of the materials but does not constitute a determination that all or any of the publications listed affect the public or have been cleared for release.

§ 298.5 Procedure for release of DIS records.

The following procedure is applicable to requests for records by the public, except that subjects of investigative files or their counsel may obtain summaries or extracts under separate procedures for use in administrative or judicial proceedings.

(a) All requests will be submitted in writing and addressed to:

PUBLIC INFORMATION OFFICER (FOIA REQUEST) INVESTIGATIVE SERVICE (D0102)

WASHINGTON, D.C. 20314

(b) All requests will contain the following information:

(1) As complete an identification as possible of the desired material, including to the extent known, the title, description, and date.

(2) With respect to material concerning investigations of personnel: the first name, middle name or initial, surname, date and place of birth, military service number, and social security account number of the individual concerned. (These items are required to consult the

index effectively.) (3) A statement as to whether the requester wishes to inspect the record or

obtain a copy of it.

(4) A statement that all costs for search and duplication, if requested (or costs to a specified limit), will be borne by the requester even if no record, or no releasable record, is found.

(5) In all third-party requests for investigative files (e.g., spouse or executor), a statement indicating what use the requester will make of the record.

(6) The full address (including ZIP

code) of the requester.

(c) A notarized request by an individual requesting investigative or other personal records may be required to avoid the risk of invasion of privacy, Requesters will be notified and furnished appropriate forms if this requirement is deemed necessary.

(d) When a request is incomplete or fails to include all of the information required, the requester will be contacted for additional information prior to be-

ginning release procedures.

(e) DIS will dispatch responses to requests within ten working days after receipt by the Information Office, unless an extension is required and the requester notified.

(f) When the release of information has been approved, a statement of estimated costs computed in accordance with the DoD Fee Schedule, or a statement waiving the fee, will be included in the notification of approval. Records approved for release will generally be mailed immediately following receipt of fees. Fees may be waived or reduced

<sup>&</sup>lt;sup>1</sup> Filed as part of original.

where release of information is determined to benefit the general public. Remittances should be in the form of a personal check, bank draft, or postal money order. Remittances are to be made payable to the Treasurer of the United States.

(g) When requests are denied in whole or part, the requester will be advised of the identity of the official making the denial, the reason for the denial, the right of appeal of the decision, and the identity of the person to whom an appeal

may be addressed.

(h) Facilities for the review or reproduction of records following approval of request or appeal are available at the DIS Public Information Office, Room 2H063, 1000 Independence Avenue, SW (James Forrestal Building), Washington, D.C. Subject to the approval of the Director, DIS, and upon request of an individual requester, arrangements may be made to review documents at one of the DIS field locations within the United States or Puerto Rico. Depending upon the field location, reproduction service may or may not be available at the specific office.

(i) Appeal of denial to DIS records

and information:

(1) All appeals will be submitted in writing and addressed to:
DIRECTOR (FOIA APPEAL)

DEFENSE INVESTIGATIVE SERVICE (D0102)

WASHINGTON, D.C. 20314

(2) All appeals will contain at least the same identification of the records requested as the original request, and a copy of the letter denying the request, if available.

(3) All appeals will be reviewed by the Director or Deputy Director, DIS. Responses to appeals will be dispatched within ten working days after receipt, unless an extension is required and appellant notified. When a request is approved on appeal, the procedures set forth in Section 298.5(f) will be followed. § 298.6 Effective date.

The effective date of this Part is February 19, 1975.

MAURICE W. ROCHE, Director, Correspondence and Directives, OASD (Comptroller).

FEBRUARY 15, 1975.

[FR Doc.75-4596 Filed 2-18-75;8:45 am]

## Title 29—Labor

#### CHAPTER I—NATIONAL LABOR RELATIONS BOARD

PART 102—RULES AND REGULATIONS, SERIES 8

Subpart K—Records and Information
FREEDOM OF INFORMATION REGULATIONS

By virtue of the authority vested in it by the National Labor Relations Act, approved July 5, 1935,¹ the National Labor

Relations Board hereby issues the following amendments to §§ 102.117 and 102.118 of its rules and regulations, Series 8, as amended, which it finds necessary to carry out the provisions of said Act.

The agency fee schedule set forth in § 102.117(c) (2) (iv) is issued pursuant to a notice of proposed rulemaking published in the Federal Register on January 14, 1975, 40 FR 2591-2592. The Board having considered the responses thereto and in the exercise of its authority under the Act has determined to adopt the rule as proposed.

1. Section 102.117 is revised to read as follows:

§ 102.117 Board materials and formal documents available for public inspection and copying; requests for described records; time limit for response; appeal from denial of request; fees for document search and duplication; files and records not subject to inspection.

(a) (1) The following materials are available to the public for inspection and copying during normal business hours: (i) All final opinions and orders made in the adjudication of cases; (ii) administrative staff manuals and instructions that affect any member of the public (excepting those establishing internal operating rules, guidelines, and procedures for the investigation, trial, and settlement of cases); (iii) a record of the final votes of each member of the Board in every agency proceeding; and (iv) a current index of final opinions and orders made in the adjudication of cases. Paragraphs (a) (1) (i) through (iv) of this section are available for inspection and copying during normal business hours at the Board's offices in Washington, D.C. Paragraphs (a) (1) (ii) and (iv) of this section are also available for inspection and copying during normal business hours at each regional, subregional, and resident office of the Board. Final opinions and orders made by regional directors in the adjudication of representation cases pursuant to the delegation of authority from the Board under section 3(b) of the act are available to the public for inspection and copying in the original office where issued.

(2) Copies of forms prescribed by the Board for the filing of charges under section 10 or petitions under section 9 may be obtained without charge from any regional, subregional, or resident office of the Board.

(b) (1) The formal documents constituting the record in a case or proceeding are matters of official record and, until destroyed pursuant to applicable statutory authority, are available to the public for inspection and copying during normal business hours, at the appropriate regional office of the Board or at the Board's office in Washington, D.C., as the case may be.

(2) The executive secretary shall certify copies of the formal documents upon request made a reasonable time in advance of need and payment of lawfully prescribed costs.

(c) (1) Requests for the inspection and copying of records other than those specified in paragraphs (a) and (b) of this section must be in writing and must reasonably describe the record in a manner to permit its identification and location. The envelope and the letter should be clearly marked to indicate that it contains a request for records under the Freedom of Information Act (FOIA). The request must contain a specific statement assuming financial liability in accordance with paragraph (c) (2) (iv) of this section, for the direct costs of the search for the requested records and their duplication. If the request is for records in a regional or subregional office of the agency, it should be made to that regional or subregional office; if for records in the office of the general counsel and located in Washington, D.C., it should be made to the Freedom of Information officer, office of the general counsel, Washington, D.C.; and if for records in the offices of the Board in Washington, D.C., to the executive secretary of the Board, Washington, D.C. Requests made to other than the appropriate office will be forwarded to that office by the receiving office, but in that event the applicable time limit for response set forth in paragraph (c) (2) of this section shall be calculated from the date of receipt by the appropriate office.

(2) (i) Within 10 working days after receipt of a request by the appropriate office of the agency a determination shall be made whether to comply with such request, and the person making the request shall be notified in writing of that determination. If the determination is to comply with the request, the records shall be made promptly available to the person making the request, upon payment of any charges due in accordance with the provisions of paragraph (c) (2) (iv) of this section. If the determination is to deny the request, the notification shall set forth the reasons therefor and the name and title or position of each person responsible for the denial, and shall notify the person making the request of the right to appeal the adverse determination under the provisions of

subparagraph (ii) of this subsection.

(ii) An appeal from an adverse determination made pursuant to paragraph (c) (2) (i) of this section must be filed within 20 working days of the receipt by the person of the notification of the adverse determination where the request is denied in its entirety; or, in the case of a partial denial, within 20 working days of the receipt of any records being made available pursuant to the request. If the adverse determination was made in a regional office, a subregional office, or by the Freedom of Information officer, office of the general counsel, the appeal shall be filed with the general counsel in Washington, D.C. If the adverse determination was made by the executive secretary of the Board, the appeal shall be filed with the chairman of

<sup>&</sup>lt;sup>1</sup>49 Stat. 449; 29 U.S.C. 151-166, as amended by act of June 23, 1947 (61 Stat. 136; 29 U.S.C. Sup. 151-167), act of October 22, 1951 (65 Stat. 601; 29 U.S.C. 158, 159, 168), and act of September 14, 1959 (73 Stat. 519; 29 U.S.C. 141-168).

80.10

the Board in Washington, D.C. Within 20 working days after the receipt of an appeal the chairman of the Board or the general counsel, as the case may be, shall make a determination with respect to such appeal and shall notify the person in writing. If the determination is to comply with the request, the records shall be made promptly available to the person making the request, upon receipt of payment of any charges due in accordance with the provisions of paragraph (c) (2) (iv) of this section. If on appeal the denial of the request for records is upheld in whole or in part, the person making the request shall be notified of the reasons for the determination, the name and title or position of each person responsible for the denial. and the provisions for judicial review of that determination under the provisions of 5 U.S.C. section 552(4)(B). Even though no appeal is filed from a denial in whole or in part of a request for records by the person making the request, the chairman of the Board or the general counsel may, without regard to the time limit for filing of an appeal, sua sponte initiate consideration of an adverse determination under this appeal procedure by written notification to the person making the request. In such event the time limit for making the determination shall commence with the issuance of such notification.

(iii) In unusual circumstances as specified in this subparagraph, the time limits prescribed in either paragraph (c) (2) (i) or (ii) of this subsection may be extended by written notice to the person requesting the record setting forth the reasons for such extension and the date on which a determination is expected to be dispatched. No such notice or notices shall specify a date or dates that would result in an extension or extensions totaling more than 10 working days with respect to a particular request. As used in this subparagraph, "unusual circumstances" means, but only to the extent reasonably necessary to the proper processing of the particular request:

(a) The need to search for and collect the requested records from field facilities or other establishments that are separate from the office processing the request;
(b) The need to search for collect, and

(b) The need to search for, collect, and appropriately examine a voluminous amount of separate and distinct records which are demanded in a single request; or

(c) The need for consultation, which shall be conducted with all practicable speed, with another agency having a substantial interest in the determination of the request or among two or more components of the agency having substantial subject-matter interest therein.

(iv) Persons requesting records from this agency shall be subject to a charge of fees for the direct cost of document search and duplication in accordance with the following schedules, procedures, and conditions:

(a) Schedule of charges:

(1) For each one-quarter hour or portion thereof of clerical time \$1.10

(2) For each one-quarter hour or portion thereof of professional time

(3) For each sheet of duplication (not to exceed 8½ x 14 inches) of requested records

(b) Each request for records shall contain a specific statement assuming financial liability, in full or to a specified maximum amount, for charges, in accordance with paragraph (c) (2) (iv) (a) of this section which may be incurred by the agency in responding to the request. If the anticipated charges exceed the maximum limit stated by the person making the request or if the request contains no assumption of financial liability for charges, the person shall be notified and afforded an opportunity to assume financial liability. The request for records shall not be deemed received for purposes of the applicable time limit for response until a written assumption of financial liability is received. When the anticipated charges exceed \$50, the person making the request, upon notification, shall deposit 50 percent of the anticipated charges with the agency. The request shall not be deemed received for purposes of the applicable time limit for response until such deposit has been

(c) Charges may be imposed even though the search discloses no records responsive to the request, or none not exempt from disclosure. The imposition of charges may be waived for the convenience of the agency, and will be reduced or waived where the agency determines that furnishing the information can be considered as primarily benefiting the general public. The agency may, by agreement with the person making the request, make arrangements with commercial firms for required services to be charged directly to the requester.

charged directly to the requester.

(v) "Working days," as used in this subsection, means calendar days excepting Saturdays, Sundays, and legal holidays.

(d) Subject to the provisions of §§ 102.31(c) and 102.66(c), all files, documents, reports, memoranda, and records of the agency, falling within the exemptions specified in 5 U.S.C. section 552(b), shall not be made available for inspection or copying, unless specifically permitted by the Board, its chairman, or its general counsel.

2. Section 102.118(a) is revised to read as follows:

§ 102.118 Board employees prohibited from producing files, records, etc., pursuant to subpoena ad testificandum or subpoena duces tecum; prohibited from testifying in regard thereto; production of witnesses' statements after direct testimony.

(a) Except as provided in § 102.117 of this part respecting requests cognizable under the Freedom of Information Act, no regional director, field examiner, administrative law judge, attorney, specially designated agent, general counsel, member of the Board, or other officer or employee of the Board shall produce or present any files, documents, re-

ports, memoranda, or records of the Board or of the general counsel, whether in response to a subpoena duces tecum or otherwise, without the written consent of the Board or the chairman of the Board if the document is in Washington, D.C., and in control of the Board; or of the general counsel if the document is in a regional office of the agency or is in Washington, D.C., and in the control of the general counsel. Nor shall any such person testify in behalf of any party to any cause pending in any court or before the Board, or any other board, commission, or other administrative agency of the United States, or of any State, territory, or the District of Columbia, or any subdivisions thereof, with respect to any information, facts, or other matter coming to his knowledge in his official capacity or with respect to the contents of any files, documents, reports, memoranda, or records of the Board or the general counsel, whether in answer to a subpoena or otherwise, without the written consent of the Board or the chairman of the Board if the person is in Washington, D.C., and subject to the supervision or control of the Board; or of the general counsel if the person is in a regional office of the agency or is in Washington, D.C., and subject to the supervision or control of the general counsel.

A request that such consent be granted shall be in writing and shall identify the documents to be produced, or the person whose testimony is desired, the nature of the pending proceeding, and the purpose to be served by the production of the document or the testimony of the official. Whenever any subpoena ad testificandum or subpoena duces tecum, the purpose of which is to adduce testimony or require the production of records as described hereinabove, shall have been served on any such person or other officer or employee of the Board, he will, unless otherwise expressly directed by the Board or the chairman of the Board or the general counsel, as the case may be, move pursuant to the applicable procedure, whether by petition to revoke, motion to quash, or otherwise, to have such subpoena invalidated on the ground that the evidence sought is privileged against disclosure by this

These amendments shall become effective on February 19, 1975.

Dated, Washington, D.C., February 14, 1975.

By direction of the Board.

John C. Truesdale, Executive Secretary.

[FR Doc.75-4624 Filed 2-18-75;8:45 am]

### Title 30—Mineral Resources

CHAPTER V—INTERIM COMPLIANCE PANEL (COAL MINE HEALTH AND SAFETY)

SUBCHAPTER A-COAL MINE HEALTH

# PART 506—FREEDOM OF INFORMATION

The Interim Compliance Panel established by section 5 of the Federal Coal Mine Health and Safety Act of 1969 (83

Stat. 744; 30 U.S.C. 804) issues regulations implementing the provisions of Pub. L. 93-502 (88 Stat. 1561) which amends the Freedom of Information Act, 5 U.S.C. 552.

Written comments on the fee schedule or objections may be submitted to the Interim Compliance Panel, Room 800, 1730 K Street NW., Washington, D.C. 20006, no later than March 21, 1975. If in the opinion of the Panel such com-ments are applicable, these regulations will be amended.

Part 506 of title 30 CFR, Subchapter A Coal Mine Health, as set forth below is herewith promulgated and shall become effective February 19, 1975.

506.1 Purpose.

506.2 General.

506.3 Responding to requests.

Certification of information. 506.4

506.5 Denial of requests.

506.6 Appeals.

506.7 Schedule of fees.

506.8 Annual report.

AUTHORITY: Sec. 508, Pub. L. 91-173, 83 Stat. 803, (30 U.S.C. 957).

### § 506.1 Purpose.

The provisions of this Part provide procedures to implement the Freedom of Information Act, 5 U.S.C. 552 as amended by Pub. L. 93-502 (88 Stat. 1561).

## § 506.2 General.

(a) Except for material specifically excluded by § 552(b) of Title 5, United States Code or other applicable statute, the Interim Compliance Panel will disclose all records and proceedings pertaining to the issuance or denial of permits for noncompliance with the interim mandatory health standards and the electric face equipment standards for underground coal mines prescribed by the Federal Coal Mine Health and Safety Act of 1969 (83 Stat. 742; 30 U.S.C. 801).

(b) Requests for information should be sent to the Interim Compliance Panel, Room 800, 1730 K Street NW, Washington, D.C. 20006.

## § 506.3 Responding to requests.

(a) Requests for information will be acted upon within ten working days after the receipt of such requests.

(b) If an unusual circumstance under section 552(a)(6)(B), United States Code, delays a decision concerning disclosures of information, the person making the request will be informed of the delay within ten working days. In no case will the decision be delayed more than 20 working days from the receipt of the initial request.

## § 506.4 Certification of information.

(a) The Chairman and Vice Chairman are the certifying officers of the Interim Compliance Panel.

(b) The Chairman or Vice Chairman may authenticate records or written instruments of the Interim Compliance Panel and may attest to the genuineness and correctness of copies of Interim

Compliance Panel records by his signature and by affixing the seal of the Interim Compliance Panel to such documents or records.

## § 506.5 Denial of requests.

(a) Denial of requests for privileged information shall be the responsibility of the Chairman or Vice Chairman.

(b) Letters denying requests for information will:

(1) Provide the person making the request with the reason for the denial;

(2) Inform the person making the request of the right to appeal the denial within 30 days: and

(3) Give the name of the official to whom the appeal may be sent.

## § 506.6 Appeals.

(a) The Interim Compliance Panel, or a quorum thereof, is the appellate authority for all denials of requests for in-formation; however, the officer who denied the initial request for information is excluded from considering the merits of the appeal.

(b) The person filing the appeal will be informed of the decision on the appeal within 20 working days after its receipt. If the denial is upheld, the person who filed the appeal will be notified concerning the provisions for judicial review of that determination.

### § 506.7 Schedule of fees.

(a) The fee schedule for processing or copying of records or proceedings is:

(1) Reproduction, duplication, copying of written records-10 cents per page; however, no charge will be made where the total amount does not exceed

(2) Certification or authentication of records-\$3 per certification or authentication.

(3) Transcription or duplication of

audio tapes-direct cost.

(4) Special handling overtime search for records, special delivery, certified, or airmail postage or other non-routine costs will be charged on a direct cost basis.

(b) The Chairman or Vice Chairman may waive all or part of any fee provided for in this section whenever he determines that waiver or reduction of fee is in the public interest because furnishing the information can be considered as primarily benefitting the general public. § 506.8 Annual report.

On or before March 1 of each calendar year the Interim Compliance Panel will submit a report covering the preceding calendar year to the Speaker of the House of Representatives and President of the Senate for referral to the appropriate committees of the Congress. The report shall include:

(a) The number of determinations made not to comply with requests for information and the reasons for each such

determination:

(b) The number of appeals filed, the result of such appeals, and the reason for the action upon each appeal that results in a denial of information;

(c) The name and title or position of each person responsible for the denial of information requested and the number of instances of participation for each:

(d) The results of each proceeding conducted pursuant to section 552(a) (4) (F), United States Code, including a report of the disciplinary action taken against the officer or employee who was primarily responsible for improperly withholding information or an explanation of why disciplinary action was not taken:

(e) A copy of every rule made in implementation of section 552 of Title 5, United

States Code;

(f) A copy of the fee schedule and the total amount of fees collected for making information available pursuant to this regulation; and

(g) Any other information which indicates efforts to administer fully section 552 of Title 5, United States Code.

Dated: February 13, 1975.

GEORGE A. HORNBECK, Chairman. Interim Compliance Panel.

[FR Doc.75-4402 Filed 2-18-75;8:45 am]

### Title 32—National Defense

### CHAPTER I-OFFICE OF THE SECRETARY OF DEFENSE

#### SUBCHAPTER P-RECORDS

## PART 292—AVAILABILITY TO THE PUBLIC OF DEFENSE INTELLIGENCE AGENCY (DIA) INFORMATION

The Director, Defense Intelligence Agency has approved the following revision of Part 292 as shown below in order to reflect the new requirements as contained in Public Law 93-502 which amended the Freedom of Information Act (5 U.S.C. 552).

Sec

292.1 Purpose.

292.2 Applicability.

292.3 Basic policy

292.4 Specific policy How the public submits requests for

records.

Schedule of fees. 292.6

Material that may be withheld from release. Filing an appeal for refusal to make 292.8

records available. 292.9 Responsibilities.

Uniform agency fees for search and duplication under the Freedom of 292.10 Information Act amendments (P.L. 93-502).

AUTHORITY: 5 U.S.C. 301, 552, as amended by the Act of 21 November 1974 (P.L. 93-502).

## § 292.1 Purpose.

This part implements the "Freedom of Information Act," (FOIA) 5 U.S.C. 552, as amended by Pub. L. 93-502, within the DIA and outline policy governing release of records to the public.

## § 292.2 Applicability.

The provisions of this part apply to all elements of the DIA and governs the release of records of these elements. This regulation is effective as of February 19, 1975.

## § 292.3 Basic policy.

(a) Upon receipt of a written request, the DIA will release to the public records concerning its operations and activities which are rightfully public information. Generally, information, other than that exempted in § 292.7, will be provided to the public. The following policy will be followed in the conduct of this program:

(1) The provisions of the Freedom of Information Act '(5 U.S.C. 552 as amended), as implemented by Part 286 of this chapter and this part, will be supported in both letter and spirit.

(2) Requested records will be withheld only when a significant and legitimate governmental purpose is served by withholding them. Records which require protection against unauthorized release in the interest of the national defense or foreign relations of the United States will not be provided.

(3) Requests from members of Congress will be governed by DoD Directive 5400.4, from the General Accounting Office by DoD Directive 7650.1, and from other agencies and courts by Part 286 of this chapter.

(4) Records will not be withheld solely because their release might result in criticism of DoD or this Agency.

(5) The applicability of the Freedom of Information Act depends on the existence of an "identifiable record" (5 U.S.C. 552(a) (3)). Accordingly, if DIA has no record containing information requested by a member of the public, it is under no obligation to compile information to create such a record.

(6) The mission of DIA does not encompass regulatory or decisionmaking matters in the sense of a public use agency, therefore, extensive reading room material for the general public is not available, however, unclassified DIA regulations and related material have been placed in the joint reading room managed by DoD Public Affairs.

(b) This basic policy is subject to the exemptions recognized in 5 U.S.C. 552(b) and discussed in § 292.7. Even where release denial is authorized by 5 U.S.C. 552(b) and § 292.7, requested records will be provided if no significant and legitimate governmental purpose is served by withholding them.

## § 292.4 Specific policy.

(a) In determining whether documentary material qualifies as a "record," 44 U.S.C., 3301 will be used as a guide. This statute, which defines the word "records" for record disposal purposes, provides that:

[It] includes all books, papers, maps, photographs, or other documentary materials, regardless of physical form or characteristics, made or received by any agency of the United States Government under Federal Law or in connection with the transaction of public business and preserved or appropriate for preservation by that agency or its legitimate successor as evidence of the

organization, functions, policies, decisions, procedures, operations, or other activities of the Government or because of the informational value of data contained therein.

 Records are not limited to permanent or historical documents but include contemporaneous documents as well.

(2) The term "records" does not include objects or articles such as structures, furniture, paintings, sculpture, three-dimensional models, vehicles, equipment, etc., whatever their historical value or value as "evidence."

(3) Records include data stored in computers.

(4) Formulae, designs, drawings, research data, computer programs, technical data packages, and so forth are not considered "records" within the Congressional intent of 5 U.S.C. 552 as amended by Pub. L. 93-502. Because of development costs, utilization, or value, these items are considered property, not preserved for informational value nor as evidence of agency functions, but as exploitable resources to be utilized in the best interest of all the public. Requests for copies of such materal shall be evaluated in accordance with policies expressly directed to the appropriate dissemination or use of such property. Requests to inspect such material to determine its content for informational purposes shall normally be granted, unless inspection is inconsistent with the obligation to protect the property value of the material, as, for example, may be true for certain formulae.

(5) The term "For Official Use Only" will not prevent a record from being released. Use of this designation is for internal convenience only, and each record containing this designation should be examined for release on its own merit.

(b) For a record to be considered "identifiable," it must exist at the time of the request.

(c) Requesters must be reasonably specific in identifying records. DIA will make a reasonable effort to locate any records that are requested.

(d) Requests for identifiable records may be denied only when the official designated in \$ 292.9 determines that such denial is authorized by this regulation.

(e) Pursuant to 5 U.S.C. 552(a) (3), requesters will be charged reasonable costs incurred by DIA in searching and copying records. Charges will be in accordance with § 292.6.

(f) In appropriate circumstances, requests requiring special attention will be referred, by the Central Reference Division Library, DIA, to the Assistant Secretary of Defense (Public Affairs) (ASD (PA)) for review and evaluation.

(g) Initial availability, releasability and cost determinations will normally be made within 10 working days of the date on which a written request for an identifiable record is received by the DIA Secretariat (FOIA). If, due to unusual circumstance, additional time is needed, a written notification of the delay will be forwarded the requester within the 10 working day period. This notification will briefly explain the circumstances for the

delay and indicate the anticipated date for a substantiative response. The period of delay may not exceed 10 additional working days.

§ 292.5 How the public submits requests for records.

(a) Requests to obtain copies of records must be made in writing. The request should contain at least the following information:

(1) Reasonable identification of the desired record including (if known) title or description, date and the issuing office.

(2) With respect to matters of official record concerning civilian or military personnel, the first name, middle name or initial, surname, date of birth, and social security number of the individual concerned if known.

(b) Persons desiring records should direct inquiry to:

Defense Intelligence Agency ATTN: SC (FOIA) Washington, D.C. 20301

### § 292.6 Schedule of fees.

Specific fees apply with respect to services rendered to the public.

(a) In general, charges may be waived when:

(1) The recipient of the benefits is engaged in a nonprofit activity designed for public safety, health or welfare.

(2) Payment of the full costs or fee by state, local government or nonprofit group would not be in the interest of the program.

(3) The incremental cost of collecting the fees would be an unduly large part of the receipts from the activity.

(b) Normally, collection of charges and fees will be made in advance of rendering the service. In some instances, it may be more practical to collect charges and fees at the time of conveying the service or property to the recipient, but only in those instances where the request specifically states that whatever cost is involved will be acceptable or acceptable up to a specified limit that covers anticipated costs. In the absence of an agreement to pay required anticipated costs, the time for responding to a request begins on resolution of willingness to pay.

(c) The schedule of fees chargeable is contained in § 292.10.

(d) Fees must be paid in full prior to search and issuance of requested copies. If uncertainty as to the existence of a record, or as to the number of sheets to be copied or certified, precludes remitting the exact fee chargeable with the request, DIA will inform the requester of the exact amount required.

(e) Remittances will be by personal check or bank draft on a bank in the United States or by U.S. postal money order. Remittances will be made payable to the "Treasurer of the United States" and forwarded to the address listed in § 292.5.

## § 292.7 Material that may be withheld from release.

(a) Detailed guidelines for withholding information under exemptions con-

<sup>&</sup>lt;sup>1</sup>Filed as part of original. Extra copies available from the U.S. Naval Publications and Forms Center, Attention Code 300, 5801 Tabor Avenue, Philadelphia, Pa. 19120.

tained in 5 U.S.C. 552 are published in 32 CFR 286.8, "Exemptions."

(b) For example, DIA is not required to release records that fall within the

categories listed below:

(1) Information requiring protection in the interest of national defense or foreign policy and authorized under criteria established by Executive Order 11652 and DoD 5200.1-R2 to be kept SECRET.

(2) Rules, regulations, orders, manuals, directives, and instructions relating to the internal personnel rules or internal practices of DIA and those providing management guidance to the Military Departments and the Unified and

Specified Commands.

(3) Information authorized or required by statute to be withheld from the public. The authorization or requirement may be found in the statute itself or in Executive orders or regulations authorized by, or in implementation of, the statute. Examples include:

(i) Documentary material referred to in 18 U.S.C. 1905-Trade and financial information provided in confidence by businesses (see subparagraph (4) of this

paragraph)

(ii) 35 U.S.C. 181-88-Records containing information relating to inventions that are the subject of patent applications for which Patent Secrecy orders have been issued.

(4) Personnel and medical files and similar files the disclosure of which constitute a clearly unwarranted invasion of

personal privacy.
(5) DIA will also withhold information which has been received from anyone, including an individual, a foreign nation, an international organization, a State or local government, a corporation, or any other organization, with the understanding, expressed or implied, that the information will be retained on a privileged or confidential basis under criteria contained in Executive Order 11652 and DoD 5200.1-R.3

(6) Exempt portions of a record may be deleted and the remaining reasonably segregable portions of the record released to the requester when the meaning of these portions is not distorted and it can be reasonably assumed that a skillful and knowledgeable person could not reconstruct the exempt information.

# § 292.8 Filing an appeal for refusal to make records available.

(a) A requester may appeal an initial decision to withhold a record. Appeals should be addressed to:

Director Defense Intelligence Agency ATTN: SC (FOIA) Washington, D.C. 20301

(b) Final determination on appeals shall normally be made within 20 working days of the receipt of the appeal at the above address. If additional time is needed to decide the appeal because of unusual circumstances, the final determination may be delayed for the number of working days, not to exceed

10, which were not utilized as additional time for responding to the initial request.

(c) When an appeal is denied, the requester will be apprised of the following:

(1) Applicable exemptions and the significant and legitimate governmental purpose served by the denial.

(2) Name, and title of position of the official responsible for the denial and of the provision for judicial review of the denial.

(3) When the denial is based upon security classification, the requester shall be advised of his optional right to seek declassification of the record by the Interdepartmental Classification Review Committee established pursuant to Executive Order 11652, March 8, 1972, in lieu of immediate judicial review.

### § 292.9 Responsibilities.

When a request for release of material is received the following will apply:

(a) The Secretariat (SC):

(1) Receives requests and assigns tasking.

(2) Maintains appropriate suspenses and authorizes all extensions of response time.

(3) Acts as the responsible official for all initial denials of access to the public.

(b) The Reference Library Branch (DS-4A), Central Reference Division, Directorate for Support:

(1) Acts as the responsible operating office for all Agency actions related to the Freedom of Information Act.

(2) Drafts and transmits responses

(i) The release of records and/or information.

(ii) Obtaining supplemental information from the requester.

(iii) Informing the requester of any fees required.

(iv) The transfer to another component or agency of the initial request.

(3) Fulfills the annual reporting requirement and maintains appropriate records.

(4) Refers unusual or precedent-setting requirements to ASD (PA).

(5) Drafts for SC:

(i) Notification of an extension of response time.

(ii) The denial of access to the record.

(c) All DIA Elements:

(1) When identified by DS-4A as the Office of Primary Interest (OPI):

(i) Review records for possible public release within the time constraints assigned.

(ii) Prepare a documented response in all cases of nonrelease.

(2) All employees are required to read this Part to insure familiarity with the requirements of the Freedom of Information Act as implemented.

(d) General Counsel (GC);

(1) Insures uniformity in the Freedom of Information Act legal positions within DIA and with DoD.

(2) Secures coordination with the DoD General Counsel on denials of public requests.

(3) Acts as the focal point in all judicial actions.

(4) Reviews all initial and final denials.

(e) Chief of Staff (CS/DP):

(1) Exercises overall staff supervision of the Freedom of Information Act activities of the Agency.

(2) Acts as the responsible official for all denials of appeals.

§ 292.10 Uniform agency fees for search and duplication under the Freedom of Information Act amendments (Pub. L. 93-502).

General fees:	
Minimum fee, per request plus	2, 00
Forms, per copy	. 05
Publications, per printed page	. 01
Microfiche, per fiche	. 06
Reports, per printed page	. 05
(Examples: Cost of 20 forms, \$3.00;	
cost of a printed publication with	
100 pages, \$3.00; cost of a micro-	
fiche publication consisting of 10	
fiche, \$2.60.)	
Office copy reproduction (when shelf	
stock is not available):	
Minimum charge up to six repro-	
duced pages	2.00
Minimum charge, first fiche	5.00
Each additional page	. 05
Each additional fiche	. 10
Other issuances:	
Minimum charge up to six pages	2.00
Each additional page	. 05
Search:	
Clerical search, per hour	5. 50
Minimum charge	2.75
Professional search (includes com-	
puter programmer time and re-	
view to determine whether a rec-	
ord comes within the scope of a	
request), per hour	11.00
Minimum charge	5.50
Computer service charges will be	
based on actual computer config-	
uration used and be based on di-	
rect costs only of the Central Pro-	,
cessing Unit plus Input/Output	
Devices plus Memory Capacity.	
Maurice W. Rochi	e.
Director Correspondence	

Directives OASD (Comptrol-

FEBRUARY 12, 1975.

[FR Doc.75-4298 Filed 2-18-75;8:45 am]

# CHAPTER XIX—CENTRAL INTELLIGENCE AGENCY

## PART 1900—PUBLIC ACCESS TO DOCU-MENTS AND RECORDS AND DECLAS-SIFICATION REQUESTS

Part 1900 of Title 32 of the Code of Federal Regulations is revised to read as follows:

GENERAL.

Sec.						
1900.1	Purpose and Authority.					
1900.3	Definitions.					
1900.5	Organization: mittals.	Requests	and	Sub-		
	REQUESTING	RECORDS				
1000 11			-			

1900.11 Free lom of Inform cations: Requirements as Form. Option to Request Records.

Pre-Request Option: Estimates of 1900.23 Charges.

1900.25 Fees for Records Services,

Filed as part of original.

## PROCESSING FREEDOM OF INFORMATION

1900.31 Screening Communications. Processing Expressions of Interest. 1900.35 Processing Requests for Records.

#### ACTIONS ON REQUESTS

1900.41 Searching for Requested Records. 1900.43 Reviewing Records. 1900.45 Expeditious Action: Extension of

1900.47 Allocation of Manpower and Resources: Agreed Extension of

1900.49 Notification and Payment: Furnishing Records.

## APPEALS

Appeal to CIA Information Review Committee. 1900.51

1900.53 Appeal to Interagency Classification Review Committee.

#### MISCELLANEOUS

1900.61 Access for Historical Research. 1900.63 Suggestions and Complaints.

AUTHORITY: National Security Act of 1947, the Central Intelligence Agency Act of 1949, the Freedom of Information Act (5 U.S.C. 552), and Executive Order 11652.

#### GENERAL

## § 1900.1 Purpose and authority.

This part is issued under the authority of, and in order to implement, section 102 of the National Security Act of 1947, as amended (50 U.S.C. 403), the Central Intelligence Agency Act of 1949, as amended (50 U.S.C. 403a et seq.), Executive Order 11652, as amended (3 CFR Revised as of January 1, 1974, p. 339) and the Freedom of Information Act, as amended (5 U.S.C. 552). It prescribes procedures for:

(a) Requesting records pursuant to the Freedom of Information Act;

(b) Requesting the declassification of documents pursuant to Executive Order 11652;

(c) Appealing any denial or refusal of any such request to an appeal authority with the Central Intelligence Agency, to the Interagency Classification Review Committee, in appropriate cases, and to the courts;

(d) The prompt and expeditious processing of such requests and appeals; and

(e) Requesting estimates and advice prior to actually requesting records, thus affording protection against unanticipated fees.

This part is also designed to assist Central Intelligence Agency management at all appropriate echelons, to allocate resources to perform the functions, duties and responsibilities of the Central Intelligence Agency prescribed by and pursuant to law, including in particular those situations where it is deemed necessary to choose among conflicting requirements, duties and responsibilities.

### § 1900.3 Definitions.

For the purpose of this part, the following terms have the meanings indi-

(a) "Agency" includes any executive department, military department or other establishment or entity included in the definition of agency in subsection

552(e) of Title 5 of the United States

Code;
(b) "Coordinator" means the Central Intelligence Agency Freedom of Information Coordinator:

(c) "Expression of interest" means a written communication submitted by a potential requester pursuant to § 1900.23 to indicate an interest in requesting records;

(d) "Freedom of Information Act" means section 552 of Title 5 of the United States Code, as amended:

(e) "National Security Council Directive of May 1972" means the directive entitled "National Security Council Directive Governing the Classification, Downgrading, Declassification and Safeguarding of National Security Information" approved on May 17, 1972 and published at page 227 of Title 3A (The President, Appendix (1972 Compilation)) of the Code of Federal Regulations;

(f) "Potential requester" means a person, organization or other entity who submits an expression of interest in ac-

cordance with § 1900.23;
(g) "Records," with reference to records of the Central Intelligence Agency, includes all papers, maps, photographs and other documentary materials regardless of physical form or characteristics made or received by the Central Intelligence Agency in pursuance of federal law or in connection with the transaction of public business and appropriate for preservation by the Central Intelligence Agency as evidence of the organization, functions, policies, decisions, procedures, operations or other activities of the Agency or because of the informational value of data contained therein. But the term does not include:

(1) Index, filing and museum documents made or acquired and preserved solely for reference, indexing, filing or

exhibition purposes;

(2) Routing and transmittal sheets and notes and filing instructions and notes which do not also include information, comment or statement of substance or policy:

(3) Books, magazines. newspapers. and similar publications and clippings and excerpts from any such publica-

tions:

(4) Documents and records prepared or originated by an agency other than the Central Intelligence Agency:

(5) Documents and records furnished by foreign governments or international organizations and held by the Central Intelligence Agency on the understanding that the information therein or the furnishing thereof be kept in confidence.

(h) "Records of interest" means records which are the subject of an expression of interest or of a request;

(i) "Work days" means calendar days other than Saturdays and Sundays and legal public holidays.

## § 1900.5 Organization: requests submittals.

The headquarters of the Central Intelligence Agency is located in Fairfax County, Va. Functions are channeled and determined by regular chain-of-

command procedures. Except as provided by this part, there are no formal or informal procedural requirements regarding public access to Agency records. Requests and other submittals may be addressed to the CIA Freedom of Information Coordinator, Central Intelligence Agency, Washington, D.C. 20505.

#### REQUESTING RECORDS

§ 1900.11 Freedom of information communications; requirements as to

(a) Any communication to the Central Intelligence Agency or to the Director of Central Intelligence under the Freedom of Information Act should be addressed

CIA Freedom of Information Coordinator, Central Intelligence Agency, Washington, D.C. 20505.

That address should appear on the envelope or other folder or package in which the communication is transmitted. It should also be included as the addressee of the letter or other communication or be clearly set forth in the text of the communication.

(b) Any request for records under the Freedom of Information Act (§ 1900.21), expression of interest in requesting records (§ 1900.23) or request for declassification of records under Executive Order 11652 shall be in writing and shall be addressed as prescribed by paragraph (a). The Coordinator may, but need not, waive the requirements as to address.

(c) The request or expression of interest shall reasonably describe the rec-

ords of interest.

(d) Any request or communication to an agency other than the Central Intelligence Agency which requests or concerns documents or records originated by the Central Intelligence Agency and is transferred by that agency to the Central Intelligence Agency shall not be a Freedom of Information Request to the Central Intelligence Agency. But the person or entity who submitted the request to that agency may submit a request for such records to the Central Intelligence Agency in accordance with the procedures prescribed by § 1900.11.

## § 1900.21 Option to request records.

Any person, organization or entity may submit a written request for records to the Coordinator, in accordance with the procedures prescribed by § 1900.11. An estimate of charges likely to be incurred may be obtained by requesting such an estimate as provided by § 1900.23.

## § 1900.23 Pre-request option: estimates of charges.

(a) In order to avoid being faced with unanticipated sizeable charges, interested persons and entities may defer the submission of requests for records and first submit a written request, in accordance with the procedures prescribed by § 1900.11, for an estimate of charges likely to be incurred if the records are requested.

(b) Notice is hereby given that a requester may be liable for the payment of search charges, in accordance with the fee schedule and provisions of § 1900.25, even if search for requested records locates no such records and even if some or all of requested records which are located are denied the requester under one or more exemptions of the Freedom of Information Act. In determining which of options §§ 1900.21 and 1900.23(a) to exercise, interested persons and entities are urged to take into consideration the fact of possible liability.

## § 1900.25 Fees for records services.

(a) Search and duplication fees shall be charged according to the schedule set out in paragraph (c) of this section for services rendered in responding to requests for Agency records under this part. Records shall be furnished without charge or at a reduced rate whenever the Coordinator determines that waiver or reduction of the charge is in the public interest because furnishing the information can be considered as primarily benefiting the general public. The Coordinator also may waive or reduce the charge whenever he determines that the interest of the government would be served thereby. Fees shall not be charged where they would amount, in the aggregate, for a request, or a series of related requests, to less than \$4.

(b) Where it is anticipated that the fees chargeable under this section will amount to more than \$25, and the requester has not indicated in advance his willingness to pay fees as high as are anticipated, the requester shall be promptly notified of the amount of the anticipated fee or of such portion thereof as can readily be estimated. In appropriate cases an advance deposit may be required. The notice or request for an advance deposit shall extend an offer to the requester to confer with the Coordinator in an attempt to revise the request in a manner which will reduce the fees and meet the needs of the requester. Dispatch of such a notice or request shall suspend the running of the period for response by the Agency until a reply is received from the requester.

(c) The schedule of fees for services performed in responding to requests for Agency records is established as follows:

(1) For each one quarter hour spent by clerical personnel in searching for a record, \$1;

(2) For each one quarter hour spent by professional personnel in searching for a record, \$2;

(3) For computer search, \$55;

(4) For copies of paper documents in sizes not larger than  $8\frac{1}{2} \times 14$  inches, \$0.10 per copy of each page; and

(5) For duplication of non-paper media or any document that cannot be reproduced on a standard office copier actual direct cost.

## PROCESSING FREEDOM OF INFORMATION COMMUNICATIONS

## § 1900.31 Screening communications.

 (a) If any Agency employee receives a written communication which the employee deems to be an apparent or intended communication under the Freedom of Information Act, he shall expedi-

tiously transmit the communication to the Coordinator and alert the Coordinator to the fact that the communication may be a communication under the Freedom of Information Act.

(b) Upon receipt of a communication in accordance with § 1900.11 or paragraph (a) of this section, the Coordinator shall promptly consult with such Agency components as he may deem appropriate and:

(1) Determine the nature of the communication—an intended expression of interest (§ 1900.21), an intended request

(§ 1900.23) or other; and

(2) If he determines the communication to be an intended expression of interest or intended request, he shall further determine whether it fails to qualify as an expression of interest or request only because it fails to reasonably describe the records of interest.

(c) The Coordinator thereupon shall take the appropriate one of the follow-

ing actions.

(1) If he determined that the communication was not an intended expression of interest or an intended request, he shall take such action with respect to the communication as he may deem appropriate.

(2) If he determined that the communication was an intended expression of interest or an intended request but failed to reasonably describe the records of interest, he shall so inform the originator of the communication promptly, in writing, and he may offer to assist the originator in revising and perfecting the description of the records of interest.

(3) The Coordinator shall determine whether any communication not acted on under subparagraph (1) or (2) is an expression of interest, or is a request made in accordance with published rules stating the procedures to be followed, as required by subsection (a)(3) of the Freedom of Information Act. The Coordinator's determination in this regard shall be based on and shall reflect the clear intent of the originator of the communication insofar as the Coordinator is able to determine that intent. When the originator's intent is not apparent to the Coordinator and when the Coordinator deems it desirable and feasible, he shall promptly communicate with the originator in order to ascertain the latter's intent.

(d) The Coordinator shall inform the requester, in writing, of his determination made under subparagraph (c) (3) and, in the case of a determination that the communication is a request, of the date of such determination. Such notification shall be given promptly and, in any case, within five work days of the date of such determination. The ten work days within which the Agency must determine whether to comply with a request, as provided by subsection (a) (6) (A) (i) of the Freedom of Information Act, shall begin as of the date of such determination.

(e) The Coordinator shall promptly process under the procedures prescribed by § 1900.33 those communications which he determines to be expressions of inter-

est. He shall promptly process under the procedures prescribed by § 1900.35 those communications which he determines to be requests.

## § 1900.33 Processing expressions of interest.

(a) Upon determining, in accordance with subparagraph (c)(3) of § 1900.31, that a communication is an expression of interest, and after promptly consulting with such Agency components as he may deem appropriate, the Coordinator, to the extent feasible, shall determine the search and duplication charges likely to be incurred by the potential requester if the potential requester ultimately requests such records. In determining such charges, the Coordinator shall take into account the nature and quantity of the work and services of people and computers and other equipment which may be required, and the applicable rates set out in the fee schedule prescribed by paragraph (c) of § 1900.25. If feasible at this stage, the Coordinator also shall determine whether to waive or reduce the fee in accordance with paragraph (a) of § 1900.25.

(b) The Coordinator thereupon shall advise the potential requester, in writing, of the likely search and duplication charges. He shall explain the bases and reasons for the charges and he shall make clear that the amounts indicated are estimates only, if such be the case, and, if there is a possibility that the charges to be incurred may be in larger amounts, he shall so inform the potential requester. If the amounts indicated are not estimates but are the amounts which in fact are to be charged if the potential requester makes a request, he shall inform the potential requester of that fact. In either event and if such be the case, he shall also inform the potential requester that search charges will be levied upon the requester even if no records fitting the description are located or if any or all records which do fit the description are denied the requester.

(c) When he deems it appropriate or when the potential requester so requests, the Coordinator shall consult with and advise the potential requester with the view to assisting the latter to determine whether and, if so, how to revise the description of the records of interest so as to cause or permit a reduction in the likely and actual search and duplication

charges.

(d) Upon receipt of such estimate and advice concerning likely charges, the

potential requester may:

(1) In accordance with § 1900.21, submit a request for records, either the records of interest indicated in his expression of interest or records encompassed in a less-inclusive description;

(2) Advise the Coordinator that he does not intend to request records; or

(3) Take no additional action.

(e) If, as a result of his consultations with the Coordinator or otherwise, the potential requester wants to request records additional to or other than those described in his expression of interest, he may submit an expression of interest

with respect to such records, in accordance with § 1900.23, or a request for such records, in accordance with § 1900.21.

#### § 1900.35 Processing requests for records.

Upon determining that a communication is a request for records, the Coordinator, after consulting with such Agency components as he may deem appropriate, shall promptly transmit a copy of the request to the component or components believed responsible for the records, if any exist, inform the components of the date of receipt of the request as determined by him pursuant to paragraph (c)(3) of § 1900.31, and alert the components to the action required of them by § 1900.41 through § 1900.47 with respect to the request.

## ACTIONS ON REQUESTS

## § 1900.41 Searching for requested records.

(a) Upon receipt of a copy of a request and an alert pursuant to \$ 1900.35, the components responsible for requested records (hereinafter the "responsible components"), shall, with such assistance as may be appropriate from the Coordinator and from such reference, indexing or filing components as may have reference, indexing or filing responsibilities with respect to any such records, undertake to locate the requested records.

(b) If no records described by the request are located, the responsible com-ponents shall so inform the Coordinator who shall promptly so inform the re-quester, in writing. The Coordinator also shall determine the charges, if any, for which the requester shall be liable in accordance with the fee schedule and provisions of § 1900.25. He shall inform the requester of the amount charged, explain the basis of computation and request prompt payment thereof.

## § 1900.43 Reviewing records.

(a) The responsible components shall review any located records in accordance with the provisions of the Freedom of Information Act and Executive Order 11652, and on the basis of other applicable law, regulations and policy, and determine which, if any, requested records, or reasonably segregable portions of records, are to be furnished the requester and which are to be denied or withheld. Any decision to furnish or to deny or withhold requested records shall be made only by employees and officials to whom authority to make such decisions has been duly delegated.

(b) In the event records require review by more than one Agency component or by more than one agency, the Coordinator or the responsible component, as may be appropirate, shall expeditiously coordinate such review.

#### § 1900.45 Expeditious action: extension of time.

Whenever feasible under the standards prescribed by § 1900.47, the search and review functions prescribed by §§ 1900.41 and 1900.43 and notice to the requester of the Agency action on the request, as 1y, the Coordinator shall promptly pre-

prescribed by paragraph (a) of § 1900.49, shall be completed within ten days of the date of Agency receipt of the request as determined by the Coordinator pursuant to subparagraph (c)(3) or § 1900.31. Whenever the Coordinator determines that "unusual circumstances," as defined by subsection (a) (6) (B) of the Freedom of Information Act, exist, he may, by written notice to the requester, authorize an additional period for completion of Agency action, but no such extension shall be for more than ten work days. His notice shall also set forth the reasons for the extension.

#### § 1900.47 Allocation of manpower and resources: agreed extension of time.

(a) Agency components shall devote such manpower and other resources to searching for, locating and reviewing records in accordance with §§ 1900.41 and 1900.43 as may be appropriate and expedient in the circumstances, taking into account:

(1) The manpower and resources avail-

able for those purposes:

(2) The right of the requester to resort to litigation if the Agency decision on the request is not made within ten work days; and

(3) All functions, duties and responsibilities assigned to those components

by, or pursuant to, law.

(b) The responsible components shall consult with the Coordinator with regard to the need to allocate resources and establish priorities, and the latter with the requester, as may be appropriate, in order to accomplish such arrangements and agreements with the requester as may be acceptable to the requester concerning the Agency's efforts and ability to act on his request expeditiously. In particular, when the Coordinator deems it feasible and of possible benefit to the requester, the public or the Agency, he shall inform the requester that more thorough or extensive search or review, or both, could be accomplished, which might be of benefit to the requester, if additional time were to be available. When appropriate in such cases, the Coordinator shall also advise the requester of the effect on charges and fees such additional search might cause. Any extensions arranged or agreed to under this section may be in extension under addition to any § 1900.45.

## § 1900.49 Notification and payment: furnishing records.

(a) The Coordinator shall promptly inform the requester, in writing, which of the requested records, or portions thereof, if any, are to be furnished the requester and those, if any, which are denied, as determined pursuant to paragraph (a) of § 1900.43. With respect to the latter, he shall also explain the reasons for the denial and he shall furnish the names and titles or positions of the persons responsible for the decision to deny access.

(b) Upon receipt of payment of all fees and charges, or upon the completion of arrangement satisfactory to the Coordinator that payment will be made prompt-

pare copies of the records, or portions of records, which are to be made available and transmit them to the requester. The Coordinator shall do likewise with respect to any records or portions of records made available to a requester by the action of the Central Intelligence Agency Information Review Committee under subparagraph (i) or (ii) of paragraph (e) of § 1900.51.

#### APPEALS

# § 1900.51 Appeal to CIA Information Review Committee.

(a) Establishment of committee. The Central Intelligence Agency Information Review Committee is established hereby, pursuant to the Freedom of Information Act and section 7(B)(2) of Executive Order 11652. The Committee shall be comprised of the Deputy Director for Administration, the Deputy Director for Intelligence, the Deputy Director for Operations, the Deputy Director for Science and Technology, the Deputy to the Director of Central Intelligence for the Intelligence Community and the Deputy to the Director of Central Intelligence for National Intelligence Officers. The Director of Central Intelligence shall appoint a chairman. The Committee, by majority vote, may delegate to one or more of its members the authority to act on any appeal or appeals under this section, and may authorize the chairman to delegate such authority.

(b) Right of appeal: notice. Whenever any requested record or any portion thereof is denied a requester, the requester may appeal the denial to the Committee and the Coordinator shall inform the requester of this right, in writing. Also, any requester for the declassification of classified documents under Executive Order 11652 who is not informed of Agency action thereon within sixty days of his request may request the Committee to declassify the documents and make them available.

(c) Appeal procedures. Any such appeal or request to the Committee shall be in writing, addressed to the CIA Freedom of Information Coordinator. The appeal or request may present such information, data and argument in support thereof as the requester may desire. The Committee shall not permit a requester to appear before the Committee or to make an oral presentation.

(d) Time for appeal: expiration of right of appeal. An appeal shall be submitted within thirty days of the date of receipt of notification of the right to appeal and the right of appeal shall cease as of the expiration of that period. But the Committee, for good cause shown, may permit an additional thirty days for the submission of an appeal.

(e) Committee action on appeals and requests. (1) The Committee shall promptly consider any appeal, together with any submissions in support thereof, and shall grant or deny the appeal or take such other action thereon as it may deem appropriate. The Committee's review, decision and action shall be based on and shall be in conformance with the Freedom of Information Act, Executive Order 11652 and other applicable law, directives, regulations and policy.

(2) The Committee shall promptly consider any requests for declassification under paragraph (a) of this section and shall declassify any such records or reasonably segregable portions of such records as it deems appropriate in accordance with Executive Order 11652.

(3) Committee action on appeals shall be completed within twenty work days of receipt of the appeal, except that the Committee may, in accordance with the provisions of § 1900.45, avail itself of an additional period of time for completion of its work on the appeal. But no such extension shall be available with respect to an appeal of a denial of a request which was the subject of an extension of time for Agency action by the Coordinator under that paragraph. In the event the Committee is unable to complete its review of an appeal within the time prescribed by the two preceding sentences it may, by agreement with the requester, extend the period for completion of such review.

(4) The Committee shall promptly inform the requester of its decisions and, with respect to any decision to withhold or deny records, it shall furnish the names and titles or positions of the persons responsible for the decision. If any record or portion thereof is denied the requester by the Committee's action, the Committee shall also inform the requester of the provision for judicial review of that determination under subsection (a) (4) of the Freedom of Information Act.

## § 1900.53 Appeal to Interagency Classification Review Committee.

In the case of any refusal to declassify records which are at least ten years old, the Committee also shall inform the requester of his right to appeal that denial to the Interagency Classification Review Committee established pursuant to section 7(A) of Executive Order 11652, as provided by section IX(D) of the National Security Council Directive of May 1972. In that event, the Committee shall also inform the requester that appeal to the Interagency Classification Review Committee shall not waive his right to judicial review as provided by subsection (a) (4) of the Act.

## MISCELLANEOUS

## § 1900.61 Access for historical research.

(a) Any person engaged in a historical research project may submit a request, in writing, to the Coordinator to be given access to information classified pursuant to an Executive order for purposes of that research project may submit a request, the nature, purpose and scope of the research project.

(b) The Coordinator may authorize access, under such conditions and at such time and place as he may deem feasible. But the Coordinator shall authorize access only with respect to documents and records prepared or originated not less than ten years prior to the date of such request and only upon the prior written

approval by the Agency Director of Security of a current security clearance of the requester and of persons associated with him in the project, in accordance with Executive Order 10450, and upon the Coordinator's further determination that:

(1) A serious professional or scholarly research project is contemplated;

(2) Such access is clearly consistent with the interests of national security;

(3) Appropriate steps have been taken to assure that classified information will not be published or otherwise compromised:

(4) The information requested is reasonably accessible and can be located and compiled with a reasonable amount of effort;

(5) The historical researcher agrees to safeguard the information in a manner consistent with Executive Order 11652 and the National Security Council Directive of May 1972; and

(6) The historical researcher agrees to authorize a prior review of his notes and manuscript by the Agency for the sole purpose of determining that no classified information is contained therein.

(c) An authorization shall be valid for the period required for the research project, as the Coordinator may determine, but in no event for more than two years. But upon renewed request in accordance with paragraph (a) of this section, authorization may be renewed in accordance with paragraph (b) and this para-

(d) The Coordinator shall cancel any authorization whenever the Director of Security cancels the security clearance of the requester or of any person associated with the requester in the research project or whenever the Coordinator determines that continued access would not be in compliance with one or more of the requirements of paragraph (b) of this section.

## § 1900.63 Suggestions and complaints.

Any person may direct any suggestion or complaint with respect to the Agency administration of Executive Order 11652 to the CIA Information Review Committee. The Committee shall consider such suggestions and complaints and shall take such action thereon as it may deem feasible and appropriate.

This regulation shall become effective on February 19, 1975.

JOHN F. BLAKE,
Deputy Director for Administration, Central Intelligence
Agency.

FEBRUARY 11, 1975.

[FR Doc.75-4254 Filed 2-18-75;8:45 am]

# CHAPTER XVI—SELECTIVE SERVICE SYSTEM

# PART 1608—PUBLIC INFORMATION Freedom of Information Act Regulations

Whereas, on January 14, 1975, the Director of Selective Service published a

Notice of Proposed Amendments to Selective Service Regulations (40 FR 2593) of January 14, 1975; and

Whereas more than thirty days have elapsed subsequent to such publication during which period comments from the public have been received and considered. The proposed amendments are being made effective without change in substance.

The revision of Part 1608 would eliminate the authority of the Director of Selective Service and the State Director of Selective Service to disclose information in a registrant's file to an official of any state or subdivision thereof and the authority of the Director of Selective Service to disclose information in a registrant's file to any person other than the officials listed; the addresses of State Directors of Selective Service would be eliminated; a charge for the search of records would be specifically prohibited: "not clearly authorized" disclosures of information could be made only upon advice of the General Counsel of the Selective Service System; and final agency action on an appeal from a denial of information would be taken by the Director of Selective Service. The present provisions of Part 1608 that are not changed in substance are rearranged for the purposes of increased clarity and convenience in use.

Now therefore by virtue of the authority vested in me by the Military Selective Service Act, as amended (50 App. U.S.C. sections 451 et seq.) and § 1604.1 of Selective Service Regulations (32 CFR 1604.1), the Selective Service Regulations, constituting a portion of Chapter XVI of Title 32 of the Code of Federal Regulations, are hereby amended, effective February 19, 1975.

Part 1603, Public Information is revised to read as follows:

Sec.

1608.1 Public information policy.

1608.2 Definitions.

1608.3 General policy on disclosure of information.

1608.4 Availability and use of information information in registrants' files.

1608.5 Waiver of confidential nature of information in registrant's files.

1608.6 Subpoena of records.

-1608.7 Available information.

1608.8 Places where information may be obtained.

1608.9 Rules governing the obtaining of information.

1608.10 Identification of information requested.

1608.11 Fees for search of records and copies of documents.

1608.12 Request for information not authorized to be disclosed.

1608.13 Review of denials of requests for information.

1608.14 Demands of courts or other authorities for records or information protected by these regulations.

AUTHORITY: Military Selective Service Act, amended (50 App. U.S.C. 451 et seq.) 32 CFR 1604.1.

## § 1608.1 Public information policy.

The Selective Service System has a positive public information policy under which information is brought to the attention of the public. The Selective Service System brings to the public, through news releases, pamphlets, educational material for distribution to high schools, and other documents, information concerning important events, and the functions of the Selective Service System.

## § 1608.2 Definitions.

When used in this part, the following words shall have the meaning ascribed to them as follows:

(a) "Disclose" shall mean an oral or written statement concerning any such record or information.

(b) "Furnish" shall mean providing in substance or verbatim a copy of any such

record or information.

"Examine" shall mean a visual inspection and examination of any such record or information at the office of the local board or appeal board as the case may be.

#### § 1608.3 General policy on disclosure of information.

(a) It is the general policy of the Selective Service System to make information available to the public unless the disclosure thereof would constitute a clearly unwarranted invasion of personal privacy or is prohibited by law or Executive order or relates to internal memoranda, letters or other documents the disclosure of which would interfere with the functions of the Selective Service System.

(b) The records in a registrant's file and the information contained in such records shall be disclosed, furnished, or examined only in accord with the pro-

visions of this part.

- (c) Technical instructions pertaining to automatic data processing, memoranda, correspondence, opinions, data, staff studies, information received in confidence, and similar documentary material prepared for the purpose of internal communication within the Selective Service System or between the Selective Service System and other organizations or persons generally are not information available to the public.
- (d) Lists of registrants may be furnished only in accordance with written instructions from the Director of Selective Service.
- (e) The addresses of registrants are confidential information.

## § 1608.4 Availability and use of information in registrants' files.

- (a) Information contained in records in a registrant's file and records pertaining to a named registrant may be disclosed or furnished to, or examined by, the following persons:
- (1) The registrant, or any person having written authority dated and signed by the registrant: Provided, That whenever the time of the expiration of such

authority is not specified therein, no information shall be disclosed, furnished, or examined under that authority after the expiration of a period of 1 year from its date.

(2) The legal representative of a deceased or incompetent registrant.

(3) All personnel of the Selective Service System while engaged in carrying out the functions of the Selective Service System.

(4) A U.S. Attorney and his duly authorized representatives, including agents of the Federal Bureau of Investigation, whenever the registrant has been reported to the U.S. Attorney as a violator for prosecution for violating the Military Selective Service Act or the rules, regulations, or directions made pursuant thereto.

(5) Any other agency, official, or employee, or class or group of officials or employees of the United States upon written request in individual cases, but only when and to the extent specifically authorized in writing by the Director of

Selective Service.

(b) No information shall be disclosed or furnished to, or examined by, any person under the provisions of this section, until such person has been properly identified as entitled to obtain such information.

(c) Persons described in paragraph (a) (1) may be furnished a copy of the registrant's file only in accord with the provisions of § 1608.11(e) or upon payment of fees prescribed in § 1608.11(b) (1) of this part. Persons described in paragraph (a) (2) may be furnished a copy of a registrant's file only upon payment of fees prescribed in § 1608.11(b) (1) of this part.

### § 1608.5 Waiver of confidential nature of information on registrants files.

The making or filing by or on behalf of a registrant of a claim or action for damages against the Government or any person, based on acts in the performance of which the record of a registrant or any part thereof was compiled, or the institution of any action against the Government or any representative thereof by or on behalf of a registrant involving his classification, selection, or induction, shall constitute a waiver of the confidential nature of all selective service records of such registrant, and, in addition, all such records shall be produced in response to the subpoena or summons of the tribunal in which such claim or action is pending.

### § 1608.6 Subpoena of records

(a) In the prosecution of a registrant or any other person for a violation of the Military Selective Service Act, the Selective Service Regulations, any orders or directions made pursuant to such act or regulations, or for perjury, all records of the registrant shall be produced in response to the subpoena or summons of the court in which such production or proceeding is pending. Any officer or employee of the Selective Service System who produces the records of a registrant in court shall be considered the custo-

dian of such records for the purpose of this section.

(b) Except as provided in paragraph (a) of this section, no officer or employee of the Selective Service System shall produce a registrant's file, or any part thereof, or testify regarding any confidential information contained therein, in response to the subpoena or summons of any court without the consent, in writing, of the registrant concerned, or of the Director of Selective Service.

(c) Whenever, under the provisions of this section, a registrant's file, or any part thereof, is produced as evidence in the proceedings of any court, such file shall remain in the personal custody of an official of the Selective Service System, and permission of the court be asked, after tender of the original file, to substitute a copy of the file with the court.

## § 1608.7 Available information.

(a) Upon request, current documents specifically identified as being printed for free distribution to the general public will be furnished without charge. Each individual requesting such documents shall be entitled to only one copy of each document.

(b) Copies of Selective Service Regulations (32 CFR Chapter XVI) and the Registrants Processing Manual are offered for sale by the Superintendent of Documents, Government Printing Office,

Washington, D.C. 20402.

(c) The Registrants Processing Manual may be inspected at the office of any local board, the office of the State Director of Selective Service for any State or at the National Headquarters, Selective Service System.

(d) Each local board maintains Classification Record (SSS Form 102) which contains the name, selective service number, and the current and past classifications for each person registered with that board. Information in this record will be supplied upon request.

(e) Any compensated employee of the Selective Service System may disclose to the former employer of a registrant who is serving in or who has been discharged from the Armed Forces whether the registrant has or has not been discharged and, if discharged, the date thereof, upon reasonable proof that the registrant left a position in the employ of the person requesting such information in order to serve in the Armed Forces.

(f) The names, position titles, grades, salaries, and duty stations of employees of the Selective Service System are pub-

lic information.

(g) The names of local board members and the names and addresses of advisors to registrants will be posted in an area available to the public at each board office to which such personnel are assigned.

(h) Personal data concerning board members that relate to their legal qualifications for appointment and/or continuation in office are a matter of official record. Upon request, the executive secretary or clerk of a local board or appeal board will verify that a member of that board was legally qualified for appointment and for continuation in office without disclosing the personal data pertaining to such member without the member's consent.

#### § 1608.8 Places where information may be obtained.

(a) Requests for information concerning a registrant shall be addressed to the local board where he is registered.

(b) Requests for information concerning the national administration of the Military Selective Service Act shall be addressed to the National Headquarters, Selective Service System, 1724 F Street, NW, Washington, D.C. 20435.

(c) Requests for information concerning the administration of the Military Selective Service Act within a particular State shall be addressed to the State Director of Selective Service involved.

### § 1608.9 Rules governing the obtaining of information.

(a) A request for information under this part must be made orally or in writing during business hours at the appropriate selective service office. When information to be furnished is not readily available; the employee responsible for obtaining the information shall advise the requester how and where it may be obtained...

(b) Although the time period allowed for inspection of identifiable documents and registrants' files must be sufficient to allow hand copying, the activity should not interfere with the daily business activities of the selective service office. Accordingly, the selective service employee handling the request for information or inspection should arrange for inspection of selective service files and documents during specified hours of the business

(c) Any person entitled under the provisions of this part to examine any record or information shall be permitted to copy it by hand, to photograph it or to copy it by using portable copying equipment so long as the use of such equipment does not disrupt the normal operations of the office.

## § 1608.10. Identification of information requested.

(a) Any person who requests information under these regulations shall provide a reasonably specific description of the information sought so that it may be located without undue search is not an identifiable record, and the request for that information may be declined.

(b) If the description is insufficient. the employee processing the request will notify the requester and, to the extent possible, indicate the additional information required. Every reasonable effort shall be made to assist a requester in the identification and location of the record or records sought. Records will not be withheld merely because it is difficult to find them.

(c) When a request is received at an office not having charge of the records, it shall promptly forward the request to the

the action taken.

### § 1608.11 Fees for search of records and copies of documents.

Fees for search of records and copies of documents are the following:

(a) Search of records is made by compensated employees of the Selective Service System without charge.

(b) The charge for copies of documents prepared on Selective Service System equipment is as follows:

(1) 25 cents per page for a copy of File Folder (SSS Form 101) and contents except as provided in paragraph (e) of this section.

(2) 25 cents per page for other identifiable records or documents.

(c) For copies of File. Folder (SSS Form 101) or other identifiable records or documents reproduced by a private concern, the requester will assume the expense of copying, The Selective Service System employee's time to monitor the reproduction, computed from the time of his departure until his return to his post, will be charged by the Selective Service System to the requester at the rate of \$1 per quarter-hour after the first quarter-hour.

(d) Copies will not be released to any requester until these fees are paid in full by money order payable to the Treas-

ury of the United States.

(e) Where a registrant has been charged under the Military Selective Service Act and must defend himself in a criminal prosecution, or where a registrant submits to induction and thereafter brings habeas corpus proceedings to test the validity of his induction, the Selective Service System will furnish to him, or to any person he may designate, one copy of his selective service file free of charge.

## § 1608.12 Request for information not authorized to be disclosed.

Whenever an employee receives a request for information or documents the disclosure of which is not clearly authorized by the provisions of this part that request will be immediately reported by telephone to the General Counsel, Selective Service System for instruction as to its disposition.

#### § 1608.13 Review of denials of requests for information.

(a) Complaints concerning possible abuse of discretion granted selective service employees under this part or failure to respond to inquiries shall be directed to the state director in the case of state headquarters or local board employees and to the Director in the case of National Headquarters employees:

(b) A requester whose request for information or documents has not been satisfied may appeal to the Director of Selective Service, 1724 F Street NW., Washington, D.C. 20435.

#### § 1608:14 Demands of courts or other authorities for records or information protected by these regulations.

No officer or employee of the Selective Service System will comply with a request, demand or order of a court or

proper office and notify the requester of other authority to produce information the disclosure of which is prohibited or restricted by the provisions of this part without the prior approval of the Director of Selective Service.

> BYRON V. PEPITONE. Director.

FEBRUARY 13, 1975.

[FR Doc.75-4451 Filed 2-18-75;8:45 am]

#### PART 286a-AVAILABILITY OF INFORMATION TO THE PUBLIC

## **Cancellation of Part**

The provisions of Part 286a have been incorporated in Part 297, Part 286a is hereby cancelled.

> MAURICE W. ROCHE, Director, Correspondence and Directives OASD (Comptroller).

FEBRUARY 14, 1975.

[FR Doc.75-4526 Filed 2-18-75;8:45 am]

#### PART 299-PUBLIC ACCESS TO RECORDS

The Director, National Security Agency/Chief, Central Security Service has approved Part 299. This Part 299 implements the requirements of 5 U.S.C. 552: as amended by Public Law 93-502 and sets out the procedures for public. access to the official records of the NSA/ CSS.

Part 299 is added to read as follows:

Sec.

299.1 Purpose.

Organization and Requests for In-299.2 formation. Indexes.

299.3 299.4 Procedures for request of records.

Determination and notification. 299.5

299.6 Appeals: Effective date.

299.7

AUTHORITY: The provisions of this Part 299 issued under 5 U.S.C. 552 as amended by Public Law 93-502.

## § 299.1 Purpose.

Pursuant to the requirements of the-Public Information Section of the Administrative Procedure Act (5 U.S.C. 552), and amendments thereto contained in Public Law 93-502, the following rules of procedure are established with respect to public access to the records of the National Security Agency/ Central Security Service.

### § 299.2 Organization and requests for information.

The Headquarters of the National Security Agency/Central Security Service is located at Fort George G. Meade. Maryland. Requests for information and decisions and other submittals may beaddressed to the Information Officer, National Security Agency/Central Security Service, Fort George G. Meade, Maryland 20755.

#### § 299.3 Indexes.

The NSA/CSS does not originate final orders, opinions, statements of policy, interpretations, staff manuals or instructions that affect a member of the public of the type covered by the indexing requirement of 5 U.S.C. 552(a)(2) or required to be published for the guidance of the public under 5 U.S.C. 552(a) (1). The Director, NSA/Chief, CSS, has therefore determined, pursuant to pertinent statutory and Executive Order requirements, that it is unnecessary and impracticable to publish an index of the type required by 5 U.S.C. 552 as amended by Public Law 93-502.

#### § 299.4 Procedures for request of records.

(a) Requests. Requests for access to records of the National Security Agency/Central Security Service may be filed by mail addressed to the Information Officer, National Security Agency/Central Security Service, Fort George G. Meade, Maryland 20755. Requests need not be made on any special form but may be by letter or other written statement identifying the request as a Freedom of Information Act request and setting forth sufficient information reasonably describing the requested record.

(b) Determination and notification. When the requested record has been located and identified, the Information Officer shall determine whether the record is one which, consistent with statutory requirements, Executive Orders and appropriate directives, may be released or should be exempted under the provisions of 5 U.S.C. 552. The Information Officer shall notify the requester of his determination within ten working days of his receipt of the request.

(c) Extension of response time. Where the requested record cannot be located within the initial response period of ten days because of unusual circumstances, the Information Officer shall notify the requester in writing within the initial response period of the delay, the reasons therefore, and a date, not to exceed ten working days, on which a determination is expected to be dispatched.

(d) Fees. If the Agency determines that the requested record can be released, the Information Officer will inform the requester as to the appropriate search and duplication fee, if any, and, upon receipt of this fee, will have the record duplicated and sent to the requester. Fees will be computed in accordance with the uniform Schedule of Fees promulgated by the Department of Defense. Fees paid in accordance with this paragraph will be paid by check or postal money order forwarded to the Information Officer and made payable to the Treasurer of the United States.

### § 299.5 Appeals.

Any person denied access to records may, within 30 days after notification of such denial, file an appeal to the Executive for Staff Services, National Security Agency/Central Security Service. Such an appeal shall be in writing addressed to the Executive for Staff Services, Na-

tional Security Agency/Central Security Service. Fort George G. Meade, Maryland 20755, shall reference the initial denial of access issued by the Agency to the requester and shall contain in sufficient detail and particularity grounds upon which the requester believes release of the information is required. The Executive for Staff Services shall respond to the appeal within 20 working days after receipt of the appeal.

### 299.6 Effective date.

This notice shall become effective upon February 19, 1975.

> MAURICE W. ROCHE. Director, Correspondence and Directives, OASD (Comptroller).

FEBRUARY 14, 1975.

[FR Doc.75-4527 Filed 2-18-75;8:45 am]

#### **PART 290-**-AVAILABILITY TO THE PUBLIC DEFENSE CONTRACT AUDIT **AGENCY INFORMATION**

The Director of the Defense Contract Audit Agency (DCAA) has approved the following revision of Part 290 as shown below in order to reflect the new requirements established by Pub. L. 93-502, which amended the Freedom of Information Act, 5 U.S.C. 552.

Sec.

290.1 Purpose. Applicability. 290,2

Policy. 290.3

Availability of records. 290,4

290.5 DCAA officials authorized to release records requested by members of the public.

290.6 Procedure for submission of requests for DCAA records by members of the public.

290.7 Material that may be withheld from

disclosure. Expedited handling required. 290.9

290.10 Appeals.

Judicial action. 290.11

AUTHORITY: . (5 U.S.C. 301, amended by Act of Nov. 21, 1974 (Pub. L. 93-502, 88 Stat. 1-3).

### § 290.1 Purpose.

This part implements 5 U.S.C. 552, as amended, and Department of Defense Directive § 5400.7 (Part 286 of this Subchapter) and describes established facilities at which, the officials from whom, and the procedures whereby members of the public may inspect and copy or obtain copies of unclassified DCAA records.

## § 290.2 Applicability.

This part is applicable to Headquarters DCAA and all DCAA Regions.

## § 290.3 Policy.

It is the policy of the DCAA to make available to the public the maximum amount of information concerning its operations and activities. This basic policy is subject to the exemptions recognized in 5 U.S.C. 552(b) and Part 286 of this Subchapter. However, notwithstanding such permissible exemptions, it is DCAA's policy that an exempt record will be made available to the public when

its disclosure would not be inconsistent with statutory requirements and when the DCAA official designated in § 290.5 determines that no significant and legitimate governmental purpose would be served by withholding the information. The latter determination is within the sole discretion of the DCAA.

### § 290.4 Availability of records.

(a) Any identifiable documentary material in the possession of the Defense Contract Audit Agency which qualifies as a "record", as discussed in paragraph (b) of this section and which is not exempt under 5 U.S.C. 552(b) and Part 286 of this Subchapter shall be made available on the request of any person.

(b) In determining whether documentary material qualifies as a "record" consideration will be given to 44 U.S.C. 366, which defines the word "record" as

follows:

(It) includes all books, papers, maps, photographs, or other documentary materials, regardless of physical form or characteristics, made or received by any agency of the United States Government in pursuance of federal law or in connection with the transaction of public business and preserved or appropriate for preservation by that agency or its legitimate successor as evidence of the organization, functions, policies, decisions, procedures, operations, or other activities of the Government or because of the informational value of data contained therein.

(1) Records are not limited to permanent or historical documents but include contemporaneous documents as

well.

(2) The term "records" does not include objects or articles such as structures, furniture, paintings, sculpture, three-dimensional models. vehicles. equipments, etc., whatever their historical value or value as "evidence."

(3) Formulae, designs, drawings, research data, computer programs, technical data packages, etc., are not considered "records" within the intent of 5 U.S.C. 552, even though maintained in documentation form because of development costs, utilization, or value, these items are considered property, not preserved for informational value nor as evidence of agency functions, but as exploitable resources to be utilized in the best interest of all the public. Requests for copies of such material shall be evaluated in accordance with policies expressly directed to the appropriate dissemination or use of such property. Requests to inspect such material to determine its content for informational purposes shall normally be granted; however, unless inspection is inconsistent with the obligation to protect the property value of the material, as, for example, may be true for certain formulae.

For a record to be considered "identifiable" it must exist at the time of the request. There is no obligation to create a record to satisfy a request for information.

(d) Requesters must reasonably describe the record sought. This requires sufficient particularity in the description to enable the DCAA to locate the record requested with reasonable effort.

(e) Pursuant to 5 U.S.C. 552(4) (A), requesters will be charged the reasonable costs of searching for, coyping and certifying records. Such charges will be determined in accordance with Part 286 of this Subchapter.

(f) Requests for identifiable records may be denied only when an official designated in § 290.5 determines that such denial is authorized by 5 U.S.C. 552 (b) and Part 286 of this Subchapter.

§ 290.5 DCAA officials authorized to release records requested by members of the public.

Authority to make initial determinations of whether to release DCAA records to members of the public is, in the case of Headquarters, DCAA vested in the Records Administrator; in the case of the DCAA Regions such authority is vested in the Regional Managers.

§ 290.6 Procedure for submission of requests for DCAA records by members of the public.

(a) Identification of records requested. Requests to inspect or obtain copies of records will normally be submitted by letter and should contain at least the following information.

(1) As complete an identification as possible of the desired record, including (if known) its title or a description thereof, date and the issuing office.

(2) A statement as to whether the requester wishes to inspect the record at a DCAA facility or obtain copies thereof by mail.

(b) Addressing requests. Requesters should address their inquiries to the appropriate official as follows:

(1) Records Administrator, Headquarters, DCAA, Cameron Station, Alexandria, Va. 22314.

(2) Regional Manager, Atlanta Region, DCAA, 3100 Maple Drive NE., Atlanta, Ga. 30305.

(3) Regional Manager, Boston Region, DCAA, Waltham Federal Center, 424 Trapelo Road, Waltham, MA 02154.

(4) Regional Manager, Chicago Region, DCAA, 527 So. LaSalle Street, Suite 652, Chicago, IL 60605.

(5) Regional Manager, Los Angeles Region, DCAA, 1340 West Sixth Street, Second Floor, Los Angeles, CA 90017.

(6) Regional Manager, New York Region, DCAA, 252 Seventh Avenue, Fourth Floor, New York, NY 10001.

(7) Regional Manager, Philadelphia Region, DCAA, Federal Building, 1421 Cherry Street, Philadelphia, Pa. 19102.

(8) Regional Manager, San Francisco Region, DCAA, 450 Golden Gate Avenue, Box 36116, San Francisco, CA 94102.

(c) Facilities for inspection and copying of records.

(1) Subject to the exemptions set forth in 5 U.S.C. 552(b) and Part 286 of this Subchapter, Headquarters, DCAA and each DCAA Region shall make available for public inspection and copying in an appropriate facility or facilities the materials described in Part 286 of this

Subchapter, unless such materials are published and copies offered for sale.

(2) The cost of copying any such opinion, order, or statement of policy or interpretation shall be imposed on the person requesting the copy in accordance with Part 286 of the Subchapter.

(3) Headquarters, DCAA and each DCAA Region shall maintain on a current basis and make available for public inspection and copying in the facilities established for that purpose current indexes providing identifying information for the public as to any matter issued, adopted or promulgated after July 4, 1967 and required by 5 U.S.C. 552 and Part 286 of this Subchapter to be indexed.

## § 290.7 Fees.

(a) The schedule of fees published in Part 286 of this Subchapter shall be applicable to all services rendered to members of the public hereunder.

(b) When no specific fee has been established for a particular service, or the service requested does not fall within a category covered by Part 286 of this Subchapter, an appropriate fee will be established therefor in accordance with Part 288 of this Subchapter.

(c) Normally, mail requests for identifiable records should be accompanied by remittance of the total fee chargeable, as well as a self-addressed, stamped envelope, if special mail services are de-

sired. (d) Fees must be paid in full prior to search for an issuance of requested copies of records. If uncertainty as to the existence of a record, or as to the number of sheets to be copied or certified, precludes remittance of the exact fee with the request, the Defense Contract Audit Agency will inform the requester of the exact amount required. The time limits for responding to requests established in 5 U.S.C. 552 and Part 286 of this Subchapter begin to run only upon receipt of any required payment, unless the requester indicates in writing a clear willingness to pay the anticipated costs.

(e) Remittances may be in the form of either a personal check or bank draft drawn on a U.S. bank, or by U.S. postal money order. All such instruments shall be drawn to the order of the Treasurer of the United States.

(f) A receipt for fees paid will be given only upon request. Refund of fees paid will be made when the requested service cannot be rendered.

§ 290.8 Material that may be withheld from disclosure.

Only those records falling within the specific exemptions listed in 5 U.S.C. 552(b), and Part 286 of this Subchapter may be withheld from members of the public. Therefore, a request for a record made pursuant to 5 U.S.C. 552 and this part may be denied only by a DCAA official designated in § 290.5 of this Chapter, and only upon a determination that:

(1) The record is subject to one or more of the exemptions set forth in 5 U.S.C. 552(b) and Part 286 of this Subchapter, and a significant and legitimate governmental purpose is served by withholding it; or

(2) The record cannot be found because it has not been described with sufficient particularity to enable the DCAA to locate it with a reasonable amount of effort; or

(3) The requester has unreasonably failed to comply with the procedural requirements imposed by this part.

## § 290.9 Expedited handling required.

The handling of all requests from members of the public to inspect or copy records will be expedited by the officials designated in § 290.5 of this chapter, as well as by all other DCAA people. The initial determination of whether to make a record available upon request will normally be made within ten (10) working days of the date a request is received by the official designated to respond for the type of record sought. Any delay beyond this initial ten day period may not exceed ten additional working days and will be authorized only for the reasons described in Part 286 of this Subchapter.

## § 290.10 Appeals.

Any person who is denied a request for a record, in whole or in part, will be given a written explanation of the basis for such determination by an appropriate official designated in § 290.5 of this Chapter, and advised of his right to appeal the denial to the Deputy for Resources Management, Headquarters, DCAA, Cameron Station, Alexandria, Va. 22314. Final determinations on appeals will normally be made within twenty (20) working days of the receipt thereof. If, due to unusual circumstances, additional time is needed to decide the appeal, the final determination may be delayed for the number of working days, not to exceed ten (10), which were not utilized as additional time for responding to the initial request. Final refusal to provide a requested record will be made in writing by the Deputy for Resources Management in accordance with the appeal procedures prescribed in Part 286 of this Subchapter and after consultation with the Counsel, DCAA and the Deputy for Audit Management.

### § 290.11 Judicial action.

5 U.S.C. 552 provides that a requester will be deemed to have exhausted his administrative remedy after he has been refused the requested record by the head of an agency, or his designee, or when the agency fails to respond to his request within the time limits prescribed by law. The requester may then seek an order from a U.S. District Court: (1) in the district in which he resides or has his principal place of business; (2) in the district in which the record is situated: or 43) in the U.S. District Court for the District of Columbia, enjoining the agency from withholding the record and ordering its production.

> MAURICE W. Roche, Director, Correspondence and Directives, OASD (Comptroller).

FEBRUARY 14, 1975.

[FR Doc.75-4528 Filed 2-18-75:8:45 am]

#### XXII-NATIONAL SECURITY CHAPTER AGENCY, CENTRAL SECURITY SERVICE

# PART 2200—AVAILABILITY OF INFORMATION

Pursuant to the requirements of the Public Information Section of the Administrative Procedure Act (5 U.S.C. 552), and amendments thereto contained in Pub. L. 93-502, the following rules of procedure are established with respect to public access to the records of the National Security Agency/Central Security Service.

Sec

2200.1 Organization and requests for infor-

2200 2 Indexes.

2200.3 Procedures for request of records. Appeals.

AUTHORITY: 5 U.S.C. 552, as amended by Pub. L. 93-502.

#### § 2200.1 Organization and requests for information.

The Headquarters of the National Security Agency/Central Security Service is located at Fort George G. Meade, Maryland. Requests for information and decisions and other submittals may be addressed to the Information Officer, National Security Agency/Central Security Service, Fort George G. Meade, Maryland 20755.

#### § 2200.2 Indexes.

The NSA/CSS does not originate final orders, opinions, statements of policy, interpretations, staff manuals or instructions that affect a member of the public of the type covered by the indexing requirement of 5 U.S.C. 552(a)(2) or required to be published for the guidance of the public under 5 U.S.C. 552(a) (1). The Director, NSA/Chief, CSS, has therefore determined, pursuant to pertinent statutory and Executive Order requirements, that it is unnecessary and impracticable to publish an index of the type required by 5 U.S.C. 552 as amended by Pub. L. 93-502.

#### § 2200.3 Procedures for request of records.

(a) Requests. Requests for access to records of the National Security Agency/ Central Security Service may be filed by mail addressed to the Information Officer, National Security Agency/Central Security Service, Fort George G. Meade, Maryland 20755. Requests need not be made on any special form but may be by letter or other written statement identifying the request as a Freedom of Information Act request and setting forth sufficient information reasonably describing the requested record.

(b) Determination and notification. When the requested record has been located and identified, the Information Officer shall determine whether the record is one which, consistent with statutory requirements, Executive Orders and appropriate directives, may be released or should be exempted under the provisions of 5 U.S.C. 552. The Information Officer shall notify the requester of his determination within ten working days of his receipt of the request.

the requested record cannot be located within the initial response period of ten days because of unusual circumstances, the Information Officer shall notify the requester in writing within the initial response period of the delay, the reasons therefor, and a date, not to exceed ten working days, on which a determination is expected to be dispatched.

(d) Fees. If the Agency determines that the requested record can be released, the Information Officer will inform the requester as to the appropriate search and duplication fee, if any, and upon receipt of this fee, will have the record duplicated and sent to the requester. Fees will be computed in accordance with the uniform Schedule of Fees promulgated by the Department of Defense. Fees paid in accordance with this paragraph will be paid by check or postal money order forwarded to the Information Officer and made payable to the Treasurer of the United States.

#### § 2200.4 Appeals.

Any person denied access to records may, within 30 days after notification of such denial, file an appeal to the Executive for Staff Services, National Security Agency/Central Security Service. Such an appeal shall be in writing addressed to the Executive for Staff Services, National Security Agency/Central Security Service, Fort George G. Meade, Maryland 20755, shall reference the initial denial of access issued by the Agency to the requester and shall contain in sufficient detail and particularly the grounds upon which the requester believes release of the information is required. The Executive for Staff Services shall respond to the appeal within 20 working days after receipt of the appeal.

Effective date. This notice shall become effective on February 19, 1975.

> LEW ALLEN, Jr. Lieutenant General, USAF Director, NSA/Chief, CSS.

[FR Doc.75-4615 Filed 2-18-75:8:45 am]

#### Title 36—Parks, Forests, and Public Property

#### -AMERICAN BATTLE MONUMENTS COMMISSION

# ART 404—PROCEDURES AND GUIDE-LINES FOR COMPLIANCE WITH THE FREEDOM OF INFORMATION ACT

On January 13, 1975, there was published in the FEDERAL REGISTER (40 FR 2447) a notice of proposed rule-making in which the American Battle Monuments Commission announced that it was planning to issue a new Part 404 to 36 CFR Chapter 4. The new Part 404 was to contain procedures and Guidelines for Compliance with the Freedom of Information Act. The publication of these procedures and guidelines is required by the November 21, 1974 amendments to the Freedom of Information Act (5 U.S.C. 552). Interested persons were given 30 days in which to submit comments and recommendations regarding the proposed the Secretary.

(c) Extension of response time. Where rule making. As a result of the comments received, § 404.1(a) was amended.

Accordingly with this change, the proposed Procedures and Guidelines are adopted as set forth below:

404.1 General. Denial of access. 404.2 404.3 Appeals.

AUTHORITY: 5 U.S.C. 552, as amended.

#### § 404.1 General.

- (a) Public requests for information from the records of the American Battle Monuments Commission should be sent to that agency at Room 4C014, The Forrestal Building, Washington, D.C. 20314. They may also be sent to its field offices at the below listed addresses:
- (1) Officer-in-Charge, European American Battle Monuments APO New York 09777 Commission,
- (2) Officer-in-Charge, Mediterranean Office, American Battle Monuments Commis-
- sion, APO New York 09794
  (3) Superintendent, Manila Am
  Cemetery, APO San Francisco 96528 American
- (b) Requests for information from the public will be honored within ten working days except where the confidentiality of such information is protected
- (c) Whenever information cannot be dispatched within ten working days after receipt of its request, an interim reply will be sent informing the requestor of the status of the request.

#### § 404.2 Denial of access.

(a) Letters denying access to confidential information will be dispatched within ten working days of receipt of the request and will be signed by one of the below listed personnel:

(1) Officer-in-Charge, ABMC European Office

(2) Officer-in-Charge ABMC Mediterranean Office (3) Directors, ABMC Washington Of-

fice

(4) Secretary, ABMC

- (b) Letters denying access to information will:
- (1) provide the requestor with the reason for denial,
- (2) inform the requestor of his or her right to appeal the denial within 30 days,

(3) give the name of the official to

whom the appeal may be sent.

(c) If an unusual circumstance delays a decision concerning access to information, the requestor will be informed of the delay within ten working days of the request's initial receipt. In no case will the decision be delayed more than 20 working days from initial receipt of the request.

(d) A copy of each denial of information will be furnished to the Secretary at the time of its dispatch.

#### § 404.3 Appeals.

(a) The Secretary is the appellate authority for all denials except those which he authors. The Chairman is the appellate authority for denials authored by

(b) The requestor will be informed of the decision on his or her appeal within 20 working days after its receipt. If the denial is upheld, the requestor will be advised that there are provisions for judicial review of such decisions under the Freedom of Information Act.

(c) In the event a court finds that the American Battle Monuments Commission has arbitrarily and capriciously withheld information from the public and a subsequent Civil Service Commission investigation finds agency personnel responsible, these personnel will be subject to disciplinary action by the American Battle Monuments Commission.

Dated: February 13, 1975,

A. J. ADAMS, Mayor General, U.S. Army, Secretary.

[FR Doc.75-4373 Filed 2-18-75:8:45 am]

#### PART 405-SCHEDULE OF FEES FOR SEARCH AND DUPLICATION OF RECORDS

On January 13, 1975, there was published in the Federal Register (40 FR 2447) a notice of proposed rule making in which the American Battle Monuments Commission announced that it was planning to issue a new Part 405 to 36 CFR Chapter 4. The new Part 405 was to contain a schedule of fees for search and duplication of records available to the public under the Freedom of Information Act. The publication of this schedule is required by the November 21, 1974 amendments to the Freedom of Information Act (5 U.S.C. 552). Interested persons were given 30 days in which to submit comments and recommendations regarding the proposed rule

No written objections have been received and the proposed schedule of fees is hereby adopted without change and is set forth below.

405.1 General. 405.2 Schedule.

AUTHORITY: 5 U.S.C. 552 as amended.

### § 405.1 General.

(a) While most information will be furnished promptly at no cost as a service to the general public, fees will be charged if the cost of search and duplication warrants.

(b) When a fee is to be charged, the individual requesting the information will be informed of the fee, and no work will be performed until he or she has agreed to pay it.

#### § 405.2 Schedule.

Fees which may be charged by this Commission for search and duplication of records are as follows:

(a) Duplication fees:

(1) \$2.00 for first 6 pages, 5¢ per page thereafter for photocopying.

(2) \$1.50 per 8x10 inch black and white print of photographs.

(3) \$2.50 per 8x10 inch color print of photographs.

(b) Search fees:

(1) \$8.00 per hour to search records for specific documents.

(2) \$215.00 for selective extracts from Commission computer tapes.

(3) Transportation costs of personnel and records arising from searches for requested information.

Dated: February 13, 1975.

A. J. ADAMS, Major General, U.S. Army Secretary.

[FR Doc.75-4372 Filed 2-18-75;8:45 am]

#### Title 43—Public Lands: Interior SUBTITLE A-OFFICE OF THE

# SECRETARY OF THE INTERIOR

### PART 2-RECORDS AND TESTIMONY Pursuant to the authority granted by

5 U.S.C. 301, 5 U.S.C. 552, 31 U.S.C. 483a and 43 U.S.C. 1460, Part 2 of Title 43, Subtitle A, Code of Federal Regulations is revoked in its entirety and a revised

Part 2 is adopted.

The 1974 Amendments to the Freedom of Information Act, Pub. L. 93-502 (November 21, 1974) significantly revised the Freedom of Information Act. In order to conform the Department of the Interior's regulations to the changes in the Act and to improve the Department's procedures for handling requests made under the Act, detailed new Freedom of Information procedures are set out in the revised Part 2. The significant features of the new procedures are as fol-

1. Specific time limits for the processing of requests are established, as required by the 1974 Amendments. Decisions on whether to grant initial requests for records must be made within 10 working days. Decisions on appeals from initial denials of requests must be decided within 20 working days. Provision is made for extension of both the initial request time limit and the appellate decision time limit. The total extension with respect to any one request may not exceed 10 working days, however.

2. Authority to process initial requests under the Freedom of Information Act is vested in the various bureaus and offices of the Department. Requests made to field installations of a bureau may be denied only by the head of the field installation or such higher authority as the head of the bureau may designate. Requests made to the headquarters of a bureau may be denied only by the head of the bureau or by an official designated

by the head of the bureau.

3. The appellate authority of the Secretary of the Interior in Freedom of Information Act cases is vested in the Assistant Secretary-Program Development and Budget, who will decide all appeals after consultation with the Solicitor, the Director of Communications and the appropriate program Assistant Secretary.

4. Specific procedures for the submission of Freedom of Information Act requests are outlined for requesters.

5. Specific standards for granting and denying requests are also established. A

request for records may be denied only if the record falls within one of the nine exemptions from the Freedom of Information Act's general disclosure requirements and the non-disclosure of the record is required by statute or Executive Order or supported by sound grounds.

Because of the length of the new Freedom of Information procedures and because it is anticipated that detailed regulations implementing the Privacy Act, P.L. 93-579, will be added to Part 2 later in 1975, it was necessary to rearrange the

remainder of the prior Part 2.
Prior § 2.3, "Declassification of classified documents," has been placed in subpart C of the revised Part 2 and redesig-

nated section 2.41.

Prior § 2.5, "Opinions in adjudication of cases; administrative manuals," has been placed in subpart A and redesignated §§ 2.1 to 2.3. The section has also been modified by deletion of certain obsolete provisions.

Prior §§ 2.6 and 2.7, "Compulsory process" and "Testimony of employees," have been placed in subpart E and designated

§§ 2.81 and 2.82.

The remaining sections of the prior Part 2 dealt with Freedom of Information matters and have been replaced by the revised Freedom of information regulations, §§ 2.11 to 2.19 of subpart B of the revised Part 2.

Subpart D of the revised Part 2 has been reserved for addition of regulations

implementing the Privacy Act.

The Appendix to the prior Part 2 has been redesignated Appendix B in the revised Part 2 and has been expanded to list the heads of the bureaus and offices of the Department and their addresses and the public information officers of the Department and their addresses.

Appendix A is reserved for addition of a uniform fee schedule. Such a schedule was proposed by a notice of proposed rulemaking published in the FEDERAL REGISTER on Friday, February 7, 1975, 40

Fed. Reg. 5783.

Because Part 2 consists of rules of agency organization, procedure and practice, adoption of the revised Part 2 through the rulemaking procedure prescribed in 5 U.S.C. 553 is not required and the revised Part 2 is accordingly adopted without resort to that procedure. In accordance with the public policy expressed in 5 U.S.C. 553 and the Department's policy statement of May 4, 1971 on rulemaking procedures, interested persons may, however, submit written comments, suggestions, data or arguments concerning the revised Part 2 to the Assistant Solicitor-General Legal Services, Office of the Solicitor, U.S. Department of the Interior, Washington, D.C. 20240, on or before April 7, 1975. All such comments received will be given full consideration.

The amendments to the Freedom of Information Act take effect on February 19, 1975. So that the Department's new procedures will be in effect when the amendments take effect or shortly thereafter, good cause for waiver of the 30day waiting period for effectiveness exists and the revised Part 2 accordingly will take effect February 19, 1975.

The revised Part 2 reads as follows: Subpart A—Opinions in Adjudication of Cases; Administrative Manuals

2.1

Purpose and scope.

Opinions in adjudication of cases. 2.2

Administrative manuals.

#### Subpart B-inspection of Records

Purpose and scope. 211 Definitions. 2.12

Records available.

2.14

Requests for records.

Action on initial requests. 2.15

Time limits on processing of initial 2.16 requests.

2.17 Appeals.

Action on appeals.

2 19 Fees.

# Subpart C—Declassification of Classified Records

2.41 Declassification of Classified Records.

#### Subpart D-[Reserved]

# Subpart E—Compulsory Process, Testimony of Employees

Compulsory process. 2.82 Testimony of employees.

Appendix B-Heads of Bureaus and Offices of

the Department of the Interior.

AUTHORITY: 5 U.S.C. 301, and 552; 31 U.S.C. 483a, and 1460.

# Subpart A—Opinions in Adjudication of Cases, Administrative Manuals

#### § 2.1 Purpose and scope.

This subpart contains the regulations of the Department of the Interior concerning the availability to the public of opinions issued in the adjudication of cases and of administrative manuals. Persons interested in obtaining access to other records are directed to the procedures for submission of Freedom of Information requests set out in Subpart B of this part.

### § 2.2 Opinions in adjudication of cases.

(a) (1) Copies of final decisions and orders issued on and after July 1, 1970, in the following categories of cases are available for inspection and coyping in the Office of Hearings and Appeals, Ballston Building No. 3, 4015 Wilson Boulevard, Arlington, Va. 22203:

(i) Contract appeals:

(ii) Appeals from decisions rendered by departmental officials relating to the use and disposition of public lands and their resources and the use and disposition of mineral resources in certain acquired lands of the United States and in the submerged lands of the Outer Continental Shelf:

(iii) Appeals from orders and decisions issued by departmental officials and administrative law judges in proceedings relating to mine health and safety; and

(iv) Appeals from orders and decisions of administrative law judges in Indian probate matters other than those involing estates of Indians of the Five Civilized Tribes and Osage Indians.

(2) Copies of final opinions and orders issued in the following categories of cases are available for inspection and copying in the Docket and Records Sec-

tion, Office of the Solicitor, Interior Building, Washington, D.C. 20240:

(i) Tort claims decided in the headquarters office of the Office of the Solicitor, and appeals from decisions of Regional Solicitors or Field Solicitors on tort claims;

(ii) Irrigation claims under Public Works Appropriation Acts (e.g., 79 Stat., 1103) or 25 U.S.C. 388 decided in the headquarters office of the Office of the Solicitor, and appeals from decisions of Regional Solicitors on irrigation claims;

(iii) Appeals under § 2.18 respecting availability of records;

(iv) Appeals from decisions of officials of the Bureau of Indian Affairs, and

Indian enrollment appeals; and (v) Appeals from decisions of officers of the Bureau of Land Management and of the Geological Survey in proceedings relating to lands or interests in land.

contract appeals, and appeals in Indian probate proceedings, issued prior to

July 1, 1970.

(3) An Index-Digest is issued by the Department. All decisions, opinions and orders issued in the categories of cases described in subdivisions paragraph (a) (1), (i), (ii), and (iii) of this section (that is, contract appeals, land appeals, and mine health and safety appeals), are covered in the Index-Digest; in addition, the Index-Digest covers the more important decisions, opinions and orders in the remaining categories of cases described in paragraphs (a)(1)(iv) and (a) (2) (i) – (iv) of this section, and the more important opinions of law issued by the Office of the Solicitor. The Index-Digest is available for use by the public in the Office of Hearings and Appeals, Ballston Building No. 3, 4015 Wilson Boulevard, Arlington, Va. 22203, in the Docket and Records Section, Office of the Solicitor, Interior Building, Washington, D.C. 20240, and in the offices of the Regional Solicitors and Field Solicitors. Selected decisions, opinions, and orders are published in a series entitled "Decisions of the United States Department of the Interior" (cited as I.D.), and copies may be obtained by subscription from the Superintendent of Documents U.S. Government Printing Office, Washington, D.C. 20402.

(4) Copies of final opinions and orders issued by Regional Solicitors on tort claims and irrigation claims, and copies of final opinions and orders on appeals in Indian probate proceedings issued by Regional Solicitors prior to July 1, 1970, are available for inspection and copying in their respective offices. Copies of final opinions and orders-issued by Field Solicitors on tort claims are available for inspection and copying in their respec-

tive offices.

(b) (1) Copies of final decisions and orders issued prior to July 1, 1970, on appeals to the Director, Bureau of Land Management, and by hearing examiners of the Bureau of Land Management, in proceedings relating to lands and interests in land are available for inspection and copying in the Office of Hearings and Appeals, Ballston Building No. 3, 4015 Wilson Boulevard, Arlington, Va.

22203, and in the offices of the Departmental administrative law judges.

(2) Copies of final decisions, opinions and orders issued on and after July 1, 1970, by departmental administrative law judges in all proceedings before them are available for inspection and copying in their respective offices and in the Office of Hearings and Appeals, Ballston Building No. 3, 4015 Wilson Boulevard, Arlington, Va. 22203.

(3) Copies of final decisions, opinions and orders issued by administrative law judges in Indian probate proceedings are available for inspection and copying in

their respective offices.

#### § 2.3 Administrative manuals.

The Departmental Manual is available for inspection in the Departmental Library, Interior Building, Washington, D.C., and at each of the regional offices of bureaus of the Department. The administrative manuals of those bureaus which have issued such documents are available for inspection at the headquarters offices and at the regional offices of the bureaus.

#### Subpart B-Inspection of Records

#### § 2.11 Purpose and scope.

(a) This subpart contains the regulations of the Department of the Interior implementing the requirement of subsection (a) (3) of the Freedom of Information Act, 5 U.S.C. Section 552(a) (3), which provides that the Department "upon any request for records which (1) reasonably describes such records and (2) is made in accordance with published rules stating the time, place, fees (if any), and procedures to be followed, shall make the records promptly available to any person." This subpart describes the procedures by which records may be obtained from all constituent units of the Department of the Interior. The procedures in this subpart are not applicable to requests for records published in the Federal Register or opinions in the adjudication of cases, statements of policy and interpretations and administrative staff manuals which have been published or made available under Subpart "A" of this part.

(b) Before invoking the formal proinformation or records of the Department may find it useful to consult with officials of the bureau possessing the information or records or the Office of Communications, U.S. Department of the Interior, Washington, D.C. 20240.

#### § 2.12 Definitions.

(a) Act. As used in this subpart, "Act" means the "Freedom of Information Act," 5 U.S.C. § 552.

(b) Bureau. As used in this subpart, "bureau" refers to all constituent bu-reaus and offices of the Department of the Interior, including the Office of the Secretary and the other Departmental offices.

### § 2.13 Records available.

(a) Department policy. It is the policy of the Department of the Interior to make the records of the Department available to the public to the greatest extent possible, in keeping with the spirit of the Freedom of Information Act.

(b) Statutory disclosure requirement. The Freedom of Information Act requires that the Department, on a request from a member of the public to inspect or copy records made in accordance with the procedures in this subpart, shall promptly make the records available.

(c) Statutory exemptions. The Act exempts nine categories of records from this disclosure requirement. The Act provides that disclosure is not required of

matters that are:

(1) Specifically authorized under criteria established by an Executive order to be kept secret in the interest of national defense or foreign policy and in fact properly classified pursuant to such Executive order:

(2) related solely to the internal personnel rules and practices of an agency;
(3) specifically exempt from disclo-

sure by statute;

(4) trade secrets and commercial or financial information obtained from a person and privileged or confidential;

(5) inter-agency or intra-agency memorandums or letters which would not be available by law to a party other than an agency in litigation with the agency;

(6) personnel and medical files and similar files the disclosure of which would constitute a clearly unwarranted

invasion of personal privacy;

(7) investigatory records compiled for law enforcement purposes, but only to the extent that production of such records would (i) interfere wies enforcement proceedings, (ii) deprive a person of a right to a fair trial or an impartial adjudication, (iii) constitute an unwarranted invasion of personal privacy, (iv) disclose the identity of a confidential source and, in the case of a record compiled by a criminal law enforcement authority in the course of a criminal investigation, or by an agency conducting a lawful national security intelligence investigation, confidential information furnished only by the confidential source, disclose investigative techniques and procedures, or (vi) endanger the life or physical safety of law enforcement personnel;

(8) contained in or related to examination, operating, or condition reports prepared by, on behalf of, or for the use of an agency responsible for the regulation or supervision of financial institu-

tions; or

(9) geological and geophysical information and data, including maps, con-

cerning wells.

- (d) Decisions on requests. It is the policy of the Department to withhold information falling within an exemption only if (1) disclosure is prohibited by statute or Executive Order or (2) sound grounds exist for invocation of the exemption.
- (e) Deletion of portions of records. If a requested record contains material within an exemption together with material not within an exemption and it is determined under the regulations in this subpart to withhold the exempt ma-

terial, any reasonably segregable nonexempt material shall be separated from the exempt material.

(f) Creation of records. This subpart applies only to records which exist at the time a request for records is made. Records are not required to be created in response to a request by combining or compiling selected items from the files or by preparing a new computer program, nor are records required to be created to provide the requester with such data as proportions, percentages, frequency distributions, trends, or comparisons.

(g) Records of concern to other departments and agencies. (1) If the release of a record would be of concern to both the Department of the Interior and another Federal agency, the record will be made available by the Department only if the interest of the Department is the primary interest. If the Department's interest is not the primary interest, the requester shall be referred in writing to the agency having the primary interest. The Department of the Interior has the primary interest in a record if the record was developed pursuant to Department regulations, directives, or request even though the record originated outside of the Department.

(2) If the release of a record in which the Department has a primary interest would be of substantial concern to another agency, the official processing the request, should, if administratively feasible and appropriate, consult with that agency before releasing the record.

(h) Records obtained from the public. If a requested record was obtained by the Department from a person or entity outside of the Government, the official responsible for processing the request shall, when it is administratively feasible to do so, seek the views of that person or entity on whether the record should be released before making a decision on the request.

#### § 2.14 Requests for records.

(a) Submission of requests. A request to inspect or copy records shall be made to the installation where the records are located. If the records are located at more than one installation or if the specific location of the records is not known to the person wishing to inspect or copy the records, he may direct his request to the head of the appropriate bureau, or the bureau's chief public information officer, if any.

(b) Assistance in submitting requests. If a requester is uncertain which bureau of the Department is responsible for a record which he wishes to inspect or copy, he may seek guidance from the Office of Communications, U.S. Department of the Interior, Washington, D.C. 20240, to assist him in determining the appropriate bureau to which to submit

his request.

(c) Form of request. (1) Requests invoking the Freedom of Information Act shall be in writing.

(2) (i) A request must reasonably describe the records requested. A request reasonably describes the records requested if it will enable an employee of

the Department familiar with the subject area of the request to locate the record with a reasonable amount of effort. If such information is available, the request should identify the subject matter of the record, the date when it was made, the place where it was made, and the person or office that made it, the present custodian of the record, and any other information which will assist in location of the requested records. If the request involves a matter known by the requester to be in litigation, the request should also state the case name and court hearing the case.

(ii) If the description of a recordsought is insufficient to allow identification and location of the record, the response denying the request on this ground shall so state and, to the extent possible, indicate what additional descriptive information, if any, would as-

sist in location of the record.

(3) A request shall state the maximum amount of fees which the requester is willing to pay. Requesters are notified that under § 2.19(d), the failure to state willingness to pay fees as high as are anticipated by the Department will delay running of the time limit and delay processing of the request, if the responsible official anticipates that the fees charge-

able may exceed \$25.00.

(4) (i) To insure expeditious handling, requests shall be prominently marked, both on the envelope and on the face of the request, with the legend "FREEDOM OF INFORMATION REQUEST." The failure of a request to bear such a legend will not disqualify a request from processing under the procedures in this subpart if the request otherwise meets the requirements of this section. A request not bearing the legend "FREEDOM OF INFORMATION REQUEST" will not, however, be deemed to have been received for purposes of the running of the time limit set out in § 2.16 until it has been identified by bureau personnel as a Freedom of Information request and marked by them with this legend.

(ii) Bureau personnel identifying a communication from the public not bearing the legend "FREEDOM OF INFORMATION REQUEST" as a request otherwise meeting the requirements of this section shall immediately (A) mark the communication with the legend "FREEDOM OF INFORMATION REQUEST," (B) date the request to reflect the date on which it was identified, and (C) take steps to assure proper processing of the request under the procedures in this

subpart.

(d) Categorical requests. (1) A request for all records falling within a reasonably specific category shall be regarded as conforming to the statutory requirement that records be reasonably described if (i) it can be determined which particular records are covered by the request and (ii) the records can be seached for, collected and produced without unduly burdening or interfering with Department operations because of the staff time consumed or the resulting disruption of the files.

(2) If a categorical request is determined under paragraph (d) (1) of this section not to reasonably describe the records requested, the response denying the request on that ground shall specify the reasons why and shall extend to the requester an opportunity to confer with knowledgeable Department personnel in an attempt to reduce the request to manageable proportions by reformulation and by agreeing on an orderly procedure for the production of the records.

#### § 2.15 Action on initial requests.

(a) Granting of requests. (1) A requested record shall be made available if (i) the record is not exempt from disclosure or (ii) the record is exempt from disclosure, but its withholding is neither required by statute or Executive order nor supported by sound grounds.

(2) A determination to release a record which is exempt from disclosure and which is located in a field installation may be made only by the head of that installation or by such higher authority as the head of the bureau may in writing designate. A determination to release a record which is exempt from disclosure and which is located at the headquarters installation of a bureau may be made only by the head of the bureau or an official whom the head of the bureau has in writing designated.

(b) Form of grant. (1) When a requested record has been determined to be available, the official processing the request shall immediately notify the person requesting the record as to where and when the record is available for inspection or, as the case may be, where and when copies will be available. If fees are due under § 2.19, the responsible official shall also state the amount or, if the exact amount cannot be determined, the approximate amount of fees due.

(2) If the record was obtained by the Department from a person or entity outside of the Government, the responsible official shall, when it is administratively feasible to do so, notify that person or entity that the record has been made available.

(c) Denial of requests. A request for a record may be denied only if it is determined that (1) the record is exempt from disclosure and (2) that withholding of the record is required by statute or Executive order or supported by sound grounds.

(d) Authority to deny requests. (1) A request made of a field installation to inspect or copy a record located at that installation may be denied only by the head of the installation or such higher authority as the head of the bureau may in writing designate and only after consultation with the appropriate field or regional solicitor.

(2) A request to inspect or copy a record located at the headquarters installation of a bureau shall be denied only by the head of the bureau or by an official whom the head of the bureau has in writing designated and only after consultation with the appropriate associate, regional, or field solicitor.

(e) Form of denial. A reply denying a request shall be in writing and shall include:

 A reference to the specific exemption or exemptions under the Freedom of Information Act authorizing the withholding of the record;

(2) If neither a statute nor an executive order requires withholding, the sound ground for withholding;

(3) A listing of the names and titles or positions of each person responsible for the denial;

(4) A statement that the denial may be appealed to the Assistant Secretary-Program Development and Budget, pursuant to § 2.17 and that such appeal must be in writing and be received by this official within 20 days (Saturdays, Sundays, and public legal holidays excepted) after the date of the denial, in the case of the denial of an entire request, or within 20 days (Saturdays, Sundays, and public legal holidays excepted) of records being made available, in the case of a partial denial, by writing to the Freedom of Information Act Officer, Office of the Assistant Secretary-Program Development and Budget, U.S. Department of the Interior, Washington, D.C. 20240.

(1) Exception. The requirements of paragraphs (c), (d), and (e) of this section do not apply to requests denied under § 2.14 on the ground that the request did not reasonably describe the records requested or to requests for records

which do not exist.

(g) Filing of denials. (1) Copies of all replies denying, in whole or part, a request for a record which are issued under this section or § 2.14 shall be promptly submitted to the Freedom of Information Act Officer, Office of the Assistant Secretary—Program Development and Budget, U.S. Department of the Interior, Washington, D.C. 20240. The Freedom of Information Act Officer shall be responsible for promptly furnishing copies of such denials to the Office of the Solicitor, the Office of Communications, and the appropriate program Assistant Sec-

(2) If a bureau has a public information office, copies of all replies denying, in whole or part, a request for a record issued by the bureau under this section or § 2.14 shall be submitted to the public information office.

# § 2.16 Time limits on processing of initial requests.

(a) Basic limit. Requests for records shall be processed promptly. A determination whether to grant or deny a request shall be made within no more than 10 days (excepting Saturdays, Sundays, and legal public holidays) after receipt of a request. This determination shall be communicated immediately to the requester.

(b) Running of basic time limit. For purposes of paragraph (a) of this section, the time limit commences to run when a request is received at an installation designated in § 2.14(a) to receive the request. If a request is received at an installation not designated in § 2.14(a) to receive the request, the officer receiv-

ing a request shall promptly transmit the request to the appropriate installation and shall notify the requester in writing of this action.

(c) Extensions of time. In the following unusual circumstances, the time limit for acting upon an initial request may be extended to the extent reasonably necessary to the proper processing of the particular request, but in no case may the time limit be extended for more than 10 working days:

 The need to search for and collect the requested records from field facilities or other establishments that are separate from the office processing the request;

(2) the need to search for, collect, and appropriately examine a voluminous amount of separate and distinct records which are demanded in a single request; or

(3) the need for consultation, which shall be conducted with all practicable speed, with another agency having a substantial interest in the determination of the request or among two or more components of the agency having substantial subject-matter interest therein.

(d) Authority to make extensions. (1) An extension of time under paragraph (c) of this section may be made only by the head of a field installation or such higher authority as the head of a bureau has in writing designated, if the records requested are located in a field installation, or by the head of a bureau or an official whom the head of the bureau has in writing designated, if the records are located in the headquarters installation of a bureau.

(2) The person requesting the records shall be notified in writing of the extension. The written notice shall state the reason for the extension and the date on which a determination on the request is expected to be dispatched.

(3) A copy of the written notice shall be forwarded to the Freedom of Information Act Officer, Office of the Assistant Secretary—Program Development and Budget, U.S. Department of the Interior, Washington, D.C. 20240. The Freedom of Information Act Officer shall be responsible for promptly furnishing copies of such notices to the Office of the Solicitor, the Office of Communications, and the appropriate program Assistant Secretary.

(e) Treatment of delay as denial. (1) If no determination has been reached at the end of the 10-day period for deciding an initial request, or the last extension thereof, the requester may deem his request denied and may exercise a right of appeal in accordance with the provisions of § 2.17.

(2) When no determination can be reached within the applicable time limit, the responsible official shall nevertheless continue to process the request. On expiration of the time limit, the responsible official shall inform the requester of the reason for the delay, of the date on which a determination may be expected to be dispatched, and of his right to treat the delay as a denial for purposes of appeal to the Assistant Secretary—Program De-

velopment and Budget in accordance with § 2.17. The requester may be asked to consider delaying use of his right to appeal until the date on which the determination is expected to be dispatched. If the requester so agrees, he is deemed not to have treated the failure to respond within the applicable time limit as a denial for purposes of the running of the 20 working-day appeal period set out in § 2.17(b). If a determination on the request is not issued by the new agreed upon date, or if the request is denied in whole or part, the requester will have available his full right of appeal under § 2.17, including the entire 20 working-day period for filing of the appeal.

#### § 2.17 Appeals.

(a) Right of appeal. Where a request for records has been denied, in whole or part, the person submitting the request may appeal the denial to the Assistant Secretary—Program Development and Budget.

(b) Time for appeal. An appeal must be received no later than twenty (20) days (Saturdays, Sundays, and public legal holidays excepted) after the date of the initial denial, in the case of a denial of an entire request, or twenty (20) days (Saturdays, Sundays, and public legal holidays excepted) after records have been made available, in the

case of a partial denial.

(c) Form of appeal. (1) An appeal shall be initiated by filing a written notice of appeal. The notice shall be accompanied by copies of the original request and the initial denial and should, in order to expedite the appellate process and give the requester an opportunity to present his arguments, contain a brief statement of the reasons why the requester believes the initial denial to have been in error.

(2) The appeal shall be addressed to Freedom of Information Act Officer, Office of the Assistant Secretary—Program Development and Budget, U.S. Department of the Interior, Washington, D.C.

20240.

(3) (i) Both the envelope containing the notice of appeal and the face of the notice shall bear the legend "FREEDOM OF INFORMATION APPEAL." The failure of an appeal to bear such a legend will not disqualify an appeal from processing under § 2.18 if the appeal otherwise meets the requirements of this section. An appeal not bearing the legend "FREEDOM OF INFORMATION APPEAL" will not, however, be deemed to have been received for purposes of the running of the time limit set out in § 2.18 until it has been identified by Department personnel as a Freedom of Information appeal and marked by them with this legend.

(ii) Department personnel identifying a communication from the public not bearing the legend "FREEDOM OF INFORMATION APPEAL" as an appeal otherwise meeting the requirements of this section shall immediately (A) mark the communication with the legend "FREEDOM OF INFORMATION APPEAL," (B) date the appeal to reflect the

date on which it was identified, and (C) take steps to assure proper processing of the appeal under the procedures in this subpart.

(4) The Freedom of Information Act Officer shall be responsible for promptly furnishing copies of such notices to the Office of the Solicitor, the Office of Communications, and the appropriate program Assistant Secretary.

#### § 2.18 Action on appeals.

(a) Authority. Appeals from initial denials of requests for records shall be decided for the Department by the Assistant Secretary—Program Development and Budget after consultation with the Solicitor, the Director of Communications and the appropriate program Assistant Secretary. If the initial denial appealed from was issued by an official required to be consulted by this paragraph, the Assistant Secretary—Program Development and Budget is not required to consult with that official.

(b) Time limit. A final determination on any appeal shall be made within 20 days (excepting Saturdays, Sundays, and public legal holidays) after receipt

of the appeal.

(c) Extensions of time. (1) If the time limit for responding to the initial request for a record was not extended under the provisions of § 2.16(c) or was extended for fewer than 10 working days, the time for processing of the appeal may be extended by the Assistant Secretary—Program Development and Budget to the extent reasonably necessary to the proper processing of the appeal, but in no event may the extension, when taken together with any extension made during processing of the initial request, result in an aggregate extension with respect to any one request of more than 10 working days. The time for processing of an appeal may be extended only if one or more of the unusual circumstances listed in § 2.16(c) requires an extension.

(2) The Assistant Secretary—Program Development and Budget shall, in writing, advise the appellant of the reasons for the extension and the date on which a final determination on the appeal is

expected to be dispatched.

(3) If no determination on the appeal has been reached at the end of the 20 working-day period for deciding an appeal, or the last extension thereof, the requester is deemed to have exhausted his administrative remedies, giving rise to a right of review in a district court of the United States as specified in 5 U.S.C. 552(a) (4). When no determination can be reached within the applicable time limit, the appeal will nevertheless continue to be processed. On expiration of the time limit the requester shall be informed of the reason for the delay, of the date on which a determination may be expected to be dispatched and of his right to seek judicial review. The requester may be asked to consider delaying resort to his right to judicial review until the date on which the determination on his appeal is expected to be dis-

(d) Form of decision: (1) The final determination on an appeal shall be in writing and shall state the basis for the determination. If the determination is to release the requested records or portions thereof, the Assistant Secretary-Program Development and Budget shall immediately make the records available or instruct the appropriate bureau official to make them immediately available. If the determination upholds in whole or part the initial denial of a request for records, the determination shall advise the requester of his right to obtain judicial review in the United States District Court for the district in which the withheld records are located, or in which the requester resides or has his principal place of business or in the United States District Court for the District of Columbia, and shall set forth the names and titles or positions of each person responsible for the denial.

(2) If the determination is to release a requested record or portions thereof and the record was obtained by the Department from a person or entity outside of the Government, the Assistant Secretary—Program Development and Budget shall, when it is administratively feasible to do so, notify the person or entity of

the release of the record.

(e) Distribution of copies. Copies of final determinations issued by the Assistant Secretary—Program Development and Budget shall be provided to the Office of the Solicitor, the Office of Communciations, and the appropriate program Assistant Secretary.

#### § 2.19 Fees.

(a) Services for which fees may be charged. (1) Unless waived pursuant to the provisions of paragraph (c) of this section, user fees shall be charged for document search and duplication costs incurred in responding to requests for records. User fees also shall be charged for the formal certification of verification attached to authenticated copies of records under the seal of the Department or any bureau.

(2) Unless waived or reduced pursuant to paragraph (c) of this section, user fees shall be charged in accordance with the schedule of charges contained in

Appendix A to this Part.

(b) Services for which fees may not be charged. No fee may be charged for any services required by the Freedom of Information Act to be performed in responding to a request for records other than those services for which fees may be charged under subsection (a) of this section. Services for which no fees may be charged include, but are not limited to, (1) examining requested records to determine whether they are exempt from mandatory disclosure or whether, even if exempt, they should nevertheless be made available in whole or part, (2) deleting exempt matter from records so that the remaining portions of the records may be made available, (3) monitoring a requester's inspection of agency records made available to him for inspection, and (4) resolving legal and policy issues affecting access to requested records.

(c) Waiver or reduction of fees. (1) Fees otherwise chargeable for document search and duplication costs incurred in responding to requests for records shall be waived or reduced, as appropriate, if the official making the records available determines that furnishing the records can be considered as primarily benefiting the public as opposed to the requester.

(2) Fees otherwise applicable for document research and duplication costs incurred in responding to requests shall be waived and not charged if the re-

quest involves:

(i) furnishing unauthenticated copies of any documents reproduced for gratui-

tous distribution;

(ii) furnishing one copy of a personal document (e.g., a birth certificate) to a person who has been required to furnish it for retention by the Department;

(iii) furnishing one copy of the transcript of a hearing before a hearing officer in a grievance or similar proceeding to the employee for whom the hear-

ing was held.

(3) Fees otherwise chargeable for document search and duplication costs incurred in responding to requests may be waived or reduced if the cost of collecting the fee would exceed the amount of the fee or if the request involves:

(i) furnishing records to press, radio and television representatives for dissemination through the media to the

general public;

(ii) furnishing records to donors with

respect to their gifts;

(iii) furnishing records to individuals or private non-profit organizations having an official voluntary or cooperative relationship with the Department to assist the individual or organization in its work with the Department:

(iv) furnishing records to state, local and foreign governments and public international organizations when to do so without charge is an appropriate courtesy, or when the recipient is carrying on a function related to that of the Department and to do so will help to accomplish the work of the Department;

(v) furnishing records when to do so saves costs and yields income equal to the direct cost of providing the records (e.g., where the Department's fee for the service would be included in a billing

against the Department);

(vi) furnishing records when to do so is in conformance with generally established business custom (e.g., furnishing personal reference data to prospective employers of former Department employees):

(vii) furnishing one copy of a record in order to assist the requester to obtain financial benefits to which he is entitled (e.g., veterans or their dependents, employees with Government employee compensation claims or persons insured by the Government).

(d) Notice of anticipated fees and prepayment. (1) Where it is anticipated that fees chargeable under this section may amount to more than \$25.00 and the requester has not indicated in advance his willingness to pay fees as high

as are anticipated, the request shall be deemed not to have been received for purposes of the time limits established by § 2.16 until the requester is advised of the fees which are anticipated and has agreed to pay these fees. Advice to requesters with respect to anticipated fees shall be provided promptly.

(2) In appropriate cases, advance payment of fees may be required before requested records are made available to

the requester.

(3) A notice of anticipated fees or notice of request for advance payment shall extend an offer to the requester to confer with appropriate personnel in an attempt to reformulate the request in a manner which will reduce the antici-pated fees and meet the needs of the requester.

(e) Form of payment. Payment of fees should be made by check or money order payable to the Department of the Interior or the bureau furnishing the information. The term United States or the initials "U.S." should not be included on the check or money order. Where appropriate, the official responsible for handling a request may require that payment by check be made in the form of a certified check.

#### Subpart C-Declassification of Classified **Documents**

§ 2.41 Declassification of classified documents.

(a) Request for classification review. (1) Requests for a classification review of a document of the Department of the Interior pursuant to section 5(c) of Executive Order 11652 (37 FR 5209, March 10, 1972) and section III B of the National Security Council Directive Governing Classification, Downgrading, Declassification and Safeguarding of National Security Information (37 FR 10053, May 1972) shall be made in accordance with the procedures established by this section.

(2) Any person desiring a classification review of a document of the Department of the Interior containing information classified as National Security Information by reason of the provisions of Executive Order 11652 (or any predecessor executive order) and which is more than 10 years old, should address such request to the Chief, Division of Enforcement and Security Management, Office of Management Operations, U.S. Department of the Interior, Washington, D.C. 20240.

(3) Requests need not be made on any special form, but shall, as specified in the executive order, describe the document with sufficient particularity to enable identification of the document requested with expenditure of no more than a reasonable amount of effort.

(4) Charges for locating and reproducing copies of records will be made when deemed applicable in accordance with Appendix A to this Part and the

requester will be notified.

(b) Action on requests for classification review. (1) The Chief, Division of Enforcement and Security Management, shall, unless the request is for a docu-

ment over 30 years old, assign the request to the bureau having custody of the requested records for action. In the case of requests for declassification of records in the custody of the Office of the Secretary and less than 30 years old, the request shall be processed by the Chief, Division of Enforcement and Security Management, Requests for declassification of documents over 30 years shall be referred directly to the Archivist of the United States. The bureau which has been assigned the request, or the Chief, Division of Enforcement and Security Management, in the case of requests assigned to him, shall immediately acknowledge the request in writing. Every effort will be made to complete action on each request within thirty (30) days of its receipt. If action cannot be completed within thirty (30) days, the requester shall be so advised.

(2) If the requester does not receive a decision on his request within sixty (60) days from the date of receipt of his request, or from the date of his most recent response to a request for more particulars, he may apply to the Department of the Interior Committee on Classification of Security Information, U.S. Department of the Interior, Washington, D.C. 20240, for a decision on his request. The Committee must render a decision within

thirty (30) days.

(c) Form of decision and appeal to Committee on Classification of Security Information. In the event that the bureau to which a request is assigned or the Chief, Division of Records and Protective Services, in the case of a request assigned to him, determines that the requested information must remain classified by reason of the provisions of Executive Order 11652, the requester shall be given prompt notification of that decision and, whenever possible, shall be provided with a brief statement as to why the information or material cannot be declassified. He shall also be advised that if he desires he may appeal the determination to the Chairman, Department of the Interior Committee on Classification of Security Information, U.S. Department of the Interior, Washington, D.C. 20240. An appeal shall include a brief statement as to why the requester disagrees with the decision which he is appealing. The Department Committee on Classification of Security Information shall render its decision within thirty (30) days of receipt of an appeal. The Departmental Committee shall be authorized to over-rule previous deter-minations in whole or in part when, in its judgment, continued protection is no longer required.

(d) Appeal to Interagency Classification Review Committee. Whenever the Department of the Interior Committee on Classification of Security Information confirms a determination for continued classification, it shall so notify the requester and advise him that he is entitled to appeal the decision to the Interagency Classification Review Committee established under section 8(A) of Executive Order 11652. Such appeals shall be addressed to the Interagency

#### **RULES AND REGULATIONS**

Classification Review Committee, the Executive Office Building, Washington, D.C. 20500.

(e) Suggestions and complaints. Any person may also direct suggestions or complaints with respect to the administration of the other provisions of Executive Order 11652 and the NSC Directive by the Department of the Interior to the Department of the Interior Committee for Classification of Security Information, U.S. Department of the Interior, Washington, D.C. 20240.

#### Subpart D—[Reserved] Subpart E-Compulsory Process and **Testimony of Employees**

#### § 2.80 Compulsory process.

(a). If the production of any record of the Department is sought by compulsory process and if it is determined in accordance with the provisions of § 2.13 that the record should not be disclosed, the person making such determination shall immediately report the matter to the Solicitor. The person to whom the compulsory process is directed shall appear in answer to the process and respectfully decline to produce the record on the ground that the disclosure, pending the receipt of instructions from the Secretary of the Interior, is prohibited by the regulations in this Subpart.

(b) The Solicitor of the Department of the Interior is authorized to exercise all of the authority of the Secretary of the Interior under this section.

### § 2.82 Testimony of employees.

(a) An officer or employee of the Department shall not testify in any judicial or administrative proceeding concerning matters related to the business of the Government without the permission of the head of the bureau, or his designee, or of the Secretary of the Interior, or his designee. If the head of a bureau or his designee, concludes that permission should be withheld, he shall report the matter immediately to the Solicitor for a determination, and the officer or employee shall appear in answer to process and respectfully decline to testify, pending the receipt of instructions from the Secretary, on the ground that testimony is prohibited by the regulations in this part. Pending instructions from the Secretary or his designee, an officer or employee in the Office of the Secretary shall follow the same procedure.

(b) Any person (including a public agency) wishing an officer or employee of the Department to testify in a judicial or administrative proceeding concerning a matter related to the business of the Government may be required to submit a statement setting forth the interest of the litigant and the information with respect to which the testimony of the officer or employee of the Department is desired, before permission to testify will be granted under this section.

(c) The Solicitor of the Department of the Interior is authorized to exercise all of the authority of the Secretary of the Interior under this section.

1. Heads of Bureaus and Offices of the Department of the Interior. (The address for all bureaus and offices, unless otherwise indicated, is U.S. Department of the Interior, Washington, D.C. 20240.)

Secretary of the Interior Office of the Secretary

Director U.S. Geological Survey The National Center Reston, Virginia 22092 Commissioner

Bureau of Indian Affairs

Director Bureau of Land Management Director Bureau of Mines

Administrator Mining Enforcement and Safety

Administration Director

National Park Service Director

Bureau of Outdoor Recreation Director

Bureau of Reclamation Director

U.S. Fish and Wildlife Service Administrator

Alaska Power Administration P.O. Box 50

Juneau, Alaska 99802

Administrator Bonneville Power Administration P.O. Box 3621

Portland, Oregon 97208

Administrator Southeastern Power Administration Samuel Elbert Bldg.

Elberton, Georgia 30635 Administrator

Southwestern Power Administration P.O. Drawer 1619

Tulsa, Oklahoma 74101 Administrator

Defense Electric Power Administration

Solicitor Office of the Solicitor

Director Office of Hearings and Appeals 4015 Wilson Boulevard Arlington, Virginia 22203

Director

Office of Legislation

Director Office for Equal Opportunity

Director

Office of Communications

Director

Office of International Activities Director

Office of Research and Development

Director

Office of Water Research and Technology

Director

Office of Land Use and Water Planning 801—19th Street, NW. Washington, D.C. 20006

Director

Office of Management Services

Director

Office of Management Operations

Director

Office of Management Consulting Director

Office of Organization and Personnel Manage-

Office of Library Services

Director Office of Manpower Training and Youth Activities

Director

Office of Audit and Investigation

Director

Office of ADP Management Director

Office of Aircraft Services 3905 Vista Avenue

Boise, Idaho 83705

Director Office of Budget

Director

Office of Policy Analysis

Director

Office of Environmental Project Review

Director

Office of Outer Continental Shelf Program Coordination

2. Public Information Officers of the Department of the Interior. (The address for all public information officers, unless otherwise indicated is U.S. Department of the Interior, Washington, D.C. 20240.)

Director

Office of Communications

U.S. Department of the Interior

Chief

Office of Minerals Information

Bureau of Mines

Director

Public Information Staff

Bureau of Indian Affairs

Chief

Office of Public Affairs Bureau of Reclamation

Information Officer U.S. Geological Survey

The National Center Reston, Virginia 22092

Chief

MESA Information Office

Mining Enforcement and Safety Administration

Ballston Tower No. 3 4015 Wilson Boulevard Arlington, Virginia 22203 Assistant to the Director

Office of Public Affairs National Park Service

Assistant Director—Public Affairs U.S. Fish and Wildlife Service

Staff Assistant for Communications Office of Land Use and Water Planning 801 19th Street NW.

Washington, D.C. 20006 Information Officer

Bonneville Power Administration P.O. Box 3621

Portland, Oregon 97208

Administrator Alaska Power Administration P.O. Box 50

Juneau, Alaska 99801 Chief

Division of Administrative Management Southeastern Power Administration

Samuel Elbert Building Elberton, Georgia 30635

Public Information Specialist Southwestern Power Administration P.O. Drawer 1619

Tulsa, Oklahoma 74101

3. Office of Hearings and Appeals-Field Offices.

Administrative Law Judge Federal Building

Sacramento, California 95825 Administrative Law Judge

Pederal Building Salt Lake City, Utah 84138

### **RULES AND REGULATIONS**

Administrative Law Judge Indian Probate Arizona Title Annex Building Phoenix, Arizona 85012 Administrative Law Judge Indian Probate Federal Building Sacramento, California 95825 Administrative Law Judge **Indian Probate** Federal Building and Courthouse Billings, Montana 59101 Administrative Law Judge Indian Probate V.E.M. Building Gallup, New Mexico 87301 Administrative Law Judge Indian Probate U.S. Post Office and Federal Office Building Tulsa, Oklahoma 74101 Administrative Law Judge Indian Probate Federal Building Portland, Oregon 97208

4. Office of the Solicitor-Field Offices.

Anchorage Region

Regional Solicitor Federal Building Anchorage, Alaska 99501 Field Solicitor U.S. Post Office and Courthouse Juneau, Alaska 99801

Atlanta Region

Regional Solicitor
148 Cain Street, N.E., Suite 410
Atlanta, Georgia 30303
Field Solicitor
Elberton, Georgia 30635
Denver Region
Regional Solicitor

Regional Solicitor
Denver Federal Center
Denver, Colorado 80225
Field Solicitor
Post Office Box 549
Aberdeen, South Dakota 57401
Field Solicitor
Post Office Box 1538
Billings, Montana 59103
Field Solicitor
Post Office Box 983
Cheyenne, Wyoming 82001

Portland Region

Regional Solicitor
Post Office Box 3621—Room 766
1002 N.E. Holladay Street
Portland, Oregon 97208
Field Solicitor
Post Office Bóx 020
Boise, Idaho 83702
Field Solicitor
Reclamation Building
Division and C Streets
Ephrata, Washington 98823

Philadelphia Region

Regional Solicitor
Merchants Exchange Building
143 South Third Street
Philadelphia, Pennsylvania 19106
Field Solicitor
686 Federal Building
Fort Snelling
Twin Cities, Minnesota 55111

Sacramento Region

Regional Solicitor 2800 Cottage Way—Room E-2753 Sacramento, California 95825 Field Solicitor Arizona Title Annex Building-Room 410 135 North 2nd Avenue Phoenix, Arizona 85003 Field Solicitor 4127 Federal Building-U.S. Courthouse 300 Booth Street Reno, Nevada 89502 Field Solicitor 3610 Central Avenue-Suite 104 Riverside, California 92506 Field Solicitor Post Office Box 36064 450 Golden Gate Avenue—Room 14126 San Francisco, California 94102 Field Solicitor Window Rock, Arizona 86515 Salt Lake City Region

Regional Solicitor Federal Building—Suite 6201 125 South State Street Salt Lake City, Utah 84111

Tulsa Region

Regional Solicitor Page Belcher Federal Building—Room 3068 333 West 4th Street Tulsa, Oklahoma 74103 Field Solicitor Federal Building & Courthouse 500 Gold Avenue, S.W. Albuquerque, New Mexico 87101 Field Solicitor Box H-4393, Herring Plaza 317 East Third Amarillo, Texas 97101 Field Solicitor Post Office Box 397 Anadarko, Oklahoma 73005 Field Solicitor Federal Building-Room 319 5th and Broadway Muskogee, Oklahoma 74401 Field Solicitor c/o Osage Agency Grandview Avenue Pawhuska, Oklahoma 74056 Field Solicitor U.S. Courthouse-Room 224 Federal Place & Washington Street Santa Fe, New Mexico 87501

KENT FRIZZELL,
Acting Secretary of the Interior.

FEBRUARY 14, 1975.

[FR Doc.75-4505 Filed 2-18-75;8:45 am]

# Title 46—Shipping CHAPTER IV—FEDERAL MARITIME COMMISSION

Docket No. 75-1; [General Order 22; Amdt. 5]
PART 503—PUBLIC INFORMATION

Pursuant to amendments to the Freedom of Information Act (5 USC 522) notice of a proposal to amend the regulations of the Federal Maritime Commission dealing with public information (46 CFR Part 503) was published in the Federal Register (40 FR 2983, January 17, 1975). The amended regulations would conform to the requirements of the Freedom of Information Act and, in accordance therewith, afforded interested parties the opportunity to submit comments on the Commission's fee schedule provided in Subpart E of the proposed regulations.

A single comment was received. That comment noted the fact that the fee proposed to be assessed by the Commission

for photocopying of documents was 30¢ per page. It suggested that, in view of a slightly lower range of charges for duplication commonly proposed by other agencies, the Commission review the cost data which justifies a 30¢ charge to assure inclusion of direct costs only, as mandated by the Freedom of Information Act amendments. The Commission reviewed the rate and its cost basis. The component costs of this charge are as follows:

The Commission does not impose a fee for the first one-half hour of search (§ 503.43(c)(3)) and the Commission has discretion to waive or reduce any fee (§ 503.43(d)(4) and (g)). In view of these provisions, and the cost data above, it has been determined that the charge of 30¢ per page for photocopying reflects only direct costs and therefore is in accord with the mandate of the Freedom of Information Act.

Therefore, the proposed regulations are hereby adopted without change and are set forth below.

Effective date. These regulations shall be effective as of February 19, 1975.

By the Commission.

FRANCIS C. HURNEY,

Secretary.

[FR Doc.75-4616 Filed 2-18-75;8:45 am]

#### Title 47—Telecommunication

# CHAPTER I—FEDERAL COMMUNICATIONS COMMISSION

[FCC 75-134]

# PART 0—COMMISSION ORGANIZATION Amendment of Rules Implementing the

Amendment of Rules Implementing the Freedom of Information Act; Report and Order

1. The Congress has recently amended the Freedom of Information Act (Pub. L. 93-502, November 21, 1974, 5 U.S.C. 552). Our purpose herein is to revise the Commission's rules and regulations to comply with the requirements of the modified statute. The amendments to both the Act and the rules apply to requests for inspection of records which are received by the Commission on or after February 19, 1975. In addition to the rules set out in the attached Appendix, which specify procedures for requesting records and acting on requests, a schedule of fees to recover the cost of staff time spent in searching for records which are requested will apply to requests submitted on or after February 19, 1975. A Notice of Proposed Rule Making requesting comment on a proposed fee schedule was released on January 22, 1975 (FCC 75-30).

2. For purposes here pertinent, the Freedom of Information Act, as amended, contains the following new or modified provisions:

(a) The amended act requires the publication of certain indexes.

(b) The amended act requires a person requesting records to "reasonably describe" those records. This replaces a provision requiring agencies to grant requests for "identifiable records."

(c) The amended act provides for

waiver of fees.

(d) The amended act specifies time limits for actions by agencies on requests for inspection of records.

(e) The amended act provides that notice of denial of a request shall set forth the names and titles or positions of

each person responsible.

(f) The amended act provides that any reasonably segregable portion of a record shall be provided to any person requesting such record after deletion of portions which are exempt from inspection

(g) The amended act modifies the description of two categories of exempt

records.

These new or modified provisions and the rules implementing them are discussed below.

- 3. The publication of indexes, 5 U.S.C. 552(a)(2). The amended act requires agencies to publish indexes of the following materials:
- (a) Final opinions, including concurring and dissenting opinions, made in the adjudication of cases;
- (b) Those statements of policy and interpretations which have been adopted by the agency and are not published in the FEDERAL REGISTER; and
- (c) Administrative staff manuals and instructions to staff that affect a member of the public.

Except in the case of administrative staff manuals, indexes for these materials are now published, as provided in § 0.445 of the rules. The manuals are listed in that section and are indexed. Since the indexes would be of no use whatsoever without the text of the manuals, no purpose would be served by publishing them. We find, therefore, pursuant to 5 U.S.C. 552, that publication of the indexes to the Commission's administrative manuals is unnecessary and impracticable.

4. Reasonable description of records requested, 5 U.S.C. 552(a) (3). In accordance with this provision, requirements in our rules which seem to require more than a reasonable description have been deleted. Our reading of what is reasonable is contained in the following two requirements:

\_The records in question must be reasonably described by the person requesting them, so as to permit their location by staff

Written requests shall \* \* \* for each document requested, \* \* \* set out all information known to the person making the request which would be helpful in identifying and locating the document.

See § 0.460 (c) and (e) and § 0.461 (a) and (b) as set out below.

5. Waiver of fees, 5 U.S.C. 552(a) (4) (A). Pursuant to this provision of the act, the proposed fee rules (§ 0.466(c)) provide as follows for the waiver of search fees:

(c) The search fee will be waived or reduced by the custodian of the records, upon a showing that waiver or reduction is in the public interest because furnishing the information is primarily of benefit to the general public.

A waiver of duplicating fees on the same grounds will be considered by the custodian of the records under present section 0.465(c) (3) of the Rules, which provides, in part, that, "When it serves the regulatory or financial interests of the U.S. Government, the Commission will make and furnish copies of its records free of charge."

6. Time limits for actions, 5 U.S.C. 552(a) (6). This part of the act provides

that agencies:

(a) Shall act on requests for records within 10 working days after receipt of the request and immediately notify the requester of the action, the reasons therefore, and of his right to appeal an adverse ruling to the head of the agency;

(b) Shall act on any appeal of an adverse ruling within 20 working days after its receipt and, in the event of an adverse ruling on appeal, notify the requester of his right to judicial review;

and

(c) May extend the aggregate period for acting on the initial request and the appeal for a maximum of 10 working days, but only for the following reasons: (1) The need to search for and collect the records from establishments that are separate from the office processing the request; (2) the need to search for, collect or examine a voluminous amount of records demanded in a single request; or (3) the need to consult with another interested agency or among components of the same agency. If the initial or appellate periods are exceeded, the requester may immediately seek an injunction in a Federal district court.

7. The new time limits are responsible for most of the rule changes set out below. The provisions are implemented by a new § 0.460, covering requests for materials which are routinely available for public inspection, and a revised § 0.461, covering requests for other materials. They have necessitated a revision of § 0.459, covering requests that materials submitted to the Commission be received

in confidence.

8. In part, new § 0.460 tracks provisions heretofore set out in § 0.451. Significant additions include the following:

(a) Paragraph (d) requires written requests for routinely available records where it appears that there will be an appreciable delay in locating or producing the documents.

(b) Paragraph (e) specifies the required contents of written requests, including a statement of the maximum search fee the person making the request is prepared to pay. (If the specified maximum is reached before all the records have been located, the search will be terminated).

(c) Paragraph (f) contains instructions for delivery of the request.

(d) Paragraph (h) provides that all requests limited to records listed in §§ 0.453 and 0.455 (those routinely avail-

able for inspection) will be granted and provides that requests for records listed in those sections shall not be combined with requests for other records.

(e) Paragraph (i) provides that the records will be produced for inspection

at the earliest possible time.

9. Significant changes in § 0.461 include the following:

(a) Paragraphs (b) and (c) specify the required contents of requests for records not routinely available, all of which must be in writing. Included is a statement of the maximum search fee the person making the request is prepared to pay.

(b) Paragraph (d) contains instructions for delivery of the request.

(c) Paragraph (g) provides for action on the request within 10 working days (if possible), for an extension of up to 10 working days (if permissible and if needed), for a further extension on consent of the requester (if needed), and for denial of the request if it cannot be acted on within the statutory period and if the requester does not agree to an extension. In the event of denial in the latter circumstance, the search for and/or assessment of the records will continue during the period provided for filing and acting on an application for review.

(d) Paragraph (h) provides for administrative and judicial review of an action granting a request for records where the records have been submitted to the Commission in confidence. Note that provision has not been made for consideration of oppositions to requests for inspection by the person who submitted the materials to the Commission. The 10 day period allowed for action is not sufficient to provide for the filing and consideration of oppositions prior to the initial action. However, as provided in paragraph (d), the person who submitted the materials will be notified when the request is filed and will be afforded an opportunity to file an application for review of an order granting the request for inspection. In view of the statutory admonition to produce the records promptly after grant of the request, the period for seeking administrative review of a grant has been reduced from 15 to 5 days and a period of 5 days has been specified for seeking a judicial stay if the Commission affirms the order granting the request for inspection. The limited period is also required because of the possibility of a partial grant, followed by applications for review by two parties, including the person who submitted the request for inspection, and the requirement that the latter application be acted on within 20 days after it is submitted.

(e) Paragraph (i) provides that the person who requested the records is afforded 30 days in which to apply for review of an adverse action.

(f) Paragraph (k) provides for action within 20 working days en an application for review, if possible, and contains essentially the same provision for statutory and consensual extensions of the 20 day

period as are provided in the case of the Part 0 of the rules and regulations is initial ruling under \$ 0.461(g).

(g) Paragraph (l) provides that, if a request is granted, the records will be made available at the earliest possible time.

10. In part, new § 0.459 tracks provisions of existing § 0.459. The main changes are that the procedures for submitting, and the criteria for acting on, requests that materials submitted to the Commission be received in confidence have been tightened and that the periods allowed for filing an application for review or for seeking a judicial stay of an adverse action have been reduced. Significant changes in § 0.459 include the following:

(a) Paragraph (a) notes that a request under § 0.459 is unnecessary if the materials in question are specifically listed in § 0.457 as materials which are not rou-

tinely available for inspection.

(b) Paragraph (c) provides that requests which do not meet the requirements of paragraphs (a) and (b) will not be considered. Among other things, it is expected that requests will contain a full statement of the facts and reasons, and of the statutory basis, for non-disclosure.

(c) Paragraph (d) provides that the request will be granted only if it presents a clear and convincing case for non-disclosure consistent with the Freedom of

Information Act.

(d) Paragraph (g) provides for the filing of an application for review of an action denying the request or non-disclosure. In view of the fact that a request for inspection of the materials may be filed at any time and must be acted on within 10 days, the period for filing an application for review has been reduced from 30 to 5 days. In addition, the period afforded for seeking a judical stay of a Commission order affirming an order denying the request has been limited to 5 days.

11. Disclosure of names of persons responsible for denying a request for inspection, 552(a) (4) (c). This provision is implemented by § 0.461(m) of the rules.

12. Deletion of the exempt portions of a record and disclosure of the remainder, 5 U.S.C. 552(b). This provision is implemented by § 0.461(e) (5).

13. Modification of substantive provisions, 5 U.S.C. 552(b) (1) and (7). These changes have been carried forward into § 0.457 (a) and (g).

14. Authority for the amendments set out below is set out in section 4(i) and 303(r) of the Communications Act of 1934, as amended, 47 U.S.C. 154(i) and 303(r), and in 5 U.S.C. 552.

15. Because the amendments are procedural in nature or carry forward to our rules mandatory provisions of the amended Freedom of Information Act. and because the revised rules must be in effect on February 19, 1975, compliance with the notice and effective date provisions of 5 U.S.C. 553 is unnecessary and impracticable.

16. In view of the foregoing, it is ordered, effective February 19, 1975, that

amended as set out below.

(Secs. 4, 303, 46 Stat., as amended, 1066, 1082 (47 U.S.C. 154, 3031)

Adopted: February 4, 1975.

Released: February 11, 1975.

FEDERAL COMMUNICATIONS COMMISSION,

[SEAL] VINCENT J. MULLINS, Secretary.

Part 0 of Chapter I of Title 47 of the Code of Federal Regulations is amended as follows:

1. Section 0.451 is revised to read as follows:

§ 0.451 Inspection of records: Generally.

(a) Records which are routinely available for public inspection. Sections 0.453 and 0.455 list those Commission records which are routinely available for public inspection and the places at which those records may be inspected. Procedures governing requests for inspection of such records are set out in § 0.460.

(b) Records which are not routinely available for public inspection. Records which are not listed in § 0.453 or § 0.455 are not routinely available for public inspection. Such records fall into two cate-

gories.

(1) The first category consists of those records or kinds of records listed in 0.457 and of particular records withheld from public inspection under § 0.459. The Commission has determined that there is a statutory basis for withholding these records from public inspection. In some cases, the Commission is prohibited from permitting the inspection of records. In other cases, the records are the property of another agency, and the Commission has no authority to permit their inspection. In still other cases, the Commission is authorized, for reason of policy, to withhold records from inspection, but is not required to do so.

(2) The second category consists of records which are not listed in § 0.453. § 0.455, or § 0.457 and have not been withheld from inspection under § 0.459. In some cases, these records have not been identified for listing. In other cases (e.g., the general correspondence files), the Commission is unable to determine either that all records in a class should be routinely available for inspection or that all records in that class should not be routinely available for inspection, and individual determination is required.

(3) Procedures governing requests for inspection of these records are set out in § 0.461.

(4) Procedures governing demands by competent authority for inspection of these records are set out in § 0.463.

(5) Except as provided in §§ 0.461 and 0.463, no officer or employee of the Commission shall permit the inspection of records which are not routinely available for public inspection under § 0.453 or § 0.455, or disclose information contained therein.

(c) Copies. Section 0.465 applies to requests for copies of Commission records

which are routinely available for public inspection under §§ 0.453 and 0.455 and those which are made available for inspection under § 0.461. Section 0.467 applies to requests for certified copies of Commission records.

(d) Search fees. Section 0.466 prescribes fees to cover the expense of searching for records made available for inspection under § 0.460 or § 0.461.

§ 0.456 [Deleted]

2. Section 0.456 is deleted.

3. In § 0.457, the portion of paragraph (a) preceding paragraph (a) (1), and paragraph (g), are revised to read as set forth below:

§ 0.457 Records not routinely available for public inspection.

(a) Materials that are specifically authorized under criteria established by executive order to be kept secret in the interest of national defense or foreign policy and are in fact properly classified pursuant to such executive order, 5 U.S.C. 552(b)(1).\*

(g) Investigatory records compiled for law enforcement purposes, to the extent that production of such records would (1) Interfere with enforcement proceedinas:

(2) Deprive a person of a right to fair trial or an impartial adjudication; (3) Constitute an unwarranted inva-

sion of personal privacy;

(4) Disclose the identity of a confidential source;

(5) Disclose investigative techniques or procedures; or

(6) Endanger the life or physical safety of law enforcement personnel. 5 U.S.C. 552(b) (7).

4. Section 0.459 is revised to read as follows:

§ 0.459 Requests that materials or information submitted to the Commission be withheld from public inspection.

(a) Any person submitting information or materials to the Commission may submit therewith a request that such information not be made routinely available for public inspection. (If the materials are specifically listed in \$ 0.457, such a request is unnecessary.) A copy of the request shall be attached to and shall cover all of the materials to which it applies and all copies of those materials. If feasible, the materials to which the request applies shall be physically separated from any materials to which the request does not apply; if this is not fea-sible, the portion of the materials to which the request applies shall be identi-

(b) Each such request shall contain a statement of the reasons for withholding the materials from inspection § 0.457) and of the facts upon which those reasons are based. If the request is that the materials be withheld from inspection for a limited period of time, that period shall be specified.

(c) Casual requests which do not comply with the requirements of paragraphs

(a) and (b) of this section will not be considered.

(d) Requests which comply with the requirements of paragraphs (a) and (b) of this section will be acted on by the appropriate Bureau or Office Chief, who is directed to grant the request only if it presents a clear and convincing case for non-disclosure consistent with the provisions of the Freedom of Information Act, 5 U.S.C. 552. If the request is granted, the ruling will be placed in a public file in lieu of the materials withheld from public inspection. A copy of the ruling shall be forwarded to the General Counsel.

(e) If the materials are submitted voluntarily (i.e., absent any direction by the Commission), the person submitting them may request the Commission to return the materials without consideration if the request for confidentiality should be denied. In that event, the materials will ordinarily be returned (e.g., an application will be returned if it cannot be considered on a confidential basis); only in the unusual instance where the public interest so requires will the materials be made available for public inspection. If submission of the materials is required by the Commission and the request for confidentiality is denied, the materials will be made available for public inspection.

(f) If no request for confidentiality is submitted, the Commission assumes no obligation to consider the need for nondisclosure but, in the unusual instance, may determine on its own motion that the materials should be withheld from public inspection. See § 0.457(g).

- (g) If a request for confidentiality is denied, the person who submitted the request may, within 5 working days, file an application for review by the Commission. If the application for review is denied, the person who submitted the request will be afforded 5 working days in which to seek a judicial stay of the ruling. If these periods expire without action by the person who submitted the request, the materials will be returned to the person who submitted them or will be placed in a public file. Notice of denial and of the time for seeking review or a judicial stay will be given by telephone, with follow-up notice in writing.
- (h) If the request is granted, the status of the materials is the same as that of materials listed in § 0.457. Any person wishing to inspect them may submit a request for inspection under § 0.461.
- 5. Section 0.460 is added, to read as follows:
- § 0.460 Requests for inspection of records which are routinely available for public inspection.
- (a) Sections 0.453 and 0.455 list those Commission records which are routinely available for public inspection and the places at which those records may be inspected. Subject to the limitations set out in this section and to the provisions of § 0.466, a person who wants to inspect such records need only appear at

the specified location and ask to see the records.

(b) A person who does not want a copy of the records must appear at the specified location during the office hours of the Commission and must inspect the records at that location. (Procedures governing requests for copies are set out in § 0.465.) However, arrangements may be made in advance, by telephone or by correspondence, to make the records available for inspection on a particular date, and there are many circumstances in which such advance arrangements will save inconvenience. If the request is for a large number of documents, for example, a delay in collecting them is predictable. Current records may be in use by the staff when the request is made. Older records may have been forwarded to another location for storage.

(c) The records in question must be reasonably described by the person requesting them so as to permit their location by staff personnel. The information needed to locate the records will vary, depending on the records requested. Advice concerning the kind of information needed to locate particular records will be furnished in advance upon request. Members of the public will not be given access to the area in which records are kept and will not be permitted to

search the files.

(d) If it appears that there will be an appreciable delay in locating or producing the records (as where a large number of documents is the subject of a single request or where an extended search for a document appears to be necessary), the request shall be submitted in writing,

either in person or by mail.

(e) Written requests shall be captioned "REQUEST FOR INSPECTION RECORDS", shall be dated, shall list the telephone number (if any) of the person making the request and, for each document requested, shall set out all information known to the person making the request which would be helpful in identifying and locating the document. Written requests shall, in addition, specify the maximum search fee the person making the request is prepared to pay (see § 0.466). If the specified maximum is reached before all of the records have been located, the search will be terminated, and the person making the request will be so notified.

(f) Written requests shall be delivered or mailed directly to the chief of the organizational unit having custody of the records, as listed in §§ 0.453 and 0.455. If the request is enclosed in an envelope, the envelope shall be marked, "REQUEST FOR INSPECTION OF

RECORDS."

(g) When a written request is received by the custodian of the records, it will be date-stamped.

- (h) All requests limited to records listed in §§ 0.453 and 0.455 will be granted. Requests for records listed in those sections shall not be combined with requests for other records.
- (i) The records will be produced for inspection at the earliest possible time.

- 6. Section 0.461 is revised to read as follows:
- § 0.461 Requests for inspection of materials not routinely available for public inspection.

Any person desiring to inspect Commission records which are not listed in § 0.453 or § 0.455 shall file a request for inspection meeting the requirements of this section.

(a) The records in question must be reasonably described by the person requesting them, so as to permit their location by staff personnel. See § 0.460(c).

(b) Requests shall be captioned, "RE-QUEST FOR INSPECTION OF REC-ORDS", shall be dated, shall list the telephone number (if any) of the person making the request and, for each document requested, shall set out all information known to the person making the request which would be helpful in identifying and locating the document. The request shall, in addition, specify the maximum search fee the person making the request is prepared to pay (see § 0.466). If the specified maximum is reached before all of the records have been located, the search will be terminated and the person making the request will be so notified.

(c) If the records are of the kinds listed in § 0.457 or if, they have been withheld from inspection under § 0.459, the request shall, in addition, contain a statement of the reasons for inspection and the facts in support thereof. In the case of other materials, no such statement need accompany the request; but the custodian of the records may require the submission of such a statement if he determines that the materials in question may lawfully be withheld from in-

spection.

(d) (1) Requests shall be delivered or mailed to the Executive Director, who will determine the identity of the custodian of the records and deliver the request to him. (If the request is to be acted on by the officer presiding at a hearing pursuant to paragraph (j) of this section, the Executive Director will deliver the request to the presiding officer rather than to the custodian of the records.)

(2) If the request is enclosed in an envelope, the envelope shall be marked, "REQUEST FOR INSPECTION

RECORDS."

(3) An original and two copies of the request shall be submitted. If the request is for materials not open to routine public inspection under § 0.457(d) § 0.459, one copy of the request will be mailed by the custodian of the records to the person who originally submitted the materials to the Commission.

(e) When the request is received by the Executive Director, it will be date-

- (f) Except as provided in paragraph (j) of this section, requests for inspection of records will be acted on as follows by the custodian of the records.
- (1) If the Commission is prohibited from disclosing the records in question,

the request for inspection will be denied with a statement setting forth the specific grounds for denial.

(2) If the records are the property of another agency, the request will be referred to that agency and the person who submitted the request will be so advised, with the reasons therefor.

(3) If it is determined that the Commission does not have authority to withhold the records from public inspection,

the request will be granted.

(4) If it is determined that the Commission does have authority to withhold the records from public inspection, the considerations favoring disclosure and non-disclosure will be weighed in the light of the facts presented, and the request will be granted, either conditionally or unconditionally, or denied.

(5) If there is a statutory basis for withholding part of a document only from inspection, that part will be deleted and the remainder will be made available

for inspection.

- (g) The custodian of the records will make every effort to act on the request within 10 working days after it is received by the Executive Director. If it is not possible to locate the records and to determine whether they should be made available for inspection within 10 working days, the custodian may, in-any of the following circumstances, extend the time for action by up to 10 working days:
- (1) It is necessary to search for and collect the requested records from field facilities or other establishments that are separate from the office processing the request.
- (2) It is necessary to search for, collect and appropriately examine a voluminous amount of separate and distinct records which are demanded in a single request; or
- (3) It is necessary to consult with another agency having a substantial interest in the determination of the request or among two or more components of the Commission.

If these circumstances are not present of if it is not possible to locate the records and make the determination within the extended period, the person who made the request will be advised of his rights and asked to consent to an extension or further extension. If he agrees to an extension, the custodian of the records will confirm the agreement in a letter specifying the length of the agreed-upon extension. If he does not agree to an extension, the request will be denied, on the ground that the custodian has not been able to locate the records and/or to make the determination within the period for an initial ruling mandated by the Freedom of Information Act, 5 U.S.C. 552. In that event, the custodian will continue to search for and/or assess the records and will advise the person who made the request of further developments; but that person may file an application for review by the Commission. When action is taken by the custodian of the records, the person who made the request and the person who originally submitted, the materials

to the Commission (if any) will be notified and advised of their rights by telephone, with followup notice in writing.

(h) In the case of records not open to routine inspection under § 0.457(d) or § 0.459, the person who originally submitted the records to the Commission may file an application for review of an action by the custodian of the records granting a request for inspection. The application for review shall be filed within 5 working days after the date of the action. No other pleadings shall be filed. If an application for review is not filed within this period, the records will be produced for inspection. If the application for review is denied, the person who submitted the records to the Commission will be notified and advised of his rights by telephone, with follow-up notice in writing, and will be afforded 5 working days in which to obtain a judicial stay of the Commission's action. If a stay is not obtained within this period, the records will be produced for inspection. Because of the statutory constraints imposed upon the Commission, no extensions of time will be granted to seek either Commission review or a judicial stay.

(i) Except as provided in paragraph (h) of this section, an application for review of an initial action on a request for inspection may be filed only by the person who made the request. The application for review shall be filed within 30 days after the date of the action by the custodian of the records. No other

pleadings shall be filed.

(j) If the request is for inspection of records for use in a proceeding which has been designated for hearing and is filed by a party to the proceeding, it will be acted on by the presiding officer under the criteria and procedures set out in this section rather than by the custodian of the records.

(k) The Commission will make every effort to act on an application for review of an action on a request for inspection of records within 20 working days after it is filed. If it is not possible to locate the records and to determine whether they should be made available for inspection within 20 working days, the General Counsel may, in the following circumstances and to the extent time has not been extended under paragraph (g) of this section, extend the time for action by up to 10 working days. (The total period of extensions taken under this paragraph and under paragraph (g) of this section without the consent of the person who submitted the request shall not exceed 10 days.):

(1) It is necessary to search for and collect the requested records from field facilities or other establishments that are separate from the office processing

the request.

(2) It is necessary to search for, collect and appropriately examine a voluminous amount of separate and distinct records which are demanded in a single request; or

(3) It is necessary to consult with another agency having a substantial interest in the determination of the request

or among two or more components of the Commission.

If these circumstances are not present or if it is not possible to locate the records and make the determination within the extended period, the person who made the request will be advised of his rights and asked to consent to an extension or further extension. If he agrees to an extension, the General Counsel will confirm the agreement in a letter specifying the length of the agreed-upon extension. If he does not agree to an extension, the Commission will continue to search for and/or assess the records and will advise the person who made the request of further developments; but that person may file a complaint in an appropriate United States district court.

[FR Doc.75-4303 Filed 2-18-75;8:45 am]

[Docket No. 20236; FCC 75-179]

#### Freedom of Information Act Regulations

In the matter of fees for searching for records requested under the Freedom of Information Act.

- 1. A notice of proposed rulemaking in this proceeding was released on January 22, 1975 (FCC 75-30, 50 FCC 2d, 40 FR 4155, January 28, 1975). The notice proposed a schedule of fees to cover the direct costs of searching for records requested under the Freedom of Information Act and invited the submission of comments on or before February 5, 1975. Comments have been submitted by GTE Service Corporation (GTE) and by the Communications Workers of America (CWA). Reply comments were not invited. Related procedural rules were released by the Commission on February 11, 1974 (FCC 74-134).
- 2. While approving generally of the fee provision,, GTE takes the position that a search fee should not be charged in the case of "informal" requests for records which are routinely available for public inspection. It has since been established that the taking of this position is based on a misconception of the Commission's intentions, due to our inability to publish procedural rules implementing the 1974 amendments to the Freedom of Information Act prior to the filing date for comments in this proceeding. The search fee is, in fact, applicable only to "formal" requests; i.e., those which the rules require to be submitted in writing. In the case of materials which are routinely available, "a person who wants to inspect such records need only appear at the specified location and ask to see the records." (§ 0.460(a), as added by FCC 74-134). A written request is required only by \$0.460(d), which reads as follows:
- (d) If it appears that there will be an appreciable delay in locating or producing the records (as where a large number of documents is the subject of a single request or where an extended search for a document ap-

pears to be necessary), the request shall be submitted in writing, either in person or by

In such circumstances, there is a possibility that we may not be able to locate the records (or all of them) within 10 days and to produce them promptly after the end of that period. There is therefore a need for a written record of the request. Since the burden of making the search would be essentially the same as the burden of searching for records which are not routinely available for inspection in the same circumstances, a search fee is equally justified. This being understood, GTE has no objection to the fee provision.

3. GTE also takes the position that the fee should not be charged in the case of searches made pursuant to the Commission's discovery procedure (47 CFR 1.311 (b) (3)), to searches for statements to be produced under the Jencks rule (47 CFR 1.362), or to searches for documents relied on by the Commission in hearing cases. We agree, and the fee provision will not be so applied. It would be inappropriate for one party to a hearing proceeding (i.e., a Commission Bureau) to impose a charge on another party for producing papers pursuant to an order by the presiding judge. Jencks statements and documents relied on by the Commission, moreover, should be readily available, and producing them should not require a search.

4. CWA takes the position that § 0.466 "will be usable to deny information to the public, or at least to delay production thereof". No explanation is offered as to how the search fee provision could be used for such purposes, and we can see no connection between the provision and such uses. Certainly, there is no intent on our part to so use the provision. The purpose of this provision is simply to compensate the Government for the direct cost of searches for records which are requested, as contemplated by 5

U.S.C. 552(a) (4) (A).

5. CWA also objects to the fact that the Rule Making Notice herein was issued, and that comments on the fee provision were required, before related procedures implementing the Freedom of Information Act amendments were issued. We note that relevant provisions of the related procedures were summarized in a footnote to § 0.466, as proposed in the Rule Making Notice. We nevertheless regret that the precedures could not be published simultaneously with the related fee provision. The necessity to publish both the procedures and the fee provision prior to February 19, 1975, the effective date of the Freedom of Information Act, and the requirement that comment be sought on the fee provision, made our actions necessary. Considering the statutory deadline of February 19, it is clearly not feasible to defer implementation of the fee provision, as urged by CWA.

6. The new search fee provision is set out below. Authority for this rule is contained in sections 4(i) and 303(r) of the Communications Act of 1934, as amended, 47 U.S.C. 154(1) and 303(r), and in 5 U.S.C. 552.

7. Because the Freedom of Information Act amendments take effect on February 19, 1975, good cause exists for making the fee provision effective on that date and less than 30 days after publication of the final rule in the FEDERAL REGISTER.

8. Accordingly, It is ordered, effective February 19, 1975, That Part 0 of the Commission's rules is amended as set forth below.

cs. 4, 303, 48 Stat., as amended, 1036, 1082; 47 U.S.C. 154, 303, and 5 U.S.C. 552)

Adopted: February 12, 1975.

Released: February 18, 1975.

FEDERAL COMMUNICATIONS COMMISSION. VINCENT J. MULLINS, [SEAL] Secretary.

In Part 0 of Chapter I of Title 47 of the Code of Federal Regulations, § 0.466 is revised to read as follows:

#### § 0.466 Search fee.

(a) Subject to the provisions of this section, a fee of \$5 per hour, and for any fraction of an hour, is charged for recovery of the direct costs of searching for records requested under § 0.460(d) or § 0.461. Sections 0.465 and 0.467 deal with charges for providing copies of records made available for inspection.

(b) No search fee will be charged if:

(1) The records are not located; (2) The records are located but are not made available for inspection;

(3) The search does not exceed one bour in duration.

(c) The search fee will be waived or

reduced by the General Counsel, upon a showing that waiver or reduction is in the public interest.

(d) The fee charged will not exceed an amount based on the time normally required to locate records of the kind requested. (For example, the person making the request will not be charged for the extra time required to locate records which have been misplaced or misfiled.)

(e) In most cases, the request for inspection shall specify the maximum search fee that the person making the request is prepared to pay. See § 0.460 (d) and (e) and § 0.461(b). If the specified maximum is reached before all of the records have been located, the search will be terminated and the person making the request will be so notified.

(f) When the search has been completed, the custodian of the records will give notice of the charges incurred to the person who made the request.

(g) The fee shall be paid to the Fee Unit, Financial Management Division, Office of Executive Director, or as otherwise directed by the Commission.

(h) Evidence of payment shall be presented to the custodian of the records before the records are made available for inspection.

(i) Records shall be inspected within 7 days after notice is given that they have been located and are available for inspec-

tion. After that period, they will be returned to storage, and additional charges may be imposed for again producing them.

(j) Record searches will not be undertaken on the request of any person who has incurred but has not paid charges for producing records.

[FR Doc.75-4417 Filed 2-18-75;8:45 am]

#### Title 32-National Defense

# CHAPTER XXI—NATIONAL SECURITY COUNCIL

PART 2101—FREEDOM OF INFORMA-TION ACT REQUESTS FOR CLASSIFIED DOCUMENTS—PROCESSING, FEES, RE-PORTS, APPLICABLE MATERIAL, DE-CLASSIFICATION CRITERIA, PARTIAL RELEASE

FEBRUARY 19, 1975

#### Subpart A-Introduction

2101.1 The amended Freedom of Information Act.

2101.2 Purpose of amended Act.

# Subpart B—Processing Requests for Classified Material

2101.11 Receipt of requests.

2101.12 Initial processing.

2101.13 Requests for time extensions. 2101.14

Initial review period. 2101.15 Initial reply to request.

Processing of appeals to "Head of Agency." 2101.16

2101.17 Appeals to Federal District Courts.

# ort C—Schedule of Fees and Method of Payment for Services Rendered

Fees schedule. 2101.21 2101.22

#### Fee payments.

Subpart D-Reports 2101 31 Quarterly Index of Publications. 2101.32 Annual Report to Congress.

## Subpart E-Applicable Material

2101.41 Primary review by NSC staff. 2101.42 Recommendations to other agencies.

#### Subpart F-Declassification Criteria

2101.51 Criteria for denying a request for release of classified material.

2101.52 Procedural criteria under E.O. 11652.

Substantive criteria under E.O. 11652 for material under ten vears old.

2101.54 Substantive criteria under E.O. 11652 for material over ten years old.

#### Subpart G-Partial Release

2101.61 Release of "reasonably segregable portion" of requested classified material.

2101.62 Downgrading of classified material reviewed under the FOI Act.

AUTHORITY: E.O. 11652, as amended by Pub. L. 93-502, unless otherwise noted.

#### Subpart A—Introduction

#### § 2101.1 The amended Freedom of Information Act.

The amended Freedom of Information Act, Pub. L. 93-502 (5 U.S.C. 552), provided, and to the extent, it is applicable to the National Security Council Staff, provides expanded opportunities for the public to secure the release of classified material under the control of, or of primary interest to, the NSC Staff. The amended Act sets more restrictive time periods within which requests must be

<sup>&</sup>lt;sup>1</sup> Commissioner Hooks dissenting: Commissioner Quello concurring in the result.

processed, modifies the criteria upon which denials of such requests can be based, and permits appeals to district courts of denials of requests. As a result of these changes, modified procedures are required from those followed in the case of requests for the release of classified material over ten years old made under Executive Order 11652. These regulations are intended to guide the NSC Staff response to requests for classified material under the amended Freedom of Information Act, insofar as it is applicable.

### § 2101.2 Purpose of amended Act.

The underlying purpose of the amended Freedom of Information Act is to insure that the public is informed to the fullest extent possible about governmental policies and actions, consistent with the requirement to protect certain categories of sensitive information. In the case of classified material, the Act is intended to protect the public against the improper classification of information, particularly to conceal inefficiency or administrative error or to prevent embarrassment to a person or department. provisions of Executive Order 11652 and the implementing NSC Directive of May 17, 1972, which regulate and control the protection and management of national security information. The provisions of these documents relating to the criteria, authority and procedures for classifying and safeguarding national security material remain unaffected by the amended Act.

#### Subpart B—Processing Requests for Classified Material

#### § 2101.11 Receipt of requests.

(a) Request for classified documents under the FOI Act should be addressed to:

National Security Council ATTN: Staff Secretary Old Executive Office Building Washington, D.C. 20506

Oral requests received by the Staff Secretary or other members of the NSC Staff will be rejected and the requester informed that his request should be made in writing and directed to the Staff Secretary.

(b) All time limits are based on the date of receipt by the NSC Staff of a request. Consequently, this "key date" must in all cases be accurately recorded.

(c) If a request is misdirected to the NSC Staff, the requestor will be promptly notified in writing.

(d) Requests for declassification of material forwarded by National Archives and Records Service (or other Agencies) should be submitted with three copies of the requested material attached.

#### § 2101.12 Initial processing.

(a) Requested material will be rapidly gathered (when not received with the request) screened, and disseminated to appropriate reviewing officers.

(1) Initial dissemination of material to reviewing officers will be made by the Secretariat, with review by the Staff Secretary.

(2) Material to be reviewed will be disseminated utilizing "highly visible" and clearly distinguishable orange folders.

(3) Deadline for submitting recommendations will be indicated in all cases and must be adhered to.

(b) Dissemination of material to other Agencies for comment will be made by the Staff Secretary to responsible officials of these Agencies.

(c) Strict accounting of status and location of all material disseminated will be maintained by the Secretariat.

#### § 2101.13 Requests for time extensions.

(a) Requestors will be promptly notified by the Staff Secretary of a time extension (indicating the new date by which a determination will be dispatched) due to the following three "unusual circumstances":

(1) need to search for or collect records from field facilities or other establishments separate from the office of

equest;

(2) need to search for, collect and examine voluminous amounts of records;

(3) need to consult with other agencies because of their having substantial subject-matter interests.

(b) Time extensions will not exceed ten working days in the aggregate, either solely in the initial or the review stage, or divided between them.

(c) The Staff Secretary will authorize

all time extensions.

(d) Prior to requesting a court authorized time extension, requestors will be contacted by the Staff Secretary in an attempt to arrange informal agreement on a time extension (any verbal agreement will be quickly formalized in writing).

(e) The Staff Secretary will authorize all requests to courts for time extensions based on "exceptional circumstances" and "due diligence". Formal approaches to courts for time extensions will be conducted by the Department of Justice.

#### § 2101.14 Initial review period.

(a) A total of ten working days is available for submitting an initial response to requests.

(1) Time period allowed reviewing officers will be the maximum period consistent with time required for administrative processing and final review of recommendations and preparation of reply by Staff Secretary.

(2) Reviewing officers will be periodically reminded of approaching due dates in order to insure recommendations are received in a timely manner.

(b) Primary reviewing officers will delegate reviewing authority in the event they are unavailable to conduct reviews.

#### § 2101.15 Initial reply to request.

(a) The Staff Secretary will review the recommendations of primary reviewing officers and approve initial replies.

(b) An initial response will always be made even if processing has not been completed.

(c) If processing has not been completed but an interim negative reply has been dispatched, processing will continue.

(d) Accurate record of replies will be maintained by Secretariat.

(e) If a negative or partial denial response is dispatched, requester will be notified of his right to appeal.

(f) If some or all material is approved for release, the Staff Secretary will insure it is expeditiously dispatched to the requestor or grant authority to NARS or other Agency to release the material.

# § 2101.16 Processing of appeals to "Head of Agency".

(a) Replies to appeals will be approved by the Assistant to the President for National Security Affairs, or the Deputy Assistant acting in the name of the Assistant to the President.

(b) The Staff Secretary will be responsible for preparing material required by the above appeal review officer in making

a final determination.

 Views of primary reviewing officers will again be solicited.

(2) Opinion of Office of the White House Counsel and/or Department of Justice on legal ramifications will be solicited.

(c) Staff Secretary will insure replies to appeals are dispatched within 20 working days of receipt or before exhaustion of time of an authorized time extension.

# § 2101.17 Appeals to Federal District Courts.

(a) Staff Secretary will coordinate the NSC Staff's support (to include background data, affidavits, etc.) to USG lawyers defending against court appeals for denied NSC classified documents.

(b) If court judgment is against USG, Staff Secretary will arrange release of

documents to litigant.

# Subpart C—Schedule of Fees and Method of Payment for Services Rendered

### § 2101.21 Fees schedule.

(a) Fees schedule for the search and reproduction of information available under the Freedom of Information Act (5 U.S.C. 552), as amended:

(1) Search for records. \$5.00 per hour when the search is conducted by a clerical employee. \$8.00 per hour when the search is conducted by a professional employee. No charge for searches of less

than 1 hour.

(2) Duplication of records. Records will be duplicated at a rate of \$.25 per page for all coyping of 4 pages or more. There is no charge for duplicating documents of 3 or less pages or the first three pages of documents of greater length.

(3) Other. When no specific fee has been established for a service, or the request for a service does not fall under one of the above categories due to the amount or type thereof, the Staff Secretary is authorized to establish an appropriate fee based on "direct costs" as provided in the Freedom of Information Act. Examples of services covered by this provision include searches involving computer time or special travel, transportation, or communications costs.

(b) If records requested under the Act are stored elsewhere than the headquarters of the National Security Council Staff at Washington, D.C., the special costs of returning such records to the headquarters for review will be added to the search costs. Search costs are due and payable even if the record which was requested cannot be located after all reasonable efforts have been made, or if the NSC Staff determines that a record which has been requested, but which is exempt from disclosure under the Act, is to be withheld. Processing of a request for records will not be undertaken until a requestor has paid in full for search and duplication charges for any previous document request under the Act.

(c) Where it is anticipated that the fees chargeable under this section will amount to more than \$25, and the requestor has not indicated in advance his willingness to pay fees as high as are anticipated, the requestor shall be promptly notified of the amount of the anticipated fee or such portion thereof as can readily be estimated. In instances where the estimated fees will greatly exceed \$25, an. advance deposit may be required. The notice or request for an advance deposit shall extend an offer to the requestor to consult with knowledgeable NSC Staff personnel-designated by the Staff Secretary—in an attempt to reformulate the request in a manner which will reduce the fees and meet the needs of the requestor. Dispatch of such a notice or request shall suspend the running of the period for response by the NSC Staff until a reply is received from the requestor. (d) Fees must be paid in full prior to

#### § 2101.22 Fee payments.

issuance of requested copies.

(a) Remittances shall be in the form either of a personal check or bank draft drawn on a bank in the United States, or a postal money order. Remittances shall be made payable to the Treasury of the United States and mailed to the Staff Secretary, National Security Council, Washington, D.C. 20506.

(b) A receipt for fees paid will be given only upon request. Refund of fees paid for services actually rendered will not

be made.

(c) The Staff Secretary, National Security Council, may in accordance with the Freedom of Information Act, as amended, waive all or part of any fee provided for in this section when it is deemed to be in either the interest of the NSC Staff or of the general public.

#### SUBPART D-REPORTS

# § 2101.31 Quarterly Index of Publica-

(a) The NSC Staff is required to publish and distribute to the public on a quarterly basis an index of the following material promulgated after July 4, 1967:

(1) Final opinions, including concurring and dissenting opinions, as well as orders, made in the adjudication of cases:

(2) Those statements of policy and interpretations which have been adopted and are not published in the FEDERAL REGISTER; and

(3) Administrative staff manuals and instructions to staff that affect a member of the public.

(b) A final order, opinion, statement of policy, interpretation, or staff manual or instruction that affects a member of the public may be relied on, used, or cited as precedent against a party other than an agency only if:

(1) It has been indexed and either made available or published as provided

by the Act: or

(2) The party has actual and timely notice of the terms thereof.

(c) The legislative history of this requirement indicated that only material having "precedential significance" are to be included in the Index.

(d) Matters specifically excluded from release under the Freedom of Information Act are not required to be included in the Index. In the case of the NSC Staff, this particularly includes matters that are:

(1) Specifically authorized under criteria established by an Executive Order to be kept secret in the interest of national defense or foreign policy and are in fact properly classified pursuant to such Executive Order (see section E, below):

(2) Related solely to the internal personnel rules and practices of an agency; (3) Specifically exempted from disclo-

sure by statute (see Part VII, below); (4) Inter-agency or intra-agency memorandums or letters which would not be available by law to a party other than an agency in litigation with the

(5) Personnel and medical files and similar files the disclosure of which would constitute a clearly unwarranted

invasion of personal privacy.

agency:

(e) Because of the considerations outlined under sections 3. and 4., above, it is anticipated that the NSC Staff will normally have little or nothing to report in its quarterly index.

#### 2101.32 Annual Report to Congress.

(a) On or before March 1st of each calendar year, a report of the NSC Staff's activities over the preceding calendar year relating to the Freedom of Information Act will be submitted to the Speaker of the House of Representatives and the President of the Senate.

(b) The above report will include:

(1) The number of determinations made by the NSC Staff not to comply with requests for records made to it under the Act and the reasons for each such determination;

(2) The number of appeals made by persons under the Act, the results of such appeals, and the reason for the action by the NSC Staff upon each appeal that results in a denial of information;

(3) The names and titles or positions of each person responsible for the denial of records requested under the Act, and the number of instances of participation for each:

(4) The results of each (Civil Service Commission) proceeding conducted pursuant to the Act, including a report of the disciplinary action taken against the officer or employee who was primarily responsible for improperly withholding rec-

ords or an explanation of why disciplinary action was not taken;

(5) A copy of every rule made by the NSC Staff regarding the Act;

(6) A copy of the fee schedule and the total amount of fees collected by the NSC Staff for making records available under the Act; and

(7) Such other information as indicates efforts by the NSC Staff to administer fully the Act. (This should include. to the extent possible, data on the costs to the NSC Staff of administering the

Act.)

(c) The NSC Staff, based in part on the information compiled for its annual report to Congress will provide-upon request-assistance to the Department of Justice in the preparation of its annual report (also due each March 1st) to Congress concerning judicial cases arising under the provisions of the Act.

#### SUBPART E-APPLICABLE MATERIAL

### § 2101.41 Primary review by NSC staff.

(a) The NSC Staff will have primary and authoritative review responsibility for Freedom of Information Act requests for the following types of classified material.

(1) Material originally classified by a properly authorized member of the NSC Staff and maintained under its control;

(2) Material produced for and processed by the NSC Staff even if not originally classified by an authorized member of the Staff:

(3) Material classified by a President, his White House Staff, the NSC Staff on his behalf, or special committee or commission appointed by him and which the Archivist of the United States has in his custody at any archival depository, in-cluding a Presidential Library, and which is concerned with "national defense or foreign policy" matters of "pri-mary subject-matter interest" to the NSC Staff:

(4) Verbatim or extensively paraphrased NSC material appearing in documents originally classified by another

agency.

(b) Requests received by other agencies for the material described above should be forwarded to the NSC Staff for action.

#### § 2101.42 Recommendations to other agencies.

The NSC Staff will, at the request of another agency, make recommendations on the release of material concerning "national defense or foreign policy" originally classified by another agency but which is of significant subjectmatter interest of the NSC Staff.

#### SUBPART F-DECLASSIFICATION CRITERIA

#### § 2101.51 Criteria for denying a request for release of classified material.

(a) The amended Freedom of Information Act does not apply to matters that are:

(1) Specifically authorized under criteria established by an Executive Order (E.O. 11652) to be kept secret in the interest of national defense or foreign

(2) Are in fact properly classified pursuant to (both procedural and substantive criteria contained in) such Executive Order;

(3) Or, are specifically exempted from disclosure by statute.

§ 2101.52 Procedural Criteria under E.O. 11652.

(a) The material must have been classified by a properly authorized indi-

vidual; (b) The material must have been properly marked. For documents prepared after the effective date of Executive Order 11652 (June 1, 1972), this must especially include its classification, whether it is subject to or exempt from the General Declassification Schedule. and the highest level individual that authorized classification in each case. Documents prepared prior to June 1, 1972, must have been properly marked pursuant to the rules of the appropriate Executive Order then applicable.

§ 2101.53 Substantive criteria under E.O. 11652 for material under ten vears old.

(a) Material must be classified "solely on the basis of national security considerations. In no case (may material be classified) in order to conceal inefficiency or administrative error, to prevent embarrassment to a person or Department, to restrain competition or independent initiative, or to prevent for any other reason the release of information which does not require protection in the interest of national security."

(b) To be classified "Top Secret," it must reasonably be expected that unauthorized disclosure of the material would "cause exceptionally grave damage to the national security." Examples are:

(1) Armed hostilities against the United States or its allies;

(2) Disruption of foreign relations vitally affecting the national security;

(3) The compromise of vital national defense plans or complex cryptologic and communications intelligence systems;

(4) The revelation of sensitive intelligence operations; and

(5) The disclosure of scientific or technological developments vital to the na-

tional security.
(c) To be classified "Secret," it must reasonably be expected that unauthorized disclosure of the material would "cause serious damage to the national security." Examples are:

(1) Disruption of foreign relations significantly affecting the national security:

(2) Significant impairment of a program or policy directly related to the national security;

(3) Revelation of significant military plans or intelligence operations; and

(4) Compromise of significant scientific or technological developments relating to national security.

(d) To be classified "Confidential," it

must be reasonably expected that unau-

thorized disclosure of the material would "cause damage to the national security."

(e) Examples of material exempt from release because it is classified pursuant to a statute are:

(1) Restricted Data (42 U.S.C. 2162);

(2) Communication information (18 U.S.C. 798);

(3) Material relating to intelligence sources and methods (50 U.S.C. 403(d) (3) and (9)).

§ 2101.54 Substantive criteria under E.O. 11652 for material over ten years old.

(a) Classified information or material furnished by foreign governments or in-ternational organizations and held by the United States on the understanding that it be kept in confidence;

(b) Classified information or material specifically covered by statute, or pertaining to cryptography, or disclosing in-

telligence sources or methods;

(c) Classified information or material disclosing a system, plan, installation, project or specific foreign relations matter the continuing protection of which is essential to the national security;

(d) Classified information or material the disclosure of which would place a person in immediate jeopardy.

## Subpart G-Partial Release

§ 2101.61 Release of "Reasonably seg-regable portion" of requested classified material.

(a) The amended Freedom of Information Act requires that "any reasonably segregable portion of a record shall be provided . . . after deletion of the portions which are exempt."

(b) Determination of which portion(s) of classified material are to be released, and what portion(s) are to remain classifled, will be made on the basis of the criteria outlined in Subpart F.

§ 2101.62 Downgrading of classified material reviewed under the FOI Act.

(a) Classified material reviewed for release under the Freedom of Information Act which no longer meets the criteria established for its original classification should be recommended for downgrading to a lower classification if appropriate.

(b) Downgraded material which retains a classification will be exempted from release.

Effective date. The amended Freedom of Information Act is effective February 19, 1975. This regulation shall become effective on that date. It is subject to revision upon review by the Interagency Classification Review Committee.

JOHN H. MURPHY. Chief, Information Support Systems, Secretariat, National Security Council.

[FR Doc.74-4468 Filed 2-18-75;2:39 pm]

Title 22-Foreign Relations PART 706-PUBLIC INFORMATION CHAPTER VII—OVERSEAS PRIVATE INVESTMENT CORPORATION SUBCHAPTER A-ADMINISTRATIVE PROVISIONS

#### Freedom of Information Act Regulations

Pursuant to the authority delegated to me by the President of the Overseas Private Investment Corporation (the "Corporation") under Redelegation of Authority No. A-71-5, December 26, 1973, subchapter A of Title 22, Chapter VII is hereby amended as follows:

#### PARTS 701-704. [RESERVED]

1. By reserving Parts 701 through 704.

#### PART 705 and 730—[REDESIGNATED]

2. By redesignating Part 730, Employee Responsibilities and Conduct, as Part 705 and thereby renumbering §§ 730.735–101 through 730.735-114 as \$\$ 705.735-101 through 705.735-114; and

#### PART 706-PUBLIC INFORMATION

3. By adding a new Part 706, Public Information, which shall be effective upon the date hereof. Although prior notice and comment with respect to the adoption of part 706 is not required under the Administrative Procedure Act (5 U.S.C. 553), interested persons may submit written comments, suggestions, data or arguments to the Vice President for Public Affairs, Overseas Private Investment Corporation, Washington, D.C. 20527, on or before April 7, 1975. Material thus submitted will be evaluated and acted upon in the same manner as if this document were a proposal. Until such time as further changes are made, however, part 706, Subchapter A, as set forth herein shall remain in effect, thus permitting the public business to proceed more expeditiously.

Part 706, Subchapter A, shall read as follows:

> Subpart A-[Reserved] Subpart B-[Reserved] Subpart C-Procedures and Fees

706.31 Requests for inspection of records. 706.32 Fees.

AUTHORITY: 5 U.S.C. 552, as amended by Pub. L. 93-502.

#### Subpart C-Procedures and Fees

§ 706.31 Requests for inspection of records.

(a) Access by any member of the public to the Corporation's public information files, or to the index thereto, which are maintained in accord with the provisions of 5 U.S.C. 552(a) (2), shall be granted by appointment only. Persons desiring such access may request an appointment either by reporting in person to the receptionist at the principal offices of the Corporation between 8:45 a.m. and 5:30 p.m., except Saturdays, Sundays and official holidays, by telephoning the Office of Public Affairs of the Corporation at (202) 632-1854 during such hours or by writing the Corporation to the attention of the Vice President of Public Affairs. Although the Corporation will endeavor to grant an appointment at the time requested, some delay may be required at times because of the small size of the Corporation's staff. Persons desiring access to the public information files should seek an appointment at least twenty-four (24) hours in advance.

(b) Access to any records of the Corporation not otherwise made available by the Corporation to the public under the provisions of 5 U.S.C. 552(a) (1) and (2), or the duplication of such records by the Corporation, shall be granted only upon specific written request to the Corporation, addressed to the attention of the Vice President for Public Affairs, which shall be deemed not to have been received by the Corporation until actual receipt thereof by the Corporation's Office of Public Affairs. Such request shall to the maximum extent required under the law, accurately describe the records to which access to or duplication of is requested such as, by way of example only, the subject matter, format, date, and where pertinent, the country, project or person involved. Any such request which does not describe such records sufficiently enough to permit the staff of the Corporation to promptly locate such records shall be returned to the requester.

#### § 706.32 Fees.

(a) The following schedule of fees representing direct costs shall be charged for services rendered by the Corporation under 5 U.S.C. 552(a) in furnishing information to members of the public:

(1) Searching for records, etc., \$15.00 per hour per person.

(2) Other facilitative assistance, \$15.00 per hour per person.

(3) Photocopy duplication, .20 per page. Certification of a true copy, \$3.50. (5) Certified statement of negative results of a search, \$3.50.

(b) Notwithstanding the provisions of paragraph (a) of this section, the Vice President for Public Affairs may waive or reduce any fee otherwise due under paragraph (a) of this section for the preparation, release or distribution of any document, instrument, publication or class thereof, for which he determines that waiver or reduction of such fee is in the public interest because furnishing the information can be considered as primarily benefiting the general public.

Issued in Washington, D.C. on February 14, 1975.

DAVID GREGG III. Executive Vice President.

[FR Doc.75-4637 Filed 2-18-75;8:45 am]

#### Title 10-Energy

# CHAPTER III—ENERGY RESEARCH AND DEVELOPMENT ADMINISTRATION

#### PART 709-PUBLIC RECORDS

The Energy Research and Develop-Administration (ERDA) established by the Energy Reorganization Act of 1974, Pub. L. 93-438, 88 Stat. 1233, and the Act was made effective on January 19, 1975 by Executive Order 11834 dated January 15, 1975, which Executive Order was published on January 17, 1975, at 40 Fed. Reg. 2971.

Consistent with the establishment of ERDA, a new Chapter III of Title 10, Code of Federal Regulations, is established consisting at this time of Part 709, Public Records, which is set forth below.

On an interim basis, the rules and regulations of the Atomic Energy Commission contained in Chapter I of Title 10, Code of Federal Regulations (Parts 0 through 170 of Chapter I of Title 10, CFR, respectively), except Part 9 thereof, approved as of the effective date of the reorganization (January 19, 1975) including any proposed rules thereunder, are applicable to ERDA to the extent they are not inconsistent with applicable law, for all ERDA activities under the Energy Reorganization Act of 1974, the Federal Nonnuclear Energy Research and Development Act of 1974 (Pub. L. 93-577), and other applicable law.

Pursuant to the authority vested in me by section 105(a) of the Energy Reorganization Act of 1974, and pursuant to the provisions of section 552 of Title 5, United States Code, as amended by Pub. L. 93-502 (the Freedom of Information Act, as amended), Part 709 of Chapter III, Title 10, Code of Federal Regulations, which Part is entitled "Public Records" and implements for ERDA the Freedom of Information Act, including the 1974 amendments thereto is hereby issued.

Except as provided below, Part 709 is a statement of ERDA policies and procedures for implementing the Freedom of Information Act, as amended, and the relevant provisions of 5 U.S.C. 553 requiring notice of proposed rule making, opportunity for public participation and delay in effective date, are inapplicable.

On page 2714 of the FEDERAL REGISTER of January 15, 1975, there was published a notice of proposed rule making regarding a proposed uniform schedule of fees for document search and duplication. No comments have been received in response to that notice and no changes have been made to the charges proposed for searching and duplicating. These charges for document search and duplication are set forth in § 709.12 below.

While these regulations are effective February 19, 1975, the public is invited to submit comments thereon, provided they are received within 90 days of publication of these regulations. Any com-ments submitted should be in writing and submitted to the Office of the Gen-

eral Counsel, Enery Research and Development Administration.

Effective date: February 19, 1975.

ROBERT C. SEAMANS, Jr. Administrator.

709.1 Applicability. Policy. Definitions. 709.2 709.3 709.4 Statement of organization, functions, procedure, and substantive rules, final opinions, statement of policy, staff manuals and instructions.

709.5 Freedom of information officers. 709.6

Requesting records. Handling of FOI requests. 709.7 709.8 Denial of records.

Matters exempt from disclosure. 709.9 709.10 Administrative appeal of initial denial of records. 709.11

Requests for classified records. Fees for search and duplication of records.

709.13 Matters in litigation. 709.14 Annual report.

AUTHORITY: 5 U.S.C. 552 and sec. 105(a), Pub. L. 93-438.

## § 709.1 Applicability.

(a) This part implements section 552 of Title 5, United States Code, and prescribes rules governing the availability to the public of the records of the Energy Research and Development Administration (ERDA).

(b) This part only applies to requests which (1) reasonably describe records and (2) are made in accordance with

this part.

(c) Under section 552 of Title 5 there is no obligation to compile or create a record solely for the purpose of satisfying a request for records.

(d) Nothing in this part shall inhibit the dissemination of scientific, technical, or other information in accordance with established ERDA policies or programs for such dissemination.

(e) Except where specifically noted otherwise, this part applies to all records whether they predate or postdate

July 4, 1967.

(f) This part supersedes any other ERDA policy, rule, or regulation to the extent inconsistent with this part.

(g) This part shall be interpreted so as to be consistent with the "Privacy Act of 1974," Pub. L. 93-579.

#### § 709.2 Policy.

(a) In implementing section 552 of Title 5, United States Code, it is the policy of ERDA to make records available to the public to the greatest extent possible in keeping with the spirit of that section. Therefore, all records of ERDA which are requested in accordance with this part shall be made available promptly, except those that ERDA specifically determines in accordance with this part (1) are exempt from disclosure under \$ 709.9, and (2) that dislosure is contrary to the public interest or will adversely affect the rights of any person.

(b) If records requested contain material exempt from disclosure, any reasonably segregable portion of a record shall be provided to a person requesting such record after deletion of the portions which are exempt from disclosure. However, the entire record will be exempt where exempt material in that record is inextricably intertwined with nonexempt material.

(c) Requests made under this part shall be considered without regard to any showing of need.

#### \$ 709.3 Definitions.

As used in this part:
(a) "ERDA" means the Energy Research and Development Administration established by the Energy Reorganization Act of 1974 (Pub. L. 93-438).

(b) "ERDA personnel" means employees, consultants, and members of advisory boards, committees and panels of ERDA; members of boards designated by the Administrator, Deputy Administra-tor, or other designee of the Administrator or Deputy Administrator to preside at adjudicatory proceedings; and officers or employees of Government agencies, including military personnel, assigned to duty at ERDA.

(c) "Administrator" means the Administrator of Energy Research and Development provided for in section 102(a) of the Energy Reorganization Act of

1974.

(d) "Deputy Administrator" means the Deputy Administrator provided for in section 102(b) of the Energy Reorgani-

zation Act of 1974.

(e) "Assistant Administrator" means an Assistant Administrator provided for in section 102(d) of the Energy Reorganization Act of 1974 or such other Assistant Administrator as the Administrator may appoint under section 102(f) of the Energy Reorganization Act of 1974.

(f) "Government" means the United

States of America.
(g) "Agency" is defined as provided in section 551(1) of Title 5, United States Code, and includes any executive department, military department, Government corporation, Government-controlled corporation, or other establishment in the executive branch of the Government (including the Executive Office of the President), or any independent regula-

tory agency.

(h) "Record" includes any book, paper, map, photograph, brochure, punch card, magnetic tape, paper tape, sound recording, pamphlet, slide, motion picture, or other documentary material regardless of form or characteristics, which evidences ERDA organization, functions, policies, decisions, procedures. operations, programs or other activities, and which was made by, in the posses sion of, or under the control of ERDA pursuant to ERDA contracts, or Federal law or in connection with the transaction of public business. "Records" do not include objects or articles such as structures, furniture, tangible objects or models, or vehicles and equipment, nor

formulas, designs, and the like which are significant not as records but as items of valuable property.

"Freedom of Information Officer" has the meaning as set forth in § 709.5

(i) Except as otherwise provided in 709.11(c), "Authorizing Official" means the Assistant Administrator for International Affairs, the Assistant Administrator for Planning and Analysis, and Directors of Offices and Divisions in ERDA Headquarters, and Directors of Energy Research Centers and Managers of Operations Offices for ERDA field organizations, who are authorized to make the initial determination for ERDA to grant or deny, in whole or in part, requests for records under this part. The officials designated above may in their discretion redelegate the above authority without power or further redelega-

(k) Except as otherwise provided in § 709.11(e) "Freedom of Information (FOI) Appeal Panel" means a panel consisting of the following three officials: (1) The Assistant Administrator for Administration or his designee, who shall be Chairman of the panel, (2) the Director of the Office of Public Affairs, or his designee, and (3) such other ERDA official as the Chairman shall designate to consider a specific appeal. The appointment of the third panel member shall be made with due regard to the particular records involved. It is contemplated that such panel members shall be Assistant Administrators, Deputy Assistant Administrators, or other high ranking ERDA officials in keeping with the policy stated in § 709.2. No member of the FOI Appeal Panel shall have been involved in the determination from which the appeal under consideration is being taken.

§ 709.4 Statement of organization, functions, procedure, and substantive rules, final opinions, statement of policy, staff manuals and instructions.

(a) As required by section 552, Title 5, United States Code, a statement as to the sources of information concerning the following subjects may be found in § 701 of this chapter: (1) A description of the central and field organization of ERDA and the established places at which, the offices from whom, and the methods whereby, the public may secure information, make submittals or requests, or obtain decisions, except for Freedom of Information requests which requests shall be made in accordance with the procedures set forth in § 709.6 of this part; (b) statements of the general course and method by which ERDA functions are channeled and determined, including the nature and requirements of all formal and informal procedures available; (c) ERDA rules of procedure, descriptions of forms available or the places at which forms may be obtained, and instructions as to the scope and contents of all papers, reports, or examinations; (d) ERDA substantive rules of general applicability adopted as authorized by law, and state-

ments of general policy or interpretations of general applicability formulated and adopted by ERDA; and (e) every amendment, revision, or repeal of the foregoing.

(b) As required by section 552, Title 5, United States Code, the following records are made available for public inspection and copying at the ERDA Public Document Room at 1717 H Street NW., Washington, D.C.:

(1) All final opinions (including concurring and dissenting opinions) and all orders in the adjudication of cases.

(2) Statements of policy and interpretations which have been adopted by ERDA and are not published in the FED-ERAL REGISTER

(3) ERDA rules and regulations.

(4) ERDA Manuals and instructions to ERDA personnel that affect any member of the public.

(c) The records specified in § 709.4(b) (3) and (4) above are also made available for public inspection and copying at any one of the ERDA field organizations identified in § 701 of this chapter.

(d) The ERDA Public Document Room will be open between 8:30 a.m. and 5:15 p.m. on Mondays through Fridays. The Public Document Room at the ERDA field organizations referenced above will be open during regular office hours on

Mondays through Fridays.

(e) Indexes (or supplements thereto) of those records specified in § 709.4(b) (1), (2) and (4) of this section, and issued, adopted, or promulgated after July 4, 1967, will be published and made available for sale to the public at no more than quarterly intervals, unless it is determined by the Administrator or Deputy Administrator, by order published in the FEDERAL REGISTER, that such publication would be unnecessary and impracticable, in which case there shall be provided copies of such index on request at a cost not to exceed the direct cost of duplication. Inclusion of a record in the index does not constitute a determination that such record is required by law to be indexed.

#### § 709.5 Freedom of information officer.

(a) There is appointed at ERDA Headquarters, Washington, D.C. 20545 a Freedom of Information (FOI) Officer whose duties shall hereinafter be de-

(b) There shall be designated by each Director of an Energy Research Center and Manager of an Operations Office for each of the following ERDA field organizations a Freedom of Information (FOI) Officer whose duties shall hereinafter be described:

(1) Bartlesville Energy Research Center, ERDA P.O. Box 1398 Bartlesville, Oklahoma 74003

(2) Grand Forks Energy Research Center, P.O. Box 8213, University Station

Grand Forks, North Dakota 58201 Laramie Research Center, ERDA P.O. Bex 3395, University Station Laramie, Wyoming 82070

- (4). Morgantown Research Center, ERDA P.O. Box 830
- Morgantown, West Virginia 26505 (5) Pittsburgh Energy Research Center, ERDA 4800 Forbes Avenue

Pittsburgh, Pennsylvania 15213

- (6) San Francisco Energy Research Center, ERDA 1429 Appraisers Building 630 Sansome Street San Francisco, California 94111
  - Albuquerque Operations Office, ERDA P.O. Box 5400 Albuquerque, New Mexico 87115 [Nore.—This office has cognizance over the following area offices: Amarillo,

Burlington, Dayton, Kansas City, Los Alamos, Pinellas, Rocky Flats, and Sandia.]

(8) Chicago Operations Office, ERDA 9800 South Cass Avenue Argonne, Illinois 60439

> INOTE -This office has cognizance over the Batavia and Brookhaven area offices and the New Brunswick laboratory.]

(9) Idaho Operations Office, ERDA 550 2nd St.

Idaho Falls, Idaho 83401 Nevada Operations Office, ERDA P.O. Box 14100

Las Vegas, Nevada 89114 (11) Oak Ridge Operations Office, ERDA P.O. Box E

Oak Ridge, Tennessee 37830 (12) Richland Operations Office, ERDA P.O. Box 550

Richland, Washington 99352 San Francisco Operations Office, ERDA 1333 Broadway, Wells Fargo Building Oakland, California 94616

(14) Savannah River Operations Office, ERDA P.O. Box "A" Aiken, South Carolina 29801

(c) The FOI Officers shall make no determinations to deny requests received under this part but shall receive and log each request by time and date, shall ascertain where the documents requested are located, shall acknowledge receipt of requests by letter to the requester, shall refer the request to the proper Authorizing Official and shall make such reports and perform such other functions as are hereinafter set forth or otherwise delegated.

### § 709.6 Requesting records.

(a) A request need not be in any particular format, but it (1) must be addressed to any one of the FOI Officers listed in § 709.5 above, (2) must be in writing, (3) must be clearly identified both on the envelope and in the letter as a Freedom of Information Act or FOIA request, (4) must reasonably describe the records sought to permit identification, and (5) must state that the requester promptly will pay the fees chargeable under § 709.12.

(b) Because of the strict statutory time limits it would be helpful if the request could be addressed to the FOI Office where the desired records are located or are believed to be located, or to the FOI Officer who is in the office which has, or it is believed has, cognizance over the records. However, it is not the responsibility of the requester to ascertain the location of the records requested.

§ 709.7 Handling of FOI requests.

(a) Promptly upon receipt of a request made in accordance with \$709.6 above, the FOI Officer shall log the date and time of receipt, ascertain the location of the records requested and the identity of the appropriate Authorizing Official, immediately dispatch the request to such Authorizing Official by the fastest available means advising him of the date and time of receipt and when the ten working days permitted for a determination will expire, and acknowledge receipt in writing of the request stating the time and date of receipt and the appropriate Authorizing Official to whom the request has been directed.

(b) Any request not made in accordance with § 709.6 shall be promptly brought to the attention of a FOI Officer by the ERDA Official or employee receiving such a request, or the requester shall be referred to the FOI Officer, as appropriate. The FOI Officer shall assist the requester in making the request in accordance with \$ 709.6.

(c) All Freedom of Information requests properly made in accordance with § 709.6 shall be deemed to have been received when date and time stamped by a FOI Officer.

(d) If a request can be immediately complied with in accordance with this part, it should be.

(e) The Authorizing Official shall advise the requester in writing within ten working days of the receipt of the request by the FOI Officer (see paragraph (c) of this section) whether the request is to be complied with. A copy of the letter to the requester shall be sent to the appropriate FOI Officer. Where a determination is made to comply with a request in whole or in part, the records shall be made available to the requester promptly in accordance with this part, subject to the requirement for payment under § 709.12.

(f) (1) The Freedom of Information Act, as amended, section 552(a)(6)(B), Title 5, United States Code, provides

In unusual circumstances . . . the time limits prescribed . . . [the 10-day initial determination limit or the 20-day administrative appeal limit] may be extended by written notice to the person making such request setting forth the reasons for such extension and the date on which a determination is expected to be dispatched. No such notice shall specify a date that would result in an extension for more than ten working days. '[U]nusual circumstances' means, but only to the extent reasonably necessary to proper processing of the particular

(i) the need to search for and collect the requested records from field facilities or other establishments that are separate from the office processing the request;

(ii) the need to search for, collect, and appropriately examine a voluminous amount of separate and distinct records which are demanded in a single request; or

(iii) the need for consultation, which shall be conducted with all practicable speed, with another agency having a substantial interest in the determination of the request or among two or more components of the

agency having substantial subject-matter interest therein.

(2) The ten working day extension provided for in unusual circumstances permits more than one extension, either divided between the initial and appeal stages (see § 709.10 below concerning appeal procedures) or within a single stage. so long as the total extended time does not exceed ten working days with respect to a particular request.

(3) The Authorizing Official has the discretion unilaterally to invoke any or all of the ten working day extension, if the unusual circumstances exist and if the additional time is considered as reasonably necessary to the proper processing of the particular request. This discretionary authority can only be invoked after consultation with counsel.

(4) The requester must be promptly notified in writing of the extension, setting forth the reasons of the extension, and the date on which a determination is expected to be dispatched. No such notice shall specify a date that would result in an extension for more than ten working days, over and above the ten working days allowed for the initial determination.

(5) The FOI Officer shall be advised by copy of a letter to the requester or otherwise of any extension invoked.

(g) The Authorizing Official or FOI Officer, shall consult with the appropriate agency when records are requested. which are in the possession of ERDA, but which were originated by another agency. Unless the other agency agrees that the document may be released by ERDA, the request shall be forwarded to the other agency for their disposition. The requester shall be advised in writing when a request is referred to another agency (see § 709.11 regarding classifled documents or information originated by another agency).

(h) Nothing in this part shall preclude the Authorizing Official and a requester from agreeing to an extension of time for the initial determination on a request. Any such agreement shall be confirmed in writing and shall clearly specify the total time agreed upon for the initial determination.

#### 709.8 Denial of records.

(a) Any record, or any portion of a record may be denied to a requester by the Authorizing Official only if he determines it contains matters specified in § 709.9 below and further determines that disclosure of such matters is contrary to the public interest or will adversely affect the rights of any person. The Authorizing Official shall make such determinations only after consultation with counsel. A subsequent request for identical records previously denied shall be considered on its own merits.

(b) Any reasonably segregable portion of a record shall be provided after deletion of the portions exempt under § 709.9 below. Those portions of documents which are deleted shall be identified in an appropriate manner, e.g., by placing

the word "DELETED" in the space where material is deleted.

(c) (1) If the Authorizing Official determines that a request will be denied in whole or in part, a written notice of the determination shall be promptly dispatched to the requester within the appropriate time period set forth in § 709.7 above. Such notice shall be sent by certified or registered mail, return receipt requested. Such notice shall appropriately advise the requester of the reasons for denying the request or any portion thereof and that the Authorizing Official signing the determination is the person responsible for that determination. If other Authorizing Officials, or appropriate officials of other agencies are responsible for denying any portion of a record requested, their names and titles or positions shall be listed in the notice of denial and it shall be clearly indicated what portion or portions they were responsible for denying.

(2) The reasons for denying the request shall where possible be more than a recitation of the statutory exemptions listed in § 709.9 below. Where material has been deleted from a record furnished, there shall also be furnished a brief explanation of why the material was deleted (each deletion may be coded as to what exemption is being used).

(3) The notification of the denial determination shall also advise the requester of the right to appeal to the FOI Appeal Panel as provided in § 709.10 below. A copy of the denial determination should be provided to the appropriate FOI Officer.

### § 709.9 Matters exempt from disclosure.

(a) The following items are the matters exempt from disclosure under the Freedom of Information Act, section 552 (b) of Title 5, United States Code, with the statutory language being underlined: (1) (i) Specifically authorized under criteria established by an Executive Order to be kept secret in the interest of national defense or foreign policy and (ii) are in fact properly classified pursuant to such Executive order; Exemption (1) matters are those matters classified as National Security Information under Executive Order 11652, or other matters authorized to be kept secret under other Executive Orders now or hereafter issued. Restricted Data and Formerly Restricted Data under the Atomic Energy Act of 1954, as amended (42 U.S.C. 2011 et seq.), are covered by this exemption as well as the exemption in (3) below.

(2) Related solely to the internal personnel rules and practices of an agency;
(3) Specifically exempted from disclosure by statute; Restricted Data and

Formerly Restricted Data come within the meaning of this exemption (see (1)

above).

(4) Trade secrets and commercial or financial information obtained from a person and privileged or confidential; Whenever a record is requested which comes within or might come within this exemption, it is necessary that the person whose information it is, be notified immediately to see if there is any ob-

jection to its release. Criteria applicable in making a determination concerning this exemption include, but are not necessarily limited to the following:

(A) Whether the information has been held in confidence by its owner;

(B) Whether the information is of a type customarily held in confidence by its owner and whether there is a reasonable basis therefor;

(C) Whether the information was transmitted to and received by the

agency in confidence; and
(D) Whether the information is avail-

able in public sources.

(ii) This exemption may be invoked for the benefit of the person who has provided the information, even though the agency has no interest in withholding the information.

(iii) Records held in confidence which identify procedures for safeguarding special nuclear material or detailed security measures for the physical protection of a facility or plant comes within this exemption except to the extent that portions thereof come within any other exemption.

(iv) Naval nuclear propulsion information which is held in confidence by ERDA; its contractors or subcontractors is deemed to come within this exemption except to the extent that portions thereof come within any other exemption. This exemption also applies to material obtained in confidence from a foreign source.

(5) Interagency or intragency memorandums or letters which would not be available by law to a party other than an agency in litigation with the agency:

(i) This exemption includes all internal memoranda which would not routinely be disclosed to a party other than a Government agency through the discovery process in litigation with ERDA or the Government, such as internal drafts or memoranda between officials or agencies: opinions, interpretations, and evaluations prepared by staff personnel, contractors or consultants for the use of ERDA; records of the deliberations of ERDA staff groups; staff criteria or guidelines used for auditing or inspection purposes; and documents or information which ERDA has received or generated before it completes the process of awarding a contract or issuing an order, decision, or regulation; ERDA plans or materials that are in process of preparation or development and are likely to be revised before being finalized, such as budget proposals, long range plans or studies, drafts of speeches or statements, or legislative proposals; ERDA plans (such as those included in budget justification material) which, even though finalized, the disclosure of which would be harmful to public or private interests, if made available in advance of the effective date; documents awaiting patent review; and working papers of ERDA attorneys and documents which come within the attorneyclient privilege.

(ii) This exemption is intended to allow the withholding the type of records indicated above to the extent they

reflect deliberative or policy making decisional processes, including advice, opinions, or recommendations that are part of ERDA's deliberative or policy making decisional processes.

(iii) While the policy set forth under \$709.2(b) applies to the exemption, where materials reflecting deliberative or policy making processes are inextricably intertwined with factual material in a particular record, the entire record may be withheld.

(iv) The withholding of records, or portions thereof, under this exemption is discretionary on the part of the Authorizing Official, and whether such exemption should be invoked is to be considered on a case-by-case basis in accordance with the policy set forth in § 709.2 above.

(6) Personnel and medical files and similar files the disclosure of which would constitute a clearly unwarranted invasion

of personal privacy;

(7) Investigatory records compiled for law enforcement purposes, but only to the extent that the production of such records would (i) interfere with enforcement proceedings, (ii) deprive a person of a right to a fair trial or an impartial adjudication, (iii) constitute an unwarranted invasion of personal privacy, (iv) disclose the identity of a confidential source, and, in the case of a record compiled by a criminal law enforcement authority in the course of a criminal investigation, or by an agency conducting a lawful national security intelligence investigation, confidential information furnished only by the confidential source, (v) disclose investigative techniques and procedures, or (vi) endanger the life or physical safety of law enforcement per-

(i) The scope of this exception against disclosure of "investigative techniques and procedures" shall not be interpreted to include routine techniques and procedures already well known to the public, such as ballistics tests, fingerprinting, and other scientific tests or commonly known techniques.

(ii) This exemption shall not be interpreted to include records falling within the scope of subsection 552(a) (2) of Title 5, United States Code, (see § 709.4 above) such as administrative staff manuals and instructions to staff that affect a member

of the public.

(iii) "Confidential source" in (7) (d) includes the identity of a person other than a paid informer who may be protected if the person provided information under an express assurance of confidentiality or in circumstances from which such an assurance could be reasonably inferred. In every case where the investigatory records sought were compiled for law enforcement purposes—either civil or criminal in nature—the agency can withhold the names, addresses, and other information that would reveal the identity of a confidential source who furnished the information. However, where the records are compiled by a criminal law enforcement authority, all of the information furnished only by a confidential source may be withheld if the information was compiled in the course of a criminal investigation. In addition, where the records are compiled by an agency conducting a lawful national security intelligence investigation, all of the information furnished only by a confidential source may also be withheld. The term "criminal law enforcement authority" is to be narrowly construed to include the Federal Bureau of Investigation and similar investigative authorities. Likewise, "national security" is to be strictly construed to refer to military security, national defense, or foreign policy. The term "intelligence" used in (7) (d) is intended to apply to positive intelligencegathering activities, counter-intelligence activities, and background security investigations by governmental units which have authority to conduct such functions. The term "an agency" is intended to include criminal law enforcement authorities as well as other agencies. Personnel, regulatory, and civil enforcement investigations are intended to be covered by the first clause authorizing withholding of information that would reveal the identity of a confidential source but are not encompassed by the second clause authorizing withholding of all confidential information under the specified circumstances.

(iv) Disclosure of information about a person to that person does not constitute

an invasion of his privacy.

(8) Contained in or related to examination, operating, or condition reports prepared by, on behalf of, or for the use of an agency responsible for the regulation or supervision of financial institutions; or

(9) Geological and geophysical information and data, including maps,

concerning wells.

By memorandum dated February 1975, the Attorney General of the United States has provided agency guidance with respect to the amendments to the Freedom of Information Act, section 552 of Title 5, United States Code.

§ 709.10 Administrative appeal of initial denial of records.

(a) Any person, whose request for records made under procedures set forth in this part, is denied in whole or in part, as provided in § 709.8, shall have the right to appeal such initial determination of denial to an FOI Appeal Panel.

(b) Appeals shall be made by written notice mailed to the Freedom of Information Officer, ERDA Headquarters, and shall be filed within sixty days after receipt of an initial denial letter by the person requesting the records. Both the letter and envelope must clearly identify that a Freedom of Information appeal is being made. The sixty-day time limit may be waived by the FOI Appeal Panel for good cause shown.

(c) Notices of appeal received by ERDA shall be processed in a manner similar to request for records under § 709.7 including mutual agreements which may be made with the requester by the FOI Officer at ERDA Headquarters or the Chairman of the FOI Appeal Panel to extend the twenty-day

period set forth in (e) below. Appeals shall be deemed to have been received when date and time stamped by the FOI Officer at ERDA Headquarters.

(d) Upon receipt of an appeal, the FOI Officer, ERDA Headquarters, will promptly advise the Chairman of the FOI Appeal Panel who will then promptly constitute the Panel as provided in § 709.3(k). The FOI Officer shall promptly notify the Authorizing Official who was responsible for the original denial and arrange to have the appropriate records transmitted to ERDA Headquarters as may be necessary to consider the appeal.

(e) The FOI Appeal Panel shall have twenty working days from date of receipt of the appeal by the FOI Officer to make a final decision on such an appeal.

(f) The FOI Appeal Panel shall consult with counsel prior to making a

final decision.

(g) A decision of the FOI Appeal Panel to uphold in whole or in part an initial decision to deny access to records shall be by a simple majority vote of the Panel members and shall be in writing, and shall make the necessary determinations stating the reasons therefore in the manner provided for in \$ 709.8, as necessary to sustain, modify, qualify, or expand the initial denial.

(h) The written decision of the FOI Appeal Panel upholding in whole or part the initial denial determination shall state that the decision is a final decision of the Administrator. The decision shall also advise the requester of the provisions for judicial review of the decision, as set forth in section 552(a) (4), of Title 5, United States Code, and shall set forth the names and titles or positions of each person responsible for the denial, including the person or persons who made the initial denial determination. The term "responsible persons" is limited to the Authorizing Official who signed the initial notice of denial and other Authorizing Officials or officials of other agencies listed in the initial notice of denial as being responsible for any portion of the initial denial (see § 709.8) and those members of the FOI Appeal Panel who concur in whole or in part with the initial denial.

(i) Decisions of the FOI Appeal Panel shall be final decisions of the

Administrator.

(j) The FOI Appeal Panel may invoke that portion of the 10-day extension period set forth in § 709.7(f), if the "unusual circumstances" as defined therein are applicable. The FOI Officer shall advise the Panel if any of the 10-day extension period had been used at the initial denial stage.

(k) The FOI Officer shall take all appropriate steps to obtain the necessary files, including the initial written denial from the Authorizing Official, for the FOI Appeal Panel to review.

§ 709.11 Requests for classified records.

(a) Requests for classified records including requests made to field organizations shall be subject to the provisions

of this part 709 with the special qualifications noted below.

(b) Any request for records made in accordance with this part, except those requests for access to classified records which are made specifically pursuant to the mandatory review provisions of Excutive Order 11652 shall be automatically considered a Freedom of Information Act request under this part.

(c) The Director, Division of Classification shall be the Authorizing Official for all requests for classified records. In this capacity he shall advise the office originating the records, or who has cognizance or responsibility for the records, of the request and consult with such office or offices prior to making a determination under this section.

(d) The written notice of a determination to deny records, or portions of records, which contain both classified material and other exempt material shall be signed by the appropriate Authorizing Official listed in (c) above. If other Authorizing Officials or appropriate officials of other agencies are responsible for denying any portion of the record, their names and titles or positions shall be listed in the notice of denial and it shall be clearly indicated what portion or portions they were responsible for denying.

(e) The Assistant Administrator for National Security or his designee shall replace the Director of the Office of Public Affairs as a member of the FOI Appeal Panel to the extent the appeal involves records or portions thereof which have been denied because they are classified. Other provisions of § 709.3 (k)

shall remain applicable.

(f) Requests for ERDA records containing classified information received from another agency, or records pre-pared jointly by ERDA and other agencies, will be treated as requests for ERDA records except that coordination will be effected by the ERDA Authorizing Official with the appropriate official of the other agency. Such coordination will be done on an expedited basis, for the purpose of determining whether the agency official wishes to deny the request (as far as it concerns another agency's classified material) and obtaining the other agency denying official's certification, signature, and identity on a record copy of the document being requested. The notice of determination to the requester, in the event part or all of the record is denied by the other agency shall cite the other agency denial offi-cial, as well as the ERDA Authorizing Official if a denial by ERDA is also involved.

(g) If the request specifically cites a document from another agency, the request will be referred directly to the proper official in the other agency and the requester will be so notified of the referral together with a copy of the referral. The notification to the requester should also include the statement that the ten-day time period will start when that request is received by the other

agency.

§ 709.12 Fees for search and duplication of records.

(a) Requests for the duplication of records at the ERDA Public Document Room located in Washington, D.C. will be honored upon payment of the following charges:

(1) Sizes up to  $8\frac{1}{2}$  x 14 inches made on office copying machines—\$0.08 per page copy. Microfiche—paper copy enlargement up to  $8\frac{1}{2}$  inches x 11 inches—

\$0.15 per page copy.

(2) Unless waived as provided below, the charge for duplicating records other than those specified above will L2 computed on the basis of ERDA's direct cost.

(3) Unless waived as provided below, the charges of duplication for requests made in accordance with § 709.6 will be the same as paragraph (a) (1) and (2) of this section.

(4) The cost of searching as provided

in § 709.12(c) below.

(b) Requests for copies of records to be duplicated and furnished by ERDA at all locations, except ERDA's Public Document Room located in Washington, D.C., will be honored upon payment of the following charges unless the charges are waived as provided below:

(1) Sizes up to  $8\frac{1}{2}$  x 14 inches made on office copying machines—\$0.10 per page copy. Larger sizes—\$0.10 for each  $8\frac{1}{2}$  x 14 inch unit or fraction thereof per

page copy.

(2) The charge for duplicating records other than those specified above will be computed on the basis of ERDA's direct cost.

(3) The cost of searching as provided

in § 709.12(c) below.

(c) If a request is for records not already available for public inspection in the Public Document Room the rates for searching are \$5.70 per hour for clerical personnel and \$16.00 per hour for nonclerical personnel. Fractional parts of an hour will be charged on a pro rata basis. When a computer search is necessary in order to fulfill a request, the computer search charge will be the actual direct cost of the computer search.

(d) No records shall be made available to a requester until all charges as provided herein are paid in full. Checks, drafts, or other negotiable instruments shall be made payable to the Energy Research and Development Administra-

tion.

(e) Transcripts by a reporting firm under contract with ERDA, may be purchased directly from the reporting firm or ERDA at the cost of reproduction as provided for in the ERDA contract with the reporting firm.

(f) Material which has been copyrighted will not be reproduced in viola-

tion of the copyright laws.

(g) Documents shall be furnished without charge or at a reduced charge where the Authorizing Official initially considering the FOI request, or the FOI Appeal Panel, determines that waiver or reduction of the fee is in the public interest because furnishing the information can be considered as primarily benefiting

the public interest. Except for records reproduced in an ERDA Public Document Room or by a reporting firm under contract with ERDA or another Government agency, no charge shall be made under this part where the cost of search and duplication of records does not exceed \$10.00.

(h) No charge will be made when no record responsive to a request has been found or when a request is denied in

whole.

(i) Unless the request made under § 709.6 specifically states that whatever cost is involved will be acceptable, or acceptable up to a specified limit that covers anticipated costs, a request will not be deemed to have been received in accordance with § 709.7 until the requester is advised of the estimated costs of searching and duplicating and agrees in writing to bear the cost.

#### § 709.13 Matters in litigation.

(a) Any request for records shall be processed to initial determination or determination on appeal without regard to the pendency of litigation on a request under this part.

(b) When a matter is in litigation at the time the initial determination is made, the Authorizing Official shall file an appeal under § 709.10 on his own motion if the initial determination is to deny in whole or in part any of the

§ 709.14 Annual report.

request.

(a) On or before March 1 of each calendar year, ERDA shall submit a report covering the preceding calendar year to the Speaker of the House of Representatives and President of the Senate for referral to the appropriate committees of the Congress. The report shall include—(1) the number of determinations made by ERDA not to comply with requests for records made to ERDA under this part and the reasons for each such determination;

(2) The number of appeals made by persons under this part, the result of such appeals, and the reason for the action upon each appeal that results

in a denial of information;

(3) The names and titles or positions of each person responsible for the denial of records requested and the number of instances of participation for each; this listing shall include those persons responsible for this initial determination to deny records, or parts thereof as well as those persons responsible for denial on appeal;

(4) The results of each proceeding conducted pursuant to subsection 552(a) (4) (F) of Title 5, United States Code, including a report of the disciplinary action taken against the officer or employee who was 'primarily responsible for improperly withholding records or an explanation of why disciplinary action was not taken:

(5) A copy of every rule made by the agency regarding the Freedom of Infor-

mation Act, section 552, Title 5, United States Code:

(6) A copy of the fee schedule and the total amount of fees collected by the agency for making records available under this part; and

(7) Such other information as indicates efforts to administer fully this part.

(b) The Freedom of Information Officer, ERDA Headquarters, shall prepare the annual report for the Administrator's signature, which shall be transmitted to the respective Houses on or before March 1 of each calendar year.

(c) The Freedom of Information Officers in the field shall forward the necessary information concerning their activities to the Freedom of Information Officer, ERDA Headquarters, within thirty days after the end of the calendar year.

[FR Doc.75-4665 Filed 2-18-75:8:45 am]

Title 18—Conservation of Power and Water Resources

CHAPTER II—TENNESSEE VALLEY
AUTHORITY

PART 301—PROCEDURES
Availability of TVA Records and
Publications

The Freedom of Information Act (5 U.S.C. 552) was recently amended by Pub. L. 93-502. These amendments change the exemptions relating to classified material and investigative records and require agencies to institute certain procedures to ensure prompt availability of nonexempt agency records. Accordingly, the Tennessee Valley Authority hereby revises § 301.1 to reflect these changes.

This revision of § 301.1 makes certain

other changes which include:

 Provision of an agency appellate procedure to be followed in case of an initial determination not to comply with a request for records;

2. Clarification of TVA's practice that the availability of certain classes of nonexempt records is deferred where premature disclosure might interfere with accomplishment of TVA's statutory responsibilities; and

3. A statement of TVA's determination that records relating to special nuclear material and nuclear facilities are exempt from public disclosure.

These regulations are effective on February 19, 1975. Since the material contained herein concerns rules involving agency procedure and public property, the relevant provisions of the Administrative Procedure Act (5 U.S.C. 553) requiring notice of proposed rulemaking, opportunity for public participation, and delay in effective date are inapplicable.

The regulations are issued under the authority of 16 U.S.C. 831-831dd and Pub. L. 93-502.

By direction of The Board of Directors. Dated: February 18, 1975.

> H. N. STROUD, Assistant General Manager.

Effective February 19, 1975, \$ 301.1 of Title 18 is revised to read as follows:

§ 301.1 Records.

(a) Records available. TVA's records will be made available for inspection and copying upon request as provided in this section, except that records are exempt and are not made available if they are (1) (i) specifically authorized under criteria established by an Executive order to be kept secret in the interest of national defense or foreign policy and (ii) are in fact properly classified pursuant to such Executive order: (2) related solely to the internal personnel rules and practices of TVA; (3) specifically exempted from disclosure by statute; (4) trade secrets and commercial or financial information obtained from any person and privileged or confidential; (5) interagency or inter-agency memorandums or letters which would not be available by law to a private party in litigation with TVA, including without limitation records relating to control and accounting for special nuclear material and to the physical security plans for the protection of TVA's nuclear facilities; (6) personnel and medical files and similar files, the disclosure of which would constitute a clearly unwarranted invasion of personal privacy; (7) investigatory records compiled for law enforcement purposes, but only to the extent that the production of such records would (i) interfere with enforcement procedings, (ii) deprive a person of a right to a fair trial or an impartial adjudication, (iii) constitute an unwarranted invasion of personal privacy, (iv) disclose the identity of a confidential source and, in the case of a record compiled by a criminal law enforcement authority in the course of a criminal investigation, or by an agency conducting a lawful national security intelligence investigation, confidential information furnished only by the confidential source. (v) disclose investigative techniques and procedures, or (v1) endanger the life or physical safety of law enforcement personnel; (8) contained in or related to examination, operation, or condition reports prepared by, on behalf of, or for the use of any agency responsible for the regulation or super-vision of financial institution; or (9) geological and geophysical information and data, including maps, concerning wells. The availability of certain classes of nonexempt records is deferred for such time as TVA may determine is reasonably necessary to avoid interference with the accomplishment of its statutory responsibilities. Such records include bids and information concerning the identity and number of bids received prior to bid opening; all nonexempt records relating to bids between the time of bid opening and award; and all nonexempt records relating to negotiations in progress involving contracts or agreements for the acquisition or disposal of real or personal property by TVA prior to the conclusion of such negotiations. Any reasonably segregable portion of an available record shall be provided to any person requesting such record after de-

letion of the portions which are exempt under this paragraph.

(b) Requests. Requests to inspect and copy TVA records shall be directed to the Director of Information, Tennessee Valley Authority, Knoxville, Tennessee 37902. A request shall:

 be in writing,
 state in the text that it is a refor records pursuant to the Freedom of Information Act or the regulations in this section.

(3) reasonably describe the records requested with sufficient specificity to permit their location and identification; and

(4) include a statement (if such is the case) that the person making the request will pay all applicable fees under \$ 301.2, or a statement of the amount of the maximum applicable fee that such person is willing to pay without further notification from TVA. Requests may be made personally or by mail, but to per mit expeditious processing, a mailed request should be identified on the envelope as a Freedom of Information Act request.

(c) Processing of requests. (1) Initial determination. (i) Within 10 days (excluding Saturdays, Sundays, and legal public holidays) after a request is received by TVA, and subject to paragraph (c) (3) of this section, TVA shall make an initial determination as to whether to comply with the request, and shall immediately give written notice of the determination to the person making the request. Initial determinations shall be made by the Director or Assistant Director of Information. If the initial determination—is not to comply with the request, the notice to the person making the request shall include a statement of the reasons for the denial of the request; a notice of the right of the person making the request to appeal the denial to TVA's General Manager, and the time limits therefor; and the name and job title of the person responsible for the initial determination.

(ii) For purposes of this paragraph, a request is deemed to be received by TVA only when it is physically delivered to Office of the Director of Information and meets all the requirements of paragraph (b) of this section. If the request does not contain a sufficient description of the record as required by paragraph (b) (3) of this section, TVA will promptly notify the person making the request of the need for clarification, and the request will not be deemed received until clarification is obtained. If the request does not contain a statement as required by paragraph (b) (4) of this section, or if the amount of anticipated search fees exceeds the higher of the amount stated or \$25, TVA will promptly notify the person making the request of the amount of the anticipated fee. Such notice shall include an offer to confer in order to determine if the request can be reformulated so as to meet the person's needs at lower cost, and the request will not be deemed received until agreement is

reached concerning fees.

(2) Appeal. (1) If the initial determination is to deny the request, the person making the request may appeal such action to TVA's General Manager. Such an appeal must be taken within 30 days after the person's receipt of the initial determination, and is taken by delivering a written notice of appeal to the General Manager, Tennessee Valley Authority, Knoxville, Tennessee 37902. Such notice shall include a statement that it is an appeal from a denial of a request under the Freedom of Information Act and shall indicate (A) the date on which the denial was issued, and (B) the date on which the denial was received by the person making the request.

(ii) Within 20 days (excluding Saturdays, Sundays, and legal public holidays) after an appeal is received, and subject to paragraph (c) (3) of this section, TVA shall make a final determination on the appeal. In making such a determination, TVA will consider whether or not to waive the provisions of any exemption contained in paragraph (a) of this section, except that without the written permission of the person involved, TVA will not waive the exemptions contained in paragraph (a) (4), (6) and (7) of this section. Determinations of appeals under this section shall be made by the General Manager, the Assistant General Manager, or an Assistant to the General Manager. If the determination on the appeal is to deny the request for records, TVA shall notify the person making the request of such determination, including the reason for the denial; a notice of the person's right to judicial review of the denial; and the name and job title of the TVA official responsible for the determination of the appeal.

(3) Extensions of time. (i) The 10day time limit provided in paragraph (c) (1) of this section may be extended by TVA for unusual circumstances as set forth in this paragraph upon written notice to the person making the request specifying the reason for such extension and the date on which an initial determination is expected to be dispatched to such person. Such extension may not exceed 10 working days, and a decision to make such extension shall be made by the Director of Information or the Assistant Director of Information.

(ii) The 20-day time limit provided in paragraph (c) (2) of this section may be extended by TVA for unusual circumstances as set forth in this paragraph upon written notice to the person appealing a denial of a request for records. The notice shall specify the reasons for the extension and the date on which a determination of the appeal is expected to be dispatched. The aggregate length of an extension under this paragraph when combined with any extension provided under paragraph (c) (3) (i) of this section shall not exceed 10 working days. A decision to make an extension under this paragraph shall be made by the General Manager, Assistant General Manager, or an Assistant to the General Manager.

(iii) For the purposes of this paragraph, unusual circumstances means, to the extent reasonably necessary to the proper processing of the particular request or appeal:

(A) Time necessary to search and collect requested records from segments of the agency separate from the office processing the request;

(B) Time necessary to search for, collect, and appropriately examine the vohuminous number of records demanded

in a single request; or

(C) Time necessary for consultation with another agency having a substantial interest in the determination of the request, or among two or more components of the agency which have an interest in the subject matter of the

request.

(d) Manner of making records available. When TVA determines to comply with a request for records, the records are made available promptly for inspection and copying at the place it con-siders most feasible. TVA will notify the person requesting the records of the place at which the records will be made available and the amount of the applicable fee pursuant to § 301.2, and will request the person to set up a mutually convenient time for inspection. Applicable fees are payable prior to actual inspection of the records or prior to receipt of requested copies after inspection, respectively. If TVA deems it more convenient to do so, it may furnish copies of available requested records in lieu of permitting physical inspection thereof. In such case payment of applicable fees is due within 30 days after receipt of the copies.

(e) Publications. TVA publishes and sells to the public at nominal cost various materials concerning its activities and other matters within its statutory responsibilities, and also provides for the sale of other materials, such as maps prepared by other agencies, at prices prescribed by such agencies. Such services are not performed under the Freedom of Information Act, and requests for the purchase of such materials are not covered by the provision of subsections (a) through (d) of this section. A price list and order form for some of the most frequently asked for TVA publications and reports is contained in TVA Form 3077, which may be obtained by writing the Director of Information, Tennessee Valley Authority, Knoxville, Tennessee 37902. Information about other informational material available for sale or distribution by TVA may be obtained at the same address.

(16 U.S.C. 831-831dd)

[FR Doc.75-4688 Filed 2-18-75;8:45 am]

#### Title 22—Foreign Relations

CHAPTER II-AGENCY FOR INTERNA-TIONAL DEVELOPMENT, DEPARTMENT OF STATE

[A.I.D. Reg. 12]

#### PART 212—PUBLIC INFORMATION

On November 21, 1974, Congress amended the Freedom of Information

Act (5 U.S.C. 552). These amendments (Pub. L. 93-502) change the directives under which the former Public Information regulations of the Agency for International Development were issued and require changes in the procedures by which information requests are processed. Accordingly, Part 212 is hereby repealed and a new Part 212 is substituted therefor setting forth new procedures for requests, appeals and processing such requests and appeals under the Freedom of Information Act.

Since the Freedom of Information amendments require persons making requests to follow published rules stating the time, place, fees, and procedures to be followed, it is considered in the public interest to publish promptly the required rules of this Agency. It would be imprac-tical to delay the effective date of the regulations beyond the effective date of the Freedom of Information amendments. Therefore, pursuant to 5 U.S.C. 553(d), these regulations shall be effective on February 19, 1975 without the normal 30-day notice being given.

However, in accordance with the spirit of the public policy set forth in 5 U.S.C. 553, interested persons may submit written comments, suggestions, data or arguments to the Director, Office of Public Affairs, Agency for International Development, Department of State, 21st and Virginia Ave., NW., Washington, D.C. 20523, on or before March 25, 1975. Material thus submitted will be evaluated and acted upon in the same manner as if this document were a proposal.

#### Subpart A-General

212.1

Statement of policy. Subpart & Publication in the Federal Register

212.11 Materials to be published. 212.12 Effect of nonpublication.

212.13 Incorporation by reference.

Subpart C—Availability of Information for Public Inspection and Copying

212.21 Public records.

Protection of personal privacy.

Current index.

212.23 212.24

Effect of noncompliance Procedures for obtaining materials under this subpart.

#### Subpart D-Access to Agency Records

212.31 212.82 Access to Agency records.

Identification of records.

212.83 Procedure for making requests.

Procedures for responding to re-

quests for records 212.35 Schedule of fees and methods of

payment for services rendered. 212.36 Denial of request for access to

records.

212.37 Procedures for agency consideration of appeals.

#### Subpart E-Exemptions from Disclosure

212.41 Exemptions from the publication and disclosure requirements of Subparts B, C, and D.

Subpart F—Opening of Records for Nonofficial Research Purposes

212.51 General policy.

AUTHORITY: The provisions of Part 212 are issued under Sec. 621, the Foreign Assistance Act of 1961, as amended, 75 Stat. 445 (22 U.S.C. 2402), Executive Order 10501, as amended by Executive Order 10816, sec. 3,

Administrative Procedure Act, 60 Stat. 237, as revised by Pub. L. 89-487, 80 Stat. 250 a amended by Pub. L. 93-502, 88 Stat. 1561 (5 U.S.C. 552).

#### Subpart A-General

#### § 212.1 Statement of policy.

(a) It is the policy of the Agency for International Development that information about its objectives and operations be freely available to the public in accordance with the provisions of Pub. L. 89-487 (80 Stat. 250) as amended by Pub. L. 93-502. The Director, Office of Public Affairs is responsible for administration of the provisions of the regulations of this part.

(b) All records of the Agency for International Development shall be made available to the public upon compliance with the procedures established in this regulation, except to the extent a determination is made to withhold a record exemptable under 5 U.S.C. 552(b). Such a determination shall be made pursuant to procedures set forth in \$\$ 212:36 and

212.37.

(c) The term "record" includes all books, papers, maps, photographs, or other documentary material, or copies thereof, regardless of physical form or characteristics, made in or received by the Agency for International Development (including its Missions abroad), and preserved as evidence of its organization, functions, policies, decisions, procedures, operations, or other activities. It does not include copies of the records of other Government agencies, foreign governments, international organizations, or non-governmental entities which do not evidence organization, functions, policies, decisions, procedures, operations, or activities of the Agency for International Development.

#### Subpart B-Publication in the Federal Register

### § 212.11 Materials to be published.

(a) The Agency separately states and currently publishes in the FEDERAL REG-ISTER for the guidance of the public:

(1) Descriptions of its central and field organization and the established places at which, the officers from whom and the methods whereby, the public may secure information, make submittals or requests, or obtain decisions;

(2) Statements of the general course and method by which its functions are channelled and determined, including the nature and requirements of all formal and informal procedures available;

(3) Rules of procedure, descriptions of forms available or the places at which forms may be obtained, and instructions as to the scope and contents of all papers, reports, or examinations:

(4) Substantive rules of general applicability adopted as authorized by law, and statements of general policy or interpretations of general applicability formulated and adopted by the Agency;

(5) Every amendment, revision or repeal of the foregoing.

(b) A.I.D. Public Notice No. 1 and the A.I.D. Regulations published in Chapter II of Title 22 and in Sub-Title A. Chapter 7 of Title 41 of the Code of Federal Regulations, implement the provisions of this section.

#### § 212.12 Effect of nonpublication.

Except to the extent that a person has actual and timely notice of the terms thereof, no person shall in any manner be required to resort to, or be adversely affected by any A.I.D. matter required to be published in the FEDERAL REGISTER and not so published.

#### § 212.13 Incorporation by reference.

For purposes of this Subpart B, A.I.D. matter which is reasonably available to the class of persons affected thereby is deemed to be published in the Federal Register when it has been incorporated by reference therein with the approval of the Director of the Federal Register.

# Subpart C—Availability of Information for Public Inspection and Copying

#### § 212.21 Public records.

In accordance with this subpart, A.I.D. makes the following information and materials available for public inspection and copying:

(a) All final opinions (including concurring and dissenting opinions), and all orders made in the adjudication of cases;

(b) Those statements of policy and interpretations which have been adopted by the Agency and are not published in the FEDERAL REGISTER; and

(c) Administrative staff manuals and instructions to staff that affect any member of the public.

### § 212.22 Protection of personal privacy.

To the extent required to prevent a clearly unwarranted invasion of personal privacy, the Agency may delete identifying details when it makes available or publishes an opinion, statement of policy, interpretation, or staff manual or instruction, and will, in each such case, explain in writing the justification for the deletion.

#### § 212.23 Current index.

The Agency maintains and makes available for public inspection and copying a current index providing identifying information for the public as to any matter which is issued, adopted, or promulgated after July 4, 1967, and which is required by § 212.21 to be made available or published. Publication of an index is deemed both unnecessary and impractical. However, copies of the index are available upon request for a fee of the direct cost of duplication.

### § 212.24 Effect of noncompliance.

No final order, opinion, statement of policy, interpretation, or staff manual or instruction that affects any member of the public will be relied upon, used, or cited as precedent by the Agency against any private party unless it has been indexed and either made available or published as provided by this subpart, or unless that private party shall have actual and timely notice of the terms thereof.

# § 212.25 Procedures for obtaining materials under this subpart.

(a) The materials required to be made available for public inspection and copying in accordance with this subpart are available to members of the public at the Agency for International Development Public Reading Room, Room 4943, Department of State Building, 21st and Virginia Ave., NW., Washington, D.C. which is open from 9 a.m. to 5 p.m., Monday through Friday, except on Holidays.

(b) Requests for materials which are available under this subpart should follow the procedures under Subpart D.

(c) The direct costs of any necessary duplication will be charged in accordance with the fee schedule set forth in § 212.35.

(d) The USAID Missions in countries abroad are not responsible for the maintenance of the index and materials available under this subpart. However, in so far as they do have these materials, they will make them available to citizens of the United States who are present in their respective countries upon either in person application or writing to the Director, USAID, c/o American Embassy in the applicable country.

# Subpart D—Access to Agency Records

# § 212.31 Availability of Agency Records.

Upon receiving a request for a record, which reasonably describes such record, and which is made in accordance with the provisions of this Subpart, the Agency for International Development will make such records promptly available to any person, except the following:

(a) matters published in the FEDERAL REGISTER pursuant to Subpart B; (b) matters made available to the public pursuant to Subpart C; and (c) matters exempt from disclosure pursuant to \$212.41.

#### § 212.32 Identification of records.

The request for a record by a member of the public must contain a reasonably specific description of the particular record sought so that an officer of the Agency who is familiar with the subject matter of the request may be able to locate the record with a reasonable amount of effort. A description that includes as much information as possible, such as the subject matter, format, approximate date, and where pertinent, the name of the country or person involved, will facilitate the search for the requested record.

# § 212.33 Procedure for making requests.

(a) Requests for reasonably described records may be made by a member of the public either in person or by writing to the Director, Office of Public Affairs, Agency for International Development, Department of State, 21st and Virginia Ave., NW., Washington, D.C. 20523. The request and the envelope must be plainly marked "FOI Request." Requests may be made in person by going to the Agency for International Development Public

Reading Room, Room 4943, Department of State Building, 21st and Virginia Ave., NW., Washington, D.C., which is open from 9 a.m. to 5 p.m., Monday through Friday, except on Holidays. Telephonic requests cannot be accepted.

(b) Requests for reasonably described records may be made directly to USAID missions abroad only by citizens of the United States who are present in their respective countries by either personal application or by written application to the Director, USAID, care of American Embassy in the applicable country. Any written request and its envelope must be plainly marked "FOI Request."

(c) If a request not properly marked "FOI Request" on the request and envelope is inadvertently delayed in reaching the Director, Office of Public Affairs, or is erroneously sent to a USAID mission abroad, it will not be deemed received by the Agency until actually received by the Director, Office of Public Affairs. In such event, the person making the request will be furnished a notice of the effective date of receipt.

# § 212.34 Procedures for responding to requests for records.

(a) Upon receipt of the request by the Director, Office of Public Affairs, a maximum of 10 working days will normally be taken to determine to what extent the Agency can provide the information requested, and upon the making of the determination, the person making the request will be immediately informed. Copies of the releasable documents shall be made available promptly thereafter upon receipt of appropriate fees as set forth in § 212.35.

(b) In unusual circumstances, the Agency may not be able to determine the availability of the requested documents within 10 working days, in which event the person making the request will be informed by written notice of the delay, the reasons for the delay and the date on which a determination may be expected. The maximum time in making a determination of availability will be 20 working days from receipt of the request. If only part of the permissible 10 working days extension is used, the Agency reserves the right to use any remainder of the time for an extension of time for the determination of an appeal, if one is made, and the unusual circumstances require an extension of time to decide an appeal. A time extension for unusual circumstances will be used as reasonably necessary for a particular request in the following instances:

(1) When there is a need to search for and collect the requested records from field facilities or other establishments that are separate from the office processing the request;

(2) When there is a need to search for, collect, and appropriately examine a voluminous amount of separate and distinct records which are demanded in a single request; or

(3) When there is a need for consultation, which shall be conducted with all practicable speed, with another agency having a substantial interest in the determination of the request or among two

or more components of the agency having substantial subject-matter interest therein.

(c) If, in spite of diligent efforts, circumstances are such that the Agency cannot make a complete and final determination even within the 10-day extension period, a denial of the request may be made. However, the Agency will continue the determination process and the denial will contain a statement to this effect with a date when final determination is expected. As soon as a final determination can be made, the person making the request will be further notified. This provision does not in any way limit or abridge the person's right to appeal the initial denial. It is only an informal process that may exist in addition to the appeal procedure in an attempt to prevent the necessity of appeal when the denial is issued without a full determination but solely in order to meet the Agency's time deadlines.

(d) If a request is made to this Agency for material that is controlled or held by another agency, the request will be referred to the appropriate agency for action and the person making the request will be immediately notified of the referral, unless the other agency has. by public regulation, delegated the release authority to this Agency. If release authority has been delegated, this Agency will follow the procedures authorized by the delegation in determining whether to release the information. If a request for material is referred to this Agency from another agency, the time period for determination of release of the information will not start until the request is received by the Director. Office of Public Affairs. The person making the request will be immediately notified of the referral and the date the request was received in this Agency. This Agency will not accept referral of requests unless and until the Director. Office of Public Affairs, determines that the material requested is actually within the scope and control of the release authority of the Agency for International Development.

(e) If only a part of a record is exempt from disclosure, then any reasonably segregable portion of such record shall be furnished after deletion of the portions which are exempt, provided that the segregable portion constitutes an intelligible record which is not distorted out of context or contradictory to the substance of the entire record before segregation.

§ 212.35 Schedule of fees and method of payment for services rendered.

(a) The following specific fees shall be applicable with respect to services rendered to members of the public under this part:

(1) Searching for records, per hour, 

otherwise (per page). (Maximum of 2 copies will be provided.)....

(3) Duplication of architectural photographs and drawings.....

(b) When no specific fee has been established for a service, for example, when the search involves computer time or special travel, transportation, or communications costs, the Director, Office of Public Affairs, is authorized to determine the direct costs of the service and include such costs in the fees chargeable under this section.

(c) Where it is anticipated that the fees chargeable under this section will amount to more than \$25, and the requester has not indicated in advance his willingness to pay fees as high as are the requester shall anticipated. promptly notified of the amount of the anticipated fee or such portion thereof as can readily be estimated. In appropriate cases an advance deposit may be required. The requester is at any time welcome to confer with knowledgeable Agency personnel in an attempt to formulate the request in a manner which will reduce the fees and meet the needs of the requester. A request will not be deemed to have been received until the requester has agreed to pay the anticipated fees and has made an advance deposit if one is required.

(d) Search costs are due and payable even if the record which was requested cannot be located after all reasonable efforts or if the Agency determines that a record which has been requested but which is exempt from disclosure under this part is to be withheld.

(e) Fees must be paid in full prior to

issuance of requested copies.

(f) Remittances shall be in the form either of a personal check or bank draft drawn on a bank in the United States, a postal money order, or cash, Remittances shall be made payable to the order of the Treasury of the United States and mailed to the Director, Freedom of Information Staff, Office of Public Affairs, Agency for International Development, Washington, D.C. 20523. The Agency will assume no responsibility for cash which is lost in the mail.

(g) A receipt for fees paid will be given only upon request. Refund of fees paid for services actually rendered will not be

made.

(h) The Director, Office of Public Affairs, or an officer designated by the Director may waive all or part of any fee provided for in this section when the Director or the designated officer deems it to be in either the Agency's interest or in the general public's interest.

§ 212.36 Denial of request for access to records.

(a) If it is determined that the Agency cannot comply with all or part of a request for records, the person making the request shall be immediately notified of the determination, the reasons for the determination, the name and title of each officer responsible for the denial, and the right of the person to appeal the adverse determination.

(b) The denial of a request for records may be made only by the Director, Office of Public Affairs or his delegate. The General Counsel shall be consulted prior to a decision to deny access.

(c) Any person who has been denied access to records pursuant to this section may appeal the adverse decision not later than 60 days after the date of the notification of denial or, in the case of a partial denial, after the date the releasable documents are actually furnished the person making the request, whichever is later. The appeal shall be in writing addressed to the Administrator or Deputy Administrator, Agency for International Development, 21st and Virginia Ave., NW., Washington, D.C. 20523. In order to make a timely response to the appeal, both the appeal and the envelope must be plainly marked "FOI Appeal". The appeal should contain a reasonable description of the record sought and withheld, a copy of the initial decision to deny access and any other information that will enable the Administrator or Deputy Administrator to make the final decision.

212.37 Procedures for Agency consideration of appeals.

(a) Upon receipt of the appeal by the Administrator or Deputy Administrator, a maximum of 20 working days will normally be taken to decide the appeal. In unusual circumstances, as defined in § 212.34, the 20 working days may be extended by 10 working days or by the number of days not used in the original denial of the request.

(b) If the appeal is granted, the person making the appeal shall be immediately notified and copies of the releaseable documents shall be made available promptly thereafter upon receipt of appropriate fees as set forth in 212.35. If the appeal is denied in whole or part, the person making the request shall be immediately notified of the decision and of the provisions of judicial review of the Agency's denial of the request.

(c) In the event a determination is not issued within the appropriate time limit and the person making the request chooses to sue the Agency, the determination process shall continue and the Administrator or Deputy Administrator may review any initial denial of the re-

quested records. (d) If appeals not properly marked "FOI Appeal" on the appeal and envelope are inadvertently delayed in reaching the Administrator or Deputy Administrator, they will not be deemed received by the Agency until actually received by the Administrator or Deputy Administrator. In such event, the person making the appeal will be furnished a notice of the effective date of receipt.

Subpart E-Exemptions From Disclosure

§ 212.41 Exemptions from the publication and disclosure requirements of Subparts B, C, and D.

None of the provisions of Subparts B. C, and D which provide for publication and disclosure of certain information and records shall be applicable to matters that are:

(a) (1) Specifically authorized under criteria established by an Executive order to be kept secret in the interest of national defense or foreign policy and (2)

are in fact properly classified pursuant to such Executive order:

(b) Related solely to the internal personnel rules and practices of the agency: (c) Specifically exempted from disclosure by statute;

(d) Trade secrets and commercial or financial information obtained from a person and privileged or confidential;

(e) Interagency or intra-agency memorandums or letters which would not be available by law to a party other than an agency in litigation with the agency;

(f) Personnel and medical files and similar files the disclosure of which would constitute a clearly unwarranted invasion

of personal privacy;

- (g) Investigatory records compiled for law enforcement purposes, but only to the extent that the production of such records would (1) interfere with enforcement proceedings, (2) deprive a person of a right to a fair trial or an impartial adjudication, (3) constitute an unwarranted invasion of personal privacy, (4) disclose the identity of a confidential source and, in the case of a record compiled by a criminal law enforcement authority in the course of a criminal investigation, or by an agency conducting a lawful national security intelligence investigation, confidential information furnished only by the confidential source, (5) disclose investigative techniques and procedures, or (6) endanger the life or physical safety of law enforcement personnel;
- (h) Contained in or related to examination, operating, or condition reports prepared by, on behalf of, or for the use of any agency responsible for the regulation or supervision of financial institutions; and
- (i) Geological and geophysical information and data (including maps) concerning wells.

#### Subpart F-Opening of Records for **Nonofficial Research Purposes**

§ 212.51 General policy.

- (1) In accordance with the policy of the Department of State, the Agency for International Development will open its records on an equitable basis to all individuals engaged in private research as soon as such action may be taken without adversely affecting the national security, the maintenance of friendly relations with other nations, the efficient operation of the agency, and the administrative feasibility of servicing requests for access to such records.
- (2) Access for research in the classified foreign policy records in A.I.D.'s custody will be governed by the regulations of the Department of State in respect thereto, as set forth in Part 6, Chapter II of Title II of the Code of Federal Regulations. Application for such access may be made to the Director, Public Affairs, Agency for International Development, 21st and Virginia Avenue NW., Washington, D.C. 20523. The Director, Public Affairs, in consultation with the Director, Historical Office, Department of State, will determine the action to be taken and advise the researcher.

fective February 19, 1975.

Dated: February 14, 1975.

JOHN E. MURPHY, Deputy Administrator, Agency for International Development. [FR Doc.75-4684 Filed 2-18-75:9:20 am1

Title 39—Postal Service CHAPTER I-UNITED STATES POSTAL SERVICE

#### PART 265-RELEASE OF INFORMATION Freedom of Information Act Regulations

The purpose of this document is to adopt revised regulations to govern the release of records to the public and to establish procedures implementing the Freedom of Information Act. amended, insofar as it applies to the Postal Service.

The revised regulations are placed in Part 265 of Title 39, Code of Federal Regulations, and supersede regulations adopted by the former Post Office Department presently appearing as Parts 261 and 262 of Title 39, Code of Federal Regulations. While continuing generally the organization of Parts 261 and 262, the new regulations have been substantially rewritten in the interest of clarity, and have been modified as necessary in the light of recent statutory developments, principally §§ 410 and 412 of Title 39, United States Code, as enacted by Pub. L. 91-375, and the Freedom of Information Act Amendments, Pub. L. 93-502.

For purposes of analysis, the new regulations may be divided roughly into two groups of sections. The first group, consisting of §§ 265.1 through 265.6, states the policy of the Postal Service on the release of records and lists the principal categories of records that are available to the public as of right and those that are exempt from mandatory disclosure. The second group, consisting of §§ 265.7 through 265.9, establishes the procedures to be followed by the public in submitting requests for records and by the Postal Service in responding to those requests.

Discussed below are the major changes effected by the revision. There are in addition minor changes, not discussed, which have been made in the light of either statutory changes or the experience of the Postal Service under the

previous regulations.

In order to conform the regulations to § 410(c) of Title 39, U.S.C., several categories of records have been added to the list of those exempt from mandatory disclosure contained in § 265.6 (formerly § 261.2). Of particular interest among these may be paragraphs (b) (3), relating to commercial information, and (d). relating to the names and addresses of postal customers. Paragraph (e) (1) of § 265.6 is a new provision added to reflect the prohibition of the disclosure by the Postal Service to the public of mailing lists, which is contained in 39 U.S.C. 412. Paragraph (e) (2) of \$ 265.6, although not altogether new, has been

Effective date. This regulation is ef- revised to reflect the changes effected by Pub. L. 93-502 in 5 U.S.C. 552(b) (1) relating to classified defense and foreign policy materials. Paragraph (c) of § 265.6, relating to disclosure of investigatory records, has similarly been revised to state the policy of the Postal Service normally to make investigatory records available to the public when disclosure would be required by 5 U.S.C. 552(b) (7) as amended by Pub. L. 93-502, even though withholding would be authorized by 39 U.S.C. 410(c) (6).

The new regulations generally continue the system of procedures for responding to Freedom of Information Act requests that has been in effect under the prior regulations. Under this system, a request is submitted to the head of the facility at which the record is maintained. This custodian makes the initial determination whether to grant or deny the request. If it is denied, an appeal may be taken to the General Counsel whose decision is final within the Postal

Service.

The sections relating to procedures and fees for requesting access to records, viz, §§ 265.7 and 265.8, have been substantially rewritten and expanded to conform to the requirements of the 1974 Amendments and to provide additional guidance to the public and to postal employees in submitting and responding to requests for records. The procedures which have been adopted are designed to take account of the fact that postal records are maintained at more than 30,000 post offices and other facilities located throughout the United States and its overseas territories. If the statutory time limits for response to requests are to be met, expeditious handling is essential. To facilitate the identification of requests for records so that priority treatment may be given them, § 265.7(a) (1) requires that a person desiring a response to his request within the statutory period label it as a Freedom of Information Act request or use a designated Postal Service form and that any envelope bear the same label. Requests not fulfilling this requirement will, of course, be given prompt treatment, but they will not be deemed as coming within the framework of these formal procedures. Such treatment seems, moreover, consistent with our experience that many inquiries which could be treated formally are, in fact, casual in nature, and can be answered informally to the greater satisfaction of the inquirer.

Paragraph (a) (2) of § 265.7 requires that requests be submitted to the custodian of the requested record, i.e., the head of the facility at which it is maintained. If it is submitted to some other facility it will, of course, be forwarded to the proper one, but it will be deemed to have been received by the Postal Service for purposes of computing the response period when it has been received by the actual custodian. If the location of the record is not known to the requester he may obtain this information from the Postal Service Records Officer or he may submit his request to the Records Officer who will forward it to the proper custodian.

The section which establishes the fee schedule (§ 265.8), was previously published in the FEDERAL REGISTER (40 FR 3220) as a proposed revision of § 262.7 of Title 39, CFR, and comments were requested from members of the public. We received one letter of comment, the suggestions of which we have adopted in part. We have redesignated the proposed §§ 262.7 as 265.8. We also made some modifications to the text.

The proposed increase in the fee for clerical search time has been reconsidered and the prior rate of \$2.00 per quarter hour will be retained. The proposed treatment of computer fees has been similarly reconsidered. We have determined to adopt the information Services Price List for computer services fees in its entirety, rather than to at-tempt to treat analysts' time and the reproduction of records separately. § 265.8 (b) (3). The list, which is also used for other purposes within the Postal Service, is subject to periodic revision in the light of experience and cost adjustments. Charges will be based on the price list in effect at the time the services are performed. The list in effect on February 19, 1975, is published as Appendix A to Part 265 to provide an illustration of the services covered and the current charges per unit of work.

A new paragraph (1) has been added to § 265.8(d) to provide that when sufficient stock exists, printed materials will be made available at their established price, or at cost if no price has been established, rather than at the per-page reproduction fee. In this respect we have adopted a suggestion made in the letter of comment.

The problem of providing meaningful advance notice of the cost of particular requests while complying with statutory deadlines for responses continues to be troublesome. The proposed rule provided for the giving of advance notice if fees were expected to exceed \$25.00. We have concluded that the \$25.00 figure would exceed the expectations of many members of the public, particularly those unfamiliar with these regulations. We consider, however, that a fee not in excess of \$10.00 would not be considered exorbi-

tant. § 265.8(e) (2).

The proposed rule provided further that the ten-day response period would not begin until the requester had been notified and had agreed to bear the estimated costs. The comment to this proposal expressed concern that this system would permit unwarranted delays in giving notice. It was suggested that notice be required within a short, fixed period and include an explanation of the basis for the estimate. We consider that this suggestion has merit and have modified the rule to require that notice of estimated costs in excess of \$10.00 be given within five working days of receipt by the custodian and that it explain briefly the basis of the estimate. The comment suggested further that the response period begin to run with the physical receipt of the request and be tolled by the giving

of notice. We are not persuaded, however, that the potential saving of time by tolling the response period, rather than starting it with the requester's response, is significant, and we believe that the approach adopted permits a simpler, more readily computed response period.

We have considered but not adopted two additional suggestions which we believe require discussion. The first is that the right of the public to obtain copies of postal records which are not exempt, as distinct from an opportunity merely to inspect the records, be made absolute. We believe, however, that this is not required either by statute or by considerations of fairness. Conditions in post offices, including the availability of copying equipment, may vary considerably. Section 265.8(c) (3) does not give the custodian unfettered discretion to refuse

The second suggestion is that fees for search time be waived in all cases in which no records are made available. The comment expressed concern that the possibility that the requester may be liable for significant search costs, even though records are not found or are withheld, may have a chilling effect on all substantial requests. Such is not, of course, the intended result and a waiver of search costs in this situation is stated as the normal rule § 265.8(e) (4). It is not reasonable, moreover, that the Postal Service, meaning the taxpayers or mail users, should be required in all possible situations to bear the costs of search.

As noted above, § 265.8 was previously published for public comment. As to the balance of the part, since many of the changes are necessary to achieve compliance with the amendments to the Freedom of Information Act which become effective February 19, 1975, there is no time to receive and evaluate public comment. We intend, nonetheless, to reexamine this part in the light of our experience and that of citizens requesting access to records pursuant to its provisions. Interested persons are also invited to submit written comments on these regulations at any time to the General Counsel, United States Postal Service, Washington, D.C. 20260. Attention: Legal Affairs Office.

> ROGER P. CRAIG, Deputy General Counsel.

The following changes become effective February 19, 1975.

Accordingly, 39 CFR Parts 261 and 262 are revised and redesignated as new Part 265, which is set forth below. 39 CFR Parts 261 and 262 are then vacated and reserved.

PART 261-[RESERVED]

PART 262-[RESERVED]

PART 265-RELEASE OF INFORMATION

Sec. 265.1 Purpose and scope. 265.2 Policy. Responsibility. Inquiries. 265 2

265.5 Public reading room. Availability of records.

Procedures for inspection and copy-265.6

ing of records. 265 8 Schedule of fees. 265.9 Annual reports.

265.10 Compliance with subpoens tecum court orders and summonses.

AUTHORITY: 39 U.S.C. 401; 5 U.S.C. 552 as amended by Pub. L. 93-502, 88 Stat. 1561.

#### § 265.1 Purpose and scope.

This part contains the regulations of the Postal Service relating to the availability to the public of Postal Service records. Included in this part are the regulations which implement section 552 of Title 5, United States Code, the "Freedom of Information Act," insofar as it applies to the Postal Service.

Official records of the Postal Service made available pursuant to the requirements of the Act shall be furnished to members of the public as prescribed by

this part.

§ 265.2 Policy.

(a) It is the policy of the Postal Service to make its official records available to the public to the maximum extent consistent with the public interest. This policy requires a practice of full disclosure subject only to the specific exceptions required or authorized by law.

(b) The exemptions from mandatory disclosure provided by section 552(b) of Title 5, and section 410(c) of Title 39, United States Code, for various types of records, reflect the fact that under some circumstances the public interest may be better served by leaving the disclosure of particular records to the discretion of the Postal Service than by requiring their disclosure. As to those records the disclosure of which is not prohibited by statute, Executive Order, or regulation, the discretion vested in the Postal Service should be exercised after giving consideration to the following: the effect of non-disclosure on the public's right to know about a particular matter; the effect of disclosure on the right of privacy of any affected individuals; the effect of disclosure on the public interest in the economical, efficient, and orderly operation of the nation's mail system; and any other factors that may be relevant under the circumstances.

#### § 265.3 Responsibility.

(a) Custodian. Official records are in the custody of the postmaster or other head of the facility, such as a postal data center, mail bag depository, sectional center facility, district office or regional headquarters, or of a group or department of Headquarters, at which they are maintained. These custodians are responsible for responding in the first instance to requests from members of the public for Postal Service records.

(b) Records officer. The Records Officer, Management Information Systems Department, United States Postal Service, Washington, D.C. 20260, is responsible for the overall administration of this part, including the issuance of detailed instructions to custodians.

(c) General counsel. The General Counsel decides timely appeals authorized by this part.

#### § 265.4 Inquiries.

Inquiries regarding the availability of Postal Service records should be directed to the head of the post office, postal data center, mail bag depository, sectional center facility, district office, regional headquarters or other facility, or group or department of Headquarters, at which the official records are maintained. If the appropriate custodian is not known, inquiries should be directed to the Records Officer, Management Information Systems Department, United States Postal Service, Washington, D.C. 20260, telephone number (202) 245-4000.

#### § 265.5 Public reading room.

The Library of the Postal Service Headquarters, 475 L'Enfant Plaza, West, SW., Washington, D.C. 20260, serves as public reading room for the materials which are listed paragraphs (a) (2) and (a) (3) of § 265.6 as available for public inspection and copying.

#### § 265.6 Availability of records.

(a) Records available to the public upon request. (1) General. Postal Service records are available for inspection or copying at the request of any person, in accordance with the provisions of this part, except as otherwise provided by law or regulations, including but not limited to, § 265.(b) through 265.(f) of this part. Certain categories of records of particular interest are available on a continuing basis as provided in paragraphs (a) (2) and (a) (3) of this section and are listed in a public index as provided in paragraph (a) (4). Access to other records may be requested on an individual basis in accordance with the procedures pro-vided in § 265.7. Official records which are maintained on an electronic storage medium will normally be made available, in accordance with this part, as an exact duplicate of the requested original in a form readable by the human eye, such as a computer print-out.

(2) Opinions. All final opinions and orders made in the adjudication of cases by the Judicial Officer, Administrative Law Judges, and Board of Contract Appeals, and all advisory opinions concerning the private express statutes issued pursuant to § 310.6 of title 39, Code of Federal Regulations, all bid protest decisions and all decisions on appeals pursuant to this part, rendered by the Law Department are on file and available for inspection and copying in the Headquarters Library, United States Postal Service, 474 L'Enfant Plaza, West, SW.; Washington, D.C. 20260.

(3) Administrative Manuals and Instructions to Staff. The manuals, instructions, and other publications of the Postal Service that affect members of the public are available through the Headquarters Library, United States Postal Service, Washington, D.C. 20260, and at many post offices and other postal facili-

ties. Those which are available to the public but are not listed for sale may be inspected in the Library or at any postal facility which maintains a copy. Copies of publications which are not listed as for sale or as available free of charge may be obtained by paying a fee in accordance with § 265.8 below.

(4) Public index. (1) A public index is maintained in the Headquarters Library of all final opinions and orders made by the Postal Service in the adjudication of cases, Postal Service policy statements which may be relied on as precedents in the disposition of cases, and administrative staff manuals and instructions that affect the public.

(ii) The index mainly contains matters issued after July 4, 1967. However, the Postal Service may include prior matters.

(iii) Any person may arrange for the inspection of any matter in the public index in accordance with the procedures of \$ 265.7

(iv) Copies of the public index and of matters listed in the public index may be purchased through the Headquarters Library upon payment of fees as listed in the index or as provided in § 265.8.

(b). Records not subject to mandatory public disclosure. Certain classes of records are exempt from mandatory disclosure under exemptions contained in the Freedom of Information Act and in section 410(c) of Title 39, United States Code. The Postal Service will exercise its discretion, in accordance with the policy stated in § 265.2, as implemented by instructions issued by the Records Officer with the approval of the General Counsel in determining whether the public interest is served by the inspection or copying of records that are:

(1) Related solely to the internal personnel rules and practices of the Postal Service.

(2) Trade secrets, or privileged or confidential commercial or financial information, obtained from any person.

(3) Information of a commercial nature, including trade secrets, whether or not obtained from a person outside the Postal Service, which under good business practice would not be publicly disclosed. This class includes, but is not limited to:

(i) Information pertaining to methods of handling valuable registered mail.

(ii) Records of money orders, except as provided in Chapter 1 of the Postal Service Manual.

(iii) Technical information concerning postage meters and prototypes submitted for Postal Service approval prior to leasing to mailers.

(iv) Reports of market surveys conducted by or under contract in behalf of the Postal Service.

(y) Records compiled within the Postal Service which would be of potential benefit to persons or firms in economic competition with the Postal Service.

(vi) Information which, if publicly disclosed, could materially increase procurement costs.

(4) Interagency or internal memoranda or letters that would not be avail-

able by law to a private party in litigation with the Postal Service.

(5) Reports and memoranda of consultants or independent contractors, except to the extent they would be required to be disclosed if prepared within the Postal Service.

(6) Files personal in nature, including medical and personnel files, the disclosure of which would constitute a clearly unwarranted invasion of personal pri-

(7) Information prepared for use in connection with proceedings under Chapter 36 of Title 39, United States Code, relating to rate, classification, and serv-

(8) Information prepared for use in connection with the negotiation of collective bargaining agreements under Chapter 12 of Title 39, United States Code, or minutes of, or notes kept during, negotiating sessions conducted under such chapter.

(9) Other matter specifically exempted from disclosure by statute.

(c) Investigatory records. (1) Investigatory files compiled for law enforcement purposes, whether or not considered closed, are exempt by statute from mandatory disclosure except to the extent otherwise available by law to a party other than the Postal Service, 39 U.S.C. 410(c) (6). As a matter of policy, however, the Postal Service will normally make investigatory records available upon request unless the production of

these records would:

(i) Interfere with enforcement proceedings,

(ii) Deprive a person of a right to a fair trial or an impartial adjudication,

(iii) Constitute an unwarranted invasion of personal privacy,

(iv) Disclose the identity of a confidential source, and, in the case of a record compiled by the Postal inspection Service in the course of a criminal investigation or of a lawful national security intelligence investigation, confidential information furnished only by the confidential source,

(v) Disclose investigative techniques and procedures, or

(vi) Endanger the life or physical safety of law enforcement personnel.

(2) Authority to disclose investigatory records to persons outside the Postal Service must be obtained from the Chief Postal Inspector, United States Postal Service, Washington, D.C. 20260, or his designee.

(d) Disclosure of names and addresses of customers. The addresses of postal customers will be made available only in accordance with the following:

(1) The new address of any specific customer who has filed a change of address order will be furnished to any person upon payment of the fee prescribed in paragraph (d) (3) of \$265.8.

in paragraph (d) (3) of § 265.8.

(2) The name and address of the holder of a bulk mail permit, permit imprint or similar permit, or postage meter permit, and the name of any person applying for a permit in behalf of a holder, will be furnished to any person upon the payment of any fees authorized by § 265.8.

(3) The recorded post office box address of a customer whose mail is redirected to a post office box will be furnished to any person upon payment of the fee prescribed in paragraph (d) (3)

of § 265.8.

(4) The business name and address of the holder of a post office box being used for the purpose of doing or soliciting business with the public, and any person applying for a box in behalf of a holder, will be furnished to any person without charge. The postmaster may furnish this information when he is satisfled from the entries appearing on Form 1093, Application for Post Office Box, or from evidence furnished by the requestor, such as an advertising circular, that a box is being used for such a business purpose. When the postmaster is unable to determine whether a business use is involved, he shall refer the request to Regional Counsel for advice.

(5) Except as provided in (d) (4) above, the name or address of the boxholder will be furnished only to:

(i) A Federal, State, or local government agency upon prior written certification that the information is required for the performance of its duties,

(ii) A person empowered by law to serve legal process upon prior written certification that the information is required to effect service, or

(iii) In compliance with a subpoena

or other court order.

(6) 'The mailing address of any customer-sought in connection with jury service, if known, will be furnished without charge upon prior written request to a court official, such as a judge, court clerk, or jury commissioner.

(7) If the location of a residence or a place of business is known to a Postal Service employee, whether as a result of his official duties or otherwise, he may, but need not, disclose the location or give directions to it. No fee shall be charged

for such information.

(e) Information not available for public disclosure. (1) The Postal Service and its officers and employees shall not make available to the public by any means or for any purpose any mailing list or other list of names or addresses (past or present) of postal patrons or other persons.

(2) Records or other documents which are classified or otherwise specifically authorized by Executive Order to be kept secret in the interest of the national defense or foreign policy are not subject to disclosure pursuant to this part.

(3) Records consisting of trade secrets or confidential financial data, the disclosure of which is prohibited by section 1905 of Title 18, United States Code, are not subject to disclosure pursuant to this part.

(4) Other records, the disclosure of which is prohibited by statute, are not subject to disclosure pursuant to this part.

(f) Protection of the right of privacy. If any record required or permitted by this part to be disclosed contains the name of, or other identifying details concerning, any person, including an em-

ployee of the Postal Service, the disclosure of which would constitute a clearly unwarranted invasion of personal privacy, the name or other identifying details shall be deleted before the record is disclosed. A notice such as, "The names of individuals and other identifying details have been deleted to prevent an unwarranted invasion of the personal privacy of the individuals involved" shall be placed on any record from which this material has been deleted.

(g) Disclosure in part of otherwise exempt record. Any reasonably segregable portion of a record shall be provided after deleting the information which is neither subject to mandatory disclosure nor available as a matter of discretion.

# § 265.7 Procedure for inspection and copying of records.

(a) Submission of requests. (1) Form and content of request. To permit expeditious handling and timely response in accordance with the provisions of this part, a request to inspect or to obtain a copy of an identifiable Postal Service record shall be in writing and bear the caption "Freedom of Information Act Request" or otherwise be clearly and prominently identified as a request for records pursuant to the Freedom of Information Act. Form 1478 Freedom of Information Request may be used for this purpose. If submitted by mail or otherwise submitted in an envelope or other cover, a request shall be clearly and prominently identified as such on the envelope or other cover. Other requests for information will be considered informal requests and will be handled as expeditiously as practicable but not necessarily within the time limitations set forth in § 265.7(b). An informal request will be granted or denied according to the substantive rules in § 265.6, if found to be a request for a record. A Freedom of Information Act request shall identify the record sought as completely as possible, by name, description, or subject matter, and be sufficient to permit the custodian to locate it with a reasonable amount of effort. The request may state the maximum amount of fees for which the requester is willing to accept liability without prior notice. See paragraph (e) (2) § 265.8. If no amount is stated the requester will be deemed willing to accept liability for fees not to exceed \$10.00.

(2) To whom submitted. A request shall be submitted to the custodian of the requested record. If the location of the record is not known, inquiry should be directed to the Records Officer, Management Information Systems Department, United States Postal Service, Washington, D.C. 20260, telephone (202) 245–4000. If a request is submitted to a facility other than that at which the record is maintained, it shall be promptly transmitted to the appropriate custodian with a copy of the transmittal furnished to the requester. A request which is not initially submitted to the appropriate custodian shall be deemed to have been received by the Postal Service for purposes of computing the time for response

is accordance with § 265.7(b) at the time that it is actually received by the appropriate custodian. If a request seeks records maintained at two or more facilities, the custodian shall be deemed to be the next senior common supervisor of the heads of the facilities, e.g., sectional center manager, district manager, Regional Postmaster General.

(3) Reasons for request. In view of the possibility that some or all of the records may be exempt from mandatory disclosure, the requestor may state any reasons why the record should nevertheless be made available to him even if

exempt.

(4) Request for waiver of fees. The requestor may ask that fees, the advance payment of fees, or the advance deposit of estimated fees be waived in whole or in part and may state any reasons in support thereof. See § 265.8(e) (6).

(5) Categorical requests. A request for all or substantially all of the records within a specific category will be deemed a reasonable description of those records only if it is possible, without further information, to determine which particular records are sought. See para. (b) (3) of § 265.7 concerning the providing of

additional information.

(6) Request for records located at numerous facilities. A request for records which are, or may be, located at all or a substantial number of post offices or other postal facilities will be deemed to be a reasonable description only of such of those records as are maintained at the post office or other facility to which the request is submitted, and of those records maintained at any other post offices or facilities specifically identified in the request. See paragraph (a) (2) of this section concerning the custodian of records of two or more facilities.

(b) Responsibilities of the custodian.

(1) The custodian of the requested record is the person responsible for determining whether to comply with or to deny the request. A custodian who is not an Officer as defined in § 221.7, however, should not deny a request, until he has obtained the advice of Regional Counsel. If denial of a request appears necessary the custodian should seek advice as soon as possible after receipt of the request so as to provide adequate time for

legal review.

(2) The custodian shall make his determination within ten working days, i.e., exclusive of Saturdays, Sundays, and holidays, of receiving the request, and more rapidly if feasible. The custodian and the requester may, by mutual agreement, preferably in writing establish a different response period from those provided in paragraph (a) of this section.

(3) If a requested record cannot be located from the information supplied, the requester should be given an opportunity to supply additional information and, if feasible, to confer with the custodian or his representative, in an attempt to provide a reasonable description of the records sought. If additional information is furnished, the request will be deemed to have been received by the

custodian when sufficient additional information to identify and locate the record with a reasonable amount of effort

has been received.

(4) The ten working day response period allowed above may be extended by the custodian, after consultation with or with the General Counsel if the custodian is at Headquarters, Regional Counsel, for a period not to exceed an additional ten working days when, and to the extent, reasonably necessary to permit the proper processing of a particular request, under one or more of the following unusual circumstances:

 (i) The request requires a search for and collection of records from a facility other than that processing the request;

(ii) The request requires the search for, and collection and appropriate examination of, a voluminous amount of separate and distinct records; or

(iii) The request requires consultation
(A) with another agency having a substantial interest in the determination of whether to comply with the request or
(B) among two or more components of the Postal Service having substantial subject matter interest in the determination of whether to comply with the request.

(5) When the custodian finds that the additional time is required, he shall acknowledge the request in writing within the initial ten day response period, state the reason for the delay, and indicate the date on which a decision as to disclosure

is expected.

(6) If the custodian has been unable to comply with the applicable time limit provisions of this section, in spite of the exercise of due diligence, he shall nonetheless make a determination as promptly as possible. He shall notify the requester of the exceptional circumstances preventing timely compliance and of the date by which it is expected that the determination will be made. A copy of this notification shall be sent to the General Counsel, U.S. Postal Service, Washington, D.C. 20260.

(7) If a requested record is known to have been destroyed, disposed of, or otherwise not to exist, the requester

shall be so notified.

- (c) Compliance with request upon affirmative determination by custodian. (1) When a requested record has been identified and is to be disclosed in whole or in part, the custodian shall ensure that the record is made available promptly and shall immediately notify the requester where and when and upon what reasonable conditions, if any, including the payment of fees, the record will be available for inspection or copies will be available. Postal Service records will normally be available for inspection and copying during regular business hours at the postal facilities at which they are maintained. The custodian may, however, designate other reasonable locations and times for inspection and copying of some or all of the records within his custody.
- (2) Any fees, authorized or required by § 265.8 shall be paid by the requester

before the record is made available or a copy is furnished unless payment is waived or deferred pursuant to \$ 265.8 (e). See also para. (e) (2) of \$ 265/8 concerning advance notice of fees.

(3) A custodian complying with a request may designate a representative to monitor any inspection or copying.

(d) Denial of request. (1) A reply denying a request in whole or in part shall be in writing, signed by the custodian or his designee, and shall include:

(i) A statement of the reason for, or justification of, the denial, e.g., the nonexistence of the record, including, if applicable, a reference to the provision or provisions of § 265.6 authorizing the withholding of the record and a brief explanation of how each provision applies to the records requested;

(ii) The name and title or position of the person responsible for the denial of the request (see paragraph (d)(2) of

§ 265.7); and

(iii) A statement of the right to appeal and of the appeal procedure within the Postal Service (described in § 265.7

(e)).

(2) The custodian is ordinarily the person responsible for the denial of the request. If the denial of a particular request has been directed by higher authority, however, the name and title or position of the person directing the denial shall be given in the reply to the requester in place of the custodian as the person responsible for the denial, and a copy of the denial shall be sent to that person.

(e) Appeal procedure. (1) If a request to inspect or to copy a record is denied, in whole or in part, if no determination is made within the period prescribed by this section, or if a request for waiver of fees is not granted, the requester may appeal to the General Counsel, U.S. Postal Service, Washington, D.C. 20260.

(2) The requester shall submit his appeal in writing within 30 days of the date of the denial or of the other action complained of, or within a reasonable time if the appeal is from a failure of the custodian to act. The General Counsel may, in his discretion, consider late

appeals.

(3) In the event of the denial of a request or of other action or failure to act on the part of a custodian from which no appeal is taken, the General Counsel may, if he considers that there is doubt as to the correctness of the custodian's action or failure to act, review the action or failure to act as though an appeal pursuant to this section had been taken.

(4) A letter of appeal should include, as applicable:

 (i) A copy of the request, of any notification of denial or other action, and of any other related correspondence;

(ii) A statement of the action, or failure to act, from which the appeal is taken:

(iii) A statement of the reasons why the requester believes the action or failure to act is erroneous; and

(iv) A statement of the relief sought.

(f) Action on appeals. (1) The decision of the General Counsel or his designee constitutes the final decision of the Postal Service on the right of the requester to inspect or copy a record. The decision will normally be made within twenty working days from the time of the receipt by the General Counsel. The 20 day response period may be extended by the General Counsel or his designee for a period not to exceed an additional ten working days when reasonably necessary to permit the proper consideration of an appeal, under one or more of the unusual circumstances set forth in paragraph (b) (4) of § 265.7 above. The aggregate number of additional working days utilized pursuant to this paragraph (f)(1) and paragraph (b) of § 265.7, however, may not exceed

(2) The decision on the appeal shall be in writing. If the decision sustains a denial of a record, in whole or in part, it shall state the justification therefor and shall specify any exemption or exemptions relied on and the manner in which they apply to the record withheld, and shall inform the requester of his right to judicial review. An indexed file of decisions on appeals shall be maintained by the General Counsel and be made available to the public.

(3) If not prohibited by or under law, the General Counsel or his designee may direct the disclosure of a record even though its disclosure is not required by

law or regulation.

### § 265.8 Schedule of fees.

(a) Policy. The purpose of this section is to establish fair and equitable fees to permit the furnishing of records to members of the public while covering the direct costs incurred by the Postal Service.

(b) Record retrieval. (1) The fee for each quarter hour spent by clerical personnel in searching for records other than by computer is \$2.00. If no more than one quarter hour of clerical search time is required in connection with a request or a series of related requests, no charge for search time shall be made.

(2) When a search cannot be performed by clerical personnel and must be performed by professional or managerial personnel, the fee for each quarter hour in searching for records other than by computer is \$2.75 for each quarter

hour.

(3) The fee for retrieving data by computer is the actual cost of the retrieval as calculated in accordance with the Information Services Price List in effect at the time that the retrieval services are performed. The list is subject to periodic revision. A copy of the list is included within the public index. (See Appendix A for the list in effect on February 19, 1975.)

(4) Paragraphs(b) (1)(c) of this section also apply to information stored within micrographic systems.

(c) Reproduction. (1) The fee for reproducing any record or publication,

other than a change of address order, is \$.10 per page. The reproduction fee is in addition to any fee authorized by paragraph (b) of this section for the retrieval

of the same records.

(2) The Postal Service may at its discretion make coin-operated copy machines available at any location or otherwise give the requester the opportunity to make copies of Postal Service records at his own expense. Unless authorized by the Records Officer, however, no offsite copying shall be permitted of records which, if lost, could not be replaced without inconvenience to the Postal Service.

(3) The Postal Service will normally furnish only one copy of any record. If duplicate copies are furnished at the request of the requester, the per-page fee shall be charged for each copy of each page. At his discretion, when it is reasonably necessary because of a lack of adequate copying facilities or other circumstances, the custodian may make the requested record available to the requester for inspection under reasonable conditions and need not furnish a copy thereof.

(d) Other costs. (1) Publications. Publications and other printed materials may, to the extent that they are available in sufficient quantity, be made available at the established price, if any, or

at cost to the Postal Service.

(2) Other charges. When a response to a request requires services or materials other than the common ones listed in paragraphs (b) and (c) of this section, the direct cost of such services or materials to the Postal Service may be charged, but only if the requester has been notified of the nature and estimated amount of such cost before it is incurred.

(3) Change of address orders. Although change of address information is not required by the Freedom of Information Act to be made available to the public, the fee for obtaining this information in accordance with § 265.641 is included in this section as a matter of convenience to the public. The fee for searching for and reproducing change of address orders is \$1.00 per change of address. The fee is not refundable.

(e) Payment and Waiver of Fees. (1) Liability and payment. The requester is responsible, subject to limitations on liability provided by this section, for the payment of all fees for services resulting from his request, whether or not any of the requested records are made available to him. Payment shall be made before any record is made available or any copy is furnished unless payment is waived or deferred pursuant to paragraph (e) (4)-(5) of this section. Checks in payment of fees shall be made payable to "United States Postal Service."

(2) Advance notice. To protect members of the public from unwittingly incurring liability for unexpectedly large fees, a request that is expected to result in fees in excess of \$10.00 will be deemed not to have been received until the requester is notified of the estimated cost and agrees to bear it. The notification shall be transmitted as soon as possible after physical receipt of the request, but

in any event within five working days, giving the best estimate then available. It shall include a brief explanatory statement of the nature and extent of the services upon which the estimate is based and shall offer the requester an opportunity to confer with the custodian or his representative in an attempt to reformulate the request so as to meet his needs at lower cost. No notification is required if the request specifically states that whatever cost is involved is acceptable or is acceptable up to a specified amount that covers estimated costs or if payment of all fees in excess of \$10.00 has been waived.

(3) Advance deposits. When it is estimated that the fees chargeable under this section will amount to more than \$100.00, an advance deposit of not less than 50 percent of estimated fees shall be required, unless the payment of fees in excess of \$100.00 has been waived. The deposit should be made within 5 working days of receipt by the requester of notice of the requirement. The determination of the availability of the records sought by the requester shall not be delayed to await this deposit, if such determination will involve search charges of less than \$25.00. In other cases, however, the determination of the availability of the records shall be delayed to await the deposit and the request will not be deemed to have been received until the deposit is received.

(4) Waiver of fees where records are not disclosed. Ordinarily, fees shall not be charged if the requested records are not found, or if all of the records located are withheld as exempt. However, search fees may be charged if the time spent in searching for the requested record warrants charges in excess of \$25.00 and the requester was notified of the estimated cost, either pursuant to paragraph (e)(2) of this section or in a separate notification, and of the fact that it could not be determined in advance whether any records would be made available and that he might be responsible for search fees even though no records are made available, unless the requester promptly withdraws or modifies his request upon receipt of such notification.

(5) Fees not charged for certain services. Fees shall not be charged where they would amount, in the aggregate, for a request or series of related requests, to less than \$3.00. This general waiver shall not apply to the fee for providing change

of address information.
(6) Waiver of fees by custodian. The custodian may waive, in whole or in part, a fee not in excess of \$25.00 or the requirement for the advance payment of such a fee, when he determines, in accordance with guidelines established by the Records Officer, that the furnishing of the records is primarily for the benefit of the general public, or that charging the fee otherwise required would impose an undue hardship or inconvenience on the requester. If the custodian determines that waiver is appropriate, but the amount of the fee or the deposit is in excess of his authority to waive, he shall

promptly submit a recommendation for such a waiver to the Officer exercising jurisdiction over his post office or facility or to the General Counsel. Until the Officer has acted on the recommendation, the custodian shall require any advance deposit or advance payment otherwise warranted by this section to be made but shall inform the requester that waiver is under consideration and may defer any action requiring deposit or payment, if desired by the requester, until waiver has been granted or denied.

(7) Waiver by officer. Any Officer of the Postal Service, as defined in § 221.7 or his designee, may waive in whole or in part any fee required by this part or the requirement for advance payment or ad-

vance deposit of any fee.

(8) Waiver of fee for changes of address. The fee prescribed by paragraph (d) of this section is waived in the following circumstances for providing change of address information for:

(i) Telegraph companies when sender of the telegram is the U.S. Gov-

ernment.

(ii) Federal, state, and local public health officials when the persons being sought are infected with or exposed to

contagious diseases.

(iii) Federal, state, and local government agencies, upon certification that the change of address is required for the performance of their duties, and all other known sources for obtaining the change of address have been exhausted. This waiver does not apply to fees for services performed in accordance with § 122.5 of

(iv) Postage meter manufacturers when they are attempting to locate a

missing meter.

(f) Accounting for fees. Custodians shall account for fees paid in accordance

with this section as follows:

(1) Deposit fees received as postal funds. Record the amounts collected as write-in entries to A/C 49299, Miscellaneous, Other than U.S. Government Agencies, in the cashbook and statement of account. Record the manner paid, the amount received, and the number of hours used to compile lists or prepare copies of other records released on the request for this information. Attach written replies to the customer's request stating the number of hours required to prepare information and the amount to be charged in lieu of the above notation, File materials chronologically.

(2) Forward fees received for information furnished by postal data centers, automatic data processing centers (ADPC), and regional offices to the disbursing officer at the appropriate postal data center for deposit, specifying the proper account number to be used for recording the amounts collected. Postal data centers, ADPC's, and Headquarters offices providing record retrieval as described in paragraph (b) (3) of this section, plus the fees covered in paragraphs (b) (1) and (2), (c) and (d) of this section will enter the fees in A/C 40990, Miscellaneous; ADPC complexes will enter in A/C 49299, Miscellaneous, Other than U.S. Government Agencies. Other installations will enter all fees deposited to a request from a non-Federal source in A/C 40990.

#### § 265.9 Annual reports.

A report concerning the administration of the Freedom of Information Act and this part for the preceding calendar year is to be submitted to the Congress by March 1 of each year. To permit the timely compilation and submission of this annual report, each custodian shall submit to the Records Officer a report of each request received pursuant to § 265.7 (a) (1) immediately following the custodian's action on it. The Records Officer shall prescribe the form and content of the report and may vary the time for the submission of reports by individual custodians to avoid undue duplication or frequency of submissions.

#### § 265.10 Compliance with subpocna duces tecum court orders and summonses.

(a) Compliance with subpoena duces tecum. (1) Except as required by Part 262, produce other records of the Postal Service only in compliance with a subpoena duces tecum or appropriate court order.

(2) Time, leave, and payroll records of postal employees are subject to production when a subpoena duces tecum or appropriate court order has been properly served. The custodian of the records may designate a postal employee to present the records. The presentation by a designee rather than the employee named in the subpoena or court order must meet with the approval of the attorneys for each side. In addition, such records may be released if authorized in writing by the employee.

(3) If the subpoena involves a jobconnected injury, the records are under the exclusive jurisdiction of the Bureau of Employees' Compensation, Department of Labor. Requests for authorization to produce these records shall be addressed to: Bureau of Employees' Compensation, U.S. Department of Labor, Washington, DC 20210. Also notify the attorney responsible for the issuance of the subpoena or court order.

(4) Employee medical records are primarily under the exclusive jurisdiction of the U.S. Civil Service Commission. The Commission has delegated authority to the Postal Service and to the Commission's Regional Directors to release medical information, in response to proper requests and upon competent medical advice, in accordance with the following criteria:

(i) Except in response to a subpoena or court order, do not release any medical information about an employee to any non-Federal entity or individual without authorization from the employee.

(ii) With authorization from the employee, the Regional Postmasters General, Postal Data Center Directors, or regional counsels will respond as follows

for medical information:

(a) If, in the opinion of a Federal medical officer, the medical information indicates the existence of a malignancy, a mental condition, or other condition about which a prudent physician would hesitate to inform a person suffering from such a condition as to its exact nature and probable outcome, do not release the medical information to the employee or to any individual designated by him, except to a physician, designated by the employee in writing. If a subpoena or court order was issued, the responding official shall caution the moving party as to the possible dangers involved if the medical information is divulged.

(b) If, in the opinion of a Federal medical officer, the medical information does not indicate the presence of any condition which would cause a prudent physician to hesitate to inform a person of the exact nature and probable outcome of his condition, release it in response to a subpoena or court order, or to the employee or to any person, firm, or organization he authorizes in writing.

(c) If a Federal medical officer is not available, refer the request to the Civil Service Commission regional office with the medical certificates or other medical reports concerned.

(5) Do not release any records contaming information as to the employee's security or loyalty.

(6) Honor subpoenas or court orders only when disclosure is authorized.

(7) When authorized to comply with a subpoena duces tecum, do not leave the original records with the court.

(b) Compliance with summons. (1) Comply with a summons requiring an appearance in court. Do not testify as to any matters for which an exemption under § 261.2(c) may be claimed. Call the Regional Counsel for instructions relating to exemptions.

(2) Do not present inspectors' reports or inspection Service records in either State or Federal courts in which the United States is not a party in interest, unless authorized by the Regional Chief Inspector, who will make a decision after consulting with Regional Counsel, If an attempt is made to compel the production of matters, decline to produce the information or matter, and state it may be exempted and cannot be disclosed or produced without specific approval of the Regional Chief Inspector, who will make a decision after consulting with Regional Counsel. The Postal Service will offer every possible assistance to the courts, but the questions of disclosing information for which an exemption may be claimed is a matter of discretion.

#### APPENDIX A

INFORMATION SERVICES PRICE LIST IN EFFECT FEBRUAR	¥ 19, 1975
System utilization charges:	
Processor utilization S/360 Model 65	
Selector channel utilization	52.40/channel hour.
Multiplexor channel utilization	18.52/channel hour.
System occupancy charges:	
Processor storage	.53/K/occupancy hour.
Extended core storage	.24/K/occupancy hour.
2314 Disk	22.38/occupancy hour.
2400 Tape drive	14.50/occupancy hour.
1288 Scanner	48.44/occupancy hour.
3330 Disk	9.12/occupancy hour.
System spooling charges:	
Local card reading	.17/1,000 cards.
Remote card reading	
Local printing	
Remote printing	
Local punching	51/1.000 cards.
Remote punching	
Peripheral charges:	
H-1200 Processing	67.25 hours.
Keypunch	
Programming support	
Programming support O/T	29.40 hours.
Systems analysis support	
Systems analysis support O/T	
Data transmission	
Special forms, 1 part	
Special forms 2 part	
Special forms, 3 part.	
Special forms, 4 part	
Special forms, 6 part	
Special forms, multilith	
Statistical services and support:	widoo per page.
Statistical system design and development	891 00 hour
System maintenance and operation	
· Special statistical projects.	
Model development and analysis	
Management science or services	
Swetam implementation	
Dyborn implementation	or of the state of

[FR Doc.75-4631 Filed 2-18-75;8:45 am]

Title 12-Banks and Banking

CHAPTER I—COMPTROLLER OF THE CURRENCY, DEPARTMENT OF THE TREASURY

PART 4—DESCRIPTION OF OFFICE, PROCEDURES, PUBLIC INFORMATION

Freedom of Information Act Regulations

On November 21, 1974, Pub. L. 93-502 was enacted. This law, referred to as the 1974 Amendments to the Freedom of Information Act, amended 5 U.S.C. 552. It is necessary for the Comptroller of the Currency to amend §§ 4.16 and 4.17 of Part 4 of 12 CFR and add a new § 4.17a to Part 4 of 12 CFR to bring the regulation into conformity with 5 U.S.C. 552 as amended by Pub. L. 93-502.

These amendments are issued under authority of the national banking law (12 U.S.C. 1 et seq.) and 5 U.S.C. 552. Since the amendments add information to and clarify the regulation setting out the availability of public information, and the procedures to acquire access to records, notice and public procedure are found to be unnecessary and not in the public interest. Accordingly, the amendments will become effective upon publication.

In 12 CFR Part 4, § 4.16 is revised to read as follows:

§ 4.16 Other records available to public; exceptions.

(a) In addition to the records referred to in § 4.15 of this part, all other records of the Comptroller of the Currency are available to any person for inspection and copying in accordance with §§ 4.17 and 4.17a, except as provided in paragraph (b) of this section.

(b) Except as specifically authorized by the Comptroller of the Currency, the following records, or portions thereof, are

not available to the public:

(1) A record, or portion thereof, which is specifically authorized under criteria established by an Executive order to be kept secret in the interest of national defense or foreign policy and which is, in fact, properly classified pursuant to such Executive order.

(2) A record, or portion thereof, relating solely to the internal personnel rules

and practices of an agency.

(3) A record, or portion thereof, specifically exempted from disclosure by

(4) A record, or portion thereof, which is privileged or contains trade secrets and commercial or financial information which relates to the business, personal, or financial affairs of any person and is furnished in confidence.

(5) An intra-agency or interagency memorandum or letter that would not be routinely available by law to a private party in litigation, including, but not limited to, memoranda, reports, and other documents prepared by the personnel of the Comptroller of the Currency and records of deliberations and discussions at meetings of the personnel of the Comptroller of the Currency.

(6) A personnel, medical, or similar record, including a financial record, or any portion thereof, the disclosure of which would constitute a clearly unwarranted invasion of personal privacy.

(7) Investigatory records compiled for law enforcement purposes, including investigatory records relating to a proceeding for the issuance of a cease and desist order or order of suspension or removal under the Financial Institutions Supervisory Act of 1966, and the granting, withholding, or revocation of any approval, permission, or authority, but only to the extent that the production of such records would (i) interfere with enforcement proceedings; (ii) deprive a person of a right to a fair trial or an impartial adjudication, (iii) constitute an unwarranted invasion of personal privacy, (iv) disclose the identity of a confidential source and, in the case of a record compiled by a criminal law enforcement authority in the course of a criminal investigation, or by an agency conducting a lawful national security intelligence investigation, confidential information furnished only by the confidential source. (v) disclose investigative techniques and procedures, or (vi) endanger the life or physical safety of law enforcement personnel.

(8) A record, or portion thereof, containing or relating to an examination, operating, or condition report prepared by, or on behalf of, or for the use of, the Comptroller of the Currency or any other agency responsible for the regulation or supervision of financial institutions, relating to the affairs of any bank or affiliate thereof, bank holding company or subsidiary, broker, finance company, or any other person engaged, or proposing to engage, in the business of banking, extending credit, or managing or controlling banks.

(9) A record, or portion thereof, which contains or is related to geological and geophysical information and data,

including maps, concerning wells. (c) Even if an exemption described in paragraph (b) of this section may be reasonably applicable to a requested record, or portion thereof, the Comptroller of the Currency may elect under the circumstances of any particular request not to apply the exemption to such requested record, or portion thereof. The fact that the exemption is not applied by the Comptroller of the Currency to any requested record, or portion thereof, has no precedential significance as to the application or nonapplication of the exemption to any other requested record. or portion thereof, no matter when the request is received.

(d) Reasonably segregable portions of a record shall be provided to any person properly requesting such record in accordance with §§ 4.17 and 4.17a of this Part after deletion of the portions which are exempt under paragraph (b) of this section.

In 12 CFR Part 4, § 4.17 is revised to read as follows:

§ 4.17 Location of public reading rooms, requests for identifiable records; and service of process.

(a) General. This section identifies the locations of the public reading rooms at which records of the Comptroller of the Currency are available for public inspection and copying, the titles of officers designated to make the initial and appellate determinations with respect to requests, the officer designated to receive service of process, the addresses for delivery of requests, appeals and service of process, and the required content of requests.

(b) Public reading rooms. Public reading rooms or public reading areas for the Comptroller of the Currency are maintained at the following locations for the records as stated in paragraph (c) of

this section:

(1) Central office. (1) Mailing address.

Special Assistant for Public Affairs

Comptroller of the Currency

Washington, D.C. 20219

(ii) Location.

Sixth Floor 490 L'Enfant Piaza East, SW Washington, D.C.

(2) Regional offices. (i) The mailing address and location for each respective region is the same and is set forth in

§ 4.1a(b) of this Part.

(c) Locations of certain records. All public records of the Comptroller of the Currency, except the public portions of applications by national banking associations to establish a branch or seasonal agency and the public portion of applications to organize a national banking association during the period such applications are in the investigatory process in the respective regions, are available in the central office listed in paragraph (b) (1) of this section. During this investigatory period, the public portions of these applications will be available in the respective regions as listed in § 4.1a(b) of this Part.

(d) Where to send requests for records. (1) Initial determinations as to whether to grant requests for (i) Consolidated Reports of Condition (Domestic), (ii) Consolidated Reports of Condition (Foreign and Domestic), (iii) Consolidated Reports of Income, (iv) Annual Reports of Equity Security Holdings for Banks having Trust Powers and Equity Security Holdings of \$75 Million or More, or (v) Quarterly Reports of Stock Transactions of Banks Submitting the Report of Equity Security Holdings listed in part (vi) of this subparagraph (1), will be made by the Director of Statistical Division or his delegate. Requests for records listed in this paragraph (d) (1) shall be submitted to:

(1) Mailing address.

Director of Statistical Division
Comptroller of the Currency

Washington, D.C. 20219

(ii) Location.

Sixth Floor 490 L'Enfant Plaza East, SW Washington, D.C.

(2) Initial determinations as to whether to grant requests for all records of the Comptroller of the Currency available under \$\$ 4.15 and 4.16 of this Part. other than those described in paragraph (d) (1) of this section and other than those described in paragraph (c) of this section as being located in the Regional Offices, will be made by the Special Assistant for Public Affairs or his delegate. Requests for all records, other than those listed in paragraph (d) (1) of this section and other than those described in paragraph (c) of this section as being located in the Regional Offices, shall be submitted to:

(i) Mailing address.

Special Assistant for Public Affairs Comptroller of the Currency Washington, D.C. 20219

(ii) Location.

Sixth Floor 490 L'Enfant Plaza East, SW Washington, D.C.

(3) Initial determination as to whether to grant requests for records described in paragraph (c) of this section as being located in the Regional Offices shall be made by the Regional Administrator of National Banks for each respective region. All requests for such records shall be submitted to the respective Regional Administrator at the location set

forth in § 4.1a(b) of this part.

(e) Administrative appeal of initial determination to deny records. Appellate determinations with respect to records of the Comptroller of the Currency will be made by the Comptroller or his delegate. All appeals shall be submitted in writing within 30 days of the initial denial and shall state the circumstances, reasons or arguments advanced for insistence upon disclosure of the requested records, and

be submitted to: (1) Mailing address.

Special Assistant for Public Affairs omptroller of the Currency Washington, D.C. 20219

(2) Location.

Sixth Floor 490 L'Enfant Plaza East, SW Washington, D.C.

- (f) Delivery of process. Service of process will be received by the Chief Counsel, Comptroller of the Currency and shall be delivered to the following location:
  - (1) Mailing address.

Chief Counsel Comptroller of the Currency Washington, D.C. 20219

(2) Location.

Office of the Chief Counsel Comptroller of the Currency Sixth Floor, 490 L'Enfant Plaza, East, SW Washington, D.C. 20219

(g) Content of request. A request for records of the Comptroller of the Currency available under §§ 4.15 and 4.16 of this Part must be in writing and state the full name, address and telephone number of the person requesting access to such records and a reasonable de-

scription of the records sought. A reasonable description shall include sufficient detail to enable the personnel of the Comptroller of the Currency who are familiar with the subject area of the request to locate the records with a reasonable amount of effort.

(h) Fees. A person requesting access to a record of the Comptroller of the Currency shall pay a fee of \$5 for the cost of locating and preparing the record for inspection and copying. Additional fees of ten cents per page will be charged for providing a copy of the record and of \$1.25 for each quarter hour spent beyond the first hour in locating, preparing, or copying the record.

In 12 CFR, Part 4 is amended by adding a new § 4.17a reading as follows:

### § 4.17a Request procedures.

(a) Date of receipt of request or appeal. (1) A request for records shall be deemed to have been received on the date which:

(i) the requirements of paragraph (g) of § 4.17 of this Part have been satisfied,

and

(ii) it is received in the office of the initial determining officer for the request as set out in § 4.17 of this Part;

except

(iii) where the provisions of paragraphs (a) (1) (i) and (ii) of this section are satisfied, but the requester has asked the Comptroller of the Currency to submit a notice of anticipated fees, in which case the date of receipt of the request shall be deemed to be the date a response in writing stating that the anticipated fees are acceptable to the requester is received in the office of the initial determining officer for the request as set out in § 4.17 of this part, or

(iv) where the provisions of subparagraph (1) (i) and (ii) of this paragraph (a) are satisfied and a request for an agreement to pay the anticipated fees or deposit is required by the Comptroller of the Currency, and the requester is notified of such requirement, the date of receipt of the request shall be deemed to be the date the affirmative response agreeing to pay the anticipated fees or the amount of the requested deposit is received in the office of the initial determining officer for the request.

(2) An appeal of a denial of a request for records, or a portion thereof, which request was deemed received by reason of satisfying the requirements of subparagraph (1) of this paragraph (a) shall be deemed received when the requirements of paragraph (e) of § 4.17 of this part have been satisfied and the appeal is received in the office designated in paragraph (e) of § 4.17 of this part.

(b) Initial determination. Initial determinations as to whether to grant a request for records will be made by the initial determining officer for the request as set out in § 4.17 of this part.

(1) If request is granted. When a request for access to records is granted, in whole or in part, a copy of the records made available pursuant to the request shall promptly be delivered to the re-

quester or made available for his inspection, whichever was requested. Records will be made available for inspection, or for duplication when copies of records are requested to be delivered, at such reasonable and proper times as not to interfere with their use by the Comp-troller of the Currency or to exclude other persons from making inspections. Reasonable limits may be placed on the number of records which may be in-

spected by a person on any given date.
(2) If request is denied. When a request for access to records is denied, in whole or in part, the requester shall be notified by mail. The notification shall contain a brief statement of the reasons for the denial, set forth the name and title or position of the official making the decision, and advise the requester of his right to an administrative appeal in

accordance with § 4.17 of this part. .

(c) Determination of appeals. (1) If an appeal is granted. When an administrative appeal is granted, in whole or in part, the request shall be treated as if it were originally granted, in whole or in part, by the initial determining officer and the provisions of paragraph (b) (1)

of this section shall apply.
(2) If appeal is denied. If an administrative appeal is denied, in whole or in part, the requester shall be notified by mail. Such notification shall contain a brief statement of the reasons for the denial, set forth the name and title or position of the official making the decision and advise the requester-appellant of his right to judicial review of the denial under 5 U.S.C. 552(a) (4)

(d) Time limits. (1) The initial determination to grant or deny a request for records shall be made and notification shall be mailed within ten (10) days (excepting Saturdays, Sundays and legal holidays) after the date the request is deemed received pursuant to paragraph (a) of this section, except as stated in paragraph (d) (3) of this section.

(2) The determination to grant or deny an administrative appeal shall be made-and notification shall be mailed within twenty (20) days (excepting Saturdays, Sundays and legal holidays) after the date the appeal is deemed received pursuant to paragraph (a) of this section, except as stated in subparagraph

(3) of this paragraph (d).

(3) Time extensions by administrative action. The time limitations set forth in paragraphs (d) (1) and (2) of this section may be extended in unusual circumstances for a maximum of ten (10) days by written notice to the person making the request or appeal. Such notification shall contain the reasons for such extension and the date on which the determination is expected to be dispatched. As used in this subparagraph (3), "unusual circumstances" means, but only to the extent reasonably necessary to the proper processing of the particular request or appeal.

(i) The need to search for and collect the requested records from field facilities or other buildings that are separate from the building in which the office of the official to whom the request or

appeal is made is located,

(ii) The need to search for, collect, and appropriately examine a voluminous amount of separate and distinct records which are demanded in a single request

or appeal, or

(iii) The need for consultation, which shall be conducted with all practicable speed, with another agency having a substantial interest in the determination of the request or among two or more components of the Comptroller of the Currency having substantial subject matter interest therein, other than consultation with the legal staff, the public affairs staff or with the Department of

(4) Time extension by agreement. A requester may agree to an extension of the time limits specified in paragraphs (d) (i) and (ii) of this section. In event of such an agreement, the respective specified time limit shall not be considered as having expired until the period of the extension of time agreed to by the

requester has expired.

(5) Time extension by judicial action. . If a suit is properly instituted against the Comptroller of the Currency by a requester, the court may, pursuant to 5 U.S.C. 552(a) (6) (c), allow the Comptroller of the Currency additional time to complete the review of the records requested.

Effective date. These amendments are effective February 19, 1975.

Dated: February 14, 1975.

THOMAS G. DESHAZO. Acting Comptroller of the Currency. [FR Doc.75-4673 Filed 2-18-75;8:45 am]

#### CHAPTER VI-FARM CREDIT **ADMINISTRATION**

#### PART 602-RELEASING INFORMATION Freedom of Information Act Regulations

The Farm Credit Administration publishes herewith amendments to its regulations in order to conform them to the requirements of the Freedom of Information Act as amended by Pub. L. 93-502, 88 Stat. 561. These amendments (1) clarify which records are exempted from disclosure, (2) determine that the publication of an index of records available for public inspection and copying is unnecessary and impracticable, (3) prescribe time limits for responses to requests, (4) prescribe a time limit within which an appeal must be taken, and (5) revise the uniform schedule of fees which may be charged for requested records. These amendments make extensive revisions in regulations pertaining to the Freedom of Information Act. Therefore, for convenience of the public, all of such regulations of the Farm Credit Administration are included in this publication.

On January 14, 1975, the Farm Credit Administration published in the FED-ERAL REGISTER (40 FR 2590) a proposed revision of the uniform schedule of fees, and stated that prior to final adoption consideration would be given to any writ-

ten comments received no later than February 14, 1975. No comments have been received.

As to the amendment pertaining to the uniform fee schedule, the relevant provisions of the Administrative Procedure Act (5 U.S.C. 553) and the Freedom of Information Act (5 U.S.C. 552(a) (4) (A)) for public participation, and delay in effective date have been followed. Because the remaining regulation amendments pertain to matters of procedure and policy and are necessary to achieve compliance with the amendments to the Freedom of Information Act which become effective on February 19, 1975, such relevant provisions of the Administrative Procedure Act are inapplicable. Accordingly, the amendments published herewith become effective on February 19, 1975.

Chapter VI of Title 12 of the Code of Federal Regulations is amended by revising Subpart B of Part 602 to read as

follows:

#### PART 602—RELEASING INFORMATION

#### Availability of Records of the Farm Credit Administration Subpart B

Sec.

602.250 Official records of the Farm Credit Administration.

602.251 Current index.

602.255 Identification of records requested. 602.260 Request for records.

602.261

Response to requests for records. 602.265 Fees for provision of records.

#### Subpart B---Availability of Records of the **Farm Credit Administration**

#### § 602.250 Official records of the Farm Credit Administration.

(a) The Farm Credit Administration shall, upon any request for records which reasonably describes them and is made in accordance with the provisions of this Subpart B, make the records available as promptly as is practicable to any person, except exempt records which include the following:

(1) Records specifically authorized under criteria established by an Executive order to be kept secret in the interest of national defense or foreign policy and are in fact properly classified pursuant

to such Executive order;

(2) Records related solely to the internal personnel rules and practices of the Farm Credit Administration, including matters which are for the guidance of agency personnel;

(3) Records which are specifically exempted from disclosure by statute; (4) Commercial or financial information obtained from any person or organi-

zation and privileged or confidential; (5) Interagency or intra-agency memorandums or letters which would not be available by law to a private party in litigation in which the United States, as real party in interest on behalf of the Farm Credit Administration, is a party, or from a bank or association supervised by the Farm Credit Administration to a private party in litigation with such

dums or letters are records of such bank or association:

(6) Personnel and similar files, the disclosure of which would constitute a clearly unwarranted invasion of personal

(7) Investigatory records compiled for law enforcement purposes, but only to the extent that the production of such records would (i) interfere with enforcement proceedings, (ii) deprive a person of a right to a fair trial or an impartial adjudication, (iii) constitute an unwarranted invasion of personal privacy, (iv) disclose the identity of a confidential source and, in the case of a record compiled by a criminal law enforcement authority in the course of a criminal investigation, or by an agency conducting a lawful national security intelligence investigation, confidential information furnished only by the confidential source, (v) disclose investigative techniques and procedures, or (vi) endanger the life or physical safety of law enforcement personnel:

(b) Any reasonably segregable portion of a record shall be provided to any person requesting such record after deletion of the portions which are exempt

under this subsection. (c) This section does not authorize withholding of information or limit the availability of records to the public, except as specifically stated in this section. This section is not authority to withhold information from Congress.

### § 602.251 Current index.

The Farm Credit Administration will make available for public inspection and copying a current index to provide identifying information as to any matter required by 5 U.S.C. 552(a)(2)(C) to be made available or published in the FED-ERAL REGISTER. Because of the infrequent requests in the past for material required to be indexed, it is determined that the publication of the index in the FEDERAL REGISTER is unnecessary and impractical. However, the Farm Credit Administration will provide a copy of such index to a member of the public upon request therefor at a cost not in excess of the direct cost of duplication.

#### § 602.255 Identification of records requested.

A member of the public who requests records from the Farm Credit Administration shall provide a reasonable description of the records sought including. where possible, specific information as to dates, titles, and subject matter, so that such records may be located without undue search or inquiry. If a record is not identified by a reasonable description, the request therefor may be denied.

#### § 602.260 Request for records.

Requests for records, other than records identified in § 602.265(a) which are available in a public reference facility in the offices of the Farm Credit Administration, shall be in writing, in an envelope clearly marked "FOIA Request," and addressed to the Director of Information,. Office of Administration, Farm Credit

bank or association if such memoran-

Administration, Washington, D.C. 20578. A request improperly addressed will be deemed not to have been received for purposes of the ten-day time period set forth in § 602.261(a) until it is received, or would have been received with the exercise of due diligence by Agency personnel, in the Office of the Director of Information. Records requested in conformance with this Subpart B and which are not exempt records may be received in person or by mail as specified in the request. Records to be received in person will be available for inspection or copying during business hours on a regular business day in the public reference facility in the offices of the Farm Credit Administration which are located in Suite 4000, 490 L'Enfant Plaza East SW, Washington, D.C. 20578.

§ 602.261 Response to requests for records.

(a) Within ten days (excluding Saturdays, Sundays, and legal public holidays), or any extension thereof as provided in paragraph (d) of this section, of the receipt of a request in the office of the Director of Information, the Director shall determine whether to comply with or to deny such request and place a notice thereof in writing in the mails addressed to the requester.

(b) Within thirty days of the receipt of a notice denying, in whole or in part, a request for records, the requester may appeal the denial. The appeal shall be in writing addressed to the Deputy Governor, Office of Administration, Farm Credit Administration, and both the letter and envelope shall be clearly marked "FOIA Appeal." An appeal improperly addressed shall be deemed not to have been received for purposes of the twenty-day time period set forth in paragraph (c) of this section until it is received, or would have been received with the exercise of due diligence by Agency personnel, in the office of the Deputy Governor, Office of Administration.

(c) Within twenty days (excluding Saturdays, Sundays, and legal public holidays), or any extension thereof as provided in paragraph (d) of this section, of the receipt of an appeal in the office of the Deputy Governor, Office of Administration, the Deputy Governor shall act upon the appeal and place a notice of the determination thereof in writing in the mails addressed to the requester. If the determination on the appeal upholds in whole or in part the denial of the request for records, or, if a determination on the appeal has not been mailed at the end of the twenty-day period or the last extension thereof, the requester is deemed to have exhausted his administrative remedies, giving rise to a right of review in a district court of the United States as specified in 5 U.S.C. 552(a) (4). When a determination cannot be mailed within the applicable time limit, the appeal will nevertheless be processed. In such case, upon the expiration of the time limit the requester will be informed of the reason for the delay, of the date on which a determination

may be expected to be mailed, and of his right to seek judicial review. The requester may be asked to forego judicial review until determination of the appeal.

(d) In unusual circumstances as specified in this paragraph the ten-day time limit prescribed in paragraph (a) of this section or the twenty-day time limit prescribed in paragraph (c), or both, may be extended by the Director of Information or the Deputy Governor, Office of Administration, as the case may be, provided that the total of all extensions shall not exceed ten days (excluding Saturdays, Sundays, and legal public holidays). Extensions shall be made by written notice to the requester setting forth the reasons for the extension and the date on which a determination is expected to be mailed. As used in this "unusual circumstances" paragraph. means, but only to the extent necessary to the proper processing of the request:

(1) the need to search for and collect the requested records from field facilities or other establishments that are separate from the office processing the

request;

(2) the need to search for, collect, and appropriately examine a voluminous amount of separate and distinct records which are demanded in a single request;

(3) the need for consultation, which shall be conducted with all practicable speed, with another agency having a substantial interest in the determination of the request or among two or more components of the agency having substantial subject-matter interest therein. 602.265 Fees for provision of records.

(a) The Farm Credit Administration furnishes a member of the public free of charge a reasonable quantity of information that has been printed or otherwise reproduced for the purpose of making it available to the public without charge.

(b) The Farm Credit Administration furnishes a member of the public free of charge information that is requested and is not exempt from disclosure when the information is readily available and can be furnished by the Farm Credit Administration without charge.

(c) When a request for information which cannot be furnished under paragraphs (a) and (b) of this section is received, fees shall be charged in accordance with the schedule contained in paragraph (d) of this section for services rendered in response to requests for Farm Credit Administration records under this Subpart B unless the Deputy Governor, Office of Administration, determines that such charges or a portion thereof are not in the public interest because furnishing the information can be considered as primarily benefiting the general public. Fees shall not be charged where they would amount, in the aggregate, for a request or series of related requests, to less than \$5. Fees shall not be charged if the records requested are not found, or if all of the records located are withheld as exempt. However, if the time expended in processing the request

is substantial, and if the requester has been notified of the estimated cost pursuant to paragraph (d) of this section and has been specifically advised that it cannot be determined in advance whether any records will be made available, fees may be charged.

(d) For the services listed below expended in locating or making available records or copies thereof, the following

fees shall be charged:

(1) For copies of documents (maximum of 10 copies will be supplied), \$1.10

per copy of each page.

(a) For each one quarter hour spent by clerical personnel in excess of the first quarter hour in searching for and producing a requested record, including services to transport personnel to places of record storage, or records to the location of personnel for the purpose of search, \$1.50.

(3) For certification of true copies,

each, \$1.
(4) For attestation under the seal of

the Administration, \$3.

(5) Where a search cannot be performed by clerical personnel, for example, where the task of determining which records fall within a request and collecting them requires the time of professional or managerial personnel, and where the amount of time that must be expended in the search and collection of the requested records by such higher level personnel is substantial, charges for the search may be made at a rate in excess of the clerical rate, namely for each one quarter hour spent in excess of the first quarter hour by such higher level personnel in searching for a requested record, \$3.15.

(6) No charge shall be made for time spent in resolving legal or policy issues affecting access to records of known contents. In addition, no charge shall be made for the time involved in examining records in connection with determining whether they are exempt from mandatory disclosure and should be withheld as

a matter of sound policy.

(7) Fees for services in processing requests maintained in whole or part in computerized form shall be in accordance with this section so far as practicable. Services of personnel in the nature of a search shall be charged for at rates prescribed in paragraph (b) (5) of this section unless the level of personnel involved permits rates in accordance with paragraph (b) (2) of this section. A charge shall be made for the computer time involved, based upon the prevailing level of costs to the Farm Credit Administration and upon the particular types of computer and associated equipment and the amounts of time on such equipment that are utilized. A charge shall also be made for any substantial amounts of special supplies or materials used to contain, present, or make available the output of computers, based upon prevailing levels of costs to the Farm Credit Administration and upon the type and amount of such supplies or materials that is used. Nothing in this paragraph shall be construed to entitle any person, as of right, to any

services in connection with computerized records, other than services to which such person may be entitled under the provisions, not including this paragraph (b), of this Subpart B.

(e) Where it is anticipated that the fees chargeable under this section will amount to more than \$25, and the requester has not indicated in advance his willingness to pay fees as high as are anticipated, the requester shall be promptly notified of the amount of the anticipated fee or such portion thereof as can readily be estimated. When the anticipated fees exceed \$50 a deposit of 50 percent of the anticipated fees must be made within 5 business days of the Farm Credit Administration's notice to the requester. Unless the request specifically states that whatever cost is involved will be acceptable, or acceptable up to a specified limit, a request that is expected to involve fees in excess of \$25 will not be deemed to have been received for purposes of this Subpart B until the requester is notified of the anticipated cost and his agreement to bear it is received. The notice or request for an advance deposit shall extend an offer to the requester to confer with identified Farm Credit Administration personnel in an attempt to reformulate the request in a manner which will reduce the fees and meet the needs of the

(f) Payment shall be made by check or money order payable to the Farm Credit

Administration.

requester.

(Secs. 5.9, 5.10, 5.12, 5.18, 85 Stat. 619, 620, 621; Pub. L. 93-502 amending 5 U.S.C. 552, 88 Stat. 1561)

W. M. HARDING. Governor Farm Credit Administration. FR Doc.75-4726 Filed 2-18-75:11:00 am l

# Title 7—Agriculture SUBTITLE A-OFFICE OF THE SECRETARY OF AGRICULTURE PART 1—ADMINISTRATIVE REGULATIONS

Subpart A-Official Records FREEDOM OF INFORMATION

Subpart A, Part 1, Subtitle A of 7 CFR is revised to amend the procedures for making and acting upon requests for information under the Freedom of Information Act, and for the disclosure of records pursuant to compulsory process as follows:

Sec.

1.1 Purpose and scope.

Public access to certain materials.

1.3 Requests for records.

Agency implementing regulations.
Agency response to requests for rec-1.4 1.5

Date of receipt of requests or appeals. 1.6

1.7 Appeals.

Extension of administrative deadlines. 1.9 Failure to meet administrative deadlines.

Fee schedule.

Exemptions and discretionary release.

1.12

Annual report.
Compilation of new records.

Authentication.

Compulsory process.

1.16 Records in formal adjudication proceedings.

AUTHORITY: (5 U.S.C. 301, 552).

#### § 1.1 Purpose and scope.

This subpart provides regulations of the Department of Agriculture implementing the Freedom of Information Act, 5 U.S.C. 552, pursuant to which official records may be obtained by any person. It also provides regulations pertaining to the disclosure of records pursuant to compulsory process. Unless prohibited by other laws, Executive Orders, or regulations, agencies of the Department may make available records which they are authorized to withhold under 5 U.S.C. 552 whenever they determine that such disclosure is in the public interest.

#### § 1.2 Public access to certain materials.

(a) In accordance with 5 U.S.C. 552 (a) (2) each agency within the Department shall make the following materials available for public inspection and copying (unless they are promptly published and copies offered for sale):

(1) Final opinions, including concurring and dissenting opinions, as well as orders, made in the adjudication of

cases;
(2) Those statements of policy and interpretation which have been adopted by the agency and are not published in the Federal Register; and

(3) Administrative staff manuals and instructions to staff that affect a mem-

ber of the public.

(b) Each agency of the Department shall also maintain and make available current indexes providing identifying information regarding any matter issued, adopted or promulgated after July 4, 1967, and required by paragraph (a) of this section to be made available or published. Each agency shall publish and make available for distribution, copies of such indexes and supplements thereto at least quarterly, unless it determines by Notice published in the FEDERAL REGISTER that publication would be unnecessary and impracticable. After issuance of such Notice, the agency shall provide copies of any index upon request at a cost not to exceed the direct cost of duplication.

#### § 1.3 Requests for records.

(a) Any person who wishes to inspect, or obtain copies of any record of an agency of the Department shall submit a request in writing and addressed to the official designated in regulations promulgated by the agency. All such requests for records shall be deemed to have been made pursuant to the freedom of Information Act, regardless of whether that Act is specifically mentioned. To facilitate processing of a request, the phrase "FOIA REQUEST" should be placed in capital letters on the front of the envelope.

(b) A request must reasonably describe the records to enable agency per-

sonnel to locate them with reasonable effort. Where possible, specific information regarding dates, titles, etc. which may help identify the records should be supplied by the requester. If the request relates to a matter in pending litigation, the court and its location should be identified.

(c) If an agency determines that a request does not reasonably describe the records, it shall inform the requester of this fact and extend an opportunity to confer promptly with knowledgeable agency personnel to attempt to identify

the records he is seeking.

(d) Nothing in this subpart shall be interpreted to preclude an agency from honoring an oral request for information, but, if the requester is dissatisfied with the response, the agency official involved shall advise him to submit a written request in accordance with paragraph (a) of this section. The "date of receipt" of such a request for purposes of § 1.5(a) of this subpart shall be the date of receipt of the written request.

(e) If a request for records made under this subpart is denied, the person making the request shall have the right to appeal the denial. This appeal must be in writing and addressed to the official designated in regulations promulgated by the agency which denied the request. To facilitate processing of an appeal, the phrase "FOIA APPEAL" should be placed in capital letters on the front of the

envelope.

#### § 1.4 Agency implementing regulations.

Each agency of the Department shall promulgate regulations setting forth the following:

(a) The location and hours of operation of the agency office or offices where members of the public may gain access to those materials required by § 1.2 of this subpart to be made available for public inspection and copying.

(b) Information regarding the publication and distribution (by sale or otherwise) of indexes and supplements thereto which are maintained in accordance with the requirements of 5 U.S.C. 552(a) (2)

and § 1.2(b) of this subpart.

(c) The title(s) and mailing address(es) of the official(s) of the agency who are authorized to receive requests for records submitted in accordance with § 1.3(a) of this subpart, and to make determinations regarding whether to grant or deny such requests. Authority to make such determinations includes authority to (1) extend the ten-day administrative deadline for reply pursuant to § 1.8 of this subpart, (2) make discretionary releases of records exempt from mandatory disclosure pursuant to § 1.11 (b) of this subpart, and (3) make determinations regarding the charging of fees pursuant to Appendix A of this subpart.

(d) The title and mailing address of the official of the agency who is authorized to receive appeals from denials of requests for records submitted in accordance with § 1.3(e) of this subpart and to make determinations regarding whether to grant or deny such appeals. authority to determine appeals includes authority to (1) extend the twenty-day administrative deadline for reply pursuant to § 1.8 of this subpart (to the extent the maximum extension authorized by § 1.8(c) was not used with regard to the initial request), (2) make discretionary releases pursuant to § 1.11(b) of this subpart, and (3) make determinations regarding the charging of fees pursuant to Appendix A of this subpart.

(e) Other information which would be of concern to a person wishing to request records from that agency in accordance

with this subpart.

# § 1.5 Agency response to requests for records.

(a) 5 U.S.C. 552(a) (6) provides that each agency of the Department to which a request for records is submitted in accordance with \$ 1.3(a) of this subpart shall inform the requester of its determination concerning that request within ten days (excepting Saturdays, Sundays, and legal public holidays), plus any extension authorized under \$1.8 of this subpart, of its date of receipt. If the agency determines to grant the request, it shall inform the requester of any conditions (e.g., payment of fees) and the approximate date upon which compliance will be effected. If it grants only a portion of the request, it shall treat the portion not granted as a denial. If it determines to deny the request, it shall immediately inform the requester of that decision and of the following:

(1) The reasons for the denial:

(2) The requester's right to appeal such denial and the title and address of the official to whom such appeal is to be addressed;

(3) The requirement that such appeal be made within forty-five days of the

date of the denial; and

(4) The name and title or position of each person responsible for denial of the

request.

(b) If the agency official designated to receive and make determinations on appeals submitted in accordance with § 1.3(e) of this subpart is a person responsible for denying an initial request, the denial shall be deemed a final agency denial, and the requester shall be so notified and furnished the information required by paragraph (e) of this section. In these circumstances a denial shall be made in accordance with § 1.7(b) of this submart.

(c) If the reason for not fulfilling the request is that the records requested are the records of another agency or Department, the agency shall inform the requester of this fact and shall forward the request to that agency or Department for processing in accordance with the latter's regulations. If the agency has no knowledge concerning the requested records it shall notify the requester of

that fact.

(d) 5 U.S.C. 552(a) (6) provides that each agency in the Department to which an appeal from a denial of an initial request for records is submitted in accordance with § 1.3(e) of this subpart shall

inform the requester of its determination concerning that appeal within twenty days (excepting Saturdays, Sundays, and legal public holidays), plus any extension authorized by \$1.8 of this subpart, of its date of receipt. If the agency determines to grant the appeal it shall inform the requester of any conditions (e.g., payment of fees) and the approximate date upon which compliance vill be effected. If it grants only a portion of the appeal, it shall treat the portion not granted as a denial. If it determines to deny the appeal, it shall inform the requester of that decision and of the following:

(1) the reasons for the denial:

(2) the right to judicial review of the denial in accordance with 5 U.S.C. 552 (a) (4); and

(3) the name and title or position of each person responsible for denial of the

appeal.

(e) If a charge is to be made for compliance with the request in accordance with Appendix A of this subpart, the agency's response shall inform the requester of the amount and basis for the charge. It may, in accordance with Appendix A of this subpart, require payment of the entire fee, or a portion thereof, before it provides the requested records.

(f) In the event compliance with the request involves inspection of records by the requester rather than the forwarding of copies, the agency response shall include the name, mailing address and telephone number of the person to be contacted to arrange a mutually convenient time for such inspection.

(g) In the event the records requested contain some portions which are exempt from mandatory disclosure and others which are not, the official responding to the request shall insure that all non-exempt portions are disclosed, and that all exempt portions are identified according to the nature of information contained and the specific exemption or exemptions which are applicable.

# § 1.6 Date of receipt of requests or appeals.

(a) The date of receipt of a request or appeal which contains the phrase FOIA REQUEST or FOIA APPEAL and is addressed in accordance with applicable agency regulations, shall be the date it is received in the mailroom serving the addressee.

(b) The date of receipt of a request or appeal which is hand-delivered to the address specified in agency regulations shall be the date of such hand delivery.

(c) The date of receipt of a request or appeal which does not comply with paragraphs (a) or (b) of this section shall be the date it is received by the official designated in agency regulations to make the applicable determination.

#### § 1.7 Appeals.

(a) Each agency shall provide for review of appeals by an official different than the official or officials designated to make initial denials.

(b) Each agency, upon a determination that it wishes to deny an appeal, shall send a copy of the records requested (when practicable) and of all correspondence relating to the request to the General Counsel: Attention, Research and Operations Division. When the volume of records is so large as to make sending a copy impracticable, the agency shall enclose an informative summary of those records. The agency shall not deny an appeal until it receives a response from the Office of the General Counsel regarding the legal issues involved in the appeal.

shall promptly review the matter (including necessary consultation with the Department of Justice and coordination with the Office of Communication) and render all necessary assistance to enable the agency to respond to the appeal within the administrative deadline or

any extension thereof.

#### § 1.8 Extension of administrative deadlines.

(a) In unusual circumstances as specified in this section either of the administrative deadlines prescribed in § 1.5 of this subpart may be extended by an authorized agency official. Written notice of the extension shall be sent to the requester within the applicable deadline, setting forth the reasons for such extension and the date a determination is expected to be dispatched. In no event shall the extension exceed a total of ten working days.

(b) As used in this section, "unusual circumstances" shall be limited to the

following:

(1) the need to search for and collect the requested records from field facilities or other establishments that are separate from the office processing the request:

(2) the need to search for, collect, and appropriately examine a voluminous amount of separate and distinct records which are demanded in a single request;

and

(3) the need for consultation, which shall be conducted with all practicable speed, with another Department or agency having a substantial interest in the determination of the request or among two or more components of the agency having substantial subjectmatter interest therein. (Note: consultation between an agency and the Office of the General Counsel, Office of Communication, or the Department of Justice, is not a basis for extension under this section).

(c) The ten-day extension authorized by this section may be divided between the initial and appellate reviews, but in no event shall the total extension exceed

ten working days.

# § 1.9 Failure to meet administrative deadlines.

In the event an agency fails to meet either of the administrative deadlines set forth in § 1.5 of this subpart, plus any extension authorized by § 1.8 of this subpart, it shall notify the requester, state the reasons for the delay, and the date by which it expects to dispatch determination. Although the requester may be deemed to have exhausted his administrative remedies under 5 U.S.C. 552(a) (6) (C) the agency shall continue processing the request as expeditiously as possible and dispatch the determination when it is reached in the same manner and form as if it had been reached within the applicable deadline.

#### § 1.10 Fee schedule.

Pursuant to authority delegated in § 2.79 of this chapter, the Director, Office of Operations, has issued regulations, following notice and public comment, setting forth a uniform schedule of fees applicable to all agencies of the Department regarding requests for records under this subpart. (See Appendix A of this subpart). Any amendments thereto will be made by the Director pursuant to notice and opportunity for comment. Said regulations provide reasonable standard charges for document search and duplication and provide for recovery of only the direct costs of such duplication. The regulations provide that documents may be furnished without charge or at a reduced charge where the agency determines that waiver or reduction of the fee is in the public interest because furnishing the information can be considered as primarily benefiting the general public.

# § 1.11 Exemptions and discretionary release.

(a) All agency records, except those specifically exempted from mandatory disclosure by one or more provisions of 5 U.S.C. 552(b), shall be made promptly available to any person submitting a request under this subpart.

(b) Except where disclosure is specifically prohibited by Executive order, statute, or applicable regulations, an agency may release records exempt from mandatory disclosure under 5 U.S.C. 552 (b) whenever it determines that such disclosure would be in the public interest.

(c) In no event shall release of any list of names and/or addresses of employees of the Department or of farmers, persons, organizations or firms, for political purposes be considered to be in the public interest; nor shall release for commercial solicitation purposes be considered to be in the public interest unless release is specifically authorized by the individuals named therein.

#### § 1.12 Annual report.

(a) Each agency of the Department shall compile the following information for each calendar year:

(1) the number of determinations made by such agency not to comply with initial requests for records made to it under § 1.3(a) of this subpart, and the reasons for each such determination;

(2) the number of appeals made by persons under § 1.3(d) of this subpart, the result of such appeals, and the reason for the action upon each appeal that results in a denial of information;

(3) the name and title or position of each person responsible for the denial of records requested under this subpart and

the number of instances of participation for each:

(4) the results of each proceeding conducted pursuant to 5 U.S.C. 552(a) (4) (F), including a report of the disciplinary action taken against the officer or employee who was primarily responsible for improperly withholding records or an explanation of why disciplinary action was not taken;

(5) a copy of every rule made by the agency regarding this subpart;

(6) the total amount of fees collected by the agency for making records available under this subpart; and

(7) such other information as indicates efforts to administer fully this subpart.

(b) Each agency shall compile the information required by (a) above for the preceding calendar year into a report and submit this report to the Director of Communication by February 1 of each year.

(c) The Director of Communication shall combine the reports from the various agencies within USDA into a Departmental report, and shall arrange for submission of this report to the Speaker of the House of Representatives and President of the Senate by March 1 of each year in accordance with 5 U.S.C. 552(d).

## § 1.13 Compilation of new records.

Nothing in 5 U.S.C. 552, or this subpart require that any agency compile a new record in order to fulfill a request for records. Such compilation may be undertaken voluntarily if the agency determines this action to be in the public interest.

#### § 1.14 Authentication.

When a request is received for an authenticated copy of a document which the agency determines to make available to the requesting party, the agency shall cause a correct copy to be prepared and sent to the Office of the General Counsel which shall certify the same and cause the seal of the Department to be affixed, except that the Judicial Officer, or the Hearing Clerk when directed by the Judicial Officer, may authenticate copies of documents in the records of the Hearing Clerk.

### § 1.15 Compulsory process.

(a) (1) In any case where it is sought by subpoena, order, or other compulsory process or demand (hereinafter in this section referred to as a "demand") to require the production or disclosure of any record or material which is exempt from disclosure under 5 U.S.C. 552(b) or information related thereto acquired by an employee of this Department in the performance of his official duties or because of his official status, the matter shall be referred to an official authorized by agency regulations to make releases pursuant to § 1.11(b) of this subpart. (2) Such official may authorize release. However, if such official determines that it would be improper to comply with the demand, he shall refer it to the agency head. The agency head may authorize release; however, if the agency head con-

curs with the initial conclusion, the matter shall be referred to the Secretary for final determination. (3) If the Secretary has determined that the records, material, or information should not be produced, or if no final determination has been made, the employee who appears in answer to the demand will respectfully decline to produce or disclose the records, material, or information demanded on the ground that the disclosure is prohibited by this section. The employee shall provide the court or other authority with a copy of this subpart and a copy (when available) of the Secretary's determination, and shall respectfully request the court or other authority to withdraw or stay the demand.

(b) (1) Whenever a demand of the type described in paragraph (a) of this sec tion is made upon an employee of this Department not authorized to make releases pursuant to § 1.11(b) of this subpart, by a court or other authority while he is appearing before, or is otherwise in the presence of the court or other authority, the employee, or other appropriate Government official or attorney acting on behalf of the employee, shall (i) immediately inform the court or other authority that this section prohibits the employee from producing or disclosing the information or material demanded and (ii) offer to refer the demand for the prompt consideration of authorized officials, providing the court or other authority a copy of this subpart and respectfully requesting that the demand be stayed pending his receipt of appropriate instructions concerning the demand. (2) If the employee is authorized to make a release pursuant to § 1.11(b) of this subpart but determines that such release would be improper, he shall offer to refer the demand for the prompt consideration of the agency head and/or Secretary and shall otherwise comply with (1) (11) above.

(c) If the court or other authority declines to stay the effect of the demand in response to a request made in accordance with paragraph (a) or (b) of this section pending the receipt by the employee of instructions or directions, or if the court or other authority rules adversely on any assertion made in conformity with the provisions of this subpart, the employee upon whom the demand has been made may tender the records, material, or information, demanded with a request they be held in camera until an appeal can be taken from the adverse ruling.

# § 1.16 Records in formal adjudication proceedings.

Records in formal adjudication proceedings are on file in the Office of the Hearing Clerk, U.S. Department of Agriculture, Washington, D.C. 20250, and shall be made available to the public.

Effective date: February 19, 1975.

Done at Washington, D.C., this 14th day of February, 1975.

EARL L. BUTZ, Secretary of Agriculture.

#### APPENDIX A-FEE SCHEDULE

The Department of Agriculture solicited in 40 FR 3604 public comment on proposed regulations implementing the 1974 Freedom of Information Act amendments as they affect the fees which may be charged requesters for services performed under that Act. The regulations also cover fees for serv-ices performed under other authority. No comments from the public were received.

The following changes have been made. The Department has determined that these changes are not such as to require further

public comment.

Sec. 5. An additional paragraph is included specifying the availability of transcripts under the Departments reporting service contract.

Sec. 7. The paragraph requiring agreement in writing to pay when an estimate exceeds

\$20.00 has been dropped.

The paragraph requiring a 50% deposit on estimates over \$150.00 has been changed by deletion of redundant language and by reducing the amount at which a deposit is required to \$50.00.

The paragraph regarding "fair prior notice"

has been deleted.

A new paragraph has been inserted advisrequesters that, in the event of the possibility of an unproductive or extensive search, the requester will be given the opportunity to confer with agency personnel to possibly reform his request to meet his needs at a lower fee.

Sec. 8.e. Information regarding the St. Louis Computer Center location and rates, and a comment on commercial time-sharing

sources have been added.

Sec. 8.h. Redundant verbiage has been removed.

Sec. 10.a. & b. Soil Conservation Service addresses have been revised.

Sec. 16.b.8. The price for cassettes was cor-

rected from \$8.00 to \$3.00. Pursuant to the authority delegated to the Director, Office of Operations in 7 CFR 2.79 (a) (3) (iii), there is published the following fee schedule to be effective February 19, 1975.
The previous fee schedule published in 39

FR 26050 as amended in 39 FR 28449 is

superseded.

#### FEE SCHEDULE

Sec. 1. General. This schedule sets forth fees to be charged for providing copies of records, including photographic reproduc-tions, microfilm, maps and mosaics, and re-lated services. The fees set forth in this schedule are applicable to all agencies and constituent units of the Department of Agriculture.

Sec. 2. Facilities. Records and related services are available at the locations specified by the agencies in their statements of organization and services. Each agency is responsible for promulgating procedures to facilitate public inspection and copying of its records. Any materials offered for sale by the Government Printing Office should be purchased from that source. Departmental agen-cies will not stock such materials for public

Agencies do not stock copies of forms and publications or maintain records at any facility which does not require these mate-

rials in its operations.

Sec. 3. Fees for materials and services. All agencies of the Department shall be guided by the fees set forth herein. Any changes or additions to this fee schedule shall be made by amendment to or revision of this schedule.

Sec. 4. Circumstances governing exceptions to the charging of fees for records and re-lated services. (For photographic reproduc-

tions, see Sec. 12.)

a. Waiver of fees for records and related services. Fees may be waived in whole or in part under the following conditions:

(1) Where individual collections are \$3.00

less.
(2) Where the furnishing of the service without charge is an appropriate courtesy to a foreign country or international organization; or comparable fees are set on a reciprocal basis with a foreign country or an international organization.

(3) Where the recipient is engaged in a nonprofit activity designed for the public

safety, health, or welfare.

(4) Where the agency determines that payment of the full fee by a State, local government, or nonprofit group would not be in the interest of the program involved.

b. Fees not to be charged for records and related services. Fees shall not be charged

under the following conditions:

(1) When the furnishing of records and related services is determined by the agency be in the public interest as primarily

benefiting the general public.
(2) When filling requests from other Departments or Government agencies for official use, provided quantities requested are reasonable in number.

(3) When members of the public provide their own copying equipment, in which case no copying fee will be charged.

When any notices, decisions, orders, or other material are required by law to be served on a party in any proceedings or matter before any Department agency.

c. Where both a and b above apply to a

matter, b shall be controlling.
Sec. 5. Limitations of copies. a. Agencies may restrict numbers of photocopies and directives furnished the public to one copy of each page. Copies of forms provided the public shall also be held to the minimum practical. Persons requiring any large quantities should be encouraged to take single copies to commercial sources for further appropriate reproduction

b. Single or multiple copies of transcripts. provided the Department under a reporting service contract, may be obtained from the contractor at a cost not to exceed \$.10 per page for the first copy and \$.05 per page for each additional copy ordered by the same person. The contractor may add a postage charge when mailing orders to the public but no other charge may be added.

Sec. 6. Search Services. a. Search services are services of agency personnel-clerical, supervisory or professional salary level-used in trying to find the records sought by the requester. They include time spent examining records for the purpose of finding rec-ords which are within the scope of the re-quest. They also include services to transport personnel to places of record storage, or records to the location of personnel for the purpose of the search, if such services are reasonably necessary.

b. Because of the nature of the Department's business and records, the normal location of a record in a file or other facility will not be considered a search. This would be the same as quickly locating a piece of material for purposes of answering a letter or telephone inquiry, and is based on the Department's obligation to respond to requests furnishing a reasonably specific description of the record.

Sec. 7. Payments of fees and charges. a. Payments will be collected to the fullest extent possible in advance or at the time the requested materials are furnished.

b. Except as otherwise stipulated by agency procedures, payment shall be made by check, draft, or money order made payable to the Treasury of the United States, but small amounts may be paid in cash, particularly where services are performed in response to a visit to a Department office.

c. Where the estimated fees to be charged exceed \$50.00, a deposit of 50 percent of the estimated amount shall be collected from

the requester before any of the requested

materials are reproduced.

d. Where a request for records indicates the necessity of an extensive search, the re-quester should be notified of that fact, and of the possibility of an unproductive search. The notification should offer the requester the opportunity to confer with agency perthe opportunity to conter with agency personnel to reform his request to meet his needs at a lower fee. When an extensive search still appears necessary, unless the agency determines that the request is in the public interest in accordance with section 4b(1), it shall inform the requester that no search will be undertaken until an agreement to pay applicable fees is received, including a deposit of 50% of the estimated fee where appropriate.

Bec. 8. Fees for records and related services. a. Photocopies, 8½" x 14" or smaller; \$0.10 for the first copy and \$0.05 for each additional copy of the same page.

b. Photocopies in excess of 8½" x 14"; \$0.25 per linear foot of the longest side of

c. Manual searches will be charged for at the rate of \$4.00 per hour for cherical time and \$9.00 per hour for supervisory or professional time. Charges will be computed to the nearest quarter hour required for the search. A search may involve both clerical and supervisory or professional time.

d. Other direct costs incurred will be assessed the requester at the actual cost to the Government, e.g., where records are required to be shipped from one office to another by commercial carrier in order to timely answer the request, the actual freight charges will be

assessed the requester.

Computer searches will be charged for at the rates established in the Users Manual or Handbook published by the computer center at which the work will be performed, except that where commercial time-sharing computer sources are the required search media, the contract rate charged by the commercial source to the Government will be charged. A listing follows showing where those rates are published and the office from which copies may be obtained or at which the rates may be examined.

Fort Collins Computer Center Users Manual Section 3.4—Administrative Policies and

Procedures—Rates, pp. 3-4 through 3-7:
Fort Collins Computer Center, U.S. Department of Agriculture, 3825 East Mulberry
Street (P.O. Box 1206), Fort Collins, Colo-

New Orleans Computer Center Users Man-ual Appendix B—New Orleans Computer Center Rates, pp. 1 through 5:

New Orleans Computer Center, U.S. Department of Agriculture, 13800 Old Gentilly Road, Building 350, New Orleans, Louisiana 70129

Kansas City Computer Center Users Manual Appendix D-2-Rates. pp. 65-80:

Kansas City Computer Center, U.S. Department of Agriculture, 8930 Ward Parkway (P.O. Box 205), Kansas City, Missouri 64141.

Washington Computer Center Users Hand-book Appendix B-Washington Computer Center Rates. pp. 1-10:

Washington Computer Center, U.S. Department of Agriculture, Room S-100, South Buildings, 12th Street and Independence Ave., S.W., Washington, D.C 20250

St. Louis Computer Center. Charges for the St. Louis Computer Center will be based on actual computer configuration used and be based on direct costs only:

St. Louis Computer Center, U.S. Department of Agriculture, Farmers Home Administration, Business Services Branch, 1520 Market Street, St. Louis, Missouri 69103.

f. The fees do not include and no charge shall be made for (a) time spent examining records to determine whether an exemption

can should be asserted, (b) time spent deleting exempt matter being withheld from rec-ords to be furnished, or (c) time spent in monitoring a requester's inspection of agency records.

g. Certifications, \$1.00 each; Authentications under Department Seal (including

aerial photographs), \$2.00 each.

h. Except as provided in section 9, for services not subject to the Freedom of Information Act, and not covered by (g) above, agencies may set their own fees in accord-ance with applicable law.

i. The fees specified in a through f of this Section apply to all requests for services un-der the Freedom of Information Act, as amended (5 USC 552), unless no fee is to be charged, or the agency has determined to waive or reduce those fees pursuant to Section 4. No higher fees nor charges in addition to those provided for in this schedule may be charged a party requesting search or duplica-tion services under the Freedom of Information Act.

j. The fees specified in g and h of this Section and in Sections 9 through 16 of this schedule apply to requests for services other than those subject to the Freedom of Information Act. The authority for establishment of these fees is 31 USC 483a and other

applicable law.

Sec. 9. Photographic reproduction, microfilm, mosaic and maps. Reproduction of such aerial or other photographic microfilm, mosaic and maps as have been obtained in connection with the authorized work of the Department may be sold at the estimated cost of furnishing such reproductions as prescribed in this schedule.

Sec. 10. Agencies which furnish photographic reproductions.—a. Aerial photo-graphic reproductions. The following agencies of the Department furnish aerial photo-

graphic reproductions:

Agricultural Stabilization and Conserva tion Service (ASCS) Room 3405 Auditor's Building, Washington, D.C. 20250. Forest Service (FS), 24 LOB RP-E, Arling-

ton, Virginia 22209, or nearest Forest Service

Regional Office.

Soil Conservation Service (SCS), Cartographic Division, SCS, Washington, D.C. 20250 or Cartographic Unit in nearest SCS Technical Service Center.

b. Other photographic reproductions. Other types of photographic reproductions may be obtained from the following agencies of the Department:

Agricultural Stabilization and Conserva-tion Service (ASCS) (Address above). Forest Service (Address above).

Office of Communication, Photographic Di-

vision, Room 536A, Washington, D.C. 20250. Soil Conservation Service, Information Division, Audio-Visual Branch, Washington, D.C. 20250.

National Agricultural Library, Information Officer, Room 204, Beltsville, Maryland 20705. Sec. 11. Photographic Sales Committee. The Photographic Sales Committee consists of representatives designated by Department agencies principally concerned with the sale

of photographic reproductions, The Committee recommends prices at which photographic and mosaic reproductions, except library material, shall be sold, and other mat-ters related to photographic reproductions. Sec. 12. Circumstances under which photo-

graphic reproductions may be provided free. Reproductions may be furnished free at the discretion of the agency, if it determines this

action to be in the public interest, to:
a. Press, radio, television, and newsreel representatives for dissemination to the gen-

b. Agencies of State and local governments carrying on a function related to that of the

Department when it will help to accomplish an objective of the Department.

c. Cooperators and others furthering agricultural programs. Generally, only one print of each photograph should be provided free. Sec. 13. Loans, Aerial photographic film

negatives or reproductions may not be loaned

outside the Federal Government.

Sec. 14. Sales of positive prints under Government contracts. The annual contract for furnishing single and double frame slide film negatives and positive prints to agencies of the Pepartment, County Extension Agents and others cooperating with the Department, carries a stipulation that the successful bidder must agree to furnish slide film positive prints to such persons, organizations, and associations as may be authorized by the Department to purchase them.

Sec. 15. Procedure for handling orders. In order to expedite handling, all orders should

contain adequate identifying information. Agencies furnishing aerial photographic re-productions require that all such orders identify the photographs. Each agency has its own procedure and order forms.

Sec. 16. Photographic reproduction prices. The prices for photographic reproductions listed here are the most generally requested

items.

a. National Agricultural Library. The following prices are applicable to National Agricultural Library items only.

Microfilm-\$1.00 for each 30 pages or fraction thereof. Photoreproduction-\$1.00 for each 10 pages.

b. General photographic reproductions. Minimum charge \$1.00 per order. All sizes are approximate. An extra charge may be necessary for excessive laboratory time caused by any section instructions from the purchaser.

Class of work	Unit	Price
I. Black and white copy negatives and film positives:		
4 x 5	Each	\$3.6
5 x 7	do	3.9
8 x 10		4. 8
11 x 14		7. 0
2. Black and white enlargements:		
Up to 8 x 10	do	2.7
11 x 14		4.9
Over 11 x 14		3. 6
3. Reductions (from any size negative)	I or odume rooms	8. 6
4. Mounting	Per smiare foot	2. 2
5. Slides: Black and white (from copy negative):	rer bquano rooter	
2 x 2 cardboard mounted	Each	2.4
3½ x 3½		3.
Original color (from flat copy)		1.1
Duplicate color (2 x 2 cardboard mounted)		-
(Duplicate color slides are slides copied from 35mm color		
slides only.) Slides made from black and white material,		
or from transparencies larger or smaller than 35mm, will		
be charged at the same rates shown for black and white		
and original color slides.		
6. Color transparencies (4 x 5)	do	8.
7. Color prints		-(
8. Current USDA slide sets in stock:		
1 to 50 frames	-	13.
51 to 60 frames		14.
61 to 75 frames		15.
76 to 95 frames		17.
96 to 105 frames		18.
106 to 130 frames		21.
	ted narrative guide)	.21.
The following can be purchased for the corresponding slide sets ab		
Cassettes		3.
		8.
Records		-
Audio-tape		1.
9. Milk sedimentation standards (5 x 7 Black and white photo-	W	
graph)		
10. Seeds and seedlings (any size)		3.

c. Aerial photographic reproductions. No minimum charge on aerial photographic reproductions. All prints are furnished unmounted and untrimmed.

1. Contact prints. The prices for contact prints are set forth below. The size refers to the approximate size of the contact print.

Size 9 by 9 on commercial grade paper\_\_ \$2 For polyester base paper, add \$0.75 per

contact print available from ASCS only). Each Size 70-mm contact prints\_\_\_ 81, 25 Size 70-mm contact transparencies.... 2.00

2. Enlargements (projection prints). The prices for enlargements of various sizes are set forth below. The size in each case refers to the approximate size of paper required to produce the enlargement ordered.

Size:	each
9 by 9 in (from 70 mm)	\$2.00
17 by 17 in	4.00
24 by 24 in	5.00
24x24 'in	_ 16.00
38 by 38 in	12.00

For larger size reproductions, add \$2.00 for each additional 12 inches or fraction thereof, linear measurement.

3. Aerial photo-index sheets.	91
Size 20 by 24 inches  Quantity:	Price each
Any quantity	\$5.00
4. Film positives. Contact printed aerial negatives, size 9 by 9 inches.	from
Quantity: Any quantity	Price each \$3.00
<ol> <li>Copy negatives. On film, aerial expesize 9 by 9 inches.</li> </ol>	osures,
Quantity: Any quantity	Price each
6. Diapositives. Prints on glass, sixe inches, thickness 0.06 inch.	9 by 9
Pros	Price
Quantity: Any quantity	*10.00
7. Aperture Cards and Printouts.	
	Each

8. Color Photography. Furnished only by the Regional Forest Service Aerial Photography Laboratories at Ogden, Utah and San Francisco, California, and the Agricultural Stabilization and Conservation Service Aerial Photography Laboratory in Salt Lake City, Utah.

Duplicate of an aperture card\_\_ \$1.00 \$0.10

Aperture card from photoindex

Printout from aperture card\_\_\_

First additional

1.00

unit

. 50

unit

Positive contact print made from negative	87.00
Enlargements 9x9 inches (from 70 mm)	
Enlargements 19x13 inches	
Enlargements 17x17 inches	
Enlargements 24x24 inches	20.00
Enlargements 88x38 inches	

# Color film transparencies (positives or negatives)

Contacts 70 mm	\$5.00
Contacts 9x9 inches	12.00
Enlargements 9x9 inches (from 70	
mm)	12.00

9. Special Needs: For special needs not covered above, persons desiring aerial photographic reproductions should contact the agencies listed in Section 10a or the Coordinator, Aerial Photographic Work of the Department of Agriculture, ASCS, Room 3405, Auditors Building, 14th Street and Independence Avenue, SW, Washington, D.C. 20250.

(5 USC 301; 5 USC 552; 31 USC 483a; and 7 CFR 2.79(a) (3) (iii))

Done at Washington, D.C., this 13th day of February, 1975.

JOHN J. KEANEY,

Director. [FR Doc.75-4753 Filed 2-18-75;12:22 pm]

#### Title 5—Administrative Personnel

#### CHAPTER III-OFFICE OF MANAGEMENT AND BUDGET

#### PART 1303—PUBLIC INFORMATION PRO-VISIONS OF THE ADMINISTRATIVE PROCEDURES ACT

Notice is hereby given that pursuant to the provisions of 5 U.S.C. 552, there is published as Chapter III, Part 1303

of Title 5, Code of Federal Regulations and as Chapter V of Title 3, Code of Federal Regulations, the following regulations as set forth below.

It is the policy of the Office of Management and Budget whenever practicable to afford the public an opportunity to participate in the rule-making process. Accordingly, interested persons may submit written comments, suggestions, or objections concerning the regulations pertaining to the Office of Management and Budget to the Assistant to the Director for Administration, Office of Management and Budget, Washington, D.C. 20503, on or before March 31, 1975. These regulations are effective immediately.

# PHILLIP D. LARSEN, Acting Assistant to the Director for Administration.

#### EXECUTIVE OFFICE OF THE PRESIDENT

Note: Until further regulations are promulgated, the remainder of the entities within the Executive Office of the President, to the extent that 5 U.S.C. 552 is applicable, shall follow the procedures set forth in the regulations applicable to the Office of Management and Budget. Requests for information from these other entities should be submitted directly to such entity.

Freedom of information regulations for:

Preedom of information regulations for: Office of Management and Budget appear at 5 CFR Ch. III:

Council on Wage and Price Stability appear at 6 CFR Ch. VII;

National Security Council appear at 32 CFR Ch. XXI.

### OFFICE OF MANAGEMENT AND BUDGET

### ORGANIZATION AND INFORMATION

Pursuant to section 552 of title 5, United States Code, as amended, the following Part 1303 is published:

#### ORGANIZATION

Sec.	
1303.1	General.
1303.2	Authority and functions.
1303.3	Organization.

# PROCEDURES

# 1303.10 Methods of operation.

#### Availability of Information

1303.20 Inspection, copying and exceptions. Charges for Search for Reproduction

1303.30 Schedule of fees and method of payment for services rendered.

AUTHORITY: 5 U.S.C. 552, as amended by Pub. L. 93-502.

#### ORGANIZATION

### § 1303.1 General.

This information is furnished for the guidance of the public and in compliance with the requirements of section 552 of title 5, United States Code, as amended.

### § 1303.2 Authority and functions.

(a) The Office of Management and Budget was established in the Executive Office of the President pursuant to Part I of Reorganization Plan No. 2 of 1970 (35 FR 7959), effective July 1, 1970. That Plan transferred to the President all functions vested by law in the Bureau of the Budget, or its Director, and designated the Bureau of the Budget as the

Office of Management and Budget. By Executive Order No. 11541 of July 1, 1970 (35 FR 10737), the President delegated all functions transferred to him by Part I of the Plan to the Director of the Office of Management and Budget.

(b) The principal statutory functions of the Office of Management and Budget are contained in the Budget and Accounting Act of 1921 (42 Stat. 20, 31 U.S.C. 1-25); the Federal Reports Act of 1942 (44 U.S.C. 3501-3511); the Government Corporation Control Act (59 Stat. 597, 31 U.S.C. 841-869); the Budget and Accounting Procedures Act of 1950 (65 Stat. 832), the Federal Advisory Committee Act (Pub. L. 92-463), the Congressional Budget and Impoundment Control Act of 1974 (Pub. L. 93-344), the Office of Federal Procurement Policy Act (Pub. L. 93-400), and the Privacy Act of 1974 (Pub. L. 93-579).

(c) The functions of the Office of Management and Budget are carried out pursuant to the provisions of the statutes cited above and the provisions of various Executive orders—principally, Executive Order No. 8248 of September 8, 1939 (CFR Cum. Supp., p. 576), outlining certain functions to be performed by the Bureau of the Budget for the President, and Executive Order No. 11230 of June 28, 1965 (3 CFR 1965 Supp., p. 146), as amended, delegating certain functions of the President to the Director of the Bureau of the Budget. Under the terms of Executive Order No. 11541 of July 1, 1970, the assignments and delegations made in the earlier orders are to be considered as assignments to the Office of Management and Budget and its Director.

## § 1303.3 Organization.

The central organization of the Office of Management and Budget consists of—

. (a) The Office of the Director, which includes the Director and the Deputy Director and their principal assistants, including the Assistant to the Director for Administration.

(b) The Administrator of the Office of Federal Procurement Policy.

(c) Three Associate and Assistant Directors with Government-wide management responsibilities in specialized areas, as follows:

Executive Development and Labor Relations Legislative Reference

Management and Operations

(d) Five program and budget Associate and Assistant Directors, as follows:
Budget Review

National Security and International Affairs Human and Community Affairs Economics and Government Natural Resources, Energy, and Science

(e) The Office has no field organization.

(f) Units of the Office of Management and Budget are presently located in the Executive Office Building, 17th Street and Pennsylvania Avenue, NW., and in the New Executive Office Building, 17th and H Streets, NW., Washington, D.C. 20503. Regular office hours are from 9

a.m. to 5:30 p.m., Monday through Friday. Both buildings are under security control. Persons desiring to visit officers or employees of the Office Management and Budget in either building will usually find it easier to do so if they write or telephone in advance for an appointment.

#### PROCEDURES

### § 1303.10 Methods of operation.

(a) The Office of Management and Budget maintains current indexes which identify information pertaining to matters issued, adopted, or promulgated after July 4, 1967, that are within the scope of 5 U.S.C. 552(a) (2). These indexes are updated quarterly and are published in the FEDERAL REGISTER. They are also available for public inspection and copying at the Office's Publications Office, Room G-236, New Executive Office Building, 17th and H Streets NW., Washington, D.C. 20503. The indexes may be examined between the hours of 9 a.m. and 5:30 p.m. on any day, except Saturdays, Sundays, and legal public holidays.

(b) The Office of Management and Budget has established an internal Committee on Freedom of Information (hereinafter referred to as the Committee), chaired by the Deputy Director. The purpose of this Committee is to oversee the Officer's administration of the Freedom of Information Act as amended and to rule on any appeals that may be filed in the event a request for information is

denied.

(c) The Assistant to the Director for Administration is responsible for acting on all initial requests. Individuals wishing to obtain any information listed on the indexes should address their request in writing to the Assistant to the Director for Administration, Office of Management and Budget, Washington, D.C. 20503, Phone 395–4790. Requests for information shall be as specific as possible. It is suggested that requestors use the format shown in either Exhibit A or B

(d) Upon receipt of any request for information or records, the Assistant to the Director for Administration will determine within 10 days (excepting Saturdays, Sundays, and legal public holidays) whether it is appropriate to grant the request and will immediately (Receipt of a request as used herein means the date the request is received in the office of the Assistant to the Director for Administration.) provide written notification to the person making the request. If the request is denied, the written notification will also include the names of other individuals who participated in the determination and a notice that an appeal may be lodged including the format and content of any such appeal within the Office of

Management and Budget.

(e) In the event the request is denied and that denial is appealed, the Assistant to the Director for Administration will refer the appeal to the Committee which

will make a determination with respect to such appeal within 20 days (excepting Saturdays, Sundays, and legal public holidays). Appeals shall be set forth in writing and addressed to the Assistant to the Director for Administration at the address specified in paragraph (c) of this section. The appeal shall include a statement explaining the basis for the appeal. Determinations of appeals will be set forth in writing and signed by the Deputy Director ,or his designee. If, on appeal, the denial is in whole or in part upheld. the written determination will also contain a notification of the provisions for judicial review and the names of the persons who participated in the determination

(f) In unusual circumstances, the time limits prescribed in paragraphs (d) and (e) of this section, may be extended for not more than 10 working days (excepting Saturdays, Sundays, or legal public holidays). Extensions may be granted by the Assistant to the Director for Administration in the case of initial requests and by the Deputy Director in the case of appeals. The extension period may be split between the initial request and the appeal but in no instance may the total period exceed 10 working days. Extensions will be by written notice to the persons making the request and will set forth the reasons for the extension and the date the determination is expected. As used herein, but only to the extent reasonably necessary to the proper processing of the particular request, the term "unusual circumstances" means:

(1) The need to search for and collect the requested records from establishments that are separate from the office

processing the request;

(2) The need to search for, collect, and examine a voluminous amount of separate and distinct records which are demanded in a single request; or

(3) The need for consultation, which shall be conducted with all practical speed, with another agency having a substantial interest in the determination of the request or among two or more components of the agency which have a substantial subject matter interest therein.

#### AVAILABILITY OF INFORMATION

§ 1303.20 Inspection, copying, and exceptions.

(a) When a request for information has been approved pursuant to section 10 above, the person making the request may make an appointment to inspect or copy the materials requested during regular business hours by writing or tele-phoning the Assistant to the Director for Administration at the address or telephone number listed in § 1303.10(c). Such materials may be copied manually without charge, and reasonable facilities will be made available for that purpose, Also, copies of individual pages of such materials will be made available at the price per page specified in paragraph (a) of § 1303.30; however, the right is reserved to limit to a reasonable quantity the copies of such materials which may

be made available in this manner when copies also are offered for sale by the Superintendent of Documents.

(b) Certain functional units of the Office of Management and Budget solely advise and assist the President, and therefore these units are not covered by 5 U.S.C. 552. However, the Director or the Deputy Director, acting on his behalf, may determine that a record which falls in one of the following categories shall be made available. These units carry out activities that provide advice and assistance to the President with regard to

(1) The formulation and preparation

of the Federal budget,

(2) The processing of enrolled bills and determinations of the relationships of pending and proposed legislation to the program of the President.

(3) The compensation of Federal em-

ployees, and

(4) The establishment and organization of new agencies and the reorganization of existing programs and agencies.
 (5) The preparation of Executive Or-

ders and Proclamations.

(c) Except to the extent that the Director or Deputy Director, acting on his behalf, determines that a record which falls within one of the following categories shall be made available, this section shall not apply to matters that are—

(1) (i) Specifically authorized under criteria established by an Executive Order to be kept secret in the interest of the national defense or foreign policy, and (ii) are in fact properly classified pursuant to such Executive Order.

(2) Related solely to the internal personnel rules and practices of the Office;
(3) Specifically exempted from dis-

closure by statute:

(4) Trade secrets and commercial or financial information obtained from a person and privileged or confidential;

(5) Interagency or intra-agency memorandums or letters which would not be available by law to a party other than an agency in litigation with the Office;

(6) Personnel and medical files and similar files the disclosure of which would constitute a clearly unwarranted

invasion of personal privacy;

(7) Investigatory records compiled for law enforcement purposes, but only to the extent that the production of such records would (i) interfere with enforcement proceedings, (ii) deprive a person of a right to a fair trial or an impartial adjudication, (iii) constitute an unwarranted invasion of personal privacy, (iv) disclose the identity of a confidential source and, in the case of a record compiled by a criminal law enforcement authority in the course of a criminal investigation, or by an agency conducting a lawful national security intelligence investigation, confidential information furnished only by the confidential source, (v) disclose investigative techniques and procedures, or (vi) endanger the life or physical safety of law enforcement personnel;

(8) Contained in or related to examination, operating, or condition reports

<sup>&</sup>lt;sup>1</sup> Exhibits A and B filed as part of the original document.

prepared by, on behalf of, or for the use of an agency responsible for the regulation or supervision of financial institutions; or

(9) Geological and geophysical information and data, including maps, concerning wells.

CHARGES FOR SEARCH AND REPRODUCTION

# § 1303.30 Schedule of fees and method of payment for services.

(a) Following is the fees schedule for the search and reproduction of information available under the Freedom of Information Act (5 U.S.C. 552), as amended.

(1) Search for records. \$5.00 per hour when the search is conducted by a clerical employee. \$8.00 per hour when the search is conducted by a professional employee. No charge for searches of less than I hour.

(2) Duplication of records. Records will be duplicated at a rate of \$.25 per page for all copying of 4 pages or more. There is no charge for duplicating 3 or less pages.

(3) Other. When no specific fee has been established for a service, or the request for a service does not fall under one of the above categories due to the amount or type thereof, the Assistant to the Director for Administration is authorized to establish an appropriate fee based on "direct costs" as provided in the Freedom of Information Act and in accordance with Office of Management and Budget Circular No. A-25, "User charges." Examples of services covered by this provision include searches involv-

ing computer time or special travel, transportation, or communications costs.

(b) If records requested under this part are stored elsewhere than the head-quarters of the Office of Management and Budget at Washington, D.C., the special costs of returning such records to the headquarters for review will be added to the search costs. Search costs are due and payable even if the record which was, requested cannot be located after all reasonable efforts have been made, or if the Office determines that a record which has been requested, but which is exempt from disclosure under this part, is to be withheld.

(c) Where it is anticipated that the fees chargeable under this section will amount to more than \$25.00, or the maximum amount specified in the request, the requester shall be promptly notified of the amount of the anticipated fee or such portion thereof as can readily be estimated. In instances where the estimated fees will greatly exceed \$25.00. an advance deposit may be required. The notice or request for an advance deposit shall extend an offer to the requester to consult with Office personnel in order to reformulate the request in a manner which will reduce the fees, yet still meet the needs of the requester. Dispatch of such a notice or request shall suspend the running of the period for response by the Office until a reply is received from

(d) Fees must be paid in full prior to issuance of requested copies. In the event the requestor is in arrears for previous requests for which the Office was

unable to find or provide the requested information (see paragraph (b)) of this section), copies of records will not be provided for any subsequent request until the arrears have been paid in full.

(e) Remittances shall be in the form either of a personal check or bank draft drawn on a bank in the United States, or a postal money order. Remittances shall be made payable to the order of the Treasury of the United States and mailed or delivered to the Assistant to the Director for Administration, Office of Management and Budget, Washington, D.C. 20503.

(f) A receipt for fees paid will be given upon request. Refund of fees paid for services actually rendered will not be made.

(g) The Assistant to the Director for Administration, or an officer designated by the Assistant to the Director for Administration may in accordance with the Freedom of Information Act, as amended, waive all or part of any fee provided for in this section when the Assistant to the Director for Administration or the designated officer deems it to be in either the Office's interest or in the general public's interest.

For the Director of the Office of Management and Budget.

PHILLIP D. LARSEN, Acting Assistant to the Director for Administration.

FEBRUARY 18, 1975.

[FR Doc.75-4777 Filed 2-18-75;3:01 p.m.]

